Higher Order Corrections To The Lifetime Of Heavy Hadrons

DISSERTATION

zur Erlangung des akademischen Grades eines Doktors der Naturwissenschaften

vorgelegt von M.Sc. Maria Laura Piscopo

eingereicht bei der Naturwissenschaftlich-Technischen Fakultät der Universität Siegen

Siegen September 2021 Betreuer und erster Gutachter Prof. Dr. Alexander Lenz Universität Siegen

Zweiter Gutachter Prof. Dr. Thomas Mannel Universität Siegen

Weitere Mitglieder der Promotionskommission

Prof. Dr. Markus Cristinziani Universität Siegen und

Prof. Dr. Alexey Petrov Wayne State University

Tag der mündlichen Prüfung 27. Oktober 2021

Abstract

In this work we discuss the theoretical status for the study of the lifetime of heavy hadrons. After presenting some introductory topics like the effective weak Hamiltonian and the heavy quark effective theory (HQET), we describe the construction of the heavy quark expansion (HQE), which constitutes the theoretical framework to systematically compute the total decay width of heavy hadrons, in terms of an expansion in inverse powers of the heavy quark mass. The structure of the HQE is discussed in detail, and the computation of the lowest dimensional contributions, explicitly outlined. Particular emphasis is put in describing the expansion of the quark propagator in the external gluon field using the Fock-Schwinger (FS) gauge, which represents a fundamental ingredient of the calculation. Moreover, the main result is the computation of the dimension-six contribution due to the Darwin operator, only recently determined and found to have a sizeable effect. Finally, we consider two phenomenological applications of the HQE in the charm sector, namely the study of the lifetime of charmed mesons and the analysis of the Glashow-Iliopoulos-Maiani (GIM) cancellations in neutral *D*-meson mixing. By comparing our results with recent measurements performed by the LHCb, Belle-II and BESIII collaborations, we conclude that the HQE is able to reproduce, within large theoretical uncertainties, the experimental pattern for the lifetimes of charmed mesons and we discuss a potential solution for the discrepancy of previous theoretical determinations of D-mixing with data.

Zusammenfassung

In dieser Arbeit erörtern wir den theoretischen Status der Untersuchung von Lebensdauern schwerer Hadronen. Nach einigen einführenden Themen wie dem schwachen effektiven Hamiltonian und der effektiven Theorie für schwere Quarks (HQET) beschreiben wir die Konstruktion der Heavy Quark Expansion (HQE), die den theoretischen Rahmen für die systematische Berechnung der totalen Zerfallsbreite schwerer Hadronen in Form einer Entwicklung in inversen Potenzen der schweren Quarkmasse bildet. Die Struktur der HQE wird im Detail diskutiert und die Berechnung der niedrigstdimensionalen Beiträge wird explizit dargestellt. Ein besonderes Augenmerk wird auf die Beschreibung der Entwicklung des Quark-Propagators in einem externen Gluon-Feld unter Verwendung der Fock-Schwinger-Eichung (FS) gelegt, die einen grundlegenden Bestandteil unserer Rechnungen darstellt. Darüber hinaus ist das Hauptergebnis die Berechnung des Beitrags des Darwin Terms mit der Massendimension sechs, der erst vor kurzem von uns erstmals bestimmt wurde und numerisch bedeutend ist. Schließlich stellen wir zwei phänomenologische Anwendungen der HQE im Charm-Sektor vor, nämlich die Untersuchung der Lebensdauern von Charm Mesonen und die Analyse der Glashow-Iliopoulos-Maiani (GIM)-Kanzellierungen in der Mischung neutraler D Mesonen. Durch den Vergleich unserer Ergebnisse mit den jüngsten Messungen der Kollaborationen LHCb, BelleII und BesIII kommen wir zu dem Schluss, dass die HQE in der Lage ist, innerhalb großer theoretischer Unsicherheiten, die experimentellen Resultate für die Lebensdauern von Charm Mesonen zu reproduzieren, und wir diskutieren eine mögliche Lösung für die Diskrepanz zwischen früheren theoretischen Bestimmungen von D-Mischung und den Daten.

Alla mia famiglia: mamma, papà e Beniamino "If the doors of perceptions were cleansed, everything would appear to man as it is, Infinite."

William Blake, The Marriage of Heaven and Hell.

Contents

Introduction			
N	otatio	ons	15
1	The	Theoretical Framework	19
	1.1	The effective weak Hamiltonian	19
	1.2	The heavy quark effective theory	27
	1.3	Expansion of the quark-propagator in the Fock-Schwinger gauge	36
	1.4	The heavy quark expansion	49
2	Pra	ctical Calculations within the HQE	61
	2.1	Contribution of two-quark operators	63
		2.1.1 Computation of $\Gamma_3^{(c\bar{u}d)}$	65
		2.1.2 Computation of $\Gamma_5^{(c\bar{u}d)}$	69
	2.2	Contribution of four-quark operators	75
		2.2.1 Computation of $\operatorname{Im}\mathcal{T}_6^{(4q)}$	77
		2.2.2 Computation of $\operatorname{Im} \mathcal{T}_7^{(4q)}$	86
3	Con	tribution of the Darwin Operator	95
	3.1	Computation of $\Gamma^{(2q)}(B)$ up to order $1/m_b^3$	96
	3.2	Operator-mixing at order $1/m_b^3$	105
	3.3	Analytical expressions for the coefficients of the Darwin operator	117
4	Phe	nomenology of Lifetime and Mixing	123
	4.1	Theoretical study of the total decay width of charmed mesons $\ . \ . \ .$	125
		4.1.1 Description of the computation $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	127
		4.1.2 Numerical results	148
	4.2	Charm mixing	154
		4.2.1 GIM cancellations in D -mixing $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	155
		4.2.2 Alternative scale setting	160

Co	Conclusion	
A	Acknowledgements	
Α	Expansion of operators containing the heavy quark momentum	171
в	Results for one-loop integrals in dimensional regularisation	175
\mathbf{C}	Decomposition of tensor integrals	181
D	Expressions for the coefficients $\mathcal{C}_0^{(q_1 \bar{q}_2 q_3)}$ and $\mathcal{C}_{G,mn}^{(q_1 \bar{q}_2 q_3)}$	187
\mathbf{E}	Expressions for the divergent functions $\mathcal{D}_{mn}^{(q_1 ar{q}_2 q_3)}$	189
\mathbf{F}	Coefficients of the Darwin operator for the charm system	191
G	Parametrisation of matrix elements of four-quark operators	193

Introduction

The standard model of particle physics (SM) describes our knowledge about the fundamental constituents of nature, quarks and leptons, and the interactions among them [1–3] and it is confirmed by numerous measurements to an astonishing precision, see e.g. textbooks like [4]. With the discovery of the Higgs boson [5–7] by the experimental collaborations ATLAS [8] and CMS [9] at the Large Hadron Collider (LHC) at CERN in 2012, the spectrum of the SM particles is complete.

Despite the enormous success, the SM leaves many important questions open, in fact, e.g. it is not able to explain the existence of ordinary matter in the Universe or that of dark matter. According to the Sakharov criteria [10], the fundamental theory of nature must incorporate C and CP violation, baryon number violating processes and a strong first order phase transition in the early Universe, to potentially explain the existence of ordinary matter. C violation is implemented by construction in the SM and CP violation is present in the Cabibbo-Kobayashi-Maskawa matrix [11,12], although typically, the amount of CP violation contained in the CKM matrix is considered to be too small to explain the matter-antimatter asymmetry [13], see, however Ref. [14] for a counter example. Baryon number is violated in the SM via sphalerons [15], but a strong first order phase transition could only occur for Higgs masses below 70 GeV [16], which is not realised in nature [8,9]¹.

Because of this, the SM is typically considered to be an effective theory, see e.g. the textbook [18], extended at higher energies with contributions that might explain some of the open questions. Numerous possible extensions of the SM have been studied in the literature, one of the simplest predicts the existence of a second Higgs doublet, see e.g. the review [19], which could provide the missing amount of CP violation and also a strong first order phase transition, see e.g. Ref. [20]. Another example is the framework presented in Ref. [14], in which it is investigated the possibility to explain the existence of matter and dark matter, through new sources of CP violation in mixing of neutral B mesons and new couplings of the B mesons with light dark matter particles.

¹In Ref. [17], we have developed a method to solve differential equations using neural networks, applied then to the study of cosmological phase transitions in the early Universe.

The search for beyond standard model (BSM) effects in particle physics, can be direct and indirect. With the former, new heavy resonances can be produced in particle collisions by increasing the center of mass energy, however, apart from the discovery of the Higgs boson, direct searches have not been successful so far at the LHC. On the other side, with indirect searches, measurements of observables with high precision, are compared with the corresponding SM predictions. In this case a robust control over the theoretical uncertainties is crucial, and the bottleneck is represented by the strong interaction, which either requires the calculation of higher order perturbative corrections or the use of non perturbative methods. Since the LHC will continue running for several years with increased luminosity and in the upcoming future there will not be a new particle accelerator with higher center of mass, in recent years there has been a progressive shift from direct to indirect searches, see e.g. Ref. [21].

Quark flavour physics is particularly well suited for indirect searches of BSM effects due to several reasons. First, many experiments are providing precise flavour data, e.g. LHCb, Belle II, BESIII, ATLAS, CMS and formerly BaBar, Belle and many more, see e.g. the extensive HFLAV report for a list of the numerous measurements [22]. Second, the theoretical description of quark flavour observables is theoretically very advanced and enables a control of the hadronic effects, see e.g. the textbooks [23-25]. The computation of higher order perturbative corrections can be systematically improved, see the recent N³LO-QCD calculation for the semileptonic $b \rightarrow c \ell \nu_{\ell}$ decay [26]. Moreover, many heavy flavour observables can be expressed in terms of a series in inverse powers of the heavy quark mass, see e.g. the review [27], and again higher order power corrections can be systematically determined, see e.g. the computation of the contributions up to order $1/m_b^5$ for semileptonic *b*-decays [28]. Non perturbative effects can be determined with theoretical tools like light-cone sum rules (LCSR) [29–31] or lattice QCD [32], which can also be systematically improved, in order to match the increasing experimental precision. Third, CP violating effects are large in the B_d -system and they are well studied, see e.g. Ref. [33]. Conversely, they are expected to be very small in the charm sector, see e.g. the review [34], and in the B_s system, see e.g. the review [35], and can then provide a useful null-hypothesis test of the SM, since any measurement of a sizeable amount of CP violation could be a clear signal for BSM effects. Finally, we currently witness a significant number of deviations between experiments and SM predictions for quark flavour observables. The most famous are the so-called "flavour anomalies" [36], observed in semileptonic loop-level decays, induced by the $b \to s\ell\ell$, with $\ell = \mu, e$, transitions and semileptonic tree-level decays, induced by the $b \to c\ell\nu$, with $\ell = \mu, \tau$, transitions. A combined statistical analysis of these anomalies points at deviations of the order of six to seven standard deviations, see e.g. Ref. [37].

The above arguments show that indirect BSM searches with quark flavour observables represent a very interesting and promising field for future investigations in elementary particle physics. However, in order to be able to unequivocally identify the signals of BSM effects, it is of primary importance to further improve the control over the theoretical predictions. In this respect, the work here presented constitutes a detailed study of the theoretical status for the determination of the lifetime of heavy hadrons, like the B- and the D-mesons. In particular, we analyse the structure of the heavy quark expansion (HQE), which provides a consistent framework to compute the total decay width of heavy hadrons in terms of a series in inverse powers of the heavy quark mass, and discuss the recent computation of higher power corrections of dimension-six. Specifically, the content presented in this work is divided into four major parts. In Chapter 1 we introduce the main theoretical ingredients required for the computation, and in particular describe the construction of the HQE. In Chapter 2 we present the explicit calculation of the lowest dimensional contributions to the HQE of a B-meson, namely due to two-quark operators up to order $1/m_b^2$ and to four-quark operators up to order $1/m_b^4$. In Chapter 3 we outline in detail the computation of the contribution of order $1/m_b^3$ due to the Darwin operator for the case of arbitrary non-leptonic decay modes of the b-quark, which has only recently been determined and found to be sizeable. Moreover, particular emphasis is put in describing the mixing between four-quark operators and the Darwin operator at dimension-six, that ensures the cancellation of the infrared divergences, arising from the emission of a soft gluon from a light quark propagator, otherwise present in the coefficients of the Darwin operator. In Chapter 4 we consider two phenomenological applications of the HQE in the charm-sector, specifically, we perform a comprehensive study of the inclusive decay width of charmed mesons and propose a possible solution to explain the large discrepancy between the theoretical determination of mixing of neutral *D*-mesons and the corresponding experimental data. Finally we conclude with a discussion of the results.

Notations

Here we list some of the notations adopted throughout this work, mostly following the textbooks [38,39]. We use the natural system of units, i.e. $c = \hbar = 1$. Indices representing all four components of a four-vector are always labelled by Greek letters e.g. $\mu = 0, 1, 2, 3$, while indices corresponding only to the three space components are labelled by Latin letters e.g. k = 1, 2, 3. Summation over repeated indices is understood unless otherwise stated. The four-dimensional Minkoswki metric tensor is $g_{\mu\nu} = diag(1, -1, -1, -1)$, so that the invariant product between two four-vectors x^{μ} and y^{μ} is given by $x \cdot y \equiv g_{\mu\nu}x^{\mu}y^{\nu} = x^{0}y^{0} - \mathbf{x} \cdot \mathbf{y}$, with three-vectors denoted in bold type. Moreover the differential operator reads

$$\partial_{\mu} = \frac{\partial}{\partial x^{\mu}} = (\partial_t, \nabla) , \qquad \partial^{\mu} = \frac{\partial}{\partial x_{\mu}} = (\partial_t, -\nabla) .$$
 (1)

The Pauli matrices are the three hermitean 2×2 matrices

$$\sigma_1 = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \quad \sigma_2 = \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix}, \quad \sigma_3 = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}, \quad (2)$$

satisfying $\sigma_j \sigma_k = \delta_{jk} + i\epsilon_{jkl}\sigma_l$, with $\epsilon_{123} = 1$. The four-dimensional gamma matrices γ^{μ} , in the standard representation, are respectively given by

$$\gamma^{0} = \begin{pmatrix} \mathbb{1}_{2} & 0\\ 0 & -\mathbb{1}_{2} \end{pmatrix}, \quad \gamma^{k} = \begin{pmatrix} 0 & \sigma_{k}\\ -\sigma_{k} & 0 \end{pmatrix}, \tag{3}$$

with

$$\{\gamma^{\mu}, \gamma^{\nu}\} = 2g^{\mu\nu} \mathbb{1}_4, \qquad \gamma^{\mu\dagger} = \gamma^0 \gamma^{\mu} \gamma^0.$$
(4)

The commutator of two gamma matrices is

$$\sigma^{\mu\nu} = \frac{i}{2} \left[\gamma^{\mu}, \gamma^{\nu} \right], \tag{5}$$

while the fifth gamma matrix is defined as

$$\gamma_5 = \gamma^5 = i\gamma^0\gamma^1\gamma^2\gamma^3 \,. \tag{6}$$

Regarding the convention for the four-dimensional Levi-Civita tensor $\epsilon^{\mu\nu\rho\sigma}$, we use $\epsilon^{0123} = 1 = -\epsilon_{0123}$. With the above definitions for γ_5 and $\epsilon^{\mu\nu\rho\sigma}$, it follows that the tensor decomposition of three gamma matrices reads

$$\gamma^{\mu}\gamma^{\nu}\gamma^{\rho} = g^{\mu\nu}\gamma^{\rho} - g^{\mu\rho}\gamma^{\nu} + g^{\nu\rho}\gamma^{\mu} + i\epsilon^{\mu\nu\rho\sigma}\gamma_{\sigma}\gamma_{5}, \qquad (7)$$

and that the trace of four gamma matrices and one γ_5 is

$$\operatorname{Tr}\left[\gamma^{\mu}\gamma^{\nu}\gamma^{\rho}\gamma^{\sigma}\gamma_{5}\right] = -4i\epsilon^{\mu\nu\rho\sigma}.$$
(8)

Quantum Chromodynamics (QCD) is a non-abelian gauge theory with the symmetry group $SU(N_c)$ and number of colours $N_c = 3$. Colour indices of fields in the adjoint representation are indicated by $a, b, c = 1, \ldots, (N_c^2 - 1)$, whereas $i, j, k = 1, \ldots, N_c$, are used to label fields in the fundamental representation. The generators in the fundamental and in the adjoint representation are respectively denoted by t^a and T^a . They satisfy the following commutation relations

$$[t^a, t^b] = i f^{abc} t^c, \qquad [T^a, T^b] = i f^{abc} T^c, \qquad (T^a)_{bc} = -i f^{abc}, \qquad (9)$$

where f^{abc} are the structure constants of the group. From the normalisation choice

$$\operatorname{Tr}[t^a t^b] = \frac{1}{2} \delta^{ab} \,, \tag{10}$$

it follows that

$$\left(t^a \cdot t^a\right)_{ij} = C_F \,\delta_{ij} \,, \qquad f^{abc} f^{dbc} = C_A \,\delta^{ad} \,, \tag{11}$$

with $C_F = (N_c^2 - 1)/2N_c$, and $C_A = N_c$. The Feynman rules for a perturbative analysis of QCD are derived from the Lagrangian ²

$$\mathcal{L}_{classical} + \mathcal{L}_{gauge-fixing} + \mathcal{L}_{ghost} \,, \tag{12}$$

 $^{^{2}}$ Note that this is an abuse of notation, it actually corresponds to the Lagrangian density.

here, without discussing $\mathcal{L}_{gauge-fixing}$ and \mathcal{L}_{ghost} , required for the renormalisation of the theory, the classical Lagrangian reads

$$\mathcal{L}_{classical} = -\frac{1}{4} G^a_{\mu\nu}(x) G^{a\mu\nu}(x) + \sum_{flavours} \bar{q}^i(x) \left(i \not\!\!D - m_q\right)_{ij} q^j(x) \,. \tag{13}$$

In Eq. (13), the gluon field strength tensor is

$$G^{a}_{\mu\nu}(x) = \partial_{\mu}A^{a}_{\nu}(x) - \partial_{\nu}A^{a}_{\mu}(x) + g_{s}f^{abc}A^{b}_{\mu}(x)A^{c}_{\nu}(x), \qquad (14)$$

 $A^a_{\mu}(x)$ denotes the corresponding gauge field and g_s is the strong coupling. Moreover, we use $G_{\mu\nu} = G^a_{\mu\nu}t^a$ and $A_{\mu} = A^a_{\mu}t^a$. Acting respectively on fields in the fundamental and adjoint representation, the covariant derivative takes the form

$$(D_{\mu})_{ij} = \partial_{\mu}\delta_{ij} - ig_s A^a_{\mu}(x)(t^a)_{ij}, \qquad (15)$$

and

$$(D_{\mu})_{ab} = \partial_{\mu}\delta_{ab} - ig_s A^c_{\mu}(x)(T^c)_{ab}.$$
(16)

Finally, the gluon field strength tensor can be expressed in terms of the commutator of two covariant derivatives in the fundamental representation, as

$$G_{\mu\nu} = G^{a}_{\mu\nu}t^{a} = \frac{i}{g_{s}} \left[D_{\mu}, D_{\nu} \right]$$

$$= \frac{i}{g_{s}} \left[\partial_{\mu} - ig_{s}A^{b}_{\mu}(x)t^{b}, \partial_{\nu} - ig_{s}A^{c}_{\nu}(x)t^{c} \right]$$

$$= \left(\partial_{\mu}A^{a}_{\nu}(x) - \partial_{\nu}A^{a}_{\mu}(x) \right)t^{a} - ig_{s}A^{b}_{\mu}(x)A^{c}_{\nu}(x) \left[t^{b}, t^{c} \right]$$

$$= \left(\partial_{\mu}A^{a}_{\nu}(x) - \partial_{\nu}A^{a}_{\mu}(x) + g_{s}f^{abc}A^{b}_{\mu}(x)A^{c}_{\nu}(x) \right)t^{a}, \qquad (17)$$

while from Eqs. (16), (15), and (9), we obtain that a covariant derivative acting on the gluon field strength tensor i.e. $D_{\rho}G_{\mu\nu} = (D_{\rho})_{ab}G^b_{\mu\nu}t^a$, see also e.g. Ref. [40], can be written as

$$D_{\rho}G_{\mu\nu} = \partial_{\rho}\delta_{ab}G^{b}_{\mu\nu}t^{a} - g_{s}A^{c}_{\rho}G^{b}_{\mu\nu}f^{cab}t^{a}$$

$$= \partial_{\rho}G_{\mu\nu} - ig_{s}\left[t^{c}, t^{b}\right]A^{c}_{\rho}G^{b}_{\mu\nu}$$

$$= \partial_{\rho}G_{\mu\nu} - ig_{s}\left[A_{\rho}, G_{\mu\nu}\right]$$

$$= \left[D_{\rho}, G_{\mu\nu}\right] = \frac{i}{g_{s}}\left[D_{\rho}, \left[D_{\mu}, D_{\nu}\right]\right], \qquad (18)$$

where the covariant derivative on the l.h.s. of Eq. (18) is in the adjoint representation and those in the last line of Eq. (18) in the fundamental. However, in the following, the coupling constant will be mostly absorbed in the definition of the gluon field i.e. $A_{\mu}(x) \rightarrow 1/g_s A_{\mu}(x)$, and $G_{\mu\nu} \rightarrow g_s G_{\mu\nu}$, so that in this case

$$G_{\mu\nu} = i \left[D_{\mu}, D_{\nu} \right], \qquad D_{\rho} G_{\mu\nu} = i \left[D_{\rho}, \left[D_{\mu}, D_{\nu} \right] \right].$$
(19)

Chapter 1

The Theoretical Framework

In this first chapter we present four of the fundamental theoretical tools necessary for the upcoming discussions and computations. Specifically, in Section 1.1, we introduce the effective weak Hamiltonian, which provides the appropriate framework to study processes like *b*- and *c*-quark decays, that happen at energy scales much lower than the *W*-boson mass. We then briefly describe in Section 1.2, the construction of the heavy quark effective theory (HQET), which represents an approximation of QCD, valid in the case of heavy quarks Q with mass $m_Q \gg \Lambda_{QCD}$, where Λ_{QCD} characterises the onset of the non perturbative regime of the strong coupling α_s . Particular emphasis is put in deriving in Section 1.3, the expansion of the quarkpropagator in the external gluon field using the Fock-Schwinger (FS) gauge, a key ingredient for the calculations presented in the subsequent chapters. Finally, in Section 1.4, we introduce the general framework in which all of the computations and results obtained in the present work are embedded, namely, the heavy quark expansion (HQE).

1.1 The effective weak Hamiltonian

The study of hadronic weak decays defines a typical multi-scale problem in which the mass of the W-boson, m_W , the one of the decaying constituent quark, m, and the hadronic non perturbative scale Λ_{QCD} , lead to the hierarchy $m_W \gg m \gg \Lambda_{QCD}$. The construction of effective field theories (EFTs) provides a general way to deal with multi-scale problems, as it allows to reduce them to a combination of simpler and single-scale ones.

In order to derive the effective Hamiltonian needed to describe weak decays of *B*- and *D*-hadrons in the sequent chapters, we consider as a paradigmatic example the $c \rightarrow s\bar{d}u$ decay. We stress that the content of this section closely follows the one



Figure 1.1: By expanding in powers of $1/m_W^2$, the non-local amplitude in the full theory (left), results in a local interaction in the effective theory (right). The crosses in vertices denote the insertion of the effective four-quark operator.

of the reviews [41–44], to which we refer for a comprehensive introduction to the effective Hamiltonians for weak decays as well as for further references on the topic.

The tree-level flavour changing transition $c \rightarrow s\bar{d}u$, proceeds through the exchange of a W-boson between the (cs) and (ud) left-handed quark currents, as it is diagrammatically shown in the left diagram of Figure 1.1. The amplitude for process is given by

$$iT = \left(i\frac{g}{2\sqrt{2}}\right)^2 V_{cs}^* V_{ud} \left\langle \int d^4x \int d^4y \,\bar{s}^i(x) \,\Gamma^\mu c^i(x) \,iD_{\mu\nu}(x,y) \,\bar{u}^j(y) \,\Gamma^\nu d^j(y) \right\rangle, \quad (1.1)$$

where g is the coupling corresponding to the $SU(2)_L$ symmetry group, $V_{q_1q_2}$ the elements of the quark-mixing Cabibbo-Kobayashi-Maskawa (CKM) matrix, and we have introduced the short-hand notation for the Lorentz structure $\Gamma_{\mu} = \gamma_{\mu}(1 - \gamma_5)$, and for the matrix element between external quark states $\langle \ldots \rangle$. Moreover, in Eq. (1.1) the propagator of the W-boson $D_{\mu\nu}(x, y)$, in the unitary gauge, admits the Fourier representation, see e.g. the textbook [38]

$$D_{\mu\nu}(x,y) = \int \frac{d^4k}{(2\pi)^4} \frac{-1}{k^2 - m_W^2} \left(g_{\mu\nu} - \frac{k_\mu k_\nu}{m_W^2}\right) e^{-ik \cdot (x-y)} \,. \tag{1.2}$$

Because of momentum conservation, the integral in Eq. (1.2) is saturated by values of k of the order of the decaying quark mass m_c , much smaller than m_W . It follows that by expanding in powers of $1/m_W^2$, the expression of the W-propagator reduces to

$$D_{\mu\nu}(x,y) = \int \frac{d^4k}{(2\pi)^4} \left[\frac{g_{\mu\nu}}{m_W^2} + \mathcal{O}\left(\frac{k^2}{m_W^4}\right) \right] e^{-ik \cdot (x-y)} \approx \frac{g_{\mu\nu}}{m_W^2} \delta^{(4)}(x-y) \,. \tag{1.3}$$

Substituting Eq. (1.3) into Eq. (1.1), and performing the integration over the variable y^{μ} , we obtain that the transition amplitude for the tree-level process, up to



Figure 1.2: Diagrams describing the decay $c \rightarrow s\bar{d}u$, at NLO-QCD in the full theory. Left-right and up-down reflected diagrams are not shown.

corrections suppressed by powers of k^2/m_W^2 in the W-propagator, reads

$$iT = -i\frac{G_F}{\sqrt{2}}V_{cs}^*V_{ud}\left\langle \int d^4x \,\bar{s}^i(x)\,\Gamma^\mu c^i(x)\bar{u}^j(x)\,\Gamma_\mu d^j(x)\right\rangle,\tag{1.4}$$

where the Fermi constant G_F is defined as

$$\frac{G_F}{\sqrt{2}} = \frac{g^2}{8m_W^2} \,. \tag{1.5}$$

The amplitude in Eq. (1.4), valid at energy scales much lower than m_W , could have been equivalently derived starting from the following effective Hamiltonian

$$\mathcal{H}_{eff}(x) = \frac{G_F}{\sqrt{2}} V_{cs}^* V_{ud} \,\bar{s}^i(x) \,\Gamma^\mu c^i(x) \bar{u}^j(x) \,\Gamma_\mu d^j(x) \,. \tag{1.6}$$

We see that, by exploiting the hierarchy $m_W \gg m_c$, the non-local product of two currents, namely the non-local operator in Eq. (1.1), has been expressed in Eq. (1.4) in terms of a local operator weighted by an effective coupling. This is schematically shown in Figure 1.1 and represents a basic illustration of the Wilson operator product expansion (OPE) [45, 46]. The next step is to include perturbative QCD corrections to the tree-level transition $c \to s \bar{d} u$, as schematically shown in Figure 1.2. In this case the effective Hamiltonian must be modified as

$$\mathcal{H}_{eff}(x) = \frac{G_F}{\sqrt{2}} V_{cs}^* V_{ud} \left(C_1 Q_1(x) + C_2 Q_2(x) \right), \qquad (1.7)$$

where the local effective four-quark operators are given by

$$Q_1(x) = \left(\bar{s}^i(x)\Gamma^\mu c^i(x)\right) \left(\bar{u}^j(x)\Gamma_\mu d^j(x)\right), \qquad (1.8)$$

$$Q_2(x) = \left(\bar{s}^i(x)\Gamma^{\mu}c^j(x)\right) \left(\bar{u}^j(x)\Gamma_{\mu}d^i(x)\right).$$
(1.9)



Figure 1.3: Diagrams describing the $c \rightarrow s\bar{d}u$ at NLO-QCD in the effective theory. Again, left-right and up-down reflected diagrams are not shown.

We see that in addition to the operator $Q_1(x)^{-1}$, already obtained in the case of tree-level transition, there is a new operator $Q_2(x)$, with different contractions of the colour indices, which arises due to the fact that the exchange of a gluon leads to two possible colour structures, because of the completeness property of the $SU(3)_c$ generators t^a , i.e.

$$t^a_{ik} t^a_{jl} = \frac{1}{2} \left(\delta_{il} \,\delta_{jk} - \frac{1}{N_c} \delta_{ik} \,\delta_{jl} \right) \,. \tag{1.10}$$

In Eq. (1.7), C_1 and C_2 denote the corresponding Wilson coefficients (WCs) of the effective operators Q_1 and Q_2 . From the result in Eq. (1.6), it follows that in the absence QCD corrections, it is $C_1 = 1$ and $C_2 = 0$.

The general prescription to determine the expression of the Wilson coefficients is to require that the amplitude in the full theory is reproduced by the corresponding one in the effective theory, which reads

$$iT = -i\frac{G_F}{\sqrt{2}}V_{cs}^*V_{ud}\left(C_1\langle Q_1\rangle + C_2\langle Q_2\rangle\right).$$
(1.11)

By computing, on one side, QCD corrections to the amplitude iT in the full theory, see Figure 1.2, and on the other side, the matrix elements of the effective operators $\langle Q_1 \rangle$ and $\langle Q_2 \rangle$, at the same order in α_s , see Figure 1.3, we can obtain the corresponding expressions for the Wilson coefficients by equating the two results and by taking into account Eq. (1.11). This procedure is called matching of the full theory onto the effective theory. Omitting the explicit calculation, we only show the final result for the renormalised amplitude in the full theory up to NLO-QCD corrections. This is

¹Note that we do not adopt the convention historically used in the literature, see e.g. Ref. [42], and instead denote by Q_1 the colour-singlet operator.

$$iT = -i\frac{G_F}{\sqrt{2}}V_{cs}^*V_{ud}\left[\left(1 + 2C_F\frac{\alpha_s}{4\pi}\log\left(\frac{\mu^2}{-p^2}\right)\right)\langle Q_1\rangle_{tree} + \frac{3}{N_c}\frac{\alpha_s}{4\pi}\log\left(\frac{m_W^2}{-p^2}\right)\langle Q_1\rangle_{tree} - 3\frac{\alpha_s}{4\pi}\log\left(\frac{m_W^2}{-p^2}\right)\langle Q_2\rangle_{tree}\right],$$
(1.12)

where $\langle Q_{1,2} \rangle_{tree}$ denote the tree level matrix elements of the operators Q_1 and Q_2 . The expression in Eq. (1.12) has been obtained in dimensional regularisation [47–50], with $D = 4-2\epsilon$ space-time dimensions, using the Feynman gauge for the gluon propagator, massless external quark states and an off-shell momentum p, see Ref. [42]. Note that Eq. (1.12) includes only logarithmic corrections of the type $\alpha_s \cdot \log$ and constant terms of order $\mathcal{O}(\alpha_s)$ have been neglected, which corresponds to the leading logarithmic approximation. Moreover, the renormalisation of the quark fields has been already implemented and has resulted in the explicit μ dependence.

Similarly, by computing the diagrams within the effective theory shown in Figure 1.3, leads to the following results for the unrenormalised matrix elements of the operators Q_1 and Q_2 , up to NLO-QCD corrections, namely

$$\langle Q_1 \rangle^{(0)} = \left[1 + 2 C_F \frac{\alpha_s}{4\pi} \left(\frac{1}{\epsilon} + \log \left(\frac{\mu^2}{-p^2} \right) \right) \right] \langle Q_1 \rangle_{tree}$$

$$+ \frac{3}{N_c} \frac{\alpha_s}{4\pi} \left(\frac{1}{\epsilon} + \log \left(\frac{\mu^2}{-p^2} \right) \right) \langle Q_1 \rangle_{tree} - 3 \frac{\alpha_s}{4\pi} \left(\frac{1}{\epsilon} + \log \left(\frac{\mu^2}{-p^2} \right) \right) \langle Q_2 \rangle_{tree} ,$$

$$(1.13)$$

and

$$\langle Q_2 \rangle^{(0)} = \left[1 + 2 C_F \frac{\alpha_s}{4\pi} \left(\frac{1}{\epsilon} + \log \left(\frac{\mu^2}{-p^2} \right) \right) \right] \langle Q_2 \rangle_{tree}$$

$$+ \frac{3}{N_c} \frac{\alpha_s}{4\pi} \left(\frac{1}{\epsilon} + \log \left(\frac{\mu^2}{-p^2} \right) \right) \langle Q_2 \rangle_{tree} - 3 \frac{\alpha_s}{4\pi} \left(\frac{1}{\epsilon} + \log \left(\frac{\mu^2}{-p^2} \right) \right) \langle Q_1 \rangle_{tree} .$$

$$(1.14)$$

The $1/\epsilon$ poles in the square brackets of Eqs. (1.13), (1.14), are removed again with the renormalisation of the quark field. However, the results are still divergent and require in addition an operator renormalisation, i.e.

$$Q_i^{(0)} = Z_{ij} Q_j, \qquad i, j = 1, 2,$$
 (1.15)

where the superscript (0) refers to unrenormalised quantities, and \hat{Z} is a 2×2 renormalisation matrix. By taking into account also the field renormalisation Z_q , the relation between the unrenormalised and renormalised matrix elements, denoted by $\langle Q_i \rangle$, is given by

$$\langle Q_i \rangle^{(0)} = Z_q^{-2} Z_{ij} \langle Q_j \rangle, \qquad (1.16)$$

and in the MS scheme [51], the explicit expression of \mathbb{Z}_q is

$$Z_q = 1 - \frac{1}{\epsilon} \frac{C_F \alpha_s}{4\pi} + \mathcal{O}(\alpha_s^2) \,. \tag{1.17}$$

Using Eq. (1.16), and Eqs. (1.13), (1.14), (1.17), yields to the following result for the \hat{Z} matrix in the MS scheme, namely

$$\hat{Z} = \mathbb{1}_2 + \frac{\alpha_s}{4\pi} \frac{1}{\epsilon} \begin{pmatrix} 3/N_c & -3\\ -3 & 3/N_c \end{pmatrix} + \mathcal{O}(\alpha_s^2), \qquad (1.18)$$

from which we obtain that the renormalised matrix elements of the local operators, respectively read

$$\langle Q_1 \rangle = \left[1 + 2 C_F \frac{\alpha_s}{4\pi} \log\left(\frac{\mu^2}{-p^2}\right) \right] \langle Q_1 \rangle_{tree} + \frac{3}{N_c} \frac{\alpha_s}{4\pi} \log\left(\frac{\mu^2}{-p^2}\right) \langle Q_1 \rangle_{tree} - 3 \frac{\alpha_s}{4\pi} \log\left(\frac{\mu^2}{-p^2}\right) \langle Q_2 \rangle_{tree} , \qquad (1.19)$$

and

$$\langle Q_2 \rangle = \left[1 + 2 C_F \frac{\alpha_s}{4\pi} \log\left(\frac{\mu^2}{-p^2}\right) \right] \langle Q_2 \rangle_{tree} + \frac{3}{N_c} \frac{\alpha_s}{4\pi} \log\left(\frac{\mu^2}{-p^2}\right) \langle Q_2 \rangle_{tree} - 3 \frac{\alpha_s}{4\pi} \log\left(\frac{\mu^2}{-p^2}\right) \langle Q_1 \rangle_{tree} \,. \tag{1.20}$$

Finally, by substituting Eqs. (1.19), (1.20), into Eq. (1.11) and taking into account Eq. (1.12), we can extract the corresponding expressions of the WCs, i.e.

$$C_1(\mu) = 1 + \frac{3}{N_c} \frac{\alpha_s}{4\pi} \log\left(\frac{m_W^2}{\mu^2}\right), \qquad C_2(\mu) = -3 \frac{\alpha_s}{4\pi} \log\left(\frac{m_W^2}{\mu^2}\right).$$
(1.21)

Setting α_s to zero in Eq. (1.21), we recover $C_1 = 1$, $C_2 = 0$. Note that this same result is obtained also by setting $\mu = m_W$, corresponding to the matching scale.

From the above description it is evident that the main property of the construction of the OPE lies in the possibility to factorise the short and long distance contributions of the full amplitude, between the Wilson coefficients and the matrix element of local operators in the effective theory. In fact, up to terms of order $\mathcal{O}(\alpha_s^2)$, we have

$$\left(1 + \frac{3}{N_c}\frac{\alpha_s}{4\pi}\log\left(\frac{m_W^2}{-p^2}\right)\right) = \left(1 + \frac{3}{N_c}\frac{\alpha_s}{4\pi}\log\left(\frac{m_W^2}{\mu^2}\right)\right)\left(1 + \frac{3}{N_c}\frac{\alpha_s}{4\pi}\log\left(\frac{\mu^2}{-p^2}\right)\right),\tag{1.22}$$

and then

$$\log\left(\frac{m_W^2}{-p^2}\right) = \log\left(\frac{m_W^2}{\mu^2}\right) + \log\left(\frac{\mu^2}{-p^2}\right). \tag{1.23}$$

By taking into account that the logarithms originate from the integration over a loop variable, it follows that we can schematically write

$$\int_{-p^2}^{m_W^2} \frac{dk^2}{k^2} = \int_{\mu^2}^{m_W^2} \frac{dk^2}{k^2} + \int_{-p^2}^{\mu^2} \frac{dk^2}{k^2} , \qquad (1.24)$$

showing that the effect of large virtual momenta in the loop, e.g. from scales $\mu \approx 1$ GeV to m_W is absorbed in the expression of the Wilson coefficients, while the low energy contributions, depending on the off-shell momentum p, are encoded into the matrix elements of the local operators.

However, it is easy to verify that at scales much smaller than m_W , the logarithms in Eq. (1.21) become large, namely

$$\alpha_s \log \frac{m_W^2}{\mu^2} = \mathcal{O}(1), \qquad \text{with} \quad \mu^2 \ll m_W^2, \qquad (1.25)$$

and therefore the series in powers of $\alpha_s \log(m_W^2/\mu^2)$ does not converge. The solution is provided by employing the renormalisation group equations (RGEs), which allow to resum the leading logarithms of the type $\alpha_s^n \log(m_W^2/\mu^2)^n$ to all orders in perturbation theory. Analogously to Eq. (1.15), we then introduce the unrenormalised Wilson coefficients

$$C_i^{(0)} = Z_{ij}^c C_j \,, \tag{1.26}$$

where Z_{ij}^c is the corresponding renormalisation matrix ². From the definition of the anomalous dimension matrix $\hat{\gamma}$

$$\hat{\gamma} = \hat{Z}^{-1} \frac{d}{d\log\mu} \hat{Z} , \qquad (1.27)$$

or explicitly, using Eq. (1.18)

$$\gamma(\alpha_s) = \frac{\alpha_s}{4\pi} \begin{pmatrix} -6/N_c & 6\\ 6 & -6/N_c \end{pmatrix}, \qquad (1.28)$$

it follows that, taking into account $\hat{Z}^{cT} = Z^{-1}$, the RGEs satisfied by the renormalised Wilson coefficients, read

$$\frac{d}{d\log\mu}C_i(\mu) = \gamma_{ij}^T(\alpha_s)C_j(\mu).$$
(1.29)

The solution of Eq. (1.29), can be formally presented as

$$C_i(\mu) = U_{ij}(\mu, m_W)C_j(m_W), \qquad (1.30)$$

with $\hat{U}(\mu, m_W)$ being the evolution matrix describing the running of the Wilson coefficients from the matching scale m_W to the lower scale μ .

We conclude by emphasising that the presence of only two operators $Q_1(x)$ and $Q_2(x)$, in the effective Hamiltonian in Eq. (1.7), follows from having considered the specific decay mode $c \to s\bar{d}u$. In fact, in the description of arbitrary *c*-quark decays, additional operators are generated by including QCD corrections, these are the penguin operators

²Because the effective Hamiltonian, proportional to $\vec{C} \cdot \langle \vec{Q} \rangle$, must be scale independent, it follows that $Z_{ij}^c = Z_{ji}^{-1}$, where Z_{ij} is given in Eq. (1.18).

$$Q_3(x) = \bar{u}^i(x)\gamma^\mu (1-\gamma_5)c^i(x) \sum_q \bar{q}^j(x)\gamma_\mu (1-\gamma_5)q^j(x), \qquad (1.31)$$

$$Q_4(x) = \bar{u}^i(x)\gamma^\mu (1-\gamma_5)c^j(x) \sum_q \bar{q}^j(x)\gamma_\mu (1-\gamma_5)q^i(x), \qquad (1.32)$$

$$Q_5(x) = \bar{u}^i(x)\gamma^\mu (1-\gamma_5)c^i(x) \sum_q \bar{q}^j(x)\gamma_\mu (1+\gamma_5)q^j(x), \qquad (1.33)$$

$$Q_6(x) = \bar{u}^i(x)\gamma^\mu (1-\gamma_5)c^j(x) \sum_q \bar{q}^j(x)\gamma_\mu (1+\gamma_5)q^i(x).$$
(1.34)

1.2 The heavy quark effective theory

The low-energy dynamics of hadrons is governed by the confining QCD interactions and the scale Λ_{QCD} , at which the strong coupling $\alpha_s(\mu)$ becomes non perturbative, provides a characteristic parameter for it. The inapplicability of standard perturbation theory poses a big challenge for the computation of any hadronic matrix element, however, simplifications usually arise when considering special limiting cases. The description of hadrons containing a heavy quark Q, where by heavy it is meant that $m_Q \gg \Lambda_{QCD}$, leads to profound consequences, because under this condition, the hadronic system can be parametrised as an almost free heavy quark surrounded by a cloud of light degrees of freedom. In particular, in the limit $m_Q \to \infty$, it is only the four-velocity v^{μ} of the infinitely heavy quark, which coincides with the hadron velocity, that characterises the bound state dynamics. The QCD interaction with the light constituents, despite changing the heavy quark momentum p_Q^{μ} , cannot affect its velocity, which is conserved because of $\Delta v^{\mu} = \Delta p_{Q}^{\mu}/m_{Q}$. In this limit, the heavy quark effectively acts as a static external colour source. The soft gluons and quarks are sensitive to the static colour field because of confinement but they are unable to resolve other quantum numbers of the heavy quark, like flavour and spin. These relativistic effects are suppressed by the heavy quark mass and can be systematically taken into account in a perturbative way, see e.g. the early review [52]. It follows that, in the heavy-quark limit, the QCD Lagrangian is approximated by an effective theory, the heavy quark effective theory (HQET) [53–63], where new symmetries, which are not present in the original theory, become manifest. Specifically, for a system with f heavy flavours, there is a SU(2f) symmetry group corresponding to rotations in spin and heavy flavour space. The possibility to exploit the existence of the heavy-quark symmetry in certain kinematical domains, leads to simplifications in the computation of hadronic matrix elements involving heavy quarks, in particular it allows to derive model-independent relations between hadronic form factors for weak decays, thus reducing significantly the number of independent input required, see e.g. the review [64].

Far from being exhaustive, the rest of this section is mainly intended as a brief introduction to the HQET, in order to derive the basic properties that will be used in the sequent chapters. The exposure closely follows the comprehensive monograph [65] and the excellent reviews [52, 64, 66, 67].

The fundamental assumption for the construction of the HQET is that a heavy quark bounded in a QCD state with light constituents carries most of the fourmomentum of the system and is quasi on-shell. Interactions with soft gluons and quarks can only change p_Q^{μ} by a fraction small compared to its large "kinetic" component. According to this picture, the heavy quark momentum is parametrised as

$$p_Q^{\mu} = m_Q v^{\mu} + k^{\mu} \,, \tag{1.35}$$

where v^{μ} is the hadron velocity with $v^2 = 1$ and the "residual" momentum k^{μ} determines by how much the heavy quark is off-shell because of the QCD interaction with the light degrees of freedom, so k is of the order of Λ_{QCD} . Substituting Eq. (1.35) into the expression for the Feynman propagator for Q, and expanding in the small quantity k/m_Q , yields

$$i\frac{\not{p}_Q + m_Q}{p_Q^2 - m_Q^2 + i\varepsilon} = i\frac{m_Q \not{\psi} + \not{k} + m_Q}{\left(m_Q^2 + 2m_Q v \cdot k + k^2 - m_Q^2 + i\varepsilon\right)}$$
$$= \frac{i}{v \cdot k + i\varepsilon} \left(\frac{1 + \not{\psi}}{2}\right) + \mathcal{O}\left(\frac{k}{m_Q}\right). \tag{1.36}$$

Eq. (1.36) shows that the propagator of a heavy quark contains a velocity dependent operator which projects onto the positive energy components of the Dirac field. In fact it is immediate to verify that the operators

$$P_{\pm} = \frac{1 \pm \psi}{2} \,, \tag{1.37}$$

fulfil $P_{\pm}^2 = P_{\pm}$, $P_{\pm}P_{\mp} = 0$, and are thus projectors. Their meaning becomes particularly transparent if we consider the rest frame of the heavy quark i.e. $v^{\mu} = (1, \mathbf{0})$, in fact in this case $P_{\pm} = (1 \pm \gamma_0)/2$, or explicitly

$$P_{+} = \begin{pmatrix} \mathbb{1}_{2} & 0\\ 0 & 0 \end{pmatrix}, \quad P_{-} = \begin{pmatrix} 0 & 0\\ 0 & \mathbb{1}_{2} \end{pmatrix}, \quad (1.38)$$

indicating that P_{\pm} respectively project onto the upper/lower two components of the Dirac spinor. Because in Eq. (1.36) only the positive energy solutions of the Dirac equation are propagated, it appears appropriate to introduce the following parametrisation for the heavy quark field:

$$Q(x) = e^{-im_Q v \cdot x} h_v(x) + \mathcal{O}\left(\frac{k}{m_Q}\right) , \qquad (1.39)$$

where the effective heavy quark $h_v(x)$ satisfies

$$h_v(x) = e^{im_Q v \cdot x} \frac{1 + \psi}{2} Q(x), \qquad (1.40)$$

and hence

$$P_{+}h_{v}(x) = h_{v}(x). (1.41)$$

The presence of the exponential prefactor in Eq. (1.39) removes the large "kinetic" part of the heavy-quark momentum, so that $h_v(x)$ contains only the small frequencies of the order of k. Notice also that due to Eq. (1.40), $h_v(x)$ is constrained to be effectively a two-component field. By expressing the QCD Lagrangian for Q, in terms of $h_v(x)$, gives

$$\mathcal{L}_{QCD} = \bar{Q}(x) \left(i \not D - m_Q \right) Q(x) = \bar{h}_v(x) i \not D h_v(x) + \mathcal{O}\left(\frac{k}{m_Q}\right)$$
$$= \bar{h}_v(x) P_+ i \not D P_+ h_v(x) + \mathcal{O}\left(\frac{k}{m_Q}\right), \qquad (1.42)$$

where we have used Eq. (1.41) and then $\psi h_v = h_v$. Taking into account the identity

$$P_{+}\gamma^{\mu}P_{+} = P_{+}P_{-}\gamma^{\mu} + P_{+}v^{\mu} = P_{+}v^{\mu}P_{+}, \qquad (1.43)$$

we obtain that in the limit of a infinitely heavy quark, Eq. (1.42) becomes

$$\mathcal{L}_{HQET} = \bar{h}_v(x)(iv \cdot D)h_v(x) = \bar{h}_v^j(x)\left(iv \cdot \partial \,\delta_{jk} + g_s v \cdot A^a \, t_{jk}^a\right)h_v^k(x)\,,\qquad(1.44)$$

and for clarity the colour indices j, k, have been explicitly indicated in the second equality. Eq. (1.44) defines the Lagrangian of the HQET. The corresponding Feynman rules for the heavy quark propagator and for the coupling of a heavy quark to the gluon field can be easily read off Eq. (1.44). They are respectively given by ³

i

$$\underbrace{\overset{v,k}{\longrightarrow}}_{j} = \frac{i}{v \cdot k} \frac{1 + v}{2} \delta_{ji}$$

$$(1.45)$$

and

$$i \quad \underbrace{\qquad }_{a,\mu}^{v} \quad j \quad = ig_s t^a_{ji} v^{\mu}$$

$$(1.46)$$

Clearly Eq. (1.45) reproduces the leading term in Eq. (1.36). It is worthwhile to emphasise that the effective heavy field $h_v(x)$, by construction, annihilates a heavy particle with velocity v^{μ} but does not create a heavy antiparticle. The conjugate field $h_v(x)$, on the other side, creates a heavy particle with velocity v^{μ} but does not annihilate a heavy antiparticle. Pair production is absent in the infinite heavy mass limit and the field-theoretic description becomes actually redundant, see e.g. Ref. [67]. Consistently, contrary to the full QCD propagator, Eq. (1.45) has a single pole since only heavy particles are propagating in space and time, see e.g. Ref. [52]. The contribution of heavy antiparticles is suppressed by the heavy quark mass and arises when power corrections are included, cf. Eq. (1.40). Note that Eq. (1.44)does not depend on the heavy quark mass, so that the theoretical description stays unchanged if the heavy quark Q is replaced by a different heavy quark Q' with the same velocity v^{μ} , provided that the condition $m_{Q'} \gg \Lambda_{QCD}$ is verified. Furthermore, since the vertex Eq. (1.46) does not contain any gamma matrix, which would act on the spin states of the heavy quark field, the interaction with the gluon is independent of the heavy quark spin. The effective theory exhibits a flavour-spin symmetry, broken by the inclusion of mass effects.

Eq. (1.39) describes only the contribution of the large component of the heavy quark field, $h_v(x)$. In order to derive power corrections to the HQET Lagrangian in Eq (1.44), we introduce the small component $\mathfrak{h}_v(x)$, defined by

$$\mathfrak{h}_{v}(x) = e^{im_{Q}v \cdot x} \, \frac{1 - \psi}{2} \, Q(x) \,, \tag{1.47}$$

³The $+i\varepsilon$ prescription is consistent with a heavy quark propagating forward in time, see Ref [58].

with

$$P_{-}\mathfrak{h}_{v}(x) = \mathfrak{h}_{v}(x). \qquad (1.48)$$

In the rest frame of the heavy quark, see Eq. (1.38), $\mathfrak{h}_v(x)$ corresponds to the lower two components of Q(x) and creates a heavy antiquark with velocity v^{μ} . Including also the effect of $\mathfrak{h}_v(x)$, Eq. (1.39) reads ⁴

$$Q(x) = e^{-im_Q v \cdot x} \Big[h_v(x) + \mathfrak{h}_v(x) \Big], \qquad (1.49)$$

and correspondingly Eq. (1.42) becomes

$$\mathcal{L}_{QCD} = \left[\bar{h}_{v}(x) + \bar{\mathfrak{h}}_{v}(x)\right] \left(m_{Q}\psi + i\not{D} - m_{Q}\right) \left[h_{v}(x) + \mathfrak{h}_{v}(x)\right]$$
$$= \bar{h}_{v}(x)(iv \cdot D)h_{v}(x) - \bar{\mathfrak{h}}_{v}(x)(iv \cdot D + 2m_{Q})\mathfrak{h}_{v}(x)$$
$$+ \bar{h}_{v}(x)P_{+}i\not{D}P_{-}\mathfrak{h}_{v}(x) + \bar{\mathfrak{h}}_{v}(x)P_{-}i\not{D}P_{+}h_{v}(x).$$
(1.50)

Notice that in deriving the second equality in Eq. (1.50), we have used Eqs. (1.41), (1.48), together with $\psi h_v(x) = h_v(x)$ and $\psi \mathfrak{h}_v(x) = -\mathfrak{h}_v(x)$. Moreover, the last line of Eq. (1.50) can be further simplified. Because D is squeezed between P_+ and P_- , the component of the covariant derivative parallel to v^{μ} vanishes and only the one orthogonal to the four-velocity actually contributes. Defining

$$D^{\mu}_{\perp} = D^{\mu} - v^{\mu} (v \cdot D) , \qquad (1.51)$$

with $v \cdot D_{\perp} = 0$, gives

$$\mathcal{L}_{QCD} = \bar{h}_v(x)(iv \cdot D)h_v(x) - \bar{\mathfrak{h}}_v(x)\big(iv \cdot D + 2m_Q\big)\mathfrak{h}_v(x) + \bar{h}_v(x)i\not{D}_{\perp}\mathfrak{h}_v(x) + \bar{\mathfrak{h}}_v(x)i\not{D}_{\perp}h_v(x).$$
(1.52)

Eq. (1.52) shows that the Lagrangian of a heavy, but not infinitely heavy, quark Q contains two independent fields $h_v(x)$ and $\mathfrak{h}_v(x)$, describing respectively massless degrees of freedom and massive excitations with mass twice as large as m_Q . These

⁴Note that the formalism introduced so far applies to the description of a bound state with a heavy quark. The case of a hadron containing a heavy antiquark is obtained by replacing the sign of the velocity i.e. $v^{\mu} \rightarrow -v^{\mu}$.



Figure 1.4: Virtual fluctuation involving the creation and annihilation of a heavy antiquark. Time flows from left to right.

fields interact due to the presence of the two terms in the second line of Eq. (1.52), so that the propagator of a heavy particle $h_v(x)$ receives virtual corrections from the coupling with the heavy antiparticle $\mathfrak{h}_v(x)$. Precisely, a heavy quark propagating forward in time can turn into a virtual heavy antiquark propagating backward in time and then turn back into a heavy quark as it is schematically shown in Figure 1.4, see also Ref. [52]. From Eq. (1.52), it follows that the propagator of the virtual antiquark is suppressed by a factor of $2m_Q$ and at energy scales of the order of Λ_{QCD} , the diagram in Figure 1.4 can be effectively described by a local interaction of the form

$$\bar{h}_v(x)i\not\!\!\!D_\perp 1/(2m_Q)i\not\!\!\!\!D_\perp h_v(x)\,, \qquad (1.53)$$

in which the heavy degrees of freedom corresponding to $\mathfrak{h}_v(x)$, appear decoupled. The process of integrating out the small component of the heavy quark field can be carried out in a systematic way by constructing an effective Lagrangian expressed only in terms of the large component $h_v(x)$. To this end, first we derive from Eq. (1.52) the equations of motion for $h_v(x)$ and $\mathfrak{h}_v(x)$ by computing $\delta \mathcal{L}_{QCD}/\delta \bar{h}_v(x)$ and $\delta \mathcal{L}_{QCD}/\delta \bar{\mathfrak{h}}_v(x)$. This yields respectively

and

$$(iv \cdot D + 2m_Q)\mathfrak{h}_v(x) = i \not\!\!D_\perp h_v(x) \,. \tag{1.55}$$

Eq. (1.55) can be inverted in order to find a relation between $\mathfrak{h}_v(x)$ and $h_v(x)$, i.e.

$$\mathfrak{h}_{v}(x) = \left(iv \cdot D + 2m_{Q} - i\varepsilon\right)^{-1} i \not\!\!{D}_{\perp} h_{v}(x), \qquad (1.56)$$

showing that $\mathfrak{h}_v(x)$ indeed represents the small component of the heavy field Q(x), as it is suppressed with respect to $h_v(x)$, by the heavy quark mass m_Q . By substituting Eq. (1.56) into the equation of motion for $h_v(x)$, Eq. (1.54), we readily arrive at

$$(iv \cdot D)h_v(x) + i\not\!\!D_\perp (iv \cdot D + 2m_Q - i\varepsilon)^{-1} i\not\!\!D_\perp h_v(x) = 0, \qquad (1.57)$$

which can be evidently traced back to the following Lagrangian ⁵

$$\mathcal{L}_{eff} = \bar{h}_v(x)(iv \cdot D)h_v(x) + \bar{h}_v(x)i\not\!\!D_\perp \left(iv \cdot D + 2m_Q - i\varepsilon\right)^{-1}i\not\!\!D_\perp h_v(x).$$
(1.58)

Eq. (1.58) provides the appropriate theory to describe the strong interactions of a heavy quark at the energy scale of the order of Λ_{QCD} . It is expressed only in terms of the effective heavy field $h_v(x)$, as the dynamics of the massive degrees of freedom becomes irrelevant at this scale. The information on $\mathfrak{h}_v(x)$ however, is contained in the second term of Eq. (1.58), which represents a non local operator. Because the action of a derivative on $h_v(x)$, returns only the "residual" momentum k^{μ} , the non local contribution in Eq. (1.58) can be consistently expanded in powers of $1/(2m_Q)$ leading to higher dimensional operators built from covariant derivatives. Correspondingly, the equation of motion satisfied by $h_v(x)$, Eq. (1.57), explicitly depends on the heavy quark mass. In order to exploit the symmetries of the effective theory in the limit of a infinitely heavy quark, it appears convenient to regard Eq. (1.58) in an alternative way, namely by treating the tower of power suppressed operators arising from the expansion of the non local term in Eq. (1.58), as perturbations to the HQET Lagrangian in Eq. (1.44) [63,69,70]. Accordingly, Eq. (1.58) is recast in the form

$$\mathcal{L}_{eff} = \mathcal{L}_{HQET} + \mathcal{L}_{power} , \qquad (1.59)$$

with

$$\mathcal{L}_{power} = \bar{h}_v(x)i\not\!\!D_\perp \frac{1}{2m_Q}i\not\!\!D_\perp h_v(x) + \bar{h}_v(x)i\not\!\!D_\perp \frac{(-iv\cdot D)}{(2m_Q)^2}i\not\!\!D_\perp h_v(x) + \dots$$
(1.60)

Here the ellipsis denote terms suppressed by higher powers of m_Q . Now, the effective heavy quark field $h_v(x)$, satisfies the equation of motion following only from the leading term of Eq. (1.59) i.e.

⁵The Lagrangian in Eq. (1.58) can be equivalently derived using the path integral formalism by integrating out the heavy degrees of freedom from the generating integral of the QCD Green functions with heavy quark fields, see Ref. [68].

$$(iv \cdot D)h_v(x) = 0,$$
 (1.61)

and in the computation of hadronic matrix elements, the contribution of \mathcal{L}_{power} in Eq. (1.60) must be included in a standard perturbative way by taking the time order product with the respective leading order operators. Similarly, by substituting Eq. (1.56) into the expression for the heavy quark field Eq. (1.49), leads to the following expansion:

$$Q(x) = e^{-im_Q v \cdot x} \left(1 + \left(iv \cdot D + 2m_Q - i\epsilon \right)^{-1} i \not{D}_\perp \right) h_v(x)$$

= $e^{-im_Q v \cdot x} \left(1 + \frac{1}{2m_Q} i \not{D}_\perp + \frac{1}{(2m_Q)^2} (-iv \cdot D) i \not{D}_\perp + \dots \right) h_v(x), \quad (1.62)$

which provides the prescription to consistently define in HQET any operator involving a heavy quark field Q(x). Consider e.g. the heavy to light vector current $\mathcal{V}^{\mu}(x) = \bar{q}(x)\gamma^{\mu}Q(x)$, with $m_q \ll m_Q$. Up to leading power corrections, $\mathcal{V}^{\mu}(x)$ can be expressed as

$$\mathcal{V}^{\mu}(x) = e^{-im_Q v \cdot x} \,\bar{q}(x) \gamma^{\mu} \left(1 + \frac{i \not{D}_{\perp}}{2m_Q} + \dots \right) h_v(x) \,. \tag{1.63}$$

Due to the equation of motion Eq. (1.61), the effective heavy field $h_v(x)$ does not contain any information about the heavy quark mass. This has the advantage that the local hadronic matrix element $\langle 0|\mathcal{V}^{\mu}(0)|M(v)\rangle$, where M is the corresponding heavy meson, defined in full QCD, admits a systematic expansion in powers of $1/m_Q$, in which the dependence on the heavy quark mass results completely factored out. In fact, from Eqs. (1.63), (1.60), it follows that

$$\langle 0|\mathcal{V}^{\mu}|M(v)\rangle_{QCD} = \langle 0|\bar{q}\gamma^{\mu}h_{v}|M(v)\rangle_{HQET} + \frac{1}{2m_{Q}}\langle 0|\bar{q}\gamma^{\mu}i\not{D}_{\perp}h_{v}|M(v)\rangle_{HQET} + \frac{1}{2m_{Q}}\langle 0|i\int d^{4}z \operatorname{T}\left\{\bar{q}\gamma^{\mu}h_{v},\mathcal{L}_{1}(z)\right\}|M(v)\rangle_{HQET} + \mathcal{O}\left(\frac{1}{m_{Q}^{2}}\right),$$
(1.64)

where we have introduced the notation

$$\mathcal{L}_{power} = \frac{1}{2m_Q} \mathcal{L}_1 + \frac{1}{4m_Q^2} \mathcal{L}_2 + \dots$$
 (1.65)

Contrary to the matrix element on the l.h.s. of Eq. (1.64), the ones on the r.h.s are independent on m_Q and can be parametrised by universal form factors [59, 70]. However, because of Eq. (1.61), the hadronic state $|M(v)\rangle_{HQET}$ differs from the original one $|M(v)\rangle_{QCD}$. This is encoded in the appearance of the time ordered product of the first term in \mathcal{L}_{power} , with the leading order part of $\mathcal{V}(0)$, which can be interpreted as a correction to the wave function of the heavy meson, see Ref. [52].

Finally, in order to identify the set of lowest dimensional operators generated by the Lagrangian in Eq. (1.60), we can employ the identity

$$P_{+} i \not{D}_{\perp} i \not{D}_{\perp} P_{+} = P_{+} i D_{\perp}^{\mu} i D_{\perp}^{\nu} \left(\frac{\{\gamma_{\mu}, \gamma_{\nu}\}}{2} + \frac{[\gamma_{\mu}, \gamma_{\nu}]}{2} \right) P_{+}$$
$$= P_{+} (i D_{\perp})^{2} P_{+} + P_{+} i D^{\mu} i D^{\nu} (-i \sigma_{\mu\nu}) P_{+} , \qquad (1.66)$$

where in the second line of Eq. (1.66) we have replaced D^{μ}_{\perp} with the total derivative D^{μ} since the component of the covariant derivative parallel to the four velocity does not contribute, in fact

$$P_{+}v^{\mu}\sigma_{\mu\nu}P_{+} = \frac{i}{2}P_{+}(\psi\gamma_{\nu} - \gamma_{\nu}\psi)P_{+} = \frac{i}{2}P_{+}(\gamma_{\nu} - \gamma_{\nu})P_{+} = 0.$$
(1.67)

Recalling the definition of the gluon field strength tensor $G_{\mu\nu} = -i [iD_{\mu}, iD_{\nu}]$, see Eq. (17), it follows that at order $1/m_Q$, two operators appear in \mathcal{L}_{power} , namely

$$\mathcal{L}_{power} = \frac{1}{2m_Q} \Big(\mathcal{O}_I(x) + \mathcal{O}_{II}(x) \Big) + \dots , \qquad (1.68)$$

with

$$\mathcal{O}_I(x) = \bar{h}_v(x)(i\not\!\!\!D_\perp)^2 h_v(x), \qquad (1.69)$$

and

$$\mathcal{O}_{II}(x) = \frac{1}{2}\bar{h}_v(x)G^{\mu\nu}\sigma_{\mu\nu}h_v(x).$$
 (1.70)

The two contributions in Eq. (1.68), describe respectively the covariant extension of the kinetic energy of the heavy quark due to its off-shell motion inside the hadron and the chromo-magnetic interaction of the heavy quark spin with the external gluon field.

1.3 Expansion of the quark-propagator in the Fock-Schwinger gauge

In light of the primary role that it will play in the following chapters, we discuss the computation of the quark propagator 6 in the presence of non perturbative QCD interactions, in a form suitable to describe the case in which the gluon field is soft, namely its characteristic momentum is much smaller than the one carried by the corresponding quark field. Under this assumption the dynamics reduces to that of a quark propagating in a weakly changing gluon background, see e.g. the lecture notes [71], and the solution of the Green function equation can be build as an operator expansion in terms of the the external gauge field. The formulation is based on the Schwinger method which was introduced in the early 50's in the context of Electrodynamics in Ref. [72]. Later it has been adapted to QCD where it has found a large number of applications, see e.g. Ref. [73]. A variation of the background field technique based on the Fock-Schwinger (FS) gauge [72, 74] has been first considered in Refs. [75–77]. This alternative method results extremely convenient for calculations in gauge theories due to the remarkable property that only gauge covariant expressions appear in the intermediate steps of the computation of gauge invariant quantities. For details on the application of the Schwinger method and of the FS gauge in QCD we refer to the technical review Ref. [78] as well as to the references within.

In the rest of the present section, after introducing the FS gauge and discussing its main features, we turn to the calculation of the quark propagator. Specifically, we use the FS gauge to compute the coefficients of the gluon operators that arise in the expansion of the quark propagator, up to terms proportional to one covariant derivative of the gluon field strength tensor $G_{\mu\nu}$. The corresponding expressions are derived both in momentum and in coordinate space.

Let us start by recalling that the vacuum expectation value of the time ordered product of two free-quark fields is defined as

$$\langle 0|\mathsf{T}\{\psi(x),\bar{\psi}(y)\}|0\rangle = iS_0(x,y).$$
(1.71)

For e.g. $x^0 > y^0$, the l.h.s. of Eq. (1.71) describes a quark emitted at the space-time point y^{μ} and subsequently annihilated at point x^{μ} , i.e. $S_0(x, y)$ denotes the propa-

 $^{^{6}}$ We now consider an arbitrary quark, without making any assumption on its mass.
gator of a free-quark. Equivalently $S_0(x, y)$ constitutes the Green function for the Dirac equation, namely it satisfies the inhomogeneous differential equation

$$(i\partial_x - m)S_0(x,y) = \delta^{(4)}(x-y).$$
 (1.72)

It is worthwhile to emphasise how the translation invariance of Eq. (1.72) implies that the free-quark propagator is also translation invariant, this is reflected by the condition $S_0(x, y) = S_0(x-y)$. Eq. (1.72) can be solved exactly. In momentum space $S_0(x-y)$ admits the well known Fourier representation, see e.g. the textbook [38]

$$S_0(x-y) = \int \frac{d^4p}{(2\pi)^4} \,\mathcal{S}_0(p) \,e^{-ip \cdot (x-y)} \,, \tag{1.73}$$

with

$$\mathcal{S}_0(p) = \frac{\not p + m}{p^2 - m^2 + i\varepsilon} \,. \tag{1.74}$$

A quark bounded in an hadronic state is subject to the long-distance interaction with the confining gluon field $A_{\mu}(x)$. Correspondingly, the quark propagator S(x, y)defines the Green function of the coupled Dirac equation, see e.g. Ref [78]⁷

$$(i\partial_x + A(x) - m)S(x, y) = \delta^{(4)}(x - y).$$
 (1.75)

Eq. (1.75) cannot be solved exactly, however in the kinematical region $k^2 \ll q^2$ where k refers to the momentum of the gluon field and $q \gg \Lambda_{QCD}$, is a large perturbative scale saturated by the quark momentum p i.e. $p^2 \sim q^2$, the quark propagates with a characteristic length scale that is much smaller than the one of the external gluon field, which effectively acts as a slowly changing background, see e.g. Ref. [79]. Under the assumption that the field $A_{\mu}(x)$ is weak and randomly orientated, it is possible to construct the solution of Eq. (1.75) in the form of the series, see e.g. Ref. [78]

$$iS(x,y) = iS_0(x-y) + iS_1(x,y) + \dots, \qquad (1.76)$$

where $S_1(x, y)$ denotes the first order correction, describing the interaction of the quark with one gluon field while the ellipsis stand for higher order terms with more than one gluon, explicitly

⁷Unless otherwise stated, in the following, the coupling constant g_s is absorbed in the definition of the gauge field $A^{\mu}(x)$.



Figure 1.5: A quark propagating from y to x in the background gluon field, scatters off $0, 1, 2, \ldots$, soft gluons.

$$iS_1(x,y) = \int d^4z \ iS_0(x-z) \ iA(z) \ iS_0(z-y).$$
(1.77)

Eq. (1.76) is schematically represented in Figure 1.5. It is straightforward to verify that Eq. (1.76) does indeed satisfy Eq. (1.75) up to terms of first order in $A_{\mu}(x)$, by substituting Eq. (1.77) and using Eq. (1.72). To fix the form of the gauge field in Eq. (1.77), it is particularly convenient to employ the FS gauge. This is defined by

$$(x^{\mu} - x_0^{\mu})A_{\mu}(x) = 0.$$
(1.78)

In Eq. (1.78) the gauge fixing parameter x_0^{μ} is an arbitrary space-time point which we set for convenience to zero. On one side, this will lead to simpler expressions, on the other, the possibility to use the independence of the final result on x_0^{μ} as a consistency check for the computation, is lost. The gauge condition then becomes

$$x^{\mu}A_{\mu}(x) = 0. (1.79)$$

Let us immediately point out that the quark propagator S(x, y) is not translation invariant anymore. First, the gauge field in Eq. (1.77) depends on the space-time coordinate. This however only apparently breaks the translation symmetry, since after averaging, the background gluon field is actually translation invariant, see Ref. [78]. The true reason for the symmetry breaking lies in the choice of the FS gauge Eq. (1.79), which gives to the origin the special role of gauge fixing parameter. In general then

$$S(x,y) \neq S(x-y)$$
. (1.80)

Eq. (1.80) can lead to differences in intermediate steps of the computation of a physical quantity although the translation invariance must be restored in any final

meaningful expression.

The FS gauge though, has many advantages. The first is that it allows for a simple relation between the gauge field $A_{\mu}(x)$ and the field strength tensor $G_{\mu\nu}(x)$. This relation reads ⁸

$$A^a_{\mu}(x) = \int_0^1 d\alpha \,\alpha x^{\rho} \,G^a_{\rho\mu}(\alpha x). \tag{1.81}$$

The proof of Eq. (1.81) proceeds as follows. We start with the identity

$$A_{\sigma}(x) = \frac{d}{dx^{\sigma}} (x \cdot A(x)) - x^{\rho} \frac{\partial}{\partial x^{\sigma}} A_{\rho}(x)$$

$$= x^{\rho} G_{\rho\sigma}(x) - x^{\rho} \frac{\partial}{\partial x^{\rho}} A_{\sigma}(x) + i x^{\rho} \left[A_{\rho}(x), A_{\sigma}(x) \right]$$

$$= x^{\rho} G_{\rho\sigma}(x) - x^{\rho} \frac{\partial}{\partial x^{\rho}} A_{\sigma}(x) , \qquad (1.82)$$

where the second and third equalities are consequence of the gauge condition Eq. (1.79). By performing the change of variable $x^{\mu} \rightarrow \alpha x^{\mu}$, it is easy to see that the dependence on $A_{\sigma}(\alpha x)$ is reduced to that of a total derivative i.e. Eq. (1.82) becomes

$$\frac{d}{d\alpha} \Big(\alpha A_{\sigma}(\alpha x) \Big) = \alpha x^{\rho} G_{\rho\sigma}(\alpha x) , \qquad (1.83)$$

which reproduces Eq. (1.81), after integrating both sides over α from 0 to 1. In order to prove another property of the FS gauge, we expand $A_{\mu}(x)$ in Eq. (1.79) around x = 0, this yields ⁹

$$x^{\mu} \Big(A_{\mu} + x^{\nu_1} \partial_{\nu_1} A_{\mu}(x) \big|_{x=0} + \frac{1}{2} x^{\nu_1} x^{\nu_2} \partial_{\nu_1} \partial_{\nu_2} A_{\mu}(x) \big|_{x=0} + \dots \Big) = 0.$$
 (1.84)

For arbitrary space-time coordinates, evidently Eq. (1.84) requires that

$$x^{\mu}A_{\mu} = x^{\mu}x^{\nu_{1}}\partial_{\nu_{1}}A_{\mu}(x)\big|_{x=0} = x^{\mu}x^{\nu_{1}}x^{\nu_{2}}\partial_{\nu_{1}}\partial_{\nu_{2}}A_{\mu}(x)\big|_{x=0} = \dots = 0.$$
(1.85)

⁸In the literature this is known as inversion formula, in reference to the fact that it inverts the usual relation in which the field strength tensor is expressed in terms of the gauge field and of its derivative, for a comprehensive overview on the FS gauge see the PhD thesis [80].

⁹Note that we often omit to explicitly write the space-time coordinate when this is zero.

Eq. (1.85) leads to the important result that in the expansion of an arbitrary function f(x), the action of the partial derivative ∂_{μ} at the origin can be replaced with that of the covariant derivative D_{μ} , in fact

$$f(x) = f + x^{\mu} \partial_{\mu} f(x) \big|_{x=0} + \frac{1}{2} x^{\mu} x^{\nu} \partial_{\mu} \partial_{\nu} f(x) \big|_{x=0} + \dots$$

= $f + x^{\mu} \big(\partial_{\mu} - iA_{\mu}(x) \big) f(x) \big|_{x=0} + x^{\mu} x^{\nu} \big(\partial_{\mu} - iA_{\mu}(x) \big) \big(\partial_{\nu} - iA_{\nu}(x) \big) f(x) \big|_{x=0} + \dots$
= $f + x^{\mu} D_{\mu} f(x) \big|_{x=0} + \frac{1}{2} x^{\mu} x^{\nu} D_{\mu} D_{\nu} f(x) \big|_{x=0} + \dots$ (1.86)

We can now derive a convenient representation for the gauge field $A_{\mu}(x)$. Expanding $G_{\rho\mu}(\alpha x)$ around x = 0 in Eq. (1.81), and taking into account Eq. (1.86), yields

$$A^{a}_{\mu}(x) = \frac{1}{2}x^{\rho}G^{a}_{\rho\mu}(0) + \frac{1}{3}x^{\alpha}x^{\rho}D_{\alpha}G^{a}_{\rho\mu}(0) + \dots , \qquad (1.87)$$

where the ellipsis denote terms with higher derivatives. Eq. (1.87) shows that the gauge field $A_{\mu}(x)$ can be expanded directly in terms of the gluon field strength tensor and of its covariant derivatives evaluated at the origin and it constitutes the main result of the FS gauge. In particular it follows that $A_{\mu}(0) = 0$. This property will reveal to be very useful in practical calculations.

To compute the first order correction to the free-quark propagator in Eq. (1.76), we substitute Eq. (1.87) into Eq. (1.77), and choose for simplicity $y^{\mu} = 0$. This gives

$$S_{1}(x,0) = \int d^{4}z \int \frac{d^{4}p}{(2\pi)^{4}} \left(\frac{i(\not p+m)}{p^{2}-m^{2}+i\varepsilon}\right) e^{-ip\cdot(x-z)}$$

$$\times \left(\frac{1}{2}\gamma^{\mu}z^{\rho}G_{\rho\mu} + \frac{1}{3}\gamma^{\mu}z^{\alpha}z^{\rho}D_{\alpha}G^{a}_{\rho\mu}\right)$$

$$\times \int \frac{d^{4}k}{(2\pi)^{4}} \left(\frac{i(\not k+m)}{k^{2}-m^{2}+i\varepsilon}\right) e^{-ik\cdot z} + \dots \qquad (1.88)$$

The functions z^{ρ} and $z^{\alpha}z^{\rho}$ in the second line of Eq. (1.88) can be conveniently rewritten using the identity ¹⁰ $z^{\mu} = i \frac{\partial}{\partial k_{\mu}} e^{-ik \cdot z}$, which leads to

¹⁰Rewriting z^{μ} in terms of a derivative with respect to k_{μ} is the simplest choice. Equivalently one could write $z^{\mu} = -i \frac{\partial}{\partial p_{\mu}} e^{i p \cdot z}$, in this case though, when integrating by parts one would have to differentiate also the function $e^{-i p \cdot x}$.

$$S_{1}(x,0) = \int d^{4}z \int \frac{d^{4}p}{(2\pi)^{4}} \left(\frac{i(\not p+m)}{p^{2}-m^{2}+i\varepsilon}\right) e^{-ip\cdot(x-z)}$$

$$\times \left[\frac{1}{2}\gamma^{\mu}G_{\rho\mu}\int \frac{d^{4}k}{(2\pi)^{4}} \left(\frac{i(\not k+m)}{k^{2}-m^{2}+i\varepsilon}\right) \left(i\frac{\partial}{\partial k_{\rho}}e^{-ik\cdot z}\right)\right]$$

$$+ \frac{1}{3}\gamma^{\mu}D_{\alpha}G_{\rho\mu}\int \frac{d^{4}k}{(2\pi)^{4}} \left(\frac{i(\not k+m)}{k^{2}-m^{2}+i\varepsilon}\right) \left(-\frac{\partial}{\partial k_{\alpha}}\frac{\partial}{\partial k_{\rho}}e^{-ik\cdot z}\right) + \dots \quad (1.89)$$

Performing a single and double integration by parts, respectively in the second and third line of Eq. (1.89), we obtain

$$S_{1}(x,0) = \int d^{4}z \int \frac{d^{4}p}{(2\pi)^{4}} \int \frac{d^{4}k}{(2\pi)^{4}} e^{-i(k-p)\cdot z} e^{-ip\cdot x}$$

$$\times \left(\frac{i(\not\!p+m)}{p^{2}-m^{2}+i\varepsilon}\right) \left[\frac{1}{2}\gamma^{\mu}G_{\rho\mu}\left(-i\frac{\partial}{\partial k_{\rho}}\right)\left(\frac{i(\not\!k+m)}{k^{2}-m^{2}+i\varepsilon}\right)\right]$$

$$-\frac{1}{3}\gamma^{\mu}D_{\alpha}G_{\rho\mu}\frac{\partial}{\partial k_{\rho}}\frac{\partial}{\partial k_{\alpha}}\left(\frac{i(\not\!k+m)}{k^{2}-m^{2}+i\varepsilon}\right)\right] + \dots , \qquad (1.90)$$

also note that we have taken into account that all the boundary terms vanish, which can be easily verified by direct inspection. The integral over the variable z^{μ} in Eq. (1.90), results in a delta function and enforces the momentum conservation $k^{\mu} = p^{\mu}$ when integrating over the variable k^{μ} . Furthermore, the first and second order derivatives of the free-quark propagator in the square brackets of Eq. (1.90), yield

$$\frac{\partial}{\partial p_{\rho}} \frac{(\not p+m)}{p^2 - m^2 + i\varepsilon} = \frac{\gamma^{\rho}}{p^2 - m^2 + i\varepsilon} - \frac{2p^{\rho}(\not p+m)}{(p^2 - m^2 + i\varepsilon)^2}, \qquad (1.91)$$

and

$$\frac{\partial}{\partial p_{\rho}}\frac{\partial}{\partial p_{\alpha}}\frac{(\not\!p+m)}{p^2-m^2+i\varepsilon} = -\frac{2(p^{\alpha}\gamma^{\rho}+p^{\rho}\gamma^{\alpha})}{(p^2-m^2+i\varepsilon)^2} - \frac{2g^{\rho\alpha}(\not\!p+m)}{(p^2-m^2+i\varepsilon)^2} + \frac{8p^{\alpha}p^{\rho}(\not\!p+m)}{(p^2-m^2+i\varepsilon)^3}.$$
(1.92)

By substituting Eqs. (1.91), (1.92), the expression in Eq. (1.90) can be written as

$$S_1(x,0) = \int \frac{d^4p}{(2\pi)^4} e^{-ip \cdot x} \left(g_2^{\rho\mu} G_{\rho\mu} + g_3^{\alpha\rho\mu} D_\alpha G_{\rho\mu} + \dots \right), \qquad (1.93)$$

where for clarity we have introduced the compact notation

$$g_{2}^{\rho\mu} = \frac{i}{2} \left[p_{\eta} \frac{\gamma^{\eta} \gamma^{\mu} \gamma^{\rho}}{(p^{2} - m^{2} + i\varepsilon)^{2}} + m \frac{\gamma^{\mu} \gamma^{\rho}}{(p^{2} - m^{2} + i\varepsilon)^{2}} - 2p_{\eta} p_{\sigma} p^{\rho} \frac{\gamma^{\eta} \gamma^{\mu} \gamma^{\sigma}}{(p^{2} - m^{2} + i\varepsilon)^{3}} - 2m^{2} p^{\rho} \frac{\gamma^{\mu}}{(p^{2} - m^{2} + i\varepsilon)^{3}} \right], \quad (1.94)$$

and

$$g_{3}^{\alpha\rho\mu} = -\frac{2}{3} \frac{1}{(p^{2} - m^{2} + i\varepsilon)^{3}} \left[\left(p_{\eta} p_{\xi} \gamma^{\eta} \gamma^{\mu} \gamma_{\tau} + m \, p_{\xi} \gamma^{\mu} \gamma_{\tau} \right) \left(g^{\xi\alpha} g^{\tau\rho} + g^{\xi\rho} g^{\tau\alpha} \right) \right. \\ \left. + \left(g^{\rho\alpha} - \frac{4p^{\rho} p^{\alpha}}{(p^{2} - m^{2} + i\varepsilon)} \right) \left(p_{\eta} p_{\sigma} \gamma^{\eta} \gamma^{\mu} \gamma^{\sigma} + m^{2} \gamma^{\mu} + 2m \, p^{\mu} \right) \right].$$

$$(1.95)$$

We can simplify Eq. (1.93) by using the tensor decomposition of three gamma matrices, see Eq. (7), together with the antisymmetry of the field strength tensor $G_{\mu\nu}$. A slightly lengthy yet simple algebraic manipulation leads to the final result for the quark propagator

$$S(x,0) = \int \frac{d^4p}{(2\pi)^4} S(p) e^{-ip \cdot x}, \qquad (1.96)$$

with

$$\mathcal{S}(p) = \mathcal{S}_0(p) + \mathcal{S}_1(p) + \dots, \qquad (1.97)$$

and

$$S_{1}(p) = -\frac{m}{2} \frac{G_{\rho\mu}}{(p^{2} - m^{2} + i\varepsilon)^{2}} \sigma^{\rho\mu} + \frac{G_{\sigma\eta}}{(p^{2} - m^{2} + i\varepsilon)^{2}} p^{\sigma} \gamma^{\eta} \gamma^{5} - \frac{2}{3} \frac{p^{\alpha} D_{\alpha} G_{\rho\mu}}{(p^{2} - m^{2} + i\varepsilon)^{3}} \gamma^{\mu} p^{\rho} + \frac{2}{3} \frac{D_{\alpha} G_{\alpha\mu}}{(p^{2} - m^{2} + i\varepsilon)^{3}} \Big[\gamma^{\mu} (p^{2} - m^{2}) - p^{\mu} (\not{p} + 2m) \Big] + 2i \frac{D_{\alpha} \tilde{G}_{\tau\eta}}{(p^{2} - m^{2} + i\varepsilon)^{3}} p^{\alpha} p^{\tau} \gamma^{\eta} \gamma^{5} + \frac{2}{3} m \frac{D_{\alpha} G_{\rho\mu}}{(p^{2} - m^{2} + i\varepsilon)^{3}} \Big(p^{\alpha} \gamma^{\rho} \gamma^{\mu} - p^{\rho} \gamma^{\mu} \gamma^{\alpha} \Big) + \dots$$
(1.98)

Here the ellipsis denote terms with higher order derivatives of the field strength tensor, while the dual field tensor is $\tilde{G}_{\mu\nu} = (1/2)\epsilon_{\mu\nu\rho\sigma}G^{\rho\sigma}$. Eq. (1.98) has a transparent meaning. The interaction with the soft gluon field introduces corrections to the free-quark propagator parametrised by operators of higher dimensions built from the gluon field strength tensor and its covariant derivatives, evaluated at the origin. Each operator of dimension-n is suppressed by n powers of the quark momentum p, with the lowest order contribution being due to the dimension-two operator $G_{\mu\nu}$. In the limit of massless quark, Eq. (1.98) reproduces the result of Ref. [78], apart from the opposite sign in front of the two terms proportional to γ_5 . We trace this back to the different convention used in the Russian literature to define the fifth gamma matrix, namely $\gamma_5 = -i \gamma^0 \gamma^1 \gamma^2 \gamma^3$, cf. Eq. (6). We stress that having fixed the notation and been consistent with it, the computation of any observable using Eq. (1.98) or the expression in Ref. [78] must lead to the very same result. Moreover, it is worthwhile to emphasise that the massless limit should be taken with care. Upon integration over p^{μ} , it is only in the domain $p^2 \sim q^2$, where q denotes a large perturbative scale, that the operator expansion in terms of the external gluon field is legitimate. However, as higher dimensional operators are considered and the power of the momentum variable in the denominator increases, the integral in Eq. (1.96) starts to be sensitive also to the long-distance region $p^2 = 0$ and to develop infrared (IR) divergences. In this case, the corresponding quark line becomes soft and the effect must be parametrised in terms of quark operators, see for details Ref. [78] or Chapter 3.

Because of Eq. (1.80), the expression obtained in Eq. (1.98) is valid only in the specific reference frame chosen, namely $y^{\mu} = 0$. In order to compute $S_1(0, y)$ we must repeat the calculation and set $x^{\mu} = 0$. In this case substituting Eq. (1.87) into Eq. (1.77) gives

$$S_{1}(0,y) = \int d^{4}z \int \frac{d^{4}p}{(2\pi)^{4}} \left(\frac{i(\not p+m)}{p^{2}-m^{2}+i\varepsilon}\right) e^{ip\cdot z}$$

$$\times \left(\frac{1}{2}\gamma^{\mu}z^{\rho}G_{\rho\mu} + \frac{1}{3}\gamma^{\mu}z^{\alpha}z^{\rho}D_{\alpha}G_{\rho\mu}^{a}\right)$$

$$\times \int \frac{d^{4}k}{(2\pi)^{4}} \left(\frac{i(\not k+m)}{k^{2}-m^{2}+i\varepsilon}\right) e^{-ik\cdot(z-y)} + \dots, \qquad (1.99)$$

where now it is convenient to rewrite z^{ρ} and $z^{\alpha}z^{\rho}$ in the second line of Eq. (1.99) using the identity $z^{\mu} = -i\frac{\partial}{\partial p_{\mu}}e^{ip\cdot z}$. The next intermediate steps proceed in analogy to the case of $S_1(x,0)$, for brevity we omit them and state here only the final result, which reads

$$S(0,y) = \int \frac{d^4p}{(2\pi)^4} \,\tilde{\mathcal{S}}(p) \,e^{ip \cdot y} \,, \qquad (1.100)$$

with

$$\tilde{\mathcal{S}}(p) = \mathcal{S}_0(p) + \tilde{\mathcal{S}}_1(p) + \dots, \qquad (1.101)$$

and

$$\tilde{\mathcal{S}}_{1}(p) = -\frac{m}{2} \frac{G_{\rho\mu}}{(p^{2} - m^{2} + i\varepsilon)^{2}} \sigma^{\rho\mu} + \frac{\tilde{G}_{\sigma\eta}}{(p^{2} - m^{2} + i\varepsilon)^{2}} p^{\sigma} \gamma^{\eta} \gamma^{5} - \frac{2}{3} \frac{p^{\alpha} D_{\alpha} G_{\rho\mu}}{(p^{2} - m^{2} + i\varepsilon)^{3}} \gamma^{\mu} p^{\rho} + \frac{2}{3} \frac{D_{\alpha} G_{\alpha\mu}}{(p^{2} - m^{2} + i\varepsilon)^{3}} \left[\gamma^{\mu} (p^{2} - m^{2}) - p^{\mu} p \right] - 2i \frac{D_{\alpha} \tilde{G}_{\tau\eta}}{(p^{2} - m^{2} + i\varepsilon)^{3}} p^{\alpha} p^{\tau} \gamma^{\eta} \gamma^{5} - \frac{2}{3} m \frac{D_{\alpha} G_{\rho\mu}}{(p^{2} - m^{2} + i\varepsilon)^{3}} \left(p^{\alpha} \gamma^{\rho} \gamma^{\mu} - p^{\rho} \gamma^{\mu} \gamma^{\alpha} \right) + \dots$$
(1.102)

The absence of translation symmetry Eq. (1.80), is then reflected in momentum space by the condition $\tilde{S}_1(p) \neq S_1(p)$, which actually holds true only starting from the operator of dimension-three $D_{\rho}G_{\mu\nu}$. In fact a comparison between Eq. (1.98) and Eq. (1.102), shows that the translation invariance is still preserved in the coefficients of the dimension-two operator $G_{\mu\nu}$.

For completeness, it is instructive to derive also an explicit representation of the quark propagator in coordinate space, by performing the anti-Fourier transform of the expressions in Eqs. (1.98), (1.102). These contain both scalar and tensor functions of the variable p^{μ} . However, it actually suffices to directly evaluate only the scalar integrals. In the case of S(x, 0) in Eq. (1.96), they have the following form

$$I_n(x) = \int \frac{d^4p}{(2\pi)^4} \frac{1}{\left(p^2 - m^2 + i\varepsilon\right)^n} e^{-ip \cdot x}, \qquad n \in \mathbb{N}.$$
(1.103)

Given the analytic expression of $I_n(x)$, it is possible to obtain the tensor integrals appearing in Eq. (1.96), by differentiating Eq. (1.103) with respect to x_{μ} , namely

$$I_n^{\mu_1\dots\mu_{n'}}(x) = \int \frac{d^4p}{(2\pi)^4} \frac{p^{\mu_1}\dots p^{\mu_{n'}}}{\left(p^2 - m^2 + i\varepsilon\right)^n} e^{-ip\cdot x} = \prod_{j=1}^{n'} \left(i\frac{d}{dx_{\mu_j}}\right) I_n(x), \qquad (1.104)$$

with $n' \leq n^{-11}$. The integral in Eq. (1.103) can be conveniently computed in Euclidean space by performing the Wick rotations $p_0 \rightarrow -ip_4$ and $x_0 \rightarrow -ix_4$, see e.g. the textbook [81] or the lecture notes [82]. This gives

$$I_n(x_E) = i \, (-1)^n \int \frac{d^4 p_E}{(2\pi)^4} \frac{1}{\left(p_E^2 + m^2\right)^n} \, e^{i p_E \cdot x_E} \,. \tag{1.105}$$

The Euclidean four-vectors are defined as $p_E^{\mu} = (\mathbf{p}, p_4)$ and $x_E^{\mu} = (\mathbf{x}, x_4)$, while the Euclidean metric reads diag(1, 1, 1, 1). The factor of $(-1)^n$ in Eq. (1.105), follows from the fact that $(p^2 - m^2) \rightarrow -(p_E^2 + m^2)$, also note that we have dropped the $+i\varepsilon$ prescription since the denominator is now positive definite. The latter can be suitably expressed in an integral form. To this end, we start by writing $\Gamma(s) = \{\mathcal{M}e^{-t}\}(s)$, where $\{\mathcal{M}g(t)\}$ denotes the Mellin transform of g(t) and $\Gamma(s)$ is the gamma function, see Ref. [83] for exhaustive tables with definitions and useful properties, i.e.

$$\Gamma(s) = \int_0^\infty dt \, t^{s-1} e^{-t} \,, \qquad \text{Re}\, s > 0 \,. \tag{1.106}$$

Performing the change of variable $t \to \lambda t$ with $\lambda > 0$, yields $\lambda^{-s} \Gamma(s) = \{\mathcal{M}e^{-\lambda t}\}(s)$, and after setting $\lambda = (p_E^2 + m^2)$ and s = n, we readily obtain that

$$\frac{1}{(p_E^2 + m^2)^n} = \frac{1}{\Gamma(n)} \int_0^\infty dt \, t^{n-1} e^{-t(p_E^2 + m^2)} \,. \tag{1.107}$$

¹¹Note that for $n \ge 2$ only n-1 powers of the four-momentum can appear in the numerator. This follows from the fact that S(p) must have mass dimension of -1 and that for each gluon operator of dimension n there are 2n powers of the momentum in the denominator.

Equivalently, Eq. (1.107) can be derived by taking into account the identity

$$\frac{1}{(p_E^2 + m^2)} = \int_0^\infty dt \, e^{-t(p_E^2 + m^2)} \,, \tag{1.108}$$

easily proved by directly computing the integral on the r.h.s. The result for n > 1 follows from differentiating n - 1 times both sides of Eq. (1.108) with respect to the parameter m^2 and using that $\Gamma(n) = (n - 1)!$. Substituting Eq. (1.107) into Eq. (1.105), we then arrive at

$$I_n(x_E) = i \, (-1)^n \int_0^\infty dt \ t^{n-1} e^{-tm^2} \int \frac{d^4 p_E}{(2\pi)^4} e^{-tp_E^2 + ip_E \cdot x_E} \,. \tag{1.109}$$

The second integral on the r.h.s. of Eq. (1.109) reduces to a standard four-dimensional Gaussian integral, see e.g. Ref [84], after shifting the integration variable by a constant Euclidean four-vector i.e. $p_E^{\mu} \rightarrow p_E^{\mu} - i x_E^{\mu}/(2t)$. The solution reads

$$\int d^4 p_E \, e^{-tp_E^2 + ip_E \cdot x_E} = \left(\frac{\pi}{t}\right)^2 e^{-\frac{x_E^2}{4t}},\tag{1.110}$$

which we insert into Eq. (1.109) to obtain

$$I_n(x_E) = i \frac{(-1)^n}{16\pi^2} \int_0^\infty dt \ t^{n-3} e^{-tm^2 - \frac{x_E^2}{4t}}.$$
 (1.111)

It is easy to show that Eq. (1.111) can be expressed in terms of the modified Bessel functions of the second type $K_{\nu}(z)$. Starting with the integral representation, see e.g. Ref. [83]

$$K_{\nu}(z) = \frac{1}{2} \left(\frac{z}{2}\right)^{\nu} \int_{0}^{\infty} dt \, t^{-\nu-1} e^{-t - \frac{z^{2}}{4t}}, \qquad |\arg z| < \frac{\pi}{2}, \ \operatorname{Re} z^{2} > 0, \qquad (1.112)$$

and performing the change of variables $t \to \alpha t, z^2 \to \alpha z^2$, with $\alpha > 0$, gives

$$\int_{0}^{\infty} dt \ t^{-\nu-1} e^{-\alpha t - \frac{z^{2}}{4t}} = 2\left(\frac{4a}{z^{2}}\right)^{\frac{\nu}{2}} K_{\nu}\left(\sqrt{a}z\right) \ . \tag{1.113}$$

From Eq. (1.113), it then follows that

$$I_n(x_E) = i \frac{(-1)^n}{8\pi^2} \left(\frac{4m^2}{x_E^2}\right)^{\frac{2-n}{2}} K_{2-n}\left(m\sqrt{x_E^2}\right), \qquad (1.114)$$

and the condition $\operatorname{Re}(mx_E^2) > 0$ translates into $x_E^2 > 0$ for positive and real values of *m* as well as real Euclidean distances. We can now rotate back to the Minkowski space-time and arrive at the final expression

$$I_n(x) = i \, \frac{(-1)^n}{8\pi^2} \left(\frac{4m^2}{-x^2}\right)^{\frac{2-n}{2}} K_{2-n}\left(m\sqrt{-x^2}\right) \,, \tag{1.115}$$

which is valid for space-like separations $x^2 < 0$. The solution corresponding to the space-time region $x^2 > 0$ can be derived from Eq. (1.115) by analytic continuation, taking into account, see e.g. Ref. [83], that

$$K_{\nu}(iz) = \frac{-\pi i}{2} e^{-\frac{\pi}{2}\nu i} H^{(2)}_{-\nu}(z) , \qquad (1.116)$$

where $H_{\nu}^{(2)}(z)$ denotes the Hankel's function of the second kind. Finally, recall that the Bessel functions $K_{\nu}(z)$ satisfy the following recursive relation

$$\frac{d}{dz}K_{\nu}(z) = \frac{\nu}{z}K_{\nu}(z) - K_{\nu+1}(z), \qquad (1.117)$$

that allows to compute the tensor integrals appearing in Eq. (1.96) according to Eq. (1.104). Let us consider explicitly the case n = 1. Eqs. (1.115), (1.104) then read

$$I_1(x) = -\frac{i}{4\pi^2} \frac{m}{\sqrt{-x^2}} K_1\left(m\sqrt{-x^2}\right) , \qquad (1.118)$$

and

$$I_1^{\mu}(x) = -\frac{1}{4\pi^2} \frac{m^2}{x^2} K_2\left(m\sqrt{-x^2}\right) x^{\mu}, \qquad (1.119)$$

from which we can readily derive the expression of the anti-Fourier transform of the free-quark propagator $S_0(p)$, namely

$$S_0(x) = -\frac{1}{4\pi^2} \frac{m^2}{x^2} K_2\left(m\sqrt{-x^2}\right) \not x - \frac{i}{4\pi^2} \frac{m^2}{\sqrt{-x^2}} K_1\left(m\sqrt{-x^2}\right) .$$
(1.120)

Proceeding in a similar way for the remaining cases n = 2, 3, it is straightforward to verify that the coordinate representation of the first order correction in Eq. (1.98), has the following form

$$S_{1}(x,0) = -\frac{\hat{G}_{\alpha\beta}}{8\pi^{2}}x^{\alpha}\gamma^{\beta}\gamma_{5}\frac{mK_{1}(m\sqrt{-x^{2}})}{\sqrt{-x^{2}}} - i\frac{G_{\alpha\beta}}{16\pi^{2}}\sigma^{\alpha\beta}mK_{0}(m\sqrt{-x^{2}}) + i\frac{D_{\alpha}G^{\alpha\beta}}{24\pi^{2}}\gamma_{\beta}K_{0}(m\sqrt{-x^{2}}) - i\frac{D_{\alpha}G^{\alpha\beta}}{48\pi^{2}}x_{\beta}\not{x}\frac{mK_{1}(m\sqrt{-x^{2}})}{\sqrt{-x^{2}}} - \frac{D_{\alpha}G^{\alpha\beta}}{24\pi^{2}}x_{\beta}mK_{0}(m\sqrt{-x^{2}}) - i\frac{D^{\alpha}G^{\beta\rho}}{48\pi^{2}}\gamma_{\rho}x_{\alpha}x_{\beta}\frac{mK_{1}(m\sqrt{-x^{2}})}{\sqrt{-x^{2}}} - \frac{D^{\alpha}\tilde{G}^{\beta\rho}}{16\pi^{2}}\gamma_{\rho}\gamma_{5}x_{\alpha}x_{\beta}\frac{mK_{1}(m\sqrt{-x^{2}})}{\sqrt{-x^{2}}} - \frac{D^{\alpha}G^{\beta\rho}}{48\pi^{2}}\gamma_{\rho}\gamma_{\alpha}x_{\beta}mK_{0}(m\sqrt{-x^{2}}) - \frac{D^{\alpha}G^{\beta\rho}}{48\pi^{2}}\gamma_{\rho}\gamma_{\beta}x_{\alpha}mK_{0}(m\sqrt{-x^{2}}) + \dots$$
(1.121)

The result in Eq. (1.121) was first derived in Ref. [85] up to terms proportional to $G_{\mu\nu}$, while the contribution of $D_{\rho}G_{\mu\nu}$ can be found in Ref. [86] ¹². In order to compute the corresponding expression for S(0, y) we can make the replacement $x^{\mu} \rightarrow -y^{\mu}$ in Eqs. (1.103), (1.104). Notice that because of Eq. (1.115), the function $I_n(x)$ is even i.e. $I_n(-x) = I_n(x)$ and from this we obtain that

$$I_n^{\mu_1\dots\mu_n}(-y) = (-1)^n I_n^{\mu_1\dots\mu_n}(y) \,. \tag{1.122}$$

The above relation allows to immediately write

$$S_0(-y) = \frac{1}{4\pi^2} \frac{m^2}{y^2} K_2\left(m\sqrt{-y^2}\right) \mathcal{Y} - \frac{i}{4\pi^2} \frac{m^2}{\sqrt{-y^2}} K_1\left(m\sqrt{-y^2}\right) , \qquad (1.123)$$

consistently with the fact that free-quark propagator is translation invariant. Moreover it is easy to check that the first order correction now reads

 $^{^{12}}$ Note that it is presented only the expression relevant for the computation described in the paper, namely with an odd number of gamma-matrices.

which clearly indicates that the translation symmetry is firstly broken in the coefficients of the dimension-three operator $D_{\rho}G_{\mu\nu}$, due to the fact that some of the quadratic functions in Eqs. (1.121), (1.124), appear with a negative relative sign.

We conclude this section with a final remark about the dependence of the quark propagator on the mass parameter m. Following the comment made on the massless limit of Eq. (1.98), the expressions in Eqs. (1.121), (1.124), become divergent for $m \rightarrow 0$. In coordinate space the singularity derives from the asymptotic behaviour of the Bessel functions in the limit of small argument, see e.g. Ref. [83], namely

$$K_{\nu}(z) \sim \begin{cases} -\log\left(\frac{z}{2}\right) - \gamma_E, & \nu = 0, \\ \frac{\Gamma(\nu)}{2} \left(\frac{2}{z}\right)^{\nu}, & \nu > 0. \end{cases}$$
(1.125)

Notice that in the specific case of Eqs. (1.121), (1.124), the divergence originates from the dimension-three contribution $D_{\alpha}G^{\alpha\beta}\gamma_{\beta}K_0(m\sqrt{-x^2})$.

1.4 The heavy quark expansion

The total decay width Γ or equivalently its inverse, the total lifetime $\tau = \Gamma^{-1}$, defines one of the fundamental properties of elementary and composite particles and hence constitutes an observable of phenomenological primary importance. In the study of lifetimes a special role is occupied by heavy hadrons, due to the interplay that strong and weak interactions have in determining their decay, see e.g. Ref. [87]. As discussed already in Section 1.2, a heavy flavour hadron H_Q is a QCD bound state that can be conveniently represented as heavy quark Q surrounded by a cloud of light quarks, antiquarks and gluons, where the distinction between heavy and light

degrees of freedom is meant with respect to the typical hadronic scale Λ_{QCD} . The description of these systems considerably simplifies by considering the limit of an infinitely heavy quark Q, i.e. $m_Q \to \infty$. In this case it is possible to neglect the effect of the non perturbative QCD interactions and the hadron dynamics results entirely determined by that of a free quark Q, with the light constituents reducing to passive spectators, see e.g. Ref. [88]. This approximation leads to the result $\Gamma(H_Q) = \Gamma_Q$, and then to the theoretical prediction that the lifetimes of hadrons containing the same heavy flavour but different spectator quarks should be equal. While among bottom hadrons this statement can be experimentally accommodated within deviations of few percent [22], the pattern in the charm family is far less monotonous and lifetime ratios of charmed hadrons can be as large as 7 $[89, 90]^{-13}$. The infinite mass limit is clearly not sufficient for a proper interpretation of the experimental data, particularly in the charm sector where deviations from the free quark decay approximation are expected to give the dominant contribution. Before discussing how the corrections to this limit can be systematically taken into account, in what is the current theoretical framework for the study of inclusive decays of heavy hadrons, it is instructive to briefly retrace the main developments that have brought to its construction. In this respect we refer to the comprehensive review [27].

The possibility to describe the decay of a heavy hadron in terms of an asymptotically free constituent heavy quark was first exploited in 1973 by Nikolaev in Ref. [91], where constraints on the decay properties of a 'supercharged' (charmed) hadron, which at the time was only a theoretical particle, were proposed as an indirect test of its existence. Furthermore, after their discovery, the description of charmed hadrons decays was initially performed by considering only the dominant partonic contribution see e.g. Refs. [92–95]. A pioneering work for the study of the lifetime of heavy hadrons is the one of Shifman and Voloshin, Ref. [87], where many of the ingredients that contributed to the formulation of the heavy quark expansion (HQE) were originally presented. The HQE is a theoretical framework in which inclusive decays of heavy hadrons can be computed in terms of an operator product expansion (OPE) [45, 46, 96], by exploiting the large scale hierarchy $m_Q \gg \Lambda_{QCD}$. This method was firstly applied in a systematic way by Chay, Georgi and Grinstein for the analysis of inclusive semileptonic decays of heavy hadrons in Ref. [97], and briefly after employed in the work of Bigi, Shifman, Uraltsev and Vainshtein [98,99] and of Blok, Koyrakh, Shifman and Vainshtein [100], for the description of inclusive non-leptonic as well as semileptonic decays, see e.g. Ref. [101].

In order to discuss the construction of the HQE, it is convenient to review some

¹³More precisely, the lifetime ratio between the longest and the shortest living *b*-hadrons, decaying weakly and containing only one heavy quark, is $\tau(\Omega_b)/\tau(\Lambda_b) = 1.11$, to be compared with $\tau(\Xi_c^0)/\tau(D^+) = 6.8$, in the charm sector [22, 89, 90].

important results in scattering theory, focusing in particular on the case of particle decay. We recall that the scattering process is described by introducing a unitary operator S, which governs the evolution of an asymptotically free state $|i\rangle$ into an asymptotically free state $|f\rangle$ for a given interaction theory, where the information about the non trivial part of the dynamics is encoded in the action of a transition operator T. The S matrix is then decomposed into

$$S_{fi} \equiv \langle f|S|i\rangle = \delta_{fi} + i T_{fi}. \qquad (1.126)$$

By taking into account that the interaction must conserve four-momentum, the transition amplitude T_{fi} can be further parametrised as

$$T_{fi} = (2\pi)^4 \delta^{(4)}(p_f - p_i) \mathcal{M}_{fi}, \qquad (1.127)$$

where p_i^{μ} , p_f^{μ} label respectively the momentum of the initial and final states and \mathcal{M}_{fi} denotes the invariant scattering amplitude. In the case that $|i\rangle$ contains only one particle, say A, the total decay width $\Gamma(A)$, is obtained by computing the amplitude squared for the process $A \to n$, where $|n\rangle$ represents an allowed *n*-particle final state, by summing over all the possible values of n and finally by accounting for the flux factor $2m_A$, namely, see e.g. the textbook [38]

$$\Gamma(A) = \frac{1}{2m_A} \sum_n \int_n (2\pi)^4 \delta^{(4)} \left(\sum_{j=1}^n p_j - p_A \right) \left| \langle n | \mathcal{M} | A \rangle \right|^2 , \qquad (1.128)$$

here

$$\int_{n} \equiv \int \prod_{j=1}^{n} \frac{d^{3} p_{j}}{(2\pi)^{3} 2E_{j}}, \qquad (1.129)$$

denotes the integration over the Lorentz invariant *n*-particle phase space while the presence of the delta function ensures that four-momentum is conserved in each decay. An equivalent and, for practical calculations, more advantageous representation of Eq. (1.128), can be obtained by employing the optical theorem, see e.g. the textbook [81], which states that in a given interaction theory, the imaginary part of the forward scattering amplitude is proportional to the total cross section for the production of all final states. The optical theorem follows from the unitarity of the scattering operator S and hence from the mathematical requirement for the conservation of probability in Quantum Field Theory. The unitarity condition reads



Figure 1.6: Schematic representation of the optical theorem: the amplitude squared for the production of all final states is proportional to the absorptive part of the forward scattering amplitude.

$$S^{\dagger}S = \mathbb{I} = SS^{\dagger} . \tag{1.130}$$

Considering the matrix element between the states $|i\rangle$, $|f\rangle$, and using the completeness relation $\sum_{n}' |n\rangle\langle n| = \mathbb{I}$, where $\sum_{n}' \equiv \sum_{n} \int_{n}$ implies the sum over all the particles in $|n\rangle$ as well as the integration over their momenta, cf. Eq. (1.129), the first equality of Eq. (1.130) can be recast as

$$\sum_{n}^{\prime} \langle f|S^{\dagger}|n\rangle \langle n|S|i\rangle = \delta_{fi} \,. \tag{1.131}$$

In the special case of forward scattering i.e. $|f\rangle = |i\rangle$, taking into account that $\langle i|S^{\dagger}|n\rangle = \langle n|S|i\rangle^{\dagger} = \langle n|S|i\rangle^{*}$, from Eqs. (1.126), (1.131), it follows that

$$\sum_{n}^{\prime} (\delta_{ni} - i T_{ni}^{*}) (\delta_{ni} + i T_{ni}) = \delta_{ii} , \qquad (1.132)$$

and expanding the l.h.s. of Eq. (1.132), readily yields

$$i(T_{ii} - T_{ii}^*) + \sum_{n}' T_{ni}^* T_{ni} = 0, \qquad (1.133)$$

or equivalently

$$2 \operatorname{Im} T_{ii} = \sum_{n}^{\prime} |T_{ni}|^2.$$
 (1.134)

Finally we can rewrite Eq. (1.134) by substituting Eq. (1.127) on both sides and by using $(\delta^{(4)}(z))^2 = \delta^{(4)}(0)\delta^{(4)}(z)$ to evaluate the square of the delta function, namely

$$2 \operatorname{Im} \mathcal{M}_{ii} = \sum_{n} \int_{n} (2\pi)^{4} \delta^{(4)} \left(\sum_{j=1}^{n} p_{j} - p_{i} \right) |\mathcal{M}_{ni}|^{2}, \qquad (1.135)$$

which corresponds to the standard formulation of the optical theorem, schematically sketched in Figure 1.6. Setting $|i\rangle = |A\rangle$, a comparison between Eq. (1.135) and Eq. (1.128), evidently gives

$$\Gamma(A) = \frac{1}{m_A} \operatorname{Im} \mathcal{M}_{AA}, \qquad (1.136)$$

showing that the total decay width $\Gamma(A)$ can be obtained by computing the imaginary part of the forward scattering amplitude $A \to A$.

We now apply the result in Eq. (1.136) to the decay of a heavy quark Q^{14} . We assume that at the renormalisation scale $\mu = m_Q$ the weak interaction is described by an effective Hamiltonian $\mathcal{H}_{eff}(x)$, governing the transition of the heavy quark into all possible lighter fermions, see Section 1.1, so that the scattering operator Scan be written as ¹⁵

$$S = \operatorname{Texp}\left\{-i\int d^4x \,\mathcal{H}_{eff}(x)\right\}\,,\qquad(1.137)$$

where T is the time-ordering operator. The first non vanishing contribution to the forward scattering amplitude \mathcal{M}_{QQ} , is obtained by expanding S to second order in the weak effective coupling. Up to terms of higher order this gives

$$T_{QQ} = \frac{1}{2} \langle Q | i \int d^4x \int d^4y \, \mathrm{T} \Big\{ \mathcal{H}_{eff}(x), \mathcal{H}_{eff}(y) \Big\} | Q \rangle.$$
(1.138)

Using that the translation invariance of the Hamiltonian operator implies

$$\mathcal{H}_{eff}(x) = e^{i\hat{P}\cdot x} \,\mathcal{H}_{eff}(0) \, e^{-i\hat{P}\cdot x} \,, \tag{1.139}$$

with $\hat{P}_{\mu} = \partial/\partial x_{\mu}$, and that $|Q\rangle$ corresponds to a state with definite momentum p_Q^{μ} , namely

$$e^{-i\hat{P}\cdot x}|Q\rangle = e^{-ip_Q\cdot x}|Q\rangle, \qquad (1.140)$$

¹⁴The choice of a heavy quark is just for future convenience, the same description applies, taking into account the proper replacements, to the weak decay of any elementary fermion.

¹⁵It is worthwhile to emphasise that \mathcal{H}_{eff} must be intended as supplemented with the QCD as well as with the QED Lagrangian, responsible for higher order corrections to the leading weak decay.

Eq. (1.138) can be simplified as, see for a similar derivation the textbook [23]

$$T_{QQ} = \frac{1}{2} \langle Q | i \int d^4x \int d^4y \,\mathrm{T} \Big\{ \underbrace{e^{-i\hat{P}\cdot y} \,\mathcal{H}_{eff}(x) \,e^{i\hat{P}\cdot y}}_{\mathcal{H}_{eff}(x-y)}, \mathcal{H}_{eff}(0) \Big\} | Q \rangle$$
$$= \frac{1}{2} (2\pi)^4 \delta^{(4)}(0) \langle Q | i \int d^4x \,\mathrm{T} \Big\{ \mathcal{H}_{eff}(x), \mathcal{H}_{eff}(0) \Big\} | Q \rangle, \qquad (1.141)$$

where in the last step we have performed the change of variable $x^{\mu} - y^{\mu} \rightarrow x^{\mu}$ under the integration over x^{μ} and used that $\int d^4y = (2\pi)^4 \delta^{(4)}(0)$. From Eqs. (1.127), (1.136), we finally obtain that

$$\Gamma(Q) = \frac{1}{2m_Q} \operatorname{Im} \langle Q | \mathcal{T} | Q \rangle, \qquad (1.142)$$

with

$$\mathcal{T} = i \int d^4 x \, \mathrm{T} \Big\{ \mathcal{H}_{eff}(x) \,, \mathcal{H}_{eff}(0) \Big\} \,. \tag{1.143}$$

We can interpret Eqs. (1.142), (1.143), as the statement that, due to the optical theorem, the total decay width of Q is proportional to the amplitude for the process $Q \to X \to Q$, describing the forward scattering of Q via the production and annihilation of all the possible intermediate states X. This corresponds to computing the imaginary part of the time ordered product of the Hamiltonian operator evaluated at two different space-time points, namely the non local operator \mathcal{T} , and to determining its expectation value between external $|Q\rangle$ states.

An analogous description for the decay of a hadronic state is plagued by the presence of the non perturbative QCD effects responsible for the confinement dynamics. However, it was first proposed by Shifman and Voloshin in Ref. [87], that the inclusive decay width of a heavy hadron, in their specific case a charmed meson, in the assumption of an infinitely heavy constituent quark, could be obtained using the partonic description, and hence by computing the probability for the free heavy quark to decay into all the lighter fermions. Following their formulation, the total decay width of a heavy hadron H_Q can be expressed as the imaginary part of the non local operator \mathcal{T} in Eq. (1.143), evaluated between external hadronic states, namely

$$\Gamma(H_Q) = \frac{1}{2m_{H_Q}} \operatorname{Im} \langle H_Q | \mathcal{T} | H_Q \rangle.$$
(1.144)

When $m_Q \to \infty$, the hadronic state and mass coincide with those of the heavy quark and Eq. (1.144) becomes $\Gamma(H_Q) = \Gamma(Q)$. However, this approximation is not sufficient for phenomenological applications and corrections to this limit must be systematically included. The HQE provides a theoretical framework to compute $\Gamma(H_Q)$ in Eq. (1.143), in the case of large, but finite, heavy quark mass m_Q . The fundamental assumption is that, inside a heavy hadron, the heavy quark, propagating in the soft background generated by the non perturbative gluon field, interacts with the light degrees of freedom exchanging momenta of the order of Λ_{QCD} , much smaller than m_Q , meaning that there is a large part in the heavy quark momentum, which is proportional to the heavy quark mass and that can be extracted by means of a field redefinition, see e.g. Ref. [102], i.e.

$$Q(x) = e^{-im_Q v \cdot x} Q_v(x), \qquad (1.145)$$

where v^{μ} denotes the hadron velocity. It is worth emphasising that despite the strong analogy, $Q_v(x)$ in Eq. (1.145) constitutes a rescaled four-component QCD field and not the two-component non relativistic field introduced in the context of the HQET, cf. Section 1.2, more details on the difference between the two methods can be found e.g. in Ref. [71]. From Eq. (1.145) it then follows that

$$iD_{\mu}Q(x) = e^{-im_{Q}v \cdot x} \Big(m_{Q}v_{\mu} + iD_{\mu} \Big) Q_{v}(x) , \qquad (1.146)$$

which combined with the equation of motion $(i\not D - m_Q)Q(x) = 0$, gives

$$P_{+}Q_{v}(x) = Q_{v}(x) - \frac{iD}{2m_{Q}}Q_{v}(x), \qquad (1.147)$$

and

$$P_{-}Q_{v}(x) = \frac{iD}{2m_{Q}}Q_{v}(x), \qquad (1.148)$$

with the projector operators P_{\pm} defined as in Eq. (1.37). Moreover, acting with P_{+} on both sides of Eq. (1.148) and using that $P_{+}i\not D = i\not D P_{-} + (iv \cdot D)$, yields

$$(iv \cdot D)Q_v(x) = -\frac{1}{2m_Q}i\not\!\!D i\not\!\!D Q_v(x).$$
(1.149)

The relations in Eqs. (1.145)-(1.149), allow to construct a systematic procedure to

compute the inclusive decay width $\Gamma(H_Q)$. Specifically, Eq. (1.144) can be evaluated in two steps, see e.g. Ref. [103]. First, by taking into account the soft interaction with the background gluon field as well as with the light spectator quarks, the imaginary part of the non local second order operator \mathcal{T} is expanded in a series of local operators \mathcal{O}_d with increasing dimension d, where the corresponding coefficients c_d are suppressed by d-3 powers of the heavy quark mass m_Q , namely

$$\operatorname{Im} \mathcal{T} = \sum_{d} c_d \, \frac{\mathcal{O}_d}{m_Q^{d-3}} \,. \tag{1.150}$$

In general all possible Lorentz and gauge invariant operators, bilinear in the heavy quark field, can appear on the r.h.s of Eq. (1.150) and for large values of m_Q , it is sufficient to consider only those of lowest dimension. These respectively are $\bar{Q}Q$, $(1/2)\bar{Q}\sigma_{\mu\nu}G^{\mu\nu}Q$, $\bar{Q}\Gamma q\bar{q}\Gamma Q$, etc., where Γ denotes a combination of gamma matrices and colour matrices. The corresponding coefficients in Eq. (1.150) are extracted by taking the matrix element of both sides of Eq. (1.150) between external quark and gluon states, see e.g. Ref. [71]. Notice that there is no dimension-four operator, since $\bar{Q}DQ$ can be reduced to $\bar{Q}Q$ by means of the equation of motion for Q [103]. The series in Eq. (1.150) starts at dimension-three with the operator $\bar{Q}Q = \bar{Q}_vQ_v$. This is not suppressed by the heavy quark mass and at leading order in $1/m_Q$ it reproduces the partonic result in Eq. (1.142). In fact, $\bar{Q}Q$ receives non perturbative corrections from higher order operators, see Ref. [98]. The proof starts with the following identity

$$\bar{Q}Q = \bar{Q}_v \psi Q_v + 2\bar{Q}_v P_- Q_v = \bar{Q}_v \psi Q_v + 2\bar{Q}_v P_- P_- Q_v , \qquad (1.151)$$

which, using Eq. (1.148) together with $\bar{Q}_v P_- = \bar{Q}_v (-i\not\!\!\!D)/2m_Q$, leads to

$$\bar{Q}Q = \bar{Q}_v \psi Q_v - 2\bar{Q}_v \frac{i\vec{\not{D}}}{2m_Q} \frac{i\vec{\not{D}}}{2m_Q} Q_v = \bar{Q}_v \psi Q_v + \bar{Q}_v \frac{(i\vec{\not{D}})^2}{2m_Q^2} Q_v + \text{total derivative}, \quad (1.152)$$

where the contribution of the total derivative can be neglected since, in forward matrix elements with zero momentum transfer, it vanishes, see e.g. Ref. [104]. The first operator on the r.h.s. of Eq. (1.152) is the generator of the conserved charge associated to the heavy flavour Q, its matrix element between external hadronic states is one, up to a normalisation factor [98]. Note that in Eq. (1.152) there are no linear terms in $1/m_Q$. These would be generated by operators of dimension-four, however, containing only one covariant derivative, they would either correspond to a total derivate, which does not contribute, as stated above, or to a derivative acting on the heavy quark field, which, by means of the equation of motion Eq. (1.149), is proportional to operators of higher order. The absence in the HQE of linear terms in $1/m_Q$ was first discussed by Chay, Georgi and Grinstein in Ref. [97], and subsequently by Bigi, Uraltsev, and Vainshtein in Ref. [98] and is known as CGG/BUV theorem, see Ref. [71] ¹⁶. First corrections to the infinite mass limit arise at dimension-five, and correspond to operators with two covariant derivatives acting on the heavy quark field. We can identify them from Eq. (1.152), i.e.

$$\bar{Q}_v \frac{(iD)^2}{2m_Q^2} Q_v = \frac{1}{2m_Q^2} \bar{Q}_v (iD_\mu)(iD^\mu) Q_v + \frac{1}{2m_Q^2} \bar{Q}_v (iD_\mu)(iD_\nu)(-i\sigma^{\mu\nu}) Q_v , \quad (1.153)$$

where we have used that $\gamma^{\mu}\gamma^{\nu} = \{\gamma^{\mu}, \gamma^{\nu}\}/2 + [\gamma^{\mu}, \gamma^{\nu}]/2$. The kinetic and chromomagnetic operators are then defined respectively as

$$\mathcal{O}_{kin} = \bar{Q}_v (iD_\mu) (iD^\mu) Q_v , \qquad (1.154)$$

$$\mathcal{O}_{mag} = \bar{Q}_v(iD_\mu)(iD_\nu)(-i\sigma^{\mu\nu})Q_v\,. \tag{1.155}$$

At dimension-six, operators generated from the action of three covariant derivatives, but also four-quark operators, contribute. The former correspond to the spin-orbit and Darwin operators, defined respectively as, see e.g. Ref. [105]

$$\mathcal{O}_{LS} = \bar{Q}_v (iD_\mu) (iv \cdot D) (iD_\nu) (-i\sigma^{\mu\nu}) Q_v , \qquad (1.156)$$

$$\mathcal{O}_{\rho_D} = \bar{Q}_v (iD_\mu) (iv \cdot D) (iD_\nu) Q_v \,. \tag{1.157}$$

Four-quark operators have the schematic form $\bar{Q}\Gamma q\bar{q}\Gamma Q$, where Γ refers to a combination of gamma matrices as well as colour matrices, compatible with the V - Astructure of the effective Hamiltonian, and q denotes a light spectator quark. It is worth mentioning that, using the equation of motion for the gluon field $D_{\mu}G^{\mu\nu a} =$ $-g_s \sum_q \bar{q}\gamma^{\nu}t^a q$, the Darwin operator can be expressed, at leading order in $1/m_Q$, in terms of four-quark operators, see e.g. Ref. [98] and also Chapter 4. Finally, operators of higher dimension are built by further expanding in the number of covariant derivatives and of light quark fields.

¹⁶In the framework of the HQET, the absence of linear terms in $1/m_Q$ to the forward matrix element of a heavy quark current is known as Luke's theorem [63], see e.g. Ref. [66].

Having constructed the series in Eq. (1.150) up to the desired order in $1/m_Q$, the second step in the calculation of Eq. (1.144) is to evaluate the matrix element of the local operators obtained, between external hadronic states, see Ref. [103]. These encode the large distance dynamics responsible for the hadronic structure and require non perturbative methods like Lattice QCD [32] or QCD Sum Rules [29,30] to be determined. In some cases they can also be extracted performing fits to the experimental data, see e.g. Ref. [106]. Alternatively, using the framework of the HQET, the dependence of the heavy quark field and of the hadronic state on the heavy quark mass, see Section 1.2, can be further factored out, the corresponding matrix elements are then expanded in inverse powers of m_Q and expressed in terms of a minimal set of elementary parameters, which must again be determined by means of the non perturbative methods mentioned above, or in some cases via spectroscopy relations [52]. For the matrix element of the dimension-three operator in Eq. (1.152), the HQET expansion has the following form [98, 105]

$$\frac{\langle H_Q | \bar{Q}Q | H_Q \rangle}{2m_{H_Q}} = 1 - \frac{\mu_\pi^2(H_Q) - \mu_G^2(H_Q)}{2m_Q^2} + \mathcal{O}\left(\frac{1}{m_Q^5}\right), \qquad (1.158)$$

where the non perturbative parameters μ_{π}^2 , μ_G^2 are related to the expectation value of the operators in Eqs. (1.154), (1.155), as

$$2m_{H_Q}\,\mu_\pi^2(H_Q) = -\langle H_Q | \mathcal{O}_{kin} | H_Q \rangle, \quad 2m_{H_Q}\,\mu_G^2(H_Q) = \langle H_Q | \mathcal{O}_{mag} | H_Q \rangle. \tag{1.159}$$

Similarly, the matrix elements of the spin-orbit and Darwin operators, are expressed in terms of the two non perturbative parameters ρ_{LS}^3 , ρ_D^3 , i.e.

$$2m_{H_Q}\rho_{LS}^3(H_Q) = -\langle H_Q|\mathcal{O}_{LS}|H_Q\rangle, \quad 2m_{H_Q}\rho_D^3(H_Q) = \langle H_Q|\mathcal{O}_D|H_Q\rangle.$$
(1.160)

In the case of four-quark operators, a simple way to estimate the matrix elements between external mesons states, is the so called 'vacuum insertion approximation' (VIA), corresponding to the assumption that the matrix elements can be saturated by the vacuum intermediate state, see e.g. Ref. [71], namely

$$\langle H_Q | \bar{Q} \Gamma q \bar{q} \Gamma Q | H_Q \rangle \stackrel{\text{VIA}}{=} \langle H_Q | \bar{Q} \Gamma q | 0 \rangle \langle 0 | \bar{q} \Gamma Q | H_Q \rangle.$$
 (1.161)

Set for definiteness $\Gamma = \gamma^{\mu}\gamma_5$ and consider H_Q to be e.g. a pseudoscalar *B* meson. It follows that the matrix element on the l.h.s. of Eq. (1.161) is parametrised in terms



Figure 1.7: Schematic representation of the HQE in Eq. (1.164). The imaginary part of the double insertion of the effective Hamiltonian (top line), is matched into a series of local operators (bottom line).

of the B meson mass m_B and decay constant f_B , where the latter is defined as

$$\langle 0|\bar{q}\gamma^{\mu}\gamma_{5}b|B\rangle = if_{B}p_{B}^{\mu}, \qquad (1.162)$$

here p_B^{μ} denotes the meson four-momentum with $p_B^2 = m_B^2$. By taking into account that the matrix element of the corresponding vector current vanishes due to parity conservation in QCD, see e.g. the textbook [39], we obtain that

$$\langle B|\bar{b}\gamma_{\mu}(1-\gamma_{5})q|0\rangle\langle 0|\bar{q}\gamma^{\mu}(1-\gamma_{5})b|B\rangle = f_{B}^{2}m_{B}^{2}.$$
(1.163)

Finally, the construction of the HQE leads to the following expansion for $\Gamma(H_Q)$

$$\Gamma(H_Q) = \Gamma_3 + \Gamma_5 \frac{\langle \mathcal{O}_5 \rangle}{m_Q^2} + \Gamma_6 \frac{\langle \mathcal{O}_6 \rangle}{m_Q^3} + \dots + 16\pi^2 \left[\tilde{\Gamma}_6 \frac{\langle \tilde{\mathcal{O}}_6 \rangle}{m_Q^3} + \tilde{\Gamma}_7 \frac{\langle \tilde{\mathcal{O}}_7 \rangle}{m_Q^4} + \dots \right] , \quad (1.164)$$

schematically sketched in Figure 1.7. Eq. (1.164) shows that, by exploiting the large hierarchy $m_Q \gg \Lambda_{QCD}$, the total decay width of a heavy hadron can be systematically computed as a series in inverse powers of the heavy quark mass. The lowest order contributions describe the effect of two- and four-quark operators and, in Eq. (1.164), the latter are labelled with a tilde. Moreover, from the diagrammatic representation in Figure 1.7, we see that while the contribution of four-quark operators are generated only at two-loop, again at LO-QCD, and this mismatch is reflected in the presence of the enhancement factor of $16\pi^2$ in front of the square brackets in Eq. (1.164) [107–110]. As already stressed in Section 1.1, the essential feature of the OPE is the separation between short- and long-distance effects. Namely, the non

perturbative dynamics is absorbed in the matrix element of local operators, whereas the short distance contribution is encoded in the corresponding coefficients. The latter, in fact, obey the perturbation expansion

$$\Gamma_d = \Gamma_d^{(0)} + \left(\frac{\alpha_s}{4\pi}\right)\Gamma_d^{(1)} + \left(\frac{\alpha_s}{4\pi}\right)^2\Gamma_d^{(2)} + \dots , \qquad (1.165)$$

and can be computed within standard perturbation theory. Extensive work has been put in this direction, here a brief summary of the current status. The complete calculation of Γ_3 up to NLO-QCD corrections has been obtained in Refs. [111–118]. Currently, also NNLO-QCD corrections are known for semileptonic decays [119–128], while for non-leptonic decays, these have only been determined, for massless final quarks and in full QCD, i.e. without using the effective Hamiltonian, in Ref. [129]. Γ_5 has been computed at LO-QCD for both non-leptonic and semileptonic decays [98,103,130,131], for the latter even NLO-QCD corrections are available [132–134]. For semileptonic decays, Γ_6 was first computed at LO-QCD in Ref. [135] and recently the NLO-QCD corrections were determined in Ref. [136], while the LO-QCD computation for non-leptonic decays has been performed for the first time in Refs. [137–139] for the *b*-system and in Ref. [140] for *c*-quark decays, see also Section 4.1. Finally $\tilde{\Gamma}_6$ is known at NLO-QCD [141–143], while $\tilde{\Gamma}_7$ only at LO-QCD [144].

We conclude by emphasising that the construction of the HQE is based on the validity of the so called quark-hadron duality (QHD). This refers to the assumption that the inclusive rate determined by summing over all the exclusive hadronic decay channels, and the one predicted by the HQE, are dual to each other, in the sense that they provide two valid representations of the same quantity, using respectively the hadron-level and the quark-level description. However, violations of QHD constitute a systematic uncertainty of the HQE, and one simple argument is the fact that, by computing the total decay width in terms of a series expansion in powers of Λ_{QCD}/m_Q , any term of the type $\exp(-m_Q/\Lambda_{QCD}) \sin(m_Q/\Lambda_{QCD})$, would be systematically neglected, since $\exp(-1/x)$ is non-analytic and its expansion around x = 0, yields identically zero. Despite deviations of HQD cannot be excluded, there is no experimental evidence so far for sizeable violations that might compromise the applicably of the HQE. For a detailed discussion of QHD see e.g. Refs. [145–147].

Chapter 2

Practical Calculations within the HQE

With the theoretical background discussed in Chapter 1, we can now show the explicit computation of the lowest-order contributions to the total decay width of a heavy hadron, Eq. (1.144). For definiteness we assume H_Q to be a B meson with $B = \{\bar{B}_d, B^-, \bar{B}_s\}$, i.e. we limit ourselves to systems containing a heavy b quark and a light antiquark $\bar{q} = \{\bar{d}, \bar{u}, \bar{s}\}$, without discussing the case of the B_c meson ¹. We stress, however, that the expressions obtained, taking into account the appropriate replacements, e.g. of the CKM factors and masses, can be also applied to the study of the charm system and of the b-baryons with one heavy quark. Furthermore, we emphasise that all the calculations presented are only at LO-QCD.

The total decay width in Eq. (1.164) can be decomposed in the sum of semileptonic and non-leptonic widths, namely

$$\Gamma(B) = \Gamma^{(\mathrm{SL})}(B) + \Gamma^{(\mathrm{NL})}(B). \qquad (2.1)$$

For simplicity, in the following, we consider only the computation of $\Gamma^{\rm NL}(B)$, again, the corresponding results for the semileptonic case can be easily derived by setting $N_c = 1, C_1 = 1$ and $C_2 = 0$. According to Eq. (1.144), the total non-leptonic decay width of a *B* meson is induced at the quark level by the flavour-changing transition $b \rightarrow q_1 \bar{q}_2 q_3$, with $q_1, q_2 = \{u, c\}$ and $q_3 = \{d, s\}$, described, at the renormalisation scale $\mu_1 \sim m_b$, by the effective weak Hamiltonian $\mathcal{H}_{eff}(x)$, see Section 1.1, i.e.

$$\mathcal{H}_{\text{eff}}(x) = \frac{G_F}{\sqrt{2}} V_{q_1 b}^* V_{q_2 q_3} \Big[C_1 Q_1(x) + C_2 Q_2(x) \Big] + \text{h.c.}$$
(2.2)

The colour-singlet and colour-rearranged local four-quark operators Q_1 , (x), $Q_2(x)$,

¹In this case the HQE must be properly generalised in order to include a double expansion in inverse powers of the bottom as well as the charm quark mass, see e.g. Refs. [148–150].

in Eq. (2.2), are respectively given by

$$Q_1(x) = \left(\bar{q}_1^i(x)\Gamma_\mu b^i(x)\right) \left(\bar{q}_3^j(x)\Gamma^\mu q_2^j(x)\right), \qquad (2.3)$$

$$Q_2(x) = \left(\bar{q}_1^i(x)\Gamma_\mu b^j(x)\right) \left(\bar{q}_3^j(x)\Gamma^\mu q_2^i(x)\right), \qquad (2.4)$$

where $\Gamma_{\mu} = \gamma_{\mu}(1 - \gamma_5)$. In Eq. (2.2), $C_1(\mu_1)$, $C_2(\mu_1)$, define the corresponding Wilson coefficients, their scale dependence is often omitted in order to simplify the notation. Note also that, being interested in discussing only the general structure of the computation, in Eq. (2.2) we have neglected the contribution of the penguin operators, however the expressions can be easily generalised to include them. Substituting Eq. (2.2) into Eq. (1.143), leads to the following decomposition for the non local second-order operator \mathcal{T} i.e.

$$\mathcal{T}^{(q_1\bar{q}_2q_3)} = C_1^2 \,\mathcal{T}_{11}^{(q_1\bar{q}_2q_3)} + \,2 \,C_1 C_2 \,\mathcal{T}_{12}^{(q_1\bar{q}_2q_3)} + C_2^2 \,\mathcal{T}_{22}^{(q_1\bar{q}_2q_3)} \,, \tag{2.5}$$

here the superscript $(q_1\bar{q}_2q_3)$ refers to the specific decay mode of the *b* quark, which for the sake of a more compact notation will be sometimes dropped, and

$$\mathcal{T}_{mn}^{(q_1\bar{q}_2q_3)} = \frac{G_F^2 |V_{q_1b}|^2 |V_{q_3q_2}|^2}{2} i \int d^4x \, \mathrm{T}\Big\{Q_m(x), Q_n^{\dagger}(0)\Big\} + (x \leftrightarrow 0) \,. \tag{2.6}$$

In Eq. (2.6), the corresponding term due to the exchange of coordinates $(x \leftrightarrow 0)$ must be considered separately, and cannot be in general reduced to a symmetry factor of 2, since the computation of power corrections will be performed in the FS gauge, which, it is worth remarking, explicitly breaks the translation invariance of the propagator, see Section 1.3.

The time-ordered product in Eq. (2.6) is written, by means of the Wick's theorem [151], as a linear combination of terms where only normal products, normal products and contractions and only contractions of fields appear, see e.g. the textbook [152]. The lowest-order contributions in the HQE correspond to two- and four-quark operators and are generated respectively from the contraction of threeand two-pairs of light quark fields while leaving the *b*-quark fields uncontracted. The first case is discussed in Section 2.1, the second in Section 2.2.

2.1 Contribution of two-quark operators

Following Refs. [130, 131], for a straightforward treatment of colour in the computation of power corrections due to the expansion of the quark propagator, it is convenient to perform in \mathcal{T}_{12} , Eq. (2.5), the change of basis

$$\left\{Q_1(x), Q_2(x)\right\} \to \left\{Q_1(x), Q_3(x)\right\},$$
 (2.7)

where $Q_3(x)$ denotes the colour-octet operator ²

$$Q_3(x) = \left(\bar{q}_1^i(x)\Gamma_\mu t_{ij}^a b^j(x)\right) \left(\bar{q}_3^l(x)\Gamma^\mu t_{lm}^a q_2^m(x)\right).$$
(2.8)

The relation between $Q_2(x)$, and $Q_3(x)$, in Eq. (2.7) is obtained by taking into account the completeness property of the $SU(3)_c$ generators, i.e.

$$t^a_{ij}t^a_{lm} = \frac{1}{2}\left(\delta_{im}\delta_{jl} - \frac{1}{N_c}\delta_{ij}\delta_{lm}\right).$$
(2.9)

Substituting $Q_2(x) = (1/N_c)Q_1(x) + 2Q_3(x)$ in \mathcal{T}_{12} in Eq. (2.5), and considering only the contribution of two-quark operators, leads to the general decomposition

$$\mathcal{T}^{(2q)} = C_1^2 \,\mathcal{T}_{11}^{(2q)} + 2 \,C_1 C_2 \,\left(\frac{1}{N_c} \mathcal{T}_{11}^{(2q)} + 2 \,\mathcal{T}_{13}^{(2q)}\right) + C_2^2 \,\mathcal{T}_{22}^{(2q)} \,, \qquad (2.10)$$

where the superscript (2q) indicates that all pairs of light quarks fields in the timeordered product in Eq. (2.6) have been contracted and replaced with the corresponding propagators, namely, without specifying the colour structure ³

$$T\left\{\bar{q}_{1}(x)\Gamma_{\mu}b(x)\bar{q}_{3}(x)\Gamma^{\mu}q_{2}(x),\bar{b}\Gamma_{\nu}q_{1}\bar{q}_{2}\Gamma^{\nu}q_{3}\right\} =: \overline{\bar{q}_{1}(x)\Gamma_{\mu}b(x)\bar{q}_{3}(x)\Gamma^{\mu}q_{2}(x)\bar{b}\Gamma_{\nu}q_{1}\bar{q}_{2}\Gamma^{\nu}q_{3}}: ,$$
(2.11)

here the two colons denote the normal product. Eq. (2.11) can be schematically visualised in Figure 2.1. Taking into account Eq. (2.11), it follows that the expres-

²Note that in Refs. [130, 131] the colour octet-operator is denoted by \tilde{Q}_1 .

³We recall that we often omit to explicitly indicate the dependence on the space-time coordinate for fields evaluated at the origin, so unless otherwise stated, we assume q = q(0), for generic q(x).

sions of the non local operators $\mathcal{T}_{mn}^{(2q)}$ in Eq. (2.10), are respectively given by

$$\mathcal{T}_{11}^{(2q)} = -\frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 i \int d^4x \ \bar{b}^j \gamma_\nu (1-\gamma_5) \, i S_{jk}^{(q_1)}(0,x) \gamma_\mu (1-\gamma_5) b^k(x) \\ \times \operatorname{Tr} \left[\gamma^\nu (1-\gamma_5) i S_{lm}^{(q_3)}(0,x) \gamma^\mu (1-\gamma_5) i S_{ml}^{(q_2)}(x,0) \right] + (x \leftrightarrow 0) , \qquad (2.12)$$

$$\mathcal{T}_{13}^{(2q)} = -\frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 i \int d^4x \ \bar{b}^j \gamma_\nu (1-\gamma_5) t_{jl}^a \, i S_{lk}^{(q_1)}(0,x) \gamma_\mu (1-\gamma_5) b^k(x) \\ \times \operatorname{Tr} \left[\gamma^\nu (1-\gamma_5) t_{mn}^a \, i S_{nr}^{(q_3)}(0,x) \gamma^\mu (1-\gamma_5) i S_{rm}^{(q_2)}(x,0) \right] + (x \leftrightarrow 0) \,, \qquad (2.13)$$

and

$$\mathcal{T}_{22}^{(2q)} = -\frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 i \int d^4x \ \bar{b}^j \gamma_\nu (1-\gamma_5) \, iS_{lm}^{(q_1)}(0,x) \gamma_\mu (1-\gamma_5) b^k(x) \\ \times \operatorname{Tr} \left[\gamma^\nu (1-\gamma_5) iS_{jk}^{(q_3)}(0,x) \gamma^\mu (1-\gamma_5) iS_{lm}^{(q_2)}(x,0) \right] + (x \leftrightarrow 0) , \qquad (2.14)$$

where the minus sign and the trace over spinor indices, derive from the fermion loop. Note that applying the Fierz identity

$$\left(\bar{q}_1^i \Gamma_\mu q_2^i\right) \left(\bar{q}_3^j \Gamma^\mu q_4^j\right) = \left(\bar{q}_3^j \Gamma_\mu q_2^i\right) \left(\bar{q}_1^i \Gamma^\mu q_4^j\right), \qquad (2.15)$$

to the four-quark operators in Eqs. (2.3), (2.4), gives

$$Q_{1,2} = Q_{2,1}^{(q_1 \leftrightarrow q_3)} \,. \tag{2.16}$$

meaning that the four-quark operator Q_1 is equal to Q_2 with the exchange $q_1 \leftrightarrow q_3$ and vice versa. This important relation implies that in Eq. (2.14) we can write

$$S_{lm,\,\alpha\beta}^{(q_1)} S_{jk,\,\gamma\delta}^{(q_3)} \to S_{lm,\,\gamma\delta}^{(q_1)} S_{jk,\,\alpha\beta}^{(q_3)},$$
 (2.17)



Figure 2.1: Diagram describing the leading order contribution to the free b-quark decay.

where the Greek letters denote spinor indices, and then that

$$\mathcal{T}_{22}^{(2q)} = \mathcal{T}_{11}^{(2q)\,(q_1 \leftrightarrow q_3)} \,. \tag{2.18}$$

By taking into account Eq. (2.18), the contribution of $Q_2 \otimes Q_2$ in Eq. (2.6) can be obtained from that of $Q_1 \otimes Q_1$ after performing the replacement $q_1 \leftrightarrow q_3$.

The coefficients of the two-quark operators up to order $1/m_b^3$, are computed in detail in Chapter 3, for generic non-leptonic decays of the *b*-quark and using the representation of the quark-propagator in momentum space, Eqs. (1.98), (1.102). However, it is instructive to perform the same calculation also using the expression of the quark-propagator in coordinate space given in Eqs. (1.121), (1.124). This is discussed in the next two sections, respectively for the case of dimension-three and dimension-five contributions, and for the single mode $b \to c\bar{u}d$.

2.1.1 Computation of $\Gamma_3^{(c\bar{u}d)}$

The leading term in Eq. (1.164) corresponds to the decay of a free *b* quark, as shown in Figure 2.1. Neglecting the interaction with the background gluon field, see Section 1.3, all the propagators in Eqs. (2.12)-(2.14), reduce to

$$S_{jk}^{(q)}(x,y) = S_0^{(q)}(x-y)\,\delta_{jk}\,,\qquad q=c,u,d\,,\qquad(2.19)$$

where $S_0^{(q)}(x-y)$ is the free-quark propagator defined in Eq. (1.72), and a superscript has been introduced in order to distinguish between the different quarks consistently with the description of the $b \to c\bar{u}d$ decay. The presence of the Kronecker delta in Eq. (2.19) leads to immediate simplifications. In fact, by enforcing the trace of the $SU(3)_c$ generators t^a in the square brackets of Eq. (2.13), we readily obtain that $\mathcal{T}_{13}^{(2q)}$ must vanish at this order i.e. ⁴

$$\mathcal{T}_{13}^{(2q)} = \mathcal{O}\left(\frac{1}{m_b^2}\right). \tag{2.20}$$

Moreover, it follows that Eqs. (2.12), (2.14), exactly coincide, so that $\mathcal{T}_{11}^{(2q)}$ and $\mathcal{T}_{22}^{(2q)}$ are equal up to higher order corrections, namely

$$\mathcal{T}_{22}^{(2q)}\Big|_{d=3} = \mathcal{T}_{11}^{(2q)}\Big|_{d=3}.$$
(2.21)

Because of Eqs. (2.20), (2.21), we need to compute only one expression, i.e.

$$\mathcal{T}_{11}^{(2q)} = -\frac{N_c G_F^2}{2} |V_{q_1 b}|^2 |V_{q_3 q_2}|^2 \int d^4 x \ \bar{b} \ \gamma_\nu (1 - \gamma_5) S_0^{(c)}(-x) \gamma_\mu (1 - \gamma_5) b(x) \times \operatorname{Tr} \left[\gamma^\nu (1 - \gamma_5) S_0^{(d)}(-x) \gamma^\mu (1 - \gamma_5) S_0^{(u)}(x) \right] + (x \leftrightarrow 0) + \dots$$
(2.22)

In Eq. (2.22), the colour factor is $\delta_{ii} = N_c$, and the ellipsis denote power suppressed contributions due to higher order terms in the quark-propagator. Notice also that the result in Eq. (2.21), combined with Eq. (2.18), shows that Eq. (2.22) must be a symmetric function under the exchange $c \leftrightarrow d$. A further simplification in the computation of the dimension-three contribution, derives from the translation invariance of the free-quark propagator, meaning that the integral in Eq. (2.22) is also translation invariant, and that the second term on the r.h.s. of Eq. (2.22), at this order, reduces to a symmetry factor of 2.

In Eq. (2.22), the corresponding expressions for the three propagators follow from Eq. (1.123). For the charm quark, it is

$$S_0^{(c)}(-x) = -\frac{i}{4\pi^2} \frac{m_c^2 K_1(m_c \sqrt{-x^2})}{\sqrt{-x^2}} + \frac{\cancel{x}}{4\pi^2} \frac{m_c^2 K_2(m_c \sqrt{-x^2})}{x^2}, \qquad (2.23)$$

and, due to the chiral structure of Eq. (2.22), only the term in Eq. (2.23) proportional to an odd number of gamma matrices contributes. Furthermore, neglecting the mass of the up- and down-quarks, we need to consider the following limits of the Bessel functions

⁴Note that the first correction to the free-quark propagator arises at order $\mathcal{O}(1/m_b^2)$.

$$\lim_{m \to 0} m^2 K_1(m\sqrt{-x^2}) = 0, \qquad \lim_{m \to 0} m^2 K_2(m\sqrt{-x^2}) = -\frac{2}{x^2}, \qquad (2.24)$$

from which we obtain that

$$S_0^{(u)}(x) = \frac{\cancel{x}}{2\pi^2 x^4} = -S_0^{(d)}(-x).$$
(2.25)

Moreover, we recall that the coefficient of the dimension-three operator $\overline{b}b$ in the OPE in Eq. (1.150), is obtained by evaluating Im $\mathcal{T}^{(2q)}$ between external b states with momentum p_b , hence we can make the following replacement in Eq. (2.22) ⁵, see also Refs. [69, 87]

$$b(x) \to e^{-ip_b \cdot x} b(0) \,. \tag{2.26}$$

By substituting Eqs. (2.23), (2.25) and (2.26) into Eq. (2.22), it is then straightforward to arrive at

$$\mathcal{T}^{(2q)} = 4 \frac{G_F^2}{\pi^6} |V_{cb}|^2 |V_{ud}|^2 \left(N_c \left(C_1^2 + C_2^2 \right) + 2 C_1 C_2 \right) \\ \times \bar{b} \left\{ \int d^4 x \, e^{-ip_b \cdot x} \, \frac{m_c^2 K_2(m_c \sqrt{-x^2})}{x^8} \, \not{x} \right\} (1 - \gamma_5) \, b + \dots \, .$$
(2.27)

The next step is to compute the imaginary part of the integral in the curly brackets of Eq. (2.27). This is easily obtained using the formalism presented in Ref. [130], based on the technique developed by Belyaev and Blok in Ref. [85] for the computation of the spectral representation of integrals appearing in the Fourier transform of the product of several massless and one massive quark propagator, expressed in coordinate space, see Ref. [130]. The result reads ⁶

$$\operatorname{Im} \int d^4x \, e^{-ip \cdot x} \, \frac{K_{\nu}(m\sqrt{-x^2})}{(\sqrt{-x^2})^n} \, x^{\mu} = G_{\nu,n}(p^2) \, p^{\mu} \,, \tag{2.28}$$

with

⁵This follows from $b(x)|b\rangle = \exp(-ip_b \cdot x)u_b(p_b)$ where $u_b(p_b) = b(0)|b\rangle$, is the *b*-quark spinor.

⁶Note that the result in Ref. [130] must be multiplied by $-\pi$. This might be due to a different convention used to define the discontinuity of a complex function.

$$G_{\nu,n}(p^2) = \frac{\pi^3}{2^{(n-4)}m^{\nu}} \frac{1}{\Gamma\left(\frac{n-\nu}{2}\right)} \sum_{k=0}^{\frac{n-\nu}{2}-1} (-1)^{\frac{n-\nu}{2}-1-k} C_{\frac{n-\nu}{2}-1}^k U_{\frac{n+\nu}{2}-k}^{n+\nu-4}(p^2) .$$
(2.29)

In Eq. (2.29), the coefficients C_n^m are given by

$$C_n^m = \binom{n}{m} = \frac{n!}{m! (n-m)!},$$
(2.30)

while the functions $U_i^{2j}(p)$ are

$$U_i^{2j}(p^2) = \frac{m^{2(i-1)}}{(j-1)!} \int_{m^2}^{p^2} dz \, \frac{(p^2-z)^{j-1}}{z^i} \,.$$
(2.31)

Setting $\nu = 2$ and n = 8 in Eq. (2.28) yields

where we have introduced the dimensionless parameter $r = m_c^2/p_b^2$. Substituting Eq. (2.32) into Eq. (2.27), we then obtain

$$\operatorname{Im} \mathcal{T}^{(2q)} = \frac{G_F^2}{192\pi^3} |V_{cb}|^2 |V_{ud}|^2 \left(N_c \left(C_1^2 + C_2^2 \right) + 2 C_1 C_2 \right) \\ \times p_b^4 p_b^{\mu} \left(1 - 8r - 12r^2 \log r + 8r^3 - r^4 \right) \bar{b} \gamma_{\mu} (1 - \gamma_5) b + \dots$$
(2.33)

Finally, Eq. (2.33) depends on the heavy quark momentum p_b , which, it is worth remarking, admits the general parametrisation $p_b^{\mu} = m_b v^{\mu} + k^{\mu}$, here v^{μ} is the hadron velocity and the 'residual' momentum k^{μ} describes the interaction of the heavy quark with the light degrees of freedom, see Chapter 1. As stated above, the dimensionthree contribution to $\text{Im}\mathcal{T}^{(2q)}$ corresponds to the decay of a free *b*-quark, meaning that all the interactions with the soft gluons and quarks can be neglected at this order. In this case the heavy quark momentum reduces to $p_b^{\mu} = m_b v^{\mu} + \mathcal{O}(1/m_b)$, and recalling the definition of the rescaled heavy quark field $b(x) = \exp(-im_b v \cdot x)b_v(x)$, see Eq. (1.145), satisfying $\psi b_v = b_v + \mathcal{O}(1/m_b)$, cf. Eq. (1.148), from Eq. (1.144) and



Figure 2.2: Soft gluon corrections to the free *b*-quark decay.

Eq. (1.158), we arrive at the well known expression

$$\Gamma_3^{(c\bar{u}d)} = \frac{G_F^2 m_b^5}{192\pi^3} |V_{cb}|^2 |V_{ud}|^2 \left(N_c \left(C_1^2 + C_2^2 \right) + 2 C_1 C_2 \right) \left(1 - 8\rho - 12\rho^2 \log \rho + 8\rho^3 - \rho^4 \right)$$
(2.34)

where $\rho = m_c^2/m_b^2$ is a dimensionless mass parameter and we have taken into account that the contribution of the axial current to the matrix element between *B* mesons states vanishes due to conservation of parity in QCD [39].

2.1.2 Computation of $\Gamma_5^{(c\bar{u}d)}$

In order to compute power corrections to the free *b*-quark decay in Eq. (2.34), we must include the effect of the QCD interaction of the heavy quark field with the soft degrees of freedom inside the heavy hadron. This results in three contributions, generated by expanding respectively, the propagator of the quarks inside the two-loop diagram, because of the large *b*-quark momentum flowing into it, the heavy-quark momentum and the relevant matrix elements up to the order in $1/m_b$ considered. In the case of two-quark operators, the starting point is represented by Eqs. (2.12)-(2.14), where now, being interested in computing dimension-five contributions, we must take into account, in the expression of the quark propagator, also terms proportional to the gluon field strength tensor $G_{\rho\sigma}$, see Section 1.3 and Figure 2.2, namely

$$S_{jk}^{(q)}(x,y) = S_0^{(q)}(x-y)\delta_{jk} + S_1^{(q)}(x-y)^a t_{jk}^a, \quad q = c, u, d, \qquad (2.35)$$

here $S_1^{(q)}(x-y)$ contains only corrections to the free-quark propagator due to operators of dimension-two, cf. Eq. (1.121) and we have explicitly indicated that up to this order the translation invariance is still preserved. The presence of the Kronecker delta and of the $SU(3)_c$ generators in Eq. (2.35), significantly simplifies the colour structure of Eqs. (2.12)-(2.14), in fact, it is easy to verify that, since $\text{Tr}[t^a] = 0$ and terms proportional to $G_{\rho\sigma}G_{\mu\nu}$ correspond to corrections of at least dimension-seven, a soft gluon cannot be radiated off every propagator for each of the expressions in Eqs. (2.12)-(2.14). Namely, only $S_1^{(c)}(-x)$ can contribute to $\mathcal{T}_{11}^{(2q)}$ and as consequence of Eq. (2.18) only $S_1^{(d)}(-x)$ can contribute to $\mathcal{T}_{22}^{(2q)}$. Finally, in Eq. (2.13) only the expansion of the two propagators inside the trace is non vanishing and both $S_1^{(u)}(x)$ and $S_1^{(d)}(-x)$ must be independently taken into account in $\mathcal{T}_{13}^{(2q)}$. However, it is clear that by substituting Eq. (2.35) into Eqs. (2.12)-(2.14), we also recover the leading order result already discussed in the previous section. This generates the remaining dimension-five contributions, once the corrections to the heavy quark momentum in Eq. (2.33) and to the matrix element of the dimension-three and dimension-four operators are respectively included. To avoid confusion we introduce the notation

$$\Gamma_5^{(c\bar{u}d)} = \Gamma_5^{(c\bar{u}d)} \Big|_{\mathrm{I}} + \Gamma_5^{(c\bar{u}d)} \Big|_{\mathrm{II}}, \qquad (2.36)$$

corresponding to the sum of the dimension-five contributions arising from the expansion of the quark-propagator on one side (I) and of the heavy-quark momentum as well as of the matrix elements of dimension lower than five, on the other (II).

Let us start by considering (I). In the case of $\mathcal{T}_{11}^{(2q)}$, as commented above, only the contribution of the gluon emitted from the charm-quark line is non vanishing and the three propagators in Eq. (2.12) are respectively given by ⁷

$$S_1^{(c)}(-x) = \frac{\tilde{G}^{\rho\eta} x_{\rho} \gamma_{\eta} \gamma_5}{8\pi^2} \frac{m_c K_1(m_c \sqrt{-x^2})}{\sqrt{-x^2}}, \qquad (2.37)$$

and

$$S_0^{(u)}(x) = \frac{1}{2\pi^2} \frac{\cancel{x}}{x^4} = -S_0^{(d)}(-x).$$
(2.38)

Substituting these expressions as well as including a symmetry factor of 2 due to the translation invariance ⁸, Eq. (2.12) becomes ⁹

 $^{^7{\}rm We}$ already take into account that in the propagator of the charm quark only terms proportional to an odd number of gamma matrices contribute.

⁸The same factor appear also in the remaining contributions.

⁹Note that the replacement in Eq. (2.26) is again introduced, since at this order the contribution from the dimension-five operator with one gluon field is obtained taking the matrix element $\langle bg | \text{Im} \mathcal{T}^{(2q)} | b \rangle$ between external *b*-quark and gluon states.

$$\mathcal{T}_{11}^{(2q)}\Big|_{d=5} = \frac{G_F^2}{64\pi^6} |V_{cb}|^2 |V_{ud}|^2 \bar{b} \left\{ \int d^4x \, e^{-ip_b \cdot x} \, \frac{\tilde{G}^{\rho\eta} \, x_\rho x_\xi x_\tau}{x^8} \frac{m_c \, K_1(m_c \sqrt{-x^2})}{\sqrt{-x^2}} \right. \\ \left. \times \left(\gamma_\mu (1-\gamma_5) \, \gamma_\eta \gamma_5 \, \gamma_\nu (1-\gamma_5) \right) \mathrm{Tr} \left[\, \gamma^\mu (1-\gamma_5) \, \gamma^\xi \, \gamma^\nu \, (1-\gamma_5) \, \gamma^\tau \, \right] \right\} b \, .$$

$$(2.39)$$

From

$$x_{\xi}x_{\tau}\left(\gamma_{\mu}\gamma_{\eta}\gamma_{\nu}\right) \operatorname{Tr}\left[\gamma^{\mu}(1-\gamma_{5})\gamma^{\xi}\gamma^{\nu}\left(1-\gamma_{5}\right)\gamma^{\tau}\right] = 32 x_{\eta} \, \sharp \,, \qquad (2.40)$$

it is clear that the contribution in Eq. (2.39) must vanish since the dual field strength tensor $\tilde{G}^{\rho\eta}$ is contracted with the symmetric tensor $x_{\rho}x_{\eta}$ i.e.

$$\mathcal{T}_{11}^{(2q)}\Big|_{d=5} = 0.$$
 (2.41)

In the case of $\mathcal{T}_{22}^{(2q)}$, in Eq. (2.14) we can only expand the propagator of the down quark while the *c*- and *u*-quark propagators are free. Using the following limits

$$\lim_{m \to 0} m K_0(m\sqrt{-x^2}) = 0, \qquad \lim_{m \to 0} m K_1(m\sqrt{-x^2}) = \frac{1}{\sqrt{-x^2}}, \qquad (2.42)$$

we evidently obtain that

$$S_0^{(u)}(x) = \frac{1}{2\pi^2} \frac{\not}{x^4}, \qquad S_1^{(d)}(-x) = -\frac{1}{8\pi^2} \frac{x^{\rho}}{x^2} \tilde{G}_{\rho\xi} \gamma^{\xi} \gamma_5, \qquad (2.43)$$

and

$$S_0^{(c)}(-x) = \frac{\cancel{x}}{4\pi^2} \frac{m_c^2 K_2(m_c \sqrt{-x^2})}{x^2} \,. \tag{2.44}$$

Substituting these expressions into Eq. (2.14) gives

$$\mathcal{T}_{22}^{(2q)}\Big|_{d=5} = \frac{G_F^2}{128\pi^6} |V_{cb}|^2 |V_{ud}|^2 \bar{b} \left\{ \int d^4x \, e^{-ip_b \cdot x} \, \frac{\tilde{G}_{\rho\xi} x^\rho x^\eta x_\tau}{x^8} m_c^2 \, K_2(m_c \sqrt{-x^2}) \right. \\ \left. \left. \left. \left(\gamma_\mu (1-\gamma_5) \, \gamma_\eta \, \gamma_\nu (1-\gamma_5) \right) \, \mathrm{Tr} \left[\, \gamma^\mu (1-\gamma_5) \, \gamma^\xi \, \gamma_5 \, \gamma^\nu \, (1-\gamma_5) \, \gamma^\tau \, \right] \right\} b \,,$$

$$(2.45)$$

where now the corresponding gamma structure can be simplified as

$$x^{\eta}x_{\tau}\left(\gamma_{\mu}\gamma_{\eta}\gamma_{\nu}(1-\gamma_{5})\right)\operatorname{Tr}\left[\gamma^{\mu}(1+\gamma_{5})\gamma^{\xi}\gamma^{\nu}(1-\gamma_{5})\gamma^{\tau}\right] = 32\,x^{\xi}\,\not(1-\gamma_{5})\,,\quad(2.46)$$

showing that also this contribution must vanish, since again the dual field tensor $\tilde{G}_{\rho\xi}$ in Eq. (2.45) is contracted with the symmetric tensor $x^{\rho}x^{\xi}$, namely

$$\mathcal{T}_{22}^{(2q)}\Big|_{d=5} = 0.$$
 (2.47)

We see that the emission of a soft gluon from both the charm and down quark propagators does not contribute at dimension-five, therefore in Eq. (2.13) it is sufficient to consider the expansion of the *u*-quark propagator only. However, we notice that the result in Eqs. (2.41), (2.47) is general and that the fact that contributions to the propagator of the q_1 - and q_3 -quarks vanish at order $1/m_b^2$, holds independently of the specific mode considered see Refs. [130, 131]. We will discuss this again in Chapter 3.

Finally in the case of $\mathcal{T}_{13}^{(2q)}$, considering only corrections to the propagator of the up quark, we must substitute the following expressions into Eq. (2.13), i.e.

$$S_1^{(u)}(x) = \frac{1}{8\pi^2} \frac{x^{\rho}}{x^2} \tilde{G}_{\rho\tau} \gamma^{\tau} \gamma_5, \qquad S_0^{(d)}(-x) = -\frac{1}{2\pi^2} \frac{x}{x^4}, \qquad (2.48)$$

and

$$S_0^{(c)}(-x) = \frac{\cancel{x}}{4\pi^2} \frac{m_c^2 K_2(m_c \sqrt{-x^2})}{x^2}, \qquad (2.49)$$

which, taking into account that $\text{Tr}[t^a t^b] = (1/2)\delta^{ab}$, leads to
$$\mathcal{T}_{13}^{(2q)}\Big|_{d=5} = \frac{G_F^2}{128\pi^6} |V_{cb}|^2 |V_{ud}|^2 \bar{b} \left\{ \int d^4x \, e^{-ip_b \cdot x} \frac{\tilde{G}_{\rho\tau} x^\rho x^\eta x_\xi}{x^8} m_c^2 K_2(m_c \sqrt{-x^2}) \right. \\ \left. \times \left(\gamma_\mu (1-\gamma_5) \, \gamma_\eta \, \gamma_\nu (1-\gamma_5) \right) \mathrm{Tr} \left[\, \gamma^\mu (1-\gamma_5) \, \gamma^\xi \, \gamma^\nu \, (1-\gamma_5) \, \gamma^\tau \, \gamma_5 \right] \right\} b \,.$$

$$(2.50)$$

The gamma structure simplifies as

$$x^{\eta} x_{\xi} (\gamma_{\mu} \gamma_{\eta} \gamma_{\nu} (1 - \gamma_{5})) \operatorname{Tr} \left[\gamma^{\mu} (1 + \gamma_{5}) \gamma^{\xi} \gamma^{\nu} (1 - \gamma_{5}) \gamma^{\tau} \right] = 32 \gamma^{\tau} x^{2} (1 - \gamma_{5}), \quad (2.51)$$

from which it follows that, since the dual field tensor $\tilde{G}_{\rho\tau}$ is now contracted with $x^{\rho}\gamma^{\tau}$, this contribution is non vanishing and equal to

$$\mathcal{T}_{13}^{(2q)}\Big|_{d=5} = \frac{G_F^2}{2\pi^6} |V_{cb}|^2 |V_{ud}|^2 \,\bar{b} \,\tilde{G}_{\rho\tau} \gamma^\tau \left\{ \int d^4x \, e^{-ip_b \cdot x} \, \frac{m_c^2 \, K_2(m_c \sqrt{-x^2})}{x^6} \, x^\rho \right\} (1 - \gamma_5) \, b \, .$$

$$(2.52)$$

The imaginary part of the integral in the curly brackets of Eq. (2.52) can be again obtained using the general result derived in Ref. [130]. Setting $\nu = 2$ and n = 6 in Eq. (2.28), yields

$$\operatorname{Im}\left\{ \int d^4x \, e^{-ip_b \cdot x} \, \frac{m_c^2 K_2(m_c \sqrt{-x^2})}{x^6} \, x^\rho \right\} = -\frac{\pi^3}{24} \, p_b^2 \, p_b^\rho \, (1-r)^3 \,, \qquad (2.53)$$

where we recall that the dimensionless parameter r is defined as $r = m_c^2/p_b^2$. In order to single out the operator appearing in Eq. (2.52), we use that $\tilde{G}_{\rho\tau} = (1/2)\epsilon_{\rho\tau\mu\nu}G^{\mu\nu}$, then, from the tensor decomposition of three gamma matrices Eq. (7) we have

$$\epsilon_{\rho\mu\nu\tau}\gamma^{\tau} = -i\gamma_{\rho}\gamma_{\mu}\gamma_{\nu}\gamma_{5} + ig_{\rho\mu}\gamma_{\nu}\gamma_{5} - ig_{\rho\nu}\gamma_{\mu}\gamma_{5} + ig_{\mu\nu}\gamma_{\rho}\gamma_{5} , \qquad (2.54)$$

which leads to

$$\bar{b}\,\tilde{G}_{\rho\tau}\gamma^{\tau}p_{b}^{\rho}(1-\gamma_{5})\,b = \frac{1}{2}\,\bar{b}\,G^{\mu\nu}\epsilon_{\rho\mu\nu\tau}\gamma^{\tau}p_{b}^{\rho}(1-\gamma_{5})\,b$$
$$= \bar{b}\left\{\frac{i}{2}\,G^{\mu\nu}\gamma_{\rho}\gamma_{\mu}\gamma_{\nu}\,p_{b}^{\rho}(1-\gamma_{5}) - i\,G^{\mu\nu}\gamma_{\mu}\,p_{b\nu}(1-\gamma_{5})\right\}b. \quad (2.55)$$

Note that, when contracted with the antisymmetric tensor $G^{\mu\nu}$, the fourth term on the r.h.s of Eq. (2.54) vanishes while the second and third terms give the same contribution. Taking into account that at this order $p_b^{\mu} = m_b v^{\mu} + \mathcal{O}(1/m_b)$ and introducing the rescaled heavy quark field $b(x) = \exp(-im_b v \cdot x)b_v(x)$, see Eq. (1.145), satisfying $b_v(x) = \psi b_v(x) + \mathcal{O}(1/m_b)$, cf. Eq. (1.148), it is easy to show that the second term on the r.h.s. of the second line in Eq. (2.55) is zero, in fact

$$-i \bar{b} G^{\mu\nu} \gamma_{\mu} p_{b\nu} b = -i m_{b} \bar{b}_{v} G^{\mu\nu} \gamma_{\mu} v_{\nu} b_{v} + \mathcal{O}(1/m_{b})$$
$$= -i m_{b} \bar{b}_{v} G^{\mu\nu} v_{\mu} v_{\nu} b_{v} + \mathcal{O}(1/m_{b})$$
$$= \mathcal{O}(1/m_{b}), \qquad (2.56)$$

where the contributions proportional to γ_5 in Eq. (2.55) can be neglected, since they vanish in matrix elements between *B* meson states due to parity conservation, and we have used the following identity

$$\bar{b}_v \gamma_\mu b_v = \bar{b}_v \psi \gamma_\mu \psi b_v = -\bar{b}_v \gamma_\mu b_v + 2\bar{b}_v v_\mu b_v , \qquad (2.57)$$

valid up to $1/m_b$ corrections, to write $\bar{b}_v \gamma_\mu b_v = \bar{b}_v v_\mu b_v$, see e.g. Ref. [153] or cf. Eq. (1.43). Conversely, the first term on the r.h.s. of the second line in Eq. (2.55) gives

$$(i/2) \,\bar{b} \,G^{\mu\nu} \gamma_{\rho} \gamma_{\mu} \gamma_{\nu} \,p_{b}^{\rho} b = (1/2) m_{b} \,\bar{b}_{v} \,G^{\mu\nu} \psi \sigma_{\mu\nu} b_{v} + \mathcal{O}(1/m_{b})$$
$$= (1/2) \,m_{b} \,\bar{b}_{v} \,G^{\mu\nu} \sigma_{\mu\nu} b_{v} + \mathcal{O}(1/m_{b})$$
$$= \mathcal{O}_{mag} + \mathcal{O}(1/m_{b}) \,, \qquad (2.58)$$

here the first equality follows from taking the antisymmetric part of $\gamma_{\mu}\gamma_{\nu}$ upon con-



Figure 2.3: Diagrams describing respectively, from left to right, the WE, PI and WA topologies.

traction with $G^{\mu\nu}$, while the chromo-magnetic operator \mathcal{O}_{mag} is defined in Eq. (1.155). From Eqs. (2.53), (2.55), (2.58), and (2.10), we finally obtain, in agreement with Ref. [130], that

$$\left. \Gamma_5^{(c\bar{u}d)} \right|_{\mathrm{I}} = -\frac{G_F^2 m_b^3}{192\pi^3} \left| V_{cb} \right|^2 \left| V_{ud} \right|^2 2 C_1 C_2 \, 8(1-\rho)^3 \, \mu_G^2(B) \,, \tag{2.59}$$

where the dimensionless mass parameter ρ is again defined as $\rho = m_c^2/m_b^2$ and the non perturbative parameter μ_G^2 is given in Eq. (1.159).

The second type of contribution (II) in Eq. (2.36) is obtained by expanding the expression in Eq. (2.33) up to order $1/m_b^2$. Using the results in Appendix A, a slightly long yet straightforward computation, which can be easily performed with e.g. Mathematica [154], leads to

$$\Gamma_{5}^{(c\bar{u}d)}\Big|_{\mathrm{II}} = \frac{G_{F}^{2}m_{b}^{3}}{192\pi^{3}} |V_{cb}|^{2} |V_{ud}|^{2} \left(N_{c} \left(C_{1}^{2}+C_{2}^{2}\right)+2 C_{1}C_{2}\right) \\ \times \left[\left(1-8\rho-12\rho^{2}\log\rho+8\rho^{3}-\rho^{4}\right)\left(-\frac{\mu_{\pi}^{2}(B)}{2}\right) +\frac{1}{2}\left(-3+8\rho-12\rho^{2}\log\rho-24\rho^{2}+24\rho^{3}-5\rho^{4}\right)\mu_{G}^{2}(B)\right], \qquad (2.60)$$

with the non perturbative parameter μ_{π}^2 given in Eq. (1.159).

2.2 Contribution of four-quark operators

We now turn to discuss the contribution of four-quark operators. By applying the Wick's theorem in Eq. (2.6), we consider all possible contractions of two pairs of

light-quark fields in the time ordered product, which leave the *b*-quark as well as a pair of light-quark fields uncontracted, namely

$$T\left\{ \left(\bar{q}_{1}(x) \Gamma_{\mu} b(x) \bar{q}_{3}(x) \Gamma^{\mu} q_{2}(x) \right), \left(\bar{b} \Gamma_{\nu} q_{1} \bar{q}_{2} \Gamma^{\nu} q_{3} \right) \right\}$$

$$=: \bar{q}_{1}(x) \Gamma_{\mu} b(x) \bar{q}_{3}(x) \Gamma^{\mu} \bar{q}_{2}(x) \bar{b} \Gamma_{\nu} q_{1} \bar{q}_{2} \Gamma^{\nu} q_{3}:$$

$$+: \bar{q}_{1}(x) \Gamma_{\mu} b(x) \bar{q}_{3}(x) \Gamma^{\mu} q_{2}(x) \bar{b} \Gamma_{\nu} q_{1} \bar{q}_{2} \Gamma^{\nu} q_{3}:$$

$$+: \bar{q}_{1}(x) \Gamma_{\mu} b(x) \bar{q}_{3}(x) \Gamma^{\mu} \bar{q}_{2}(x) \bar{b} \Gamma_{\nu} q_{1} \bar{q}_{2} \Gamma^{\nu} q_{3}:. \qquad (2.61)$$

Note that for simplicity we have omitted to specify the colour structure. The three terms on the r.h.s. of Eq. (2.61) generate different topologies, usually referred to, respectively, as Weak Exchange (WE), Pauli Interference (PI) and Weak Annihilation (WA)¹⁰, schematically shown in Figure 2.3. In analogy to Eq. (2.10) we then introduce the notation

$$\mathcal{T}^{(4q)} = \mathcal{T}^{WE} + \mathcal{T}^{PI} + \mathcal{T}^{WA}, \qquad (2.62)$$

where the superscript (4q) indicates that four-quark fields are not contracted and

$$\mathcal{T}^{X} = \left(C_{1}^{2} \,\mathcal{T}_{11}^{X} + 2 \,C_{1} \,C_{2} \,\mathcal{T}_{12}^{X} + C_{2}^{2} \,\mathcal{T}_{22}^{X} \right), \tag{2.63}$$

with X labelling the specific topology. We emphasise that the three contributions in Eq. (2.63) correspond to the same Dirac structure and differ only by the contraction of the colour indices, namely, starting with the case of Weak Exchange, we can conveniently define the following tensor in colour space

$$\mathcal{A}_{klrs}^{ijmn} = \frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 i \int d^4x \left(\bar{b}^i \gamma_\nu (1 - \gamma_5) i S_{kl}^{(q_1)}(0, x) \gamma_\mu (1 - \gamma_5) b^j(x) \right) \\ \times \bar{q}_3^m(x) \gamma^\mu (1 - \gamma_5) i S_{rs}^{(q_2)}(x, 0) \gamma^\nu (1 - \gamma_5) q_3^n + (x \leftrightarrow 0), \qquad (2.64)$$

¹⁰It is worth mentioning that for the description of baryons, the WE and PI topologies are interchanged.

so that the three terms in Eq. (2.63) are respectively given by

$$\mathcal{T}_{11}^{\mathrm{WE}} = \mathcal{A}_{klrs}^{ijmn} \,\delta_k^i \delta_l^j \delta_r^m \delta_s^n \,, \quad \mathcal{T}_{12}^{\mathrm{WE}} = \mathcal{A}_{klrs}^{ijmn} \,\delta_s^i \delta_l^j \delta_r^m \delta_k^n \,, \quad \mathcal{T}_{22}^{\mathrm{WE}} = \mathcal{A}_{klrs}^{ijmn} \,\delta^{in} \delta^{jm} \delta_{ks} \delta_{lr} \,.$$

$$(2.65)$$

Similarly, for the Pauli Interference topology, we introduce the tensor

$$\mathcal{B}_{klrs}^{ijmn} = \frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 \ i \int d^4x \left(\bar{b}^i \gamma_\nu (1 - \gamma_5) \ i S_{kl}^{(q_1)}(0, x) \gamma_\mu (1 - \gamma_5) b^j(x) \right) \\ \times \ \bar{q}_2^m \ \gamma^\nu (1 - \gamma_5) \ i S_{rs}^{(q_3)}(0, x) \ \gamma^\mu (1 - \gamma_5) \ q_2^n(x) \right) + (x \leftrightarrow 0) , \qquad (2.66)$$

and the three contributions in Eq. (2.63) read

$$\mathcal{T}_{11}^{\mathrm{PI}} = \mathcal{B}_{klrs}^{ijmn} \,\delta_k^i \delta_l^j \delta_r^m \,\delta_s^n \,, \quad \mathcal{T}_{12}^{\mathrm{PI}} = \mathcal{B}_{klrs}^{ijmn} \,\delta_k^i \delta_l^j \delta_l^m \,\delta_s^n \,, \quad \mathcal{T}_{22}^{\mathrm{PI}} = \mathcal{T}_{11}^{\mathrm{PI}} \,. \tag{2.67}$$

Finally, in the case of Weak Annihilation, we define

$$\mathcal{C}_{klrs}^{ijmn} = -\frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 \ i \int d^4x \left(\bar{b}^i \gamma_\nu (1-\gamma_5) q_1^j \bar{q}_1^m(x) \gamma_\mu (1-\gamma_5) b^n(x) \right) \\ \times \operatorname{Tr} \left[\gamma^\nu (1-\gamma_5) i S_{kl}^{(q_2)}(0,x) \gamma^\mu (1-\gamma_5) i S_{rs}^{(q_3)}(x,0) \right] + (x \leftrightarrow 0) , \qquad (2.68)$$

where the minus sign and the trace over spinor indices follow from the fermion-loop and

$$\mathcal{T}_{11}^{\mathrm{WA}} = \mathcal{C}_{klrs}^{ijmn} \,\delta^{ij} \delta^{mn} \delta_{ks} \delta_{lr} \,, \quad \mathcal{T}_{12}^{\mathrm{WA}} = \mathcal{C}_{klrs}^{ijmn} \,\delta^{i}_{k} \delta^{j}_{s} \delta_{lr} \delta^{mn} \,, \quad \mathcal{T}_{22}^{\mathrm{WA}} = \mathcal{C}_{klrs}^{ijmn} \,\delta^{i}_{k} \delta^{j}_{s} \delta^{m}_{r} \delta^{n}_{l} \,.$$

$$(2.69)$$

2.2.1 Computation of $\text{Im}\mathcal{T}_6^{(4q)}$

To compute the leading power corrections to the WE, PI and WA topologies, namely the dimension-six contribution to the four-quark operators, we can ignore the QCD interaction with the background field and set in Eqs. (2.64), (2.66), (2.68)

$$S_{jk}^{(q_i)}(x,y) = S_0^{(q_i)}(x-y)\,\delta_{jk}\,, \qquad i = 1, 2, 3\,, \tag{2.70}$$

where $S_0^{(q_i)}(x-y)$ is the free-quark propagator associated to q_i , and note that we consider now the general case of $b \to q_1 \bar{q}_2 q_3$ transition. Higher order terms in Eq. (2.70), cf. Eqs. (1.76), (1.121), describe soft gluon corrections to the four-quark operators and lead to contributions to the HQE of at least dimension-eight. These will not be discussed in the present work. However now, the loop-computation is more easily performed in momentum space and we will use, contrary to what we have done in the previous section, the Fourier representation of the propagator in Eq. (2.70). Substituting Eq. (1.73) into the first of the three expressions in Eq. (2.65) yields ¹¹

$$\mathcal{T}_{11}^{\text{WE}} = -G_F^2 |V_{q_1b}|^2 |V_{q_3q_2}|^2 i \int d^4x \int \frac{d^4l}{(2\pi)^4} \int \frac{d^4k}{(2\pi)^4} e^{-i(p+k-l)\cdot x} \left(\bar{b}^i \gamma_\nu (1-\gamma_5) \right) \\ \times \left(\frac{l}{l^2 - m_1^2 + i\varepsilon}\right) \gamma_\mu (1-\gamma_5) b^i \bar{q}_3^j \gamma^\mu (1-\gamma_5) \left(\frac{l}{k^2 - m_2^2 + i\varepsilon}\right) \gamma^\nu (1-\gamma_5) q_3^j ,$$

$$(2.71)$$

where $p^{\mu} = p_b^{\mu} + p_{q_3}^{\mu}$. In deriving Eq. (2.71) we have used that b(x) and $\bar{q}_3(x)$ describe respectively, an incoming quark with momentum p_b^{μ} and an incoming antiquark with momentum $p_{q_3}^{\mu}$ ¹². We notice that a symmetry factor of 2 due to translation invariance has been already included and that the chiral structure of Eq. (2.71) implies that all terms in the propagator proportional to an even number of gamma matrices must vanish. Performing the integration over the variables x^{μ} and k^{μ} , leads to

$$\mathcal{T}_{11}^{\text{WE}} = -4 G_F^2 |V_{q_1 b}|^2 |V_{q_3 q_2}|^2 i \int \frac{d^4 l}{(2\pi)^4} \frac{\left(l^{\rho} l^{\sigma} - p^{\rho} l^{\sigma}\right)}{(l^2 - m_1^2 + i\varepsilon)((l-p)^2 - m_2^2 + i\varepsilon)} \\ \times g_{\sigma\xi} \left(\bar{b}^i \gamma_{\nu} (1-\gamma_5) \gamma_{\rho} \gamma_{\mu} b^i\right) \left(\bar{q}_3^j \gamma^{\mu} \gamma^{\xi} \gamma^{\nu} (1-\gamma_5) q_3^j\right), \qquad (2.72)$$

where the structure of the four-quark operator in Eq. (2.72) can be simplified taking into account the tensor decomposition of three gamma matrices in Eq. (7) and the Fierz identity in Eq. (2.15), i.e.

¹¹Note that we take into account the replacement in Eq. (2.26), supplemented with the corresponding one for the light spectator antiquark.

¹²For baryons, being both b and q_3 quarks, it would be $p^{\mu} = p_b^{\mu} - p_{q_3}^{\mu}$.

$$\left(\bar{b}^{i}\gamma_{\nu}(1-\gamma_{5})\gamma_{\rho}\gamma_{\mu}b^{i}\right)\left(\bar{q}_{3}^{j}\gamma^{\mu}\gamma^{\xi}\gamma^{\nu}(1-\gamma_{5})q_{3}^{j}\right) = 4\left(\bar{b}^{i}\gamma^{\xi}(1-\gamma_{5})q_{3}^{j}\right)\left(\bar{q}_{3}^{j}\gamma_{\rho}(1-\gamma_{5})b^{i}\right).$$
(2.73)

Using the Passarino-Veltman reduction algorithm [155], the one-loop tensor integral in Eq. (2.72) can be decomposed in terms of one- and two-point one-loop scalar integrals. Note that the latter are ultraviolet (UV) divergent and must be regularised. Performing the computation in dimensional regularisation [47–50] with $D = 4 - 2\epsilon$ space-time dimensions, one obtains the results listed in Appendix B, in which the singularity appears as a single pole in ϵ , see Eqs. (B.4), (B.5). However, being interested only in the imaginary part of \mathcal{T}_{11}^{WE} which, cf. Eqs. (B.8), (B.9), is finite for $\epsilon \to 0$, we can use the expressions in Eqs. (B.10)-(B.20) setting D = 4. This gives

$$\operatorname{Im}\left(i\int\frac{d^{4}l}{(2\pi)^{4}}\frac{\left(l^{\rho}l^{\sigma}-p^{\rho}l^{\sigma}\right)}{(l^{2}-m_{1}^{2}+i\varepsilon)((l-p)^{2}-m_{2}^{2}+i\varepsilon)}\right) = \frac{p^{2}\sqrt{\lambda(1,r_{1},r_{2})}}{48\pi}\left[g^{\rho\sigma}\left(2-r_{1}-r_{2}-(r_{1}-r_{2})^{2}\right)+2\frac{p^{\rho}p^{\sigma}}{p^{2}}\left(2(r_{1}-r_{2})^{2}-(1+r_{1}+r_{2})\right)\right]$$

$$(2.74)$$

where $r_i = m_i^2/p^2$ and $\lambda(a, b, c) \equiv (a - b - c)^2 - 4bc$ is the Källen function. Notice that Eq. (2.74) is a symmetric function under the exchange $m_1 \leftrightarrow m_2$. Substituting Eqs. (2.73), (2.74) into Eq. (2.72), we then obtain

$$\operatorname{Im} \mathcal{T}_{11}^{\mathrm{WE}} = \frac{G_F^2}{12\pi} |V_{q_1b}|^2 |V_{q_3q_2}|^2 p^2 \sqrt{\lambda(1, r_1, r_2)} \\ \times \left[\left((r_1 - r_2)^2 + r_1 + r_2 - 2 \right) \left(\bar{b}^i \Gamma_\mu q_3^j \right) \left(\bar{q}_3^j \Gamma^\mu b^i \right) \right. \\ \left. - 2 \frac{p^\mu p^\nu}{p^2} \left(2(r_1 - r_2)^2 - (1 + r_1 + r_2) \right) \left(\bar{b}^i \Gamma_\mu q_3^j \right) \left(\bar{q}_3^j \Gamma_\nu b^i \right) \right].$$
(2.75)

Finally, we can express the colour-rearranged operators in Eq. (2.75) in terms of colour-singlet and colour-octet ones using Eq. (2.9), this yields

$$\operatorname{Im} \mathcal{T}_{11}^{\mathrm{WE}} = \frac{G_F^2}{12\pi} |V_{q_1b}|^2 |V_{q_3q_2}|^2 p^2 \sqrt{\lambda(1, r_1, r_2)} \left\{ \left((r_1 - r_2)^2 + r_1 + r_2 - 2 \right) \right. \\ \left. \times \left[\frac{1}{N_c} \left(\bar{b}^i \Gamma_{\nu} q_3^i \right) \left(\bar{q}_3^j \Gamma^{\nu} b^j \right) + 2 \left(\bar{b}^i \Gamma_{\nu} t_{ij}^a q_3^j \right) \left(\bar{q}_3^l \Gamma^{\nu} t_{lm}^a b^m \right) \right] \right. \\ \left. - 2 \frac{p^{\mu} p^{\nu}}{p^2} \left(2(r_1 - r_2)^2 - (1 + r_1 + r_2) \right) \left[\frac{1}{N_c} \left(\bar{b}^i \Gamma_{\mu} q_3^i \right) \left(\bar{q}_3^j \Gamma_{\nu} b^j \right) \right. \\ \left. + 2 \left(\bar{b}^i \Gamma_{\mu} q_3^i \right) \left(\bar{q}_3^j \Gamma_{\nu} b^j \right) \right] \right\}.$$

$$(2.76)$$

To compute the remaining colour structures we substitute Eq. (1.73) into the second and third term of Eq. (2.65), obtaining respectively

$$\mathcal{T}_{12}^{\text{WE}} = -G_F^2 |V_{q_1b}|^2 |V_{q_3q_2}|^2 i \int d^4x \int \frac{d^4l}{(2\pi)^4} \int \frac{d^4k}{(2\pi)^4} e^{-i(p+k-l)\cdot x} \left[\bar{b}^i \gamma_\nu (1-\gamma_5) \right] \\ \times \left(\frac{l}{l^2 - m_1^2 + i\varepsilon} \right) \gamma_\mu (1-\gamma_5) b^j \bar{q}_3^j \gamma^\mu (1-\gamma_5) \left(\frac{k}{k^2 - m_2^2 + i\varepsilon} \right) \gamma^\nu (1-\gamma_5) q_3^i \right],$$
(2.77)

and

$$\mathcal{T}_{22}^{\rm WE} = N_c \, \mathcal{T}_{12}^{\rm WE} \,,$$
 (2.78)

with the colour factor in Eq. (2.78) following from $\delta_{ii} = N_c$. The calculation of Eq. (2.77) proceeds in the very same way as for Eq. (2.71). For brevity we only state here the final expression which reads

$$\operatorname{Im} \mathcal{T}_{12}^{\mathrm{WE}} = \frac{G_F^2}{12\pi} |V_{q_1b}|^2 |V_{q_3q_2}|^2 p^2 \sqrt{\lambda(1, r_1, r_2)} \left[\left((r_1 - r_2)^2 + r_1 + r_2 - 2 \right) \right. \\ \left. \times \left(\bar{b}^i \Gamma_{\nu} q_3^i \right) \left(\bar{q}_3^j \Gamma^{\nu} b^j \right) - 2 \frac{p^{\mu} p^{\nu}}{p^2} \left(2(r_1 - r_2)^2 - (1 + r_1 + r_2) \right) \left(\bar{b}^i \Gamma_{\mu} q_3^i \right) \left(\bar{q}_3^j \Gamma_{\nu} b^j \right) \right],$$

$$(2.79)$$

note that in this case, the result contains already colour-singlet and colour-rearranged operators and we do not need to further use Eq. (2.9). Substituting Eqs. (2.76), (2.79), and (2.78), into Eq. (2.63), we readily arrive at

$$\operatorname{Im} \mathcal{T}^{\mathrm{WE}} = \frac{G_F^2}{12\pi} |V_{q_1 b}|^2 |V_{q_3 q_2}|^2 p^2 \sqrt{\lambda(1, r_1, r_2)} \left\{ k_1 \left[\omega_1(r_1, r_2) \left(\bar{b}^i \Gamma_\mu q_3^i \right) \left(\bar{q}_3^j \Gamma^\mu b^j \right) - 2 \frac{p^\mu p^\nu}{p^2} \omega_2(r_1, r_2) \left(\bar{b}^i \Gamma_\mu q_3^i \right) \left(\bar{q}_3^j \Gamma_\nu b^j \right) \right] + k_2 \left[\omega_1(r_1, r_2) \left(\bar{b}^i \Gamma_\mu t_{ij}^a q_3^j \right) \left(\bar{q}_3^l \Gamma^\mu t_{lm}^a b^m \right) - 2 \frac{p^\mu p^\nu}{p^2} \omega_2(r_1, r_2) \left(\bar{b}^i \Gamma_\mu t_{ij}^a q_3^j \right) \left(\bar{q}_3^l \Gamma_\nu t_{lm}^a b^m \right) \right] \right\} + \dots,$$

$$(2.80)$$

where the ellipsis refer to power suppressed contributions arising from corrections to the propagator of order $\mathcal{O}(1/m_b^2)$ and for the sake of a more compact notation we have introduced the following combinations of Wilson coefficients

$$k_1 = \frac{1}{N_c} C_1^2 + 2 C_1 C_2 + N_c C_2^2, \qquad k_2 = 2 C_1^2, \qquad (2.81)$$

while

$$\omega_1(a,b) = (a-b)^2 + a + b - 2, \qquad \omega_2(a,b) = 2(a-b)^2 - (1+a+b).$$
(2.82)

We now turn to consider the contribution of Pauli Interference. Inserting Eq. (1.73) into the first expression in Eq. (2.67), gives

$$\mathcal{T}_{11}^{\mathrm{PI}} = -G_F^2 |V_{q_1b}|^2 |V_{q_3q_2}|^2 i \int d^4x \int \frac{d^4l}{(2\pi)^4} \int \frac{d^4k}{(2\pi)^4} e^{-i(p-k-l)\cdot x} \left(\bar{b}^i \gamma_\nu (1-\gamma_5) \right) \\ \times \left(\frac{k}{k^2 - m_1^2 + i\varepsilon} \right) \gamma_\mu (1-\gamma_5) b^i \bar{q}_2^j \gamma^\nu (1-\gamma_5) \left(\frac{l}{l^2 - m_3^2 + i\varepsilon} \right) \gamma^\mu (1-\gamma_5) q_2^j ,$$
(2.83)

here $p^{\mu} = p^{\mu}_b - p^{\mu}_{q_2}$, which follows from the fact that $q_2(x)$ describes an outgoing an-

tiquark with momentum $p_{q_2}^{\mu \ 13}$. The integration over the variables x^{μ} and k^{μ} yields

$$\mathcal{T}_{11}^{\mathrm{PI}} = 4 \, G_F^2 |V_{q_1 b}|^2 |V_{q_3 q_2}|^2 \, i \, \int \frac{d^4 l}{(2\pi)^4} \, \frac{(l^{\rho} l^{\sigma} - p^{\rho} l^{\sigma})}{(l^2 - m_3^2 + i\varepsilon)((l-p)^2 - m_1^2 + i\varepsilon)} \\ \times \, g_{\sigma\xi} \left(\bar{b}^i \gamma_{\nu} (1 - \gamma_5) \gamma_{\rho} \gamma_{\mu} b^i \right) \left(\bar{q}_3^j \gamma^{\nu} \gamma^{\xi} \gamma^{\mu} (1 - \gamma_5) q_3^j \right), \tag{2.84}$$

where the four-quark operator can be simplified as in Eq. (2.73), again using Eq. (2.9), while the imaginary part of the one-loop integral in Eq. (2.84) is obtained, taking into account the symmetry under the exchange $m_1 \leftrightarrow m_3$, by replacing $r_2 \rightarrow r_3$ into Eq. (2.74). A straightforward calculation leads to

$$\operatorname{Im} \mathcal{T}_{11}^{\mathrm{PI}} = \frac{G_F^2}{2\pi} |V_{q_1 b}|^2 |V_{q_3 q_2}|^2 p^2 \sqrt{\lambda(1, r_1, r_3)} \left(1 - r_1 - r_3\right) \\ \times \left(\frac{1}{N_c} \left(\bar{b}^i \Gamma_{\mu} q_2^i\right) \left(\bar{q}_2^j \Gamma^{\mu} b^j\right) + 2 \left(\bar{b}^i \Gamma_{\mu} t_{ij}^a q_2^j\right) \left(\bar{q}_2^l \Gamma^{\mu} t_{lm}^a b^m\right)\right).$$
(2.85)

The remaining colour structure is obtained by substituting Eq. (1.73) into the second term of Eq. (2.67), i.e.

$$\mathcal{T}_{12}^{\mathrm{PI}} = -G_F^2 |V_{q_1b}|^2 |V_{q_3q_2}|^2 i \int d^4x \int \frac{d^4l}{(2\pi)^4} \int \frac{d^4k}{(2\pi)^4} e^{-i(p-k-l)\cdot x} \left(\bar{b}^i \gamma_\nu (1-\gamma_5) \right) \\ \times \left(\frac{k}{k^2 - m_1^2 + i\varepsilon} \right) \gamma_\mu (1-\gamma_5) b^j \bar{q}_2^j \gamma^\nu (1-\gamma_5) \left(\frac{l}{l^2 - m_3^2 + i\varepsilon} \right) \gamma^\mu (1-\gamma_5) q_2^i \right).$$
(2.86)

All the steps proceed analogously to the case of $\mathcal{T}_{11}^{\text{PI}}$, the only difference being that the result is already expressed in terms of colour-singlet operators and we do not need to use Eq. (2.9). It is in fact easy to show that

$$\operatorname{Im} \mathcal{T}_{12}^{\mathrm{PI}} = \frac{G_F^2}{2\pi} |V_{q_1b}|^2 |V_{q_3q_2}|^2 p^2 \sqrt{\lambda(1, r_1, r_3)} (1 - r_1 - r_3) \left(\bar{b}^i \Gamma_{\mu} q_2^i\right) \left(\bar{q}_2^j \Gamma^{\mu} b^j\right), \quad (2.87)$$

¹³For baryons, being both b and q_2 quarks, it would be $p^{\mu} = p_b^{\mu} + p_{q_2}^{\mu}$.

and from Eqs. (2.85), (2.87), (2.63) we readily obtain that

$$\operatorname{Im} \mathcal{T}^{\mathrm{PI}} = \frac{G_F^2}{2\pi} |V_{q_1 b}|^2 |V_{q_3 q_2}|^2 p^2 \sqrt{\lambda(1, r_1, r_3)} \left(1 - r_1 - r_3\right) \\ \times \left[k_3 \left(\bar{b}^i \Gamma_\mu q_2^i\right) \left(\bar{q}_2^j \Gamma^\mu b^j\right) + k_4 \left(\bar{b}^i \Gamma_\mu t_{ij}^a q_2^j\right) \left(\bar{q}_2^l \Gamma^\mu t_{lm}^a b^m\right) \right] + \dots, \qquad (2.88)$$

with

$$k_3 = \frac{1}{N_c} \left(C_1^2 + C_2^2 \right) + 2 C_1 C_2, \qquad k_4 = 2 \left(C_1^2 + C_2^2 \right). \tag{2.89}$$

Finally we discuss the Weak Annihilation topology. Substituting Eq. (1.73) into the first expression of Eq. (2.69) yields

$$\mathcal{T}_{11}^{\text{WA}} = N_c \, G_F^2 |V_{q_1 b}|^2 |V_{q_3 q_2}|^2 i \int d^4 x \int \frac{d^4 l}{(2\pi)^4} \int \frac{d^4 k}{(2\pi)^4} e^{-i(p+k-l)\cdot x} \left(\bar{b}^i \gamma_\nu (1-\gamma_5) q_1^i\right) \\ \times \left(\bar{q}_1^j \gamma_\mu (1-\gamma_5) b^j\right) \text{Tr} \left[\gamma^\nu (1-\gamma_5) \left(\frac{\not l}{l^2 - m_2^2 + i\varepsilon}\right) \gamma^\mu (1-\gamma_5) \left(\frac{\not k}{k^2 - m_3^2 + i\varepsilon}\right)\right]$$
(2.90)

where $p^{\mu} = p_b^{\mu} + p_{q_1}^{\mu}$, due to the fact that $\bar{q}_1(x)$ describes an incoming antiparticle with momentum $p_{q_1}^{\mu}$ ¹⁴, and we have taken into account that only terms in the propagator proportional to an odd number of gamma matrices contribute to the trace. Performing the integration over the variables x^{μ} and k^{μ} and evaluating the trace in Eq. (2.90), leads to

$$\mathcal{T}_{11}^{\text{WA}} = 8N_c G_F^2 |V_{q_1 b}|^2 |V_{q_3 q_2}|^2 i \int \frac{d^4 l}{(2\pi)^4} \frac{l_\rho l_\sigma - p_\rho l_\sigma}{(l^2 - m_2^2 + i\varepsilon)((l-p)^2 - m_3^2 + i\varepsilon)} \\ \times \left(\bar{b}^i \gamma_\nu (1-\gamma_5) q_1^i\right) \left(\bar{q}_1^j \gamma_\mu (1-\gamma_5) b^j\right) \left(g^{\nu\rho} g^{\mu\sigma} + g^{\nu\sigma} g^{\mu\rho} - g^{\nu\mu} g^{\rho\sigma} + i\epsilon^{\nu\rho\sigma\mu}\right).$$
(2.91)

Again, since the one-loop integral in Eq. (2.91) is symmetric under the exchange

¹⁴For baryons, being both b and q_1 quarks, it would be $p^{\mu} = p_b^{\mu} - p_{q_1}^{\mu}$.

 $m_2 \leftrightarrow m_3$, its imaginary part can be obtained from Eq. (2.74) making the replacement $r_1 \rightarrow r_3$, namely

$$\operatorname{Im} \mathcal{T}_{11}^{\mathrm{WA}} = \frac{N_c}{12\pi} G_F^2 |V_{q_1 b}|^2 |V_{q_3 q_2}|^2 p^2 \sqrt{\lambda(1, r_3, r_2)} \left[\left((r_3 - r_2)^2 + r_3 + r_2 - 2 \right) \right. \\ \left. \times \left(\bar{b}^i \Gamma_\nu q_1^i \right) \left(\bar{q}_1^j \Gamma^\nu b^j \right) - 2 \frac{p^\mu p^\nu}{p^2} \left(2(r_3 - r_2)^2 - (1 + r_3 + r_2) \right) \left(\bar{b}^i \Gamma_\mu q_1^i \right) \left(\bar{q}_1^j \Gamma_\nu b^j \right) \right]$$

$$(2.92)$$

Note that Eq. (2.92) reproduces $\text{Im}\mathcal{T}_{22}^{\text{WE}}$ in Eq. (2.78) with the exchange $q_1 \leftrightarrow q_3$. This result is consequence of the Fierz identity in Eq. (2.15), and by taking into account Eq. (2.16) it follows that

$$\mathcal{T}_{mn}^{\mathrm{WA}} = \mathcal{T}_{nm}^{\mathrm{WE}(q_1 \leftrightarrow q_3)} \,. \tag{2.93}$$

From Eq. (2.93), we see that the expression for $\text{Im}\mathcal{T}^{\text{WA}}$ can be immediately obtained from Eq. (2.80) by replacing $C_1 \leftrightarrow C_2$, and $q_1 \leftrightarrow q_3$, namely

$$\operatorname{Im} \mathcal{T}^{\mathrm{WA}} = \frac{G_F^2}{12\pi} |V_{q_1b}|^2 |V_{q_3q_2}|^2 p^2 \sqrt{\lambda(1, r_3, r_2)} \left\{ k_5 \left[\omega_1(r_3, r_2) \left(\bar{b}^i \Gamma_{\mu} q_1^i \right) \left(\bar{q}_1^j \Gamma^{\mu} b^j \right) - 2 \frac{p^{\mu} p^{\nu}}{p^2} \omega_2(r_3, r_2) \left(\bar{b}^i \Gamma_{\mu} q_1^i \right) \left(\bar{q}_1^j \Gamma_{\nu} b^j \right) \right] + k_6 \left[\omega_1(r_3, r_2) \left(\bar{b}^i \Gamma_{\mu} t_{ij}^a q_1^j \right) \left(\bar{q}_1^l \Gamma^{\mu} t_{lm}^a b^m \right) - 2 \frac{p^{\mu} p^{\nu}}{p^2} \omega_2(r_3, r_2) \left(\bar{b}^i \Gamma_{\mu} t_{ij}^a q_1^j \right) \left(\bar{q}_1^l \Gamma_{\nu} t_{lm}^a b^m \right) \right] \right\} + \dots, \qquad (2.94)$$

with

$$k_5 = N_c C_1^2 + 2 C_1 C_2 + \frac{1}{N_c} C_2^2, \qquad k_6 = 2 C_2^2.$$
(2.95)

In deriving Eqs. (2.80), (2.88) and (2.94) we have only neglected power corrections due to the expansion of the quark propagator, however, because p^{μ} depends on the residual component of the heavy quark momentum k^{μ} as well as on the soft momentum of the light spectator quark $p_{q_i}^{\mu}$, the expressions above contain also the information about the interaction of the heavy quark with the light degrees of freedom. In order to single out the dimension-six result, we set $p^{\mu} = m_b v^{\mu} + \mathcal{O}(1/m_b)$ and introduce the rescaled field $b(x) = \exp(-im_b v \cdot x)b_v(x)$, see Eq. (1.145), satisfying $b_v(x) = \psi b_v(x) + \mathcal{O}(1/m_b)$, cf. Eq. (1.148). We then obtain that

$$\frac{p^{\mu}p^{\mu}}{p^2} \left(\bar{b}\gamma_{\mu}(1-\gamma_5)q_i\right) \left(\bar{q}_i\gamma_{\nu}(1-\gamma_5)b\right) = \left(\bar{b}_v(1-\gamma_5)q_i\right) \left(\bar{q}_i(1+\gamma_5)b_v\right) + \mathcal{O}\left(\frac{1}{m_b}\right),$$
(2.96)

where for simplicity we have not specified the colour structure, and that the dimensionsix contributions to $\text{Im}\mathcal{T}^{\text{WE}}$, $\text{Im}\mathcal{T}^{\text{PI}}$ and $\text{Im}\mathcal{T}^{\text{WA}}$ are given respectively by

$$\operatorname{Im} \mathcal{T}_{6}^{\mathrm{WE}} = \frac{G_{F}^{2}}{12\pi} |V_{q_{1}b}|^{2} |V_{q_{3}q_{2}}|^{2} m_{b}^{2} \sqrt{\lambda(1, x_{1}, x_{2})} \left\{ k_{1} \Big[\omega_{1}(x_{1}, x_{2}) O_{1}^{(q_{3})} - 2\omega_{2}(x_{1}, x_{2}) O_{2}^{(q_{3})} \Big] + k_{2} \Big[\omega_{1}(x_{1}, x_{2}) \tilde{O}_{1}^{(q_{3})} - 2\omega_{2}(x_{1}, x_{2}) \tilde{O}_{2}^{(q_{3})} \Big] \right\},$$

$$(2.97)$$

$$\operatorname{Im} \mathcal{T}_{6}^{\mathrm{PI}} = \frac{G_{F}^{2}}{2\pi} |V_{q_{1}b}|^{2} |V_{q_{3}q_{2}}|^{2} m_{b}^{2} \sqrt{\lambda(1, x_{1}, x_{3})} \left(1 - x_{1} - x_{3}\right) \left[k_{3} O_{1}^{(q_{2})} + k_{4} \tilde{O}_{1}^{(q_{2})}\right],$$
(2.98)

and

$$\operatorname{Im} \mathcal{T}_{6}^{\operatorname{WA}} = \frac{G_{F}^{2}}{12\pi} |V_{q_{1}b}|^{2} |V_{q_{3}q_{2}}|^{2} m_{b}^{2} \sqrt{\lambda(1, x_{3}, x_{2})} \left\{ k_{5} \left[\omega_{1}(x_{3}, x_{2}) O_{1}^{(q_{1})} - 2\omega_{2}(x_{3}, x_{2}) O_{2}^{(q_{1})} \right] + k_{6} \left[\omega_{1}(x_{3}, x_{2}) \tilde{O}_{1}^{(q_{1})} - 2\omega_{2}(x_{3}, x_{2}) \tilde{O}_{2}^{(q_{1})} \right] \right\}.$$

$$(2.99)$$

In Eqs. (2.97)-(2.99), x_i denotes the dimensionless mass parameters $x_i = m_i^2/m_b^2$, and the following basis for the dimension-six four-quark operators has been introduced

$$O_1^{(q)} = \left(\bar{b}_v^i \gamma_\nu (1-\gamma_5) q^i\right) \left(\bar{q}^j \gamma^\nu (1-\gamma_5) b_v^j\right), \qquad (2.100)$$

$$O_2^{(q)} = \left(\bar{b}_v^i (1 - \gamma_5) q^i\right) \left(\bar{q}^j (1 + \gamma_5) b_v^j\right), \qquad (2.101)$$

$$\tilde{O}_{1}^{(q)} = \left(\bar{b}_{v}^{i}\gamma_{\nu}(1-\gamma_{5})t_{ij}^{a}q^{j}\right)\left(\bar{q}^{l}\gamma^{\nu}(1-\gamma_{5})t_{lm}^{a}b_{v}^{m}\right),$$
(2.102)

$$\tilde{O}_{2}^{(q)} = \left(\bar{b}_{v}^{i}(1-\gamma_{5})t_{ij}^{a}q^{j}\right)\left(\bar{q}^{l}(1+\gamma_{5})t_{lm}^{a}b_{v}^{m}\right).$$
(2.103)

2.2.2 Computation of $\text{Im}\mathcal{T}_7^{(4q)}$

By including $1/m_b$ corrections to the incoming four-momentum p^{μ} as well as to the heavy quark field b(x), in Eqs. (2.80), (2.88), and (2.94), leads to dimension-seven contributions to the WE, PI, and WA topologies. In the following, we discuss in detail only the first two cases since the corresponding expression for WA can be immediately obtained using Eq. (2.93). However, before considering the specific contributions separately, it is convenient to derive some general results which will facilitate the computation. We recall that the incoming momentum p^{μ} is the sum (for WE and WA), or the difference (for PI), of the *b*-quark and of the light-quark momentum, i.e. $p^{\mu} = p_b^{\mu} \pm p_q^{\mu}$, here $p_b^{\mu} = m_b v^{\mu} + k^{\mu}$, while $q = q_1, q_2, q_3$, respectively for the case of WA, PI and WE. Taking into account that $k \sim p_q \ll m_b$, the square of the incoming momentum can be written, up to terms of order $\mathcal{O}(1/m_b^2)$, as $p^2 = m_b^2 (1 + z)$, with the small parameter z given by

$$z = 2 \frac{v \cdot k}{m_b} \pm 2 \frac{v \cdot p_q}{m_b} \ll 1.$$
 (2.104)

Correspondingly, the heavy quark field b(x), using the framework of the HQET, see Section 1.2, can be expanded in powers of $1/(2m_b)$, as

$$b(x) = e^{-im_b v \cdot x} \left(h_v(x) + \frac{i \not D_\perp}{2m_b} h_v(x) + \mathcal{O}\left(\frac{1}{m_b^2}\right) \right) , \qquad (2.105)$$

where $h_v(x)$ denotes the effective heavy quark field, which coincides with $b_v(x)$ at leading order in $1/m_b$, and which obeys to the equation of motion $(iv \cdot D)h_v(x) = 0$, following from the HQET Lagrangian in Eq. (1.44). However, in the original QCD Lagrangian in Eq. (1.58), there are also subleading contributions suppressed by the heavy quark mass, which are treated as perturbations to \mathcal{L}_{HQET} , cf. \mathcal{L}_{power} in Eq. (1.59), so that, in the expansion of the matrix element of an operator containing the heavy quark field, their effect must be included in a standard perturbative way by taking the time ordered product of \mathcal{L}_{power} with the corresponding leading order operators, see Eq. (1.64). It then follows that

$$\bar{b}\Gamma_{\mu}q\bar{q}\Gamma_{\nu}b \simeq \bar{h}_{v}\Gamma_{\mu}q\bar{q}\Gamma_{\nu}h_{v} + \frac{1}{2m_{b}}\left[\bar{h}_{v}\Gamma_{\mu}q\bar{q}\Gamma_{\nu}i\not{D}h_{v} + \bar{h}_{v}(-i\not{D})\Gamma_{\mu}q\bar{q}\Gamma_{\nu}h_{v}\right] + \frac{1}{m_{b}}i\int d^{4}y \,\mathrm{T}\left\{\bar{h}_{v}\Gamma_{\mu}q\bar{q}\Gamma_{\nu}h_{v},\mathcal{L}_{1}(y)\right\} + \mathcal{O}\left(\frac{1}{m_{b}^{2}}\right), \qquad (2.106)$$

and for brevity we have not specified the colour structure. Note that in Eq. (2.106), the equation of motion for $h_v(x)$ has been used to replace the action of D^{μ}_{\perp} with that of D_{μ} and that the symbol \simeq refers to the fact that left and right hand side of the equation must be evaluated respectively between QCD and HQET states, cf. Eq. (1.64) and see for more details Ref. [109] ¹⁵. Moreover, the non local operator in the second line of Eq. (2.106), parametrises the contribution of the first power correction to the QCD Lagrangian, i.e. \mathcal{L}_1 in Eq. (1.65).

Taking into account that the action of a derivative on $h_v(x)$, returns only small frequencies of the order of k and that q(x) describes an outgoing antiquark with momentum p_q^{μ} , we respectively have ¹⁶

$$v \cdot k \left(\bar{h}_v \Gamma_\mu q \bar{q} \Gamma_\nu h_v \right) = \lim_{x \to 0} \left(\bar{h}_v(x) \Gamma_\mu q(x) \bar{q}(x) \Gamma_\nu i v \cdot D h_v(x) \right) = 0, \qquad (2.107)$$

while

$$v \cdot p_q \left(\bar{h}_v \Gamma_\mu q \bar{q} \Gamma_\nu h_v \right) = \lim_{x \to 0} \left(\bar{h}_v(x) \Gamma_\mu(-iv \cdot D) q(x) \bar{q}(x) \Gamma_\nu h_v(x) \right) = -\bar{h}_v \Gamma_\mu iv \cdot D q \bar{q} \Gamma_\nu h_v ,$$
(2.108)

where Eq. (2.107) vanishes due to the equation of motion for h_v , and in Eq. (2.108), the small momentum p_q has resulted in a dimension-seven operator with a covariant derivative acting on q. Consider now the expansion of the product of two momenta, namely

¹⁵However, for simplicity, in the following we will just use an equal sign.

¹⁶Recall that in the FS gauge $A_{\mu}(0) = 0$, and that the action of a partial derivative at the origin can be replaced with that of a covariant derivative, see Section 1.3 and also Appendix A.

$$p^{\mu}p^{\nu} = m_b^2 \left(v^{\mu}v^{\nu} + \frac{v^{\mu}k^{\nu}}{m_b} + \frac{v^{\nu}k^{\mu}}{m_b} + \frac{v^{\mu}p_q^{\nu}}{m_b} + \frac{v^{\nu}p_q^{\mu}}{m_b} + \mathcal{O}\left(\frac{1}{m_b^2}\right) \right) , \qquad (2.109)$$

which, combined with the result in Eq. (2.106), yields

$$p^{\mu}p^{\nu}\left(\bar{b}\Gamma_{\mu}q\bar{q}\Gamma_{\nu}b\right)$$

$$= m_{b}^{2}\left(\bar{h}_{v}(1-\gamma_{5})q\bar{q}(1+\gamma_{5})h_{v} + \frac{1}{m_{b}}\bar{h}_{v}(1-\gamma_{5})q\bar{q}(1+\gamma_{5})i\not\!\!Dh_{v}\right)$$

$$+ \frac{1}{m_{b}}i\int d^{4}y T\left\{\bar{h}_{v}(1-\gamma_{5})q\bar{q}(1+\gamma_{5})h_{v}, \mathcal{L}_{1}(y)\right\} - 2\frac{m_{q}}{m_{b}}\bar{h}_{v}(1-\gamma_{5})q\bar{q}(1-\gamma_{5})h_{v} + \dots\right),$$

$$(2.110)$$

where the ellipsis stand for terms of order $\mathcal{O}(1/m_b^2)$. Notice that in Eq. (2.110), the anti-commutation relation $\psi \not D = -\not D \psi + 2v \cdot D$, and the corresponding equations of motion for $h_v(x)$ and for the light field, i.e. $(i\not D - m_q)q(x) = 0$, have been used. Moreover, we have taken into account that operators related by Hermitian conjugation lead to the same matrix element, hence a factor of 2 has been included. Specifically these are

and

$$\bar{h}_v(1+\gamma_5)q\bar{q}(1+\gamma_5)h_v = \left[\bar{h}_v(1-\gamma_5)q\bar{q}(1-\gamma_5)h_v\right]^{\dagger}.$$
 (2.112)

Finally, we introduce the dimension-seven four-quark operator basis, which for clarity, we split into three categories, namely

$$\mathcal{P}_{1}^{(q)} = m_{q} \left(\bar{h}_{v}^{i} (1 - \gamma_{5}) q^{i} \right) \left(\bar{q}^{j} (1 - \gamma_{5}) h_{v}^{j} \right), \qquad (2.113)$$

$$\mathcal{P}_2^{(q)} = \left(\bar{h}_v^i \gamma_\mu (1-\gamma_5) iv \cdot Dq^i\right) \left(\bar{q}^j \gamma^\mu (1-\gamma_5) h_v^j\right), \qquad (2.114)$$

$$\mathcal{P}_{3}^{(q)} = \left(\bar{h}_{v}^{i}(1-\gamma_{5})iv \cdot Dq^{i}\right) \left(\bar{q}^{j}(1+\gamma_{5})h_{v}^{j}\right), \qquad (2.115)$$

parametrising the effect of the light-quark momentum p_q ,

$$\mathcal{R}_1^{(q)} = \left(\bar{h}_v^i \gamma_\mu (1-\gamma_5) q^i\right) \left(\bar{q}^j \gamma^\mu (1-\gamma_5) i \not\!\!D h_v^j\right), \qquad (2.116)$$

$$\mathcal{R}_2^{(q)} = \left(\bar{h}_v^i(1-\gamma_5)q^i\right) \left(\bar{q}^j(1+\gamma_5)i\mathcal{D}h_v^j\right), \qquad (2.117)$$

describing local contributions due to the expansion of the heavy-quark field, and

$$\mathcal{M}_1^{(q)} = i \int d^4 y \mathrm{T} \left\{ \mathcal{O}_1^{(q)}, \mathcal{O}_I(y) \right\}, \qquad (2.118)$$

$$\mathcal{M}_2^{(q)} = i \int d^4 y \mathrm{T} \left\{ \mathcal{O}_1^{(q)}, \mathcal{O}_{II}(y) \right\}, \qquad (2.119)$$

$$\mathcal{M}_3^{(q)} = i \int d^4 y \mathrm{T} \left\{ \mathcal{O}_2^{(q)}, \mathcal{O}_I(y) \right\}, \qquad (2.120)$$

$$\mathcal{M}_4^{(q)} = i \int d^4 y \mathrm{T} \left\{ \mathcal{O}_2^{(q)}, \mathcal{O}_{II}(y) \right\}, \qquad (2.121)$$

with $\mathcal{O}_{I,II}$ defined in Eqs. (1.69), (1.70), corresponding to the non-local contributions generated by taking the time ordered product of the $1/m_b$ correction in the Lagrangian in Eq. (1.59), with the dimension-six local HQET operators

$$\mathcal{O}_{1}^{(q)} = \left(\bar{h}_{v}^{i}\gamma_{\nu}(1-\gamma_{5})q^{i}\right)\left(\bar{q}^{j}\gamma^{\nu}(1-\gamma_{5})h_{v}^{j}\right), \qquad (2.122)$$

$$\mathcal{O}_{2}^{(q)} = \left(\bar{h}_{v}^{i}(1-\gamma_{5})q^{i}\right) \left(\bar{q}^{j}(1+\gamma_{5})h_{v}^{j}\right), \qquad (2.123)$$

$$\tilde{\mathcal{O}}_{1}^{(q)} = \left(\bar{h}_{v}^{i}\gamma_{\nu}(1-\gamma_{5})t_{ij}^{a}q^{j}\right)\left(\bar{q}^{l}\gamma^{\nu}(1-\gamma_{5})t_{lm}^{a}h_{v}^{m}\right),$$
(2.124)

$$\tilde{\mathcal{O}}_{2}^{(q)} = \left(\bar{h}_{v}^{i}(1-\gamma_{5})t_{ij}^{a}q^{j}\right)\left(\bar{q}^{l}(1+\gamma_{5})t_{lm}^{a}h_{v}^{m}\right).$$
(2.125)

Note that the $1/m_b$ contributions arising from the expansion of the heavy quark momentum vanish due to the equation of motion for $h_v(x)$. Apart from the operators in Eqs. (2.113)-(2.121), the basis includes also the corresponding colour-octet operators $\tilde{\mathcal{P}}_1^{(q)}, \tilde{\mathcal{P}}_2^{(q)}, \tilde{\mathcal{P}}_3^{(q)}, \tilde{\mathcal{R}}_1^{(q)}, \tilde{\mathcal{R}}_2^{(q)}$, containing respectively the colour matrices t^a , and $\tilde{\mathcal{M}}_1^{(q)}, \tilde{\mathcal{M}}_2^{(q)}, \tilde{\mathcal{M}}_3^{(q)}, \tilde{\mathcal{M}}_4^{(q)}$, in which we must replace $\mathcal{O}_i \to \tilde{\mathcal{O}}_i$. For brevity however, we omit to explicitly show them.

With the above results, it is straightforward to obtain the expansion of Eq. (2.80), up to order $\mathcal{O}(1/m_b^2)$, namely

$$\operatorname{Im} \mathcal{T}^{\mathrm{WE}} = \frac{G_F^2}{12\pi} |V_{q_1b}|^2 |V_{q_3q_2}|^2 m_b^2 \sqrt{\lambda(1, x_1, x_2)} \left(1 + \frac{(1 - x_1 - x_2)}{\lambda(1, x_1, x_2)} z \right) \\ \times \left\{ k_1 \left[\left(\omega_1(x_1, x_2) - y_1(x_1, x_2) z \right) \left(\mathcal{O}_1^{(q_3)} + \frac{\mathcal{R}_1^{(q_3)}}{m_b} + \frac{\mathcal{M}_1^{(q_3)}}{m_b} + \frac{\mathcal{M}_2^{(q_3)}}{m_b} \right) \right. \\ \left. - 2 \left(\omega_2(x_1, x_2) - y_2(x_1, x_2) z \right) (1 - z) \left(\mathcal{O}_2^{(q_3)} + \frac{\mathcal{R}_2^{(q_3)}}{m_b} + \frac{\mathcal{M}_3^{(q_3)}}{m_b} + \frac{\mathcal{M}_4^{(q_3)}}{m_b} - 2 \frac{\mathcal{P}_1^{(q_3)}}{m_b} \right) \right] \\ \left. + \left(k_1 \to k_2, \text{ singlet} \to \text{ octet} \right) + \mathcal{O} \left(\frac{1}{m_b^2} \right) \right\},$$

$$(2.126)$$

where

$$y_1(a,b) = 2(a-b)^2 + (a+b), \qquad y_2(a,b) = 4(a-b)^2 - (a+b), \qquad (2.127)$$

and the small parameter z is defined in Eq. (2.104) (with the plus sign). The leading order result in Eq. (2.126) reproduces the dimension-six expression obtained in Eq. (2.97), but with the QCD operators, i.e. $Q_i^{(q_3)}, \tilde{Q}_i^{(q_3)}$, see Eqs. (2.100)-(2.103), replaced by the corresponding HQET ones i.e. $\mathcal{O}_i^{(q_3)}, \tilde{\mathcal{O}}_i^{(q_3)}$, see Eqs. (2.122)-(2.125). Moreover, using Eqs. (2.107), (2.108), to rewrite z in terms of derivatives acting on light-quark field, we arrive at the final dimension-seven contribution, which reads

$$\operatorname{Im} \mathcal{T}_{7}^{\mathrm{WE}} = \frac{G_{F}^{2}}{12\pi} |V_{q_{1}b}|^{2} |V_{q_{3}q_{2}}|^{2} m_{b} \sqrt{\lambda(1, x_{1}, x_{2})} \left\{ k_{1} \left[\omega_{1}(x_{1}, x_{2}) \left(\mathcal{R}_{1}^{(q_{3})} + \mathcal{M}_{1}^{(q_{3})} + \mathcal{M}_{2}^{(q_{3})} \right) \right. \\ \left. + 2 \frac{(1 - x_{1} - x_{2})}{\lambda(1, x_{1}, x_{2})} \left(2 \omega_{2}(x_{1}, x_{2}) \mathcal{P}_{3}^{(q_{3})} - \omega_{1}(x_{1}, x_{2}) \mathcal{P}_{2}^{(q_{3})} \right) \right. \\ \left. + 2 \omega_{2}(x_{1}, x_{2}) \left(2 \left(\mathcal{P}_{1}^{(q_{3})} - \mathcal{P}_{3}^{(q_{3})} \right) - \left(\mathcal{R}_{2}^{(q_{3})} + \mathcal{M}_{3}^{(q_{3})} + \mathcal{M}_{4}^{(q_{3})} \right) \right) \right. \\ \left. + 2 \left(y_{1}(x_{1}, x_{2}) \mathcal{P}_{2}^{(q_{3})} - 2 y_{2}(x_{1}, x_{2}) \mathcal{P}_{3}^{(q_{3})} \right) \right] \\ \left. + \left(k_{1} \rightarrow k_{2}, \operatorname{singlet} \rightarrow \operatorname{octet} \right) \right\}.$$

$$(2.128)$$

Similarly, it follows that the expansion of Eq. (2.88) up order $\mathcal{O}(1/m_b^2)$, yields

$$\operatorname{Im} \mathcal{T}^{\mathrm{PI}} = \frac{G_F^2}{2\pi} |V_{q_1b}|^2 |V_{q_3q_2}|^2 m_b^2 \sqrt{\lambda(1, x_1, x_3)} (1 - x_1 - x_3) \\ \times \left[1 + \left(\frac{(x_1 + x_3)}{(1 - x_1 - x_3)} + \frac{(1 - x_1 - x_3)}{\lambda(1, x_1, x_3)} \right) z \right] \\ \times \left\{ k_3 \left[\mathcal{O}_1^{(q_2)} + \frac{\mathcal{R}_1^{(q_2)}}{m_b} + \frac{\mathcal{M}_1^{(q_2)}}{m_b} + \frac{\mathcal{M}_2^{(q_2)}}{m_b} \right] \\ + \left(k_3 \to k_4, \text{ singlet} \to \text{ octet} \right) + \mathcal{O} \left(\frac{1}{m_b^2} \right) \right\}, \qquad (2.129)$$

with the small parameter z defined in Eq. (2.104) (with the minus sign). Again, the leading order contribution in Eq. (2.129) reproduces to the dimension-six expression obtained in Eq. (2.98), but with the QCD operators, i.e. $Q_i^{(q_2)}, \tilde{Q}_i^{(q_2)}$, see Eqs. (2.100)-(2.103) replaced by the corresponding HQET ones i.e. $\mathcal{O}_i^{(q_2)}, \tilde{\mathcal{O}}_i^{(q_2)}$, see Eqs. (2.122)-(2.125). Moreover, using Eqs. (2.107), (2.108), to rewrite z in terms of derivatives acting on light-quark field we arrive at final dimension-seven result, which reads

$$\operatorname{Im} \mathcal{T}_{7}^{\operatorname{PI}} = \frac{G_{F}^{2}}{2\pi} |V_{q_{1}b}|^{2} |V_{q_{3}q_{2}}|^{2} m_{b} \sqrt{\lambda(1, x_{1}, x_{3})} (1 - x_{1} - x_{3})$$

$$\times \left\{ k_{3} \left[\mathcal{R}_{1}^{(q_{2})} + \mathcal{M}_{1}^{(q_{2})} + \mathcal{M}_{2}^{(q_{2})} + 2 \left(\frac{(x_{1} + x_{3})}{(1 - x_{1} - x_{3})} + \frac{(1 - x_{1} - x_{3})}{\lambda(1, x_{1}, x_{3})} \right) \mathcal{P}_{2}^{(q_{2})} \right] + \left(k_{3} \rightarrow k_{4}, \operatorname{singlet} \rightarrow \operatorname{octet} \right) \right\}.$$

$$(2.130)$$

Finally, the corresponding dimension-seven contribution to the WA topology is simply obtained by replacing $C_1 \leftrightarrow C_2$, and $q_1 \leftrightarrow q_3$, into Eq. (2.128), namely

$$\operatorname{Im} \mathcal{T}_{7}^{\mathrm{WA}} = \frac{G_{F}^{2}}{12\pi} |V_{q_{1}b}|^{2} |V_{q_{3}q_{2}}|^{2} m_{b} \sqrt{\lambda(1, x_{3}, x_{2})} \left\{ k_{5} \left[\omega_{1}(x_{3}, x_{2}) \left(\mathcal{R}_{1}^{(q_{1})} + \mathcal{M}_{1}^{(q_{1})} + \mathcal{M}_{2}^{(q_{1})} \right) \right. \\ \left. + 2 \frac{(1 - x_{3} - x_{2})}{\lambda(1, x_{3}, x_{2})} \left(2 \omega_{2}(x_{3}, x_{2}) \mathcal{P}_{3}^{(q_{1})} - \omega_{1}(x_{3}, x_{2}) \mathcal{P}_{2}^{(q_{1})} \right) \right. \\ \left. + 2 \omega_{2}(x_{3}, x_{2}) \left(2 \left(\mathcal{P}_{1}^{(q_{1})} - \mathcal{P}_{3}^{(q_{1})} \right) - \left(\mathcal{R}_{2}^{(q_{1})} + \mathcal{M}_{3}^{(q_{1})} + \mathcal{M}_{4}^{(q_{1})} \right) \right) \right. \\ \left. + 2 \left(y_{1}(x_{3}, x_{2}) \mathcal{P}_{2}^{(q_{1})} - 2 y_{2}(x_{3}, x_{2}) \mathcal{P}_{3}^{(q_{1})} \right) \right] \\ \left. + \left(k_{5} \rightarrow k_{6}, \operatorname{singlet} \rightarrow \operatorname{octet} \right) \right\}.$$

$$(2.131)$$

We emphasise that the results in Eqs. (2.128), (2.130), and (2.131), can be applied without any difference, to the description of baryons as well. In fact, in this case, the sign change in front of the light-quark momentum i.e. $p^{\mu} = p_b^{\mu} \mp p_q^{\mu}$, is compensated by the same sign change in front of the operators $\mathcal{P}_i^{(q)}$, i = 1, 2, 3, since now q(x)describes an incoming quark with momentum p_q^{μ} .

To conclude, we note that we have derived the dimension-seven contribution in terms of operators defined in HQET, the corresponding expressions in terms of QCD fields are obtained by expanding Eqs. (2.80), (2.88), and (2.94), only in the small momentum p_q of the light spectator quark, but not in the HQET field. This results in the following basis

$$P_1^{(q)} = m_q \left(\bar{b}^i (1 - \gamma_5) q^i \right) \left(\bar{q}^j (1 - \gamma_5) b^j \right), \qquad (2.132)$$

$$P_2^{(q)} = \frac{1}{m_b} \left(\bar{b}^i D_\nu \gamma_\mu (1 - \gamma_5) D^\nu q^i \right) \left(\bar{q}^j \gamma^\mu (1 - \gamma_5) b^j \right), \qquad (2.133)$$

$$P_3^{(q)} = \frac{1}{m_b} \left(\bar{b}^i \tilde{D}_\nu (1 - \gamma_5) D^\nu q^i \right) \left(\bar{q}^j (1 + \gamma_5) b^j \right), \qquad (2.134)$$

together with the colour-octet operators $\tilde{P}_i^{(q)}$. Due to the presence, in Eqs. (2.133), (2.134), of a covariant derivative acting on the *b*-quark field, which scales as m_b at this order, there is no explicit power counting, differently to the HQET basis. The corresponding QCD result for Eqs. (2.128), (2.130), (2.131), can be immediately obtained by setting in these expressions, $\mathcal{R}_i^{(q)}, \mathcal{R}_i^{(q)}, \mathcal{M}_i^{(q)}, \mathcal{M}_i^{(q)}$ to zero and by replacing $\mathcal{P}_i^{(q)} \to P_i^{(q)}$ and $\tilde{\mathcal{P}}_i^{(q)} \to \tilde{P}_i^{(q)}$. We stress that, in this case, the difference in the operator basis is compensated by the different parametrisation of the corresponding matrix elements in QCD and in HQET, cf. Chapter 4 and Appendix G.

Chapter 3

Contribution of the Darwin Operator

In Section 2.1 we have discussed the calculation of the dimension-three and dimensionfive contributions to the HQE of a B meson due to the single mode $b \rightarrow c\bar{u}d$ and using the representation of the quark-propagator in coordinate space given in Eqs. (1.121), (1.124). In the present chapter, we generalise the above results by computing the coefficients of the two-quark operators in the HQE, for the generic $b \rightarrow q_1 \bar{q}_2 q_3$ transition and up to terms of order $1/m_b^3$, where the latter describe the contribution of the Darwin operator, see Eq. $(1.157)^{-1}$. The representation derived in Ref. [130] for the calculation of the imaginary part of integrals containing Bessel functions, is not sufficient for the case in which more than one propagator is massive, hence now the whole computation is performed in momentum space, using the expression of the quark-propagator given in Eqs. (1.98), (1.102). Specifically, we start in Section 3.1, by computing the expansion of $\text{Im}\mathcal{T}^{(2q)}$ up to order $1/m_b^3$. We will find that the coefficients of the Darwin operator develop IR divergences in correspondence of the emission of a soft gluon from a light quark propagator. In particular, the singularities originate from the expansion of the propagator of the up, down, and strange quarks, which we consider massless, and are logarithmic, namely the corresponding coefficients have the asymptotic form $\sim \log(m_q^2/m_b^2)$ in the limit $m_q \to 0$, with q = u, d, s. As described e.g. in Ref. [73], these logarithmic infrared divergences are due to the mixing between operators of the same dimension under renormalisation. This is discussed in Section 3.2, where we compute the one-loop diagram describing the mixing of the four-quark and the Darwin operators and perform their renormalisation in order to ensure the cancellation of the IR divergences. Finally, the complete expressions for the coefficients of the Darwin operator are presented in Section 3.3 together with a discussion of the results. The structure closely follows the one of Ref. [137].

¹We stress that again, the computation is performed only at LO-QCD.

3.1 Computation of $\Gamma^{(2q)}(B)$ up to order $1/m_b^3$

The starting point for the computation of the contribution of two-quark operators is Eq. (2.10), which we state here again for practicality, i.e. ²

$$\mathcal{T}^{(2q)} = C_1^2 \,\mathcal{T}_{11}^{(2q)} + \,2 \,C_1 C_2 \,\left(\frac{1}{N_c} \mathcal{T}_{11}^{(2q)} + 2 \,\mathcal{T}_{13}^{(2q)}\right) + C_2^2 \,\mathcal{T}_{22}^{(2q)} \,, \tag{3.1}$$

with Eqs. (2.12), (2.13), compactly written as

$$\mathcal{T}_{11\{13\}}^{(2q)} = -\frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 i \int d^4x \ \bar{b} \gamma_\nu (1-\gamma_5) \{t^a\} i S^{(q_1)}(0,x) \gamma_\mu (1-\gamma_5) b(x)$$

$$\times \operatorname{Tr} \left[\gamma^\nu (1-\gamma_5) \{t^a\} i S^{(q_3)}(0,x) \gamma^\mu (1-\gamma_5) i S^{(q_2)}(x,0) \right] + (x \leftrightarrow 0) . \quad (3.2)$$

Note that only in the case of $\mathcal{T}_{13}^{(2q)}$ the $SU(3)_c$ generators t^a appear on the r.h.s. of Eq. (3.2), as indicated by the curly brackets, and that a summation over colour indices is understood. Furthermore, the corresponding expression for $\mathcal{T}_{22}^{(2q)}$ has been omitted, since it can be derived from that of $\mathcal{T}_{11}^{(2q)}$ by taking into account the result in Eq. (2.18).

Being interested in the expansion of $\mathcal{T}^{(2q)}$ up to order $1/m_b^3$, we must now substitute in Eq. (3.2), the complete expression of the quark propagator derived in Section 1.3, namely including also terms proportional to one covariant derivative of the gluon field strength tensor $D_{\rho}G_{\mu\nu}$. It is worth remarking that the colour structure of Eq. (3.2) allows for a straightforward treatment of colour, in complete analogy to what has already been discussed in Section 2.1.2. To this end, it is convenient to single out the colour structure of the propagator in Eq. (1.97), which schematically reads

$$\mathcal{S}_{jk}^{(q_i)}(p) = \mathcal{S}_0^{(q_i)}(p)\,\delta_{jk} + \mathcal{S}_1^{(q_i)\,a}(p)\,t_{jk}^a + \mathcal{O}\big(t^a t^b\big)\,, \qquad i = 1, 2, 3\,, \qquad (3.3)$$

where $\mathcal{S}_0^{(q_i)}(p)$ denotes the Fourier transform of the free-quark propagator given in Eq. (1.74), while $\mathcal{S}_1^{(q_i)}(p)$ includes higher order corrections due to operators of dimension-two and dimension-three describing the emission of one soft gluon field,

²Note that the superscript $(q_1\bar{q}_2q_3)$ is often omitted for the sake of a cleaner notation, however it must be always understood.



Figure 3.1: Two-loop diagrams describing power corrections due to the expansion of the quark propagator up to order $1/m_b^3$, from the $Q_1 \otimes Q_1$ (left) and $Q_2 \otimes Q_2$ (right) contributions.



Figure 3.2: Two-loop diagrams describing power corrections due to the expansion of the quark propagator up to order $1/m_b^3$, from the $Q_1 \otimes Q_3$ contribution.

see Eq. (1.98), and again the superscript (q_i) has been introduced in order to distinguish between different quarks. As consequence of fact that $\text{Tr}[t^a] = 0$ and that terms quadratic in the gluon field strength tensor correspond to operators in the HQE of at least dimension-seven, a soft gluon cannot be emitted from all the propagators in Eq. (3.2). Specifically, in the case of $\mathcal{T}_{11}^{(2q)}$ only the contribution of $S_1^{(q_1)}$ is non zero and hence, because of Eq. (2.18), only $S_1^{(q_3)}$ contributes to $\mathcal{T}_{22}^{(2q)}$, see Figure 3.1. Finally, in the case of $\mathcal{T}_{13}^{(2q)}$, we can independently expand the two propagators inside the trace and both $S_1^{(q_2)}$ and $S_1^{(q_3)}$ give a non vanishing contribution, see Figure 3.2. Substituting Eqs. (1.96), (1.100) into Eq. (3.2), and taking into account Eq. (2.18), we respectively obtain ³

³Note that the replacement in Eq. (2.26) is used.

$$\begin{aligned} \mathcal{T}_{11}^{(2q)} &= -\frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 \int d^4x \int \frac{d^4l_1}{(2\pi)^4} \int \frac{d^4l_2}{(2\pi)^4} \int \frac{d^4l_3}{(2\pi)^4} \left\{ e^{-i(p_b+l_2-l_1-l_3)\cdot x} \\ &\times \bar{b}\gamma_\nu (1-\gamma_5) \left(\mathcal{S}_0^{(q_1)}(l_1) + \tilde{\mathcal{S}}_1^{(q_1)}(l_1) \right) \gamma_\mu (1-\gamma_5) b \\ &\times \operatorname{Tr} \left[\gamma^\nu (1-\gamma_5) \mathcal{S}_0^{(q_3)}(l_3) \gamma^\mu (1-\gamma_5) \mathcal{S}_0^{(q_2)}(l_2) \right] \\ &+ \bar{b}\gamma_\nu (1-\gamma_5) \left(\mathcal{S}_0^{(q_1)}(l_1) + \mathcal{S}_1^{(q_1)}(l_1) \right) \gamma_\mu (1-\gamma_5) b \\ &\times \operatorname{Tr} \left[\gamma^\nu (1-\gamma_5) \mathcal{S}_0^{(q_3)}(l_3) \gamma^\mu (1-\gamma_5) \mathcal{S}_0^{(q_2)}(l_2) \right] e^{i(p_b+l_2-l_1-l_3)\cdot x} \right\} + \mathcal{O} \left(\frac{1}{m_b^4} \right), \end{aligned}$$
(3.4)

$$\mathcal{T}_{22}^{(2q)} = -\frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 \int d^4x \int \frac{d^4l_1}{(2\pi)^4} \int \frac{d^4l_2}{(2\pi)^4} \int \frac{d^4l_3}{(2\pi)^4} \left\{ e^{-i(p_b+l_2-l_1-l_3)\cdot x} \\ \times \bar{b}\gamma_\nu (1-\gamma_5) \, \mathcal{S}_0^{(q_1)}(l_1) \, \gamma_\mu (1-\gamma_5) b \\ \times \operatorname{Tr} \left[\gamma^\nu (1-\gamma_5) \left(\mathcal{S}_0^{(q_3)}(l_3) + \tilde{\mathcal{S}}_1^{(q_3)}(l_3) \right) \gamma^\mu (1-\gamma_5) \mathcal{S}_0^{(q_2)}(l_2) \right] \\ + e^{i(p_b+l_2-l_1-l_3)\cdot x} \, \bar{b}\gamma_\nu (1-\gamma_5) \mathcal{S}_0^{(q_1)}(l_1) \gamma_\mu (1-\gamma_5) b \\ \times \operatorname{Tr} \left[\gamma^\nu (1-\gamma_5) \left(\mathcal{S}_0^{(q_3)}(l_3) + \mathcal{S}_1^{(q_3)}(l_3) \right) \gamma^\mu (1-\gamma_5) \mathcal{S}_0^{(q_2)}(l_2) \right] \right\} + \mathcal{O} \left(\frac{1}{m_b^4} \right) ,$$
(3.5)

and

$$\begin{aligned} T_{13}^{(2q)} &= -\frac{G_F^2}{2} |V_{q_1b}|^2 |V_{q_3q_2}|^2 \int d^4x \int \frac{d^4l_1}{(2\pi)^4} \int \frac{d^4l_2}{(2\pi)^4} \int \frac{d^4l_3}{(2\pi)^4} \left\{ e^{-i(p_b+l_2-l_1-l_3)\cdot x} \\ &\times \bar{b}\gamma_\nu (1-\gamma_5) \mathcal{S}_0^{(q_1)}(l_1)\gamma_\mu (1-\gamma_5) b \\ &\times \left\{ \mathrm{Tr} \left[\gamma^\nu (1-\gamma_5) \mathcal{S}_1^{(q_3)}(l_3)\gamma^\mu (1-\gamma_5) \mathcal{S}_0^{(q_2)}(l_2) \right] \right\} \\ &+ \mathrm{Tr} \left[\gamma^\nu (1-\gamma_5) \mathcal{S}_0^{(q_3)}(l_3)\gamma^\mu (1-\gamma_5) \mathcal{S}_1^{(q_2)}(l_2) \right] \right\} \\ &+ e^{i(p_b+l_2-l_1-l_3)\cdot x} \bar{b}\gamma_\nu (1-\gamma_5) \mathcal{S}_0^{(q_1)}(l_1)\gamma_\mu (1-\gamma_5) b \\ &\times \left\{ \mathrm{Tr} \left[\gamma^\nu (1-\gamma_5) \mathcal{S}_1^{(q_3)}(l_3)\gamma^\mu (1-\gamma_5) \mathcal{S}_0^{(q_2)}(l_2) \right] \right\} \\ &+ \mathrm{Tr} \left[\gamma^\nu (1-\gamma_5) \mathcal{S}_0^{(q_3)}(l_3)\gamma^\mu (1-\gamma_5) \mathcal{S}_1^{(q_2)}(l_2) \right] \right\} \end{aligned}$$

where, owing to the fact that starting with the dimension-three operator $D_{\rho}G_{\mu\nu}$, the translation invariance of the quark-propagator is broken, see Section 1.3, the second term in Eq. (3.2) must be explicitly computed and cannot be derived from symmetry arguments. Note also that in writing Eq. (3.6) we have already taken into account that the contribution of the free-quark propagator alone, vanishes because of the traceless property of the colour matrices, cf. Eq. (2.20).

It is worth remarking that the expression of the propagator in Eqs. (1.121), (1.124) is infrared divergent in the limit of massless quark, cf. Eq. (1.125). This point will be further discussed later on, for the moment it is important to stress that by setting the mass of the up, down and strange quarks to zero, the logarithmic divergences would appear in Eqs. (3.4)-(3.6) from the contribution of $S_1^{(u)}, S_1^{(d)}$ and $S_1^{(s)}$, starting at order $1/m_b^3$. In the following, in order to regularise the integrals, we keep the light-quark masses finite and only in the final expressions, once the infrared divergences have been subtracted, we take the limit $m_q \rightarrow 0$, where q = u, d, s. Moreover, the computation is performed in D = 4 space-time dimensions since the imaginary part of the integrals Eqs. (3.4)-(3.6) is ultraviolet (UV) finite at LO-QCD. Alternatively, Eqs. (3.4)-(3.6) could be calculated in dimensional regularisation setting in this case $m_q = 0$ from the beginning [138, 139].

The manipulation of Eqs. (3.4)-(3.6) proceeds in a similar way and can be con-

veniently performed using e.g. the Mathematica package *FeynCalc* [156]. After integrating over the variables x^{μ} and l_{3}^{μ} , Eqs. (3.4)-(3.6) reduce to a linear combination of two-point two-loop tensor integrals with one external momentum p_{b}^{μ} and of possible rank $r = 1, \ldots, 4$, of the type

$$\int \frac{d^4 l_1}{(2\pi)^4} \int \frac{d^4 l_2}{(2\pi)^4} \frac{\left\{ l_1^{\mu}, l_2^{\mu}, l_1^{\mu} l_2^{\nu}, \dots, l_1^{\mu} l_1^{\nu} l_2^{\rho}, \dots, l_1^{\mu} l_1^{\nu} l_1^{\rho} l_2^{\sigma}, \dots \right\}}{\left[l_1^2 - m_1^2 + i\varepsilon \right]^{a_1} \left[l_2^2 - m_2^2 + i\varepsilon \right]^{a_2} \left[(l_1 + l_2 - p_b)^2 - m_3^2 + i\varepsilon \right]^{a_3}},$$
(3.7)

with $a_i = 1, 2, 3$. The tensor structure of the integrals in Eq. (3.7) can be simplified using the procedure discussed in Appendix C for D = 4. As a result, each integral of rank-*r* in Eq. (3.7), is expressed in terms of a linear combination of tensors of the same rank built from the metric tensor $g^{\mu\nu}$ and the external momentum p_b^{μ} where the corresponding coefficients represent scalar integrals of the form

$$\int \frac{d^4 l_1}{(2\pi)^4} \int \frac{d^4 l_2}{(2\pi)^4} \frac{\left\{l_1^2, l_1 \cdot l_2, l_1 \cdot p_b, \dots\right\}}{\left[l_1^2 - m_1^2 + i\varepsilon\right]^{a_1} \left[l_2^2 - m_2^2 + i\varepsilon\right]^{a_2} \left[(l_1 + l_2 - p_b)^2 - m_3^2 + i\varepsilon\right]^{a_3}}.$$
(3.8)

Note that all the possible scalar products of the three momenta $l_1^{\mu}, l_2^{\mu}, p_b^{\mu}$ appear in the numerator of Eq. (3.8). In the next step, we use the Mathematica package *LiteRed* [157,158] to perform the reduction of the set of scalar integrals obtained, to a liner combination of master integrals (MIs). Let us introduce the notation

$$\mathcal{I}_{n_1 n_2 n_3} \equiv \int \frac{d^4 l_1}{(2\pi)^4} \int \frac{d^4 l_2}{(2\pi)^4} \frac{1}{\left[l_1^2 - m_1^2 + i\varepsilon\right]^{n_1} \left[l_2^2 - m_2^2 + i\varepsilon\right]^{n_2} \left[(l_1 + l_2 - p_b)^2 - m_3^2 + i\varepsilon\right]^{n_3}} \tag{3.9}$$

with $n_i \in \mathbb{N}_0$, then the set of master integrals reads

$$\left\{ \mathcal{I}_{111}, \, \mathcal{I}_{211}, \, \mathcal{I}_{121}, \, \mathcal{I}_{112}, \, \mathcal{I}_{011}, \, \mathcal{I}_{101}, \, \mathcal{I}_{110} \right\}.$$
(3.10)

The first four integrals in Eq. (3.10) correspond to the MIs of the sunrise graph with three different masses [159, 160]. From the definition in Eq. (3.9), it evidently follows that once the expression of \mathcal{I}_{111} is known, the solution for the remaining three MIs can be obtained by differentiating \mathcal{I}_{111} with respect to the appropriate mass parameter, i.e.

$$\mathcal{I}_{211} = \frac{\partial}{\partial m_1^2} \mathcal{I}_{111}, \quad \mathcal{I}_{121} = \frac{\partial}{\partial m_2^2} \mathcal{I}_{111}, \quad \mathcal{I}_{112} = \frac{\partial}{\partial m_3^2} \mathcal{I}_{111}.$$
(3.11)

Furthermore, the last three integrals in Eq. (3.10) factorise into the product of two scalar tadpoles, in fact, e.g. \mathcal{I}_{101} , can be rewritten as

$$\mathcal{I}_{101} = \int \frac{d^4 l_1}{(2\pi)^4} \frac{1}{l_1^2 - m_1^2 + i\varepsilon} \int \frac{d^4 l_2}{(2\pi)^4} \frac{1}{l_2^2 - m_3^2 + i\varepsilon}, \qquad (3.12)$$

by performing the change of variable $l_1^{\mu} + l_2^{\mu} - p_b^{\mu} \rightarrow l_2^{\mu}$, and similarly for \mathcal{I}_{011} and \mathcal{I}_{110} . However, we immediately point out that these do not contribute to $\text{Im}\mathcal{T}^{(2q)}$, since the imaginary part of the product of two tadpoles vanishes, cf. Eq. (B.8), namely

$$\operatorname{Im} \mathcal{I}_{101} = \operatorname{Im} \mathcal{I}_{110} = \operatorname{Im} \mathcal{I}_{011} = 0.$$
(3.13)

From Eq. (3.11), (3.13), it then follows that in order to compute the imaginary part of the set of integrals in Eq. (3.8), it suffices to know the solution of the master integral \mathcal{I}_{111} in the physical decay region $p_b^2 \ge (m_1 + m_2 + m_3)^2$, in correspondence of which its integrand develops a discontinuity. Using the result presented in Ref. [161], we obtain that ⁴

$$\operatorname{Im} \mathcal{I}_{111} = \frac{1}{256\pi^3} \int_{(m_2+m_3)^2}^{(\sqrt{s}-m_1)^2} dt \, \frac{\sqrt{\lambda(t,m_2^2,m_3^2)\,\lambda(s,t,m_1^2)}}{t\,s} \,, \tag{3.14}$$

where $s = p_b^2$. However, the integral in Eq. (3.14) admits a simple analytical expression if at most two masses are non zero. For three non vanishing masses, its complexity highly increases and the solution involves elliptic functions, see e.g. Refs. [162–165]. We emphasise that in the approximation of massless u, d and s quarks, it is always possible to set at least one mass to zero and to compute all the corresponding master integrals analytically, except in the case of $b \rightarrow c\bar{c}s$ transition where, as discussed above, we need to keep the mass of the *s*-quark finite in order to regularise the IR divergence originating from $S_1^{(s)}$. It follows that for this specific mode, we do not provide an analytical expression for all the corresponding MIs, and our results still require a numerical integration.

The integrals in Eqs. (3.14)-(3.11), are scalar functions of the external momen-

⁴Note that we set d = 4 in the result of Ref. [161].

tum p_b^{μ} and depend on the dimensionless parameters $r_i = m_i^2/p_b^2$. At this point of the computation then, taking into account that $G_{\mu\nu} = -i [iD_{\mu}, iD_{\nu}]$, see Eq. (17), together with $D_{\rho}G_{\mu\nu} = -[iD_{\rho}, [iD_{\mu}, iD_{\nu}]]$, see Eq. (19), the imaginary part of Eqs. (3.4)-(3.5) can be schematically written in the following form ⁵

$$\operatorname{Im} \mathcal{T}_{mn}^{(2q)} = \mathcal{F}_{mn}(p_b, r_i) \, \bar{b}b + \mathcal{G}_{mn}^{\mu\nu}(p_b, r_i) \, \bar{b}(iD_{\mu})(iD_{\nu})b + \mathcal{D}_{mn}^{\mu\nu\rho}(p_b, r_i) \, \bar{b}(iD_{\mu})(iD_{\nu})(iD_{\rho})b + \mathcal{O}\left(\frac{1}{m_b^4}\right) \,, \qquad (3.15)$$

corresponding, respectively, to the sum of the contributions due to the free quark propagator and to the two lowest dimensional corrections to this, namely the ones proportional to the gluon field strength tensor and to its first covariant derivative. We notice that $\mathcal{F}_{13}(p_b, r_i) = 0$, since the free-quark propagator is colour singlet and the contribution of the colour octet operator Q_3 , cf. Eq. (2.8), vanishes. Moreover, another feature of Eq. (3.15) is that

$$\mathcal{F}_{22}(p_b, r_i) = \mathcal{F}_{11}(p_b, r_i), \qquad (3.16)$$

which is the generalisation of the result given in Eq. (2.21), to the case of an arbitrary decay mode of the *b* quark. Singling out the colour factor due to $\delta_{ii} = N_c$, we can write

$$\mathcal{F}_{11}(p_b, r_i) = N_c \,\tilde{\mathcal{F}}_{11}(p_b, r_i) \,, \tag{3.17}$$

and then from Eq. (3.1) we readily obtain that

$$\operatorname{Im} \mathcal{T}^{(2q)}\Big|_{d=3}^{(q_1\bar{q}_2q_3)} = \left(N_c \left(C_1^2 + C_2^2 \right) + 2 C_1 C_2 \right) \tilde{\mathcal{F}}_{11}^{(q_1\bar{q}_2q_3)}(p_b, r_i) \, \bar{b}b \,, \tag{3.18}$$

which again generalises the result in Eq. (2.33). Furthermore, Eq. (3.15) presents also the important property, already encountered in Section 2.1.2 for the specific case of $b \rightarrow c\bar{u}d$ transition, that at order $1/m_b^2$ only the expansion of the q_2 quark propagator gives a non vanishing contribution, namely

$$\mathcal{G}_{11}^{\mu\nu}(p_b, r_i) = \mathcal{G}_{22}^{\mu\nu}(p_b, r_i) = 0, \quad \mathcal{G}_{13}^{\mu\nu}(p_b, r_i) \neq 0.$$
(3.19)

⁵Note that for brevity spinor and colour indices are not shown.

This result has relevant numerical consequences, since the coefficient of C_1^2 at order $1/m_b^2$, is strongly suppressed. So far only power corrections deriving from the expansion of the quark-propagator have been taken into account. The coefficient functions in Eq. (3.15) depend on the heavy quark momentum p_b^{μ} explicitly and implicitly through the variable r_i . Introducing the standard parametrisation $p_b^{\mu} = m_b v^{\mu} + k^{\mu}$, see Section 1.2, and recalling the definition of the rescaled heavy quark field $b(x) = \exp(-im_b v \cdot x)b_v(x)$, see Eq. (1.145), each term in Eq. (3.15) can be further expanded in powers of $1/m_b$, resulting in higher dimensional operators with additional covariant derivatives acting on the b_v field. Specifically, in our case, the expansion must be performed up to order $1/m_b^3$, hence power corrections to the dimension-six coefficients $\mathcal{D}_{ab}^{\mu\nu\rho}(p_b, r_i)$ in Eq. (3.15), can be neglected as these would lead to contributions of order $\mathcal{O}(1/m_b^4)$. Using the procedure described in Appendix A, we then obtain

$$\mathcal{F}_{mn}(p_b, r_i) \,\overline{b}b = \mathcal{F}_{mn}(m_b v, \rho_i) \,\overline{b}_v b_v + \mathcal{K}^{\mu}_{mn}(m_b v, \rho_i) \,\overline{b}_v i D_{\mu} b_v + \mathcal{G}^{\prime\mu\nu}_{mn}(m_b v, \rho_i) \,\overline{b}_v i D_{\mu} i D_{\nu} b_v + \mathcal{D}^{\prime\mu\nu\rho}_{mn}(m_b v, \rho_i) \,\overline{b}_v i D_{\mu} i D_{\nu} i D_{\rho} b_v + \dots ,$$
(3.20)

$$\mathcal{G}_{mn}^{\mu\nu}(p_b, r_i)\,\bar{b}iD_{\mu}iD_{\nu}b = \mathcal{G}_{mn}^{\mu\nu}(m_bv, \rho_i)\,\bar{b}_viD_{\mu}iD_{\nu}b_v + \mathcal{D}_{mn}^{\prime\prime\mu\nu\rho}(m_bv, \rho_i)\,\bar{b}_viD_{\mu}iD_{\nu}iD_{\rho}b_v + \dots$$
(3.21)

and

$$\mathcal{D}_{mn}^{\mu\nu\rho}(p_b, r_i)\,\bar{b}iD_{\mu}iD_{\nu}iD_{\rho}b = \mathcal{D}_{mn}^{\mu\nu\rho}(m_b v, \rho_i)\,\bar{b}_v iD_{\mu}iD_{\nu}iD_{\rho}b_v + \dots\,,\tag{3.22}$$

where the ellipsis denote power suppressed contributions of order $\mathcal{O}(1/m_b^4)$ and we have introduced the dimensionless mass parameters $\rho_i = m_i^2/m_b^2$. Finally, in order to compute $\Gamma^{(2q)}(B)$, we must evaluate the matrix element of $\text{Im}\mathcal{T}^{(2q)}$ between external *B* states. This can be conveniently done in the framework of the HQET, in which, the residual mass dependence of the b_v field and of the *B* meson state, can be systematically extracted, leading to a further expansion in $1/m_b$. A consistent procedure to determine the forward matrix element of operators containing multiple covariant derivatives acting on the heavy quark field and to express them in terms of a minimal set of non perturbative parameters, has been presented in Ref. [105]. Using their results, we readily arrive at the final expression for the contribution of two-quark operators to the total decay width of a B meson, namely

$$\Gamma^{(2q)}(B) = \Gamma_0 \left[\left(N_c C_1^2 + 2 C_1 C_2 + N_c C_2^2 \right) \mathcal{C}_0^{(q_1 \bar{q}_2 q_3)} \left(1 - \frac{\mu_\pi^2(B)}{2m_b^2} \right) + \left(N_c C_1^2 \ \mathcal{C}_{G,11}^{(q_1 \bar{q}_2 q_3)} + 2 C_1 C_2 \ \mathcal{C}_{G,12}^{(q_1 \bar{q}_2 q_3)} + N_c C_2^2 \ \mathcal{C}_{G,22}^{(q_1 \bar{q}_2 q_3)} \right) \frac{\mu_G^2(B)}{m_b^2} + \left(N_c C_1^2 \ \mathcal{C}_{D,11}^{(q_1 \bar{q}_2 q_3)} + 2 C_1 C_2 \ \mathcal{C}_{D,12}^{(q_1 \bar{q}_2 q_3)} + N_c C_2^2 \ \mathcal{C}_{D,22}^{(q_1 \bar{q}_2 q_3)} \right) \frac{\rho_D^3(B)}{m_b^3} \right] + \mathcal{O} \left(\frac{1}{m_b^4} \right),$$
(3.23)

where we have introduced

$$\Gamma_0 = \frac{G_F^2 m_b^5}{192 \,\pi^3} \, |V_{q_1 b}|^2 |V_{q_2 q_3}|^2 \,, \tag{3.24}$$

and the non perturbative parameters $\mu_{\pi}^2(B)$, $\mu_G^2(B)$ and $\rho_D^3(B)$ are defined as in Eqs. (1.159), (1.160). In Eq. (3.23), $C_0^{(q_1\bar{q}_2q_3)}$ refers to the partonic-level coefficient, which coincides, up to a factor of (-1/2), with that of the kinetic operator $\mathcal{O}_{kin}/m_b^{2.6}$, cf. Eq. (1.154), while $\mathcal{C}_{G,mn}^{(q_1\bar{q}_2q_3)}$ and $\mathcal{C}_{D,mn}^{(q_1\bar{q}_2q_3)}$, respectively describe the contribution of the chromo-magnetic and of the Darwin operators. Note that having adopted a covariant definition for these operators, the coefficient of the spin-orbit operator \mathcal{O}_{LS} , cf. Eq. (1.156), is identically zero, see for more details e.g. Refs. [105, 138, 166]. Moreover, we stress that while $\mathcal{C}_0^{(q_1\bar{q}_2q_3)}$ and $\mathcal{C}_{G,nm}^{(q_1\bar{q}_2q_3)}$ are finite functions of at most one dimensionless mass parameter $\rho = m_c^2/m_b^2$, and we list their complete expressions in Appendix D, as previously discussed, the coefficients of the Darwin operator still depend on the infrared regulator m_q , with q = u, d, s. These in fact, have the following schematic form

$$\mathcal{C}_{D,mn}^{(q_1\bar{q}_2q_3)} = \mathcal{R}_{mn}^{(q_1\bar{q}_2q_3)} + \mathcal{D}_{mn}^{(q_1\bar{q}_2q_3)}, \qquad (3.25)$$

where $\mathcal{R}_{mn}^{(q_1\bar{q}_2q_3)}$ are finite functions of at most the dimensionless parameter ρ , while $\mathcal{D}_{nm}^{(q_1\bar{q}_2q_3)}$ absorb the remaining divergent contributions, namely

$$\mathcal{D}_{mn}^{(q_1\bar{q}_2q_3)} \sim \log\left(\frac{m_q^2}{m_b^2}\right) \,, \tag{3.26}$$

⁶This follows from the reparametrisation invariance of the HQE, see e.g. Ref. [166].



Figure 3.3: Schematic representation of the OPE at order $1/m_b^3$. At LO-QCD, the Darwin operator is generated at two-loop, whereas the four-quark operators arise already at one-loop.

and their explicit expression can be found in Appendix E. Eq. (3.26) shows that the functions $C_{D,nm}^{(q_1\bar{q}_2q_3)}$ are logarithmically sensitive to the light quark mass $m_q \ll m_b$. However, the advantage of constructing the operator product expansion lies in the introduction of a factorisation scale μ , in terms of which the dependence on the hard, i.e. μ_h and soft i.e. μ_s scales with $m_b \ge \mu_h \ge \mu$, and $\mu_s \le \mu$, is respectively factorised between short distance coefficients and matrix elements of local operators, see e.g. Ref. [71]. Eq. (3.23) alone, including the effect of two-quark operators only, does not correspond to the complete OPE up to dimension-six because starting at this order also four-quark operators contribute. In fact, the logarithmic infrared divergence in Eq. (3.26) reflects the mixing between operators at order $1/m_b^3$ under renormalisation. It follows that by solving the corresponding RGEs, the dependence on the light quark mass in the coefficients in Eq. (3.25) can be correctly absorbed in the matrix element of local operators, making then manifest the factorisation between hard and soft scales. This is discussed in detail in the next section.

3.2 Operator-mixing at order $1/m_b^3$

To understand the origin of the IR divergences in Eq. (3.26), and how these are properly subtracted, we study the structure of Eq. (1.144) at order $1/m_b^3$. Within the HQE, the imaginary part of time ordered product of the double insertion of the effective Hamiltonian is expanded in a series of local operators with new effective



Figure 3.4: One-loop diagram describing the mixing of four-quark operators with the Darwin operator.

couplings. This can be schematically written as ⁷

$$\left\langle \operatorname{Im} \mathcal{T} \right\rangle \Big|_{d=6} = c_{\rho_D}(\mu_0) \left\langle \mathcal{O}_{\rho_D} \right\rangle(\mu_0) + \sum_{q=q_1,q_2,q_3} \vec{c}_{4q}^{(q)}(\mu_0) \cdot \left\langle \vec{\mathcal{O}}_{4q}^{(q)} \right\rangle(\mu_0), \quad (3.27)$$

where the shorthand $\langle \ldots \rangle$ denotes a matrix element between external *B* meson states and the dependence on the scale μ_0 at which the matrix element of the local dimension-six operators are renormalised, is now explicitly indicated. In Eq. (3.27), the Darwin operator \mathcal{O}_{ρ_D} is defined as in Eq. (1.157) and we have introduced a compact notation for the four-quark operators listed in Eqs. (2.100)-(2.103), namely

$$\vec{\mathcal{O}}_{4q}^{(q)} = \left(O_1^{(q)}, \, O_2^{(q)}, \, \tilde{O}_1^{(q)}, \, \tilde{O}_2^{(q)} \right). \tag{3.28}$$

In the following we want to determine the short distance coefficients $c_{\rho D}(\mu_0)$ and $\vec{c}_{4q}^{(q)}(\mu_0)$. The OPE leading to Eq. (3.27) can be schematically visualised as in Figure 3.3. It has the peculiarity that the order of the Darwin and of the four-quark operators, in terms of the loop- and α_s -expansion, does not coincide. Specifically, while the four-quark operators are generated at one-loop at order α_s^0 , the Darwin operator arises only at two-loop again at order α_s^0 . It follows that the one-loop correction to the four-quark operators, shown in Figure 3.4, is of the same order in terms of loop- and α_s -expansion as the coefficient of the Darwin operator and must be included to obtain the complete contribution to $c_{\rho D}(\mu_0)$ at LO-QCD, see e.g. Refs. [167–169]. We note that operator mixing at zeroth-order in α_s has been extensively discussed for the $b \rightarrow s\gamma$ effective Hamiltonian, see e.g. Refs. [170–172]. In order to evaluate the diagram in Figure 3.4 for all four operators in Eq. (3.28), we start by considering the one-loop matrix element of the colour-singlet operators in the presence of a soft gluon field $A_{\mu}(x)$, which is obtained by computing the time

⁷Note that the superscript $(q_1\bar{q}_2q_3)$ and the subscript mn are omitted for the sake of a cleaner notation, however they must be always understood.

ordered product of $\mathcal{O}_{4q,j}^{(q)}$, j = 1, 2, with the interaction part of the QCD Lagrangian $\mathcal{L}_{QCD}^{int}(x) = \sum_{q} \bar{q} \mathcal{A}(x) q$. This reads ⁸

$$\langle \mathcal{O}_{4q,j}^{(q)} \rangle_{1-loop} = \langle i \int d^4 z \, \mathrm{T} \left\{ \vec{\mathcal{O}}_{4q,j}^{(q)}, \sum_{q'} \vec{q}'(z) \mathcal{A}(z) q'(z) \right\} \rangle + \dots$$
$$= \langle \bar{b}_v \bar{\Gamma}_j \left(\int d^4 z \, i S_0^{(q)}(-z) i \mathcal{A}(z) i S_0^{(q)}(z) \right) \Gamma_j b_v \rangle + \dots, \qquad (3.29)$$

where the ellipsis denote terms with more than one gluon field, $S_0^{(q)}(x-y)$ is the free-quark propagator and for convenience we have introduced the notation

$$\Gamma_1 = \gamma_\mu (1 - \gamma_5) \qquad \Gamma_2 = (1 + \gamma_5), \qquad \overline{\Gamma}_j = \gamma_0 \Gamma_j^{\dagger} \gamma_0. \tag{3.30}$$

We recognise in the term in round brackets of Eq. (3.29) the first order correction to the free-quark propagator for vanishing space-time separation, cf. Eq. (1.77), namely

$$\int d^4 z \, i S_0^{(q)}(-z) i \mathcal{A}(z) i S_0^{(q)}(z) = \lim_{x \to 0} \, i S_1^{(q)}(x,0) \,, \tag{3.31}$$

which, in the FS gauge, admits the Fourier representation

$$\lim_{x \to 0} S_1^{(q)}(x,0) = \int \frac{d^4 l}{(2\pi)^4} \,\mathcal{S}_1^{(q)}(l) \,, \tag{3.32}$$

with $S_1^{(q)}(l)$ given in Eq. (1.98). The integral in Eq. (3.32) has both UV- and IR-divergences in the limit of massless quark q. To regularise the former we use dimensional regularisation, setting the number of space-time dimension to $D = 4 - 2\epsilon$, for the latter we must choose the same regularisation scheme applied to the computation of the coefficients in Eq. (3.25), hence we keep the mass of the light-quark running into the loop finite. The algebra of gamma matrices in D dimensions is computed using the naive dimensional regularisation (NDR) scheme, with

$$\{\gamma_{\mu}, \gamma_{\nu}\} = 2g_{\mu\nu}, \qquad g_{\mu\nu}g^{\mu\nu} = D, \qquad \{\gamma_{\mu}, \gamma_5\} = 0.$$
(3.33)

Taking into account in the expression for the quark-propagator $S_1^{(q)}(l)$ that $G_{\mu\nu} = -i [iD_{\mu}, iD_{\nu}]$ and that $D_{\rho} G_{\mu\nu} = -[iD_{\rho}, [iD_{\mu}, iD_{\nu}]]$, Eq. (3.29) becomes

⁸Note that summation over colour indices is understood.

$$\langle \mathcal{O}_{4q,j}^{(q)} \rangle_{1-loop} = \langle \bar{b}_v \bar{\Gamma}_j i \int \frac{d^D l}{(2\pi)^D} \left\{ -\frac{i}{2} \frac{[D^\sigma, D^\tau]}{(l^2 - m_q^2 + i\varepsilon)^2} l^\rho \gamma^\mu \epsilon_{\rho\mu\sigma\tau} \right.$$

$$+ i \frac{2}{3} \frac{[D_\rho, [D^\rho, D^\mu]]}{(l^2 - m_q^2 + i\varepsilon)^2} \left(\gamma_\mu - l_\mu \frac{\not{l}}{(l^2 - m_q^2 + i\varepsilon)} \right)$$

$$- i \frac{2}{3} \left[D_\nu, [D_\rho, D_\mu] \right] \frac{l^\nu l^\rho \gamma^\mu}{(l^2 - m_q^2 + i\varepsilon)^3}$$

$$+ i \frac{[D_\nu, [D^\sigma, D^\tau]]}{(l^2 - m_q^2 + i\varepsilon)^3} l^\nu l^\rho \gamma^\mu \epsilon_{\rho\mu\sigma\tau} \left\{ \Gamma_j b_v \right\} + \dots$$

$$(3.34)$$

We stress that due to the chiral structure of Eq. (3.34), terms in the propagator proportional to an even number of gamma matrices do not contribute and that we have used that $\gamma_5\Gamma_j = \Gamma_j$. Moreover, the first integral on the r.h.s of Eq. (3.34) vanishes, being the integrand an odd function of l^{μ} and the integration domain even. This result is independent of the colour structure of the operator inserted in the vertex of the diagram in Figure 3.4 and applies also to the matrix element of the remaining colour-octet operators in Eq. (3.28). Hence, the four-quark operators $\vec{\mathcal{O}}_{4q}^{(q)}$ do not mix with the chromo-magnetic operator at order α_s^{0-9} . Eq. (3.34) then simplifies to

$$\langle \mathcal{O}_{4q,j}^{(q)} \rangle_{1-loop} = -\langle \bar{b}_v \bar{\Gamma}_j \int \frac{d^D l}{(2\pi)^D} \left\{ \frac{[D_\nu, [D^\sigma, D^\tau]]}{(l^2 - m_q^2 + i\varepsilon)^3} l^\nu l^\rho \gamma^\mu \epsilon_{\rho\mu\sigma\tau} \right. \\ \left. + \frac{2}{3} \frac{[D_\rho, [D^\rho, D^\mu]]}{(l^2 - m_q^2 + i\varepsilon)^2} \left(\gamma_\mu - l_\mu \frac{l}{(l^2 - m_q^2 + i\varepsilon)} \right) \right. \\ \left. - \frac{2}{3} [D_\nu, [D_\rho, D_\mu]] \frac{l^\nu l^\rho \gamma^\mu}{(l^2 - m_q^2 + i\varepsilon)^3} \right\} \Gamma_j b_v \rangle + \dots .$$
(3.35)

The dimensional analysis of Eq. (3.35) reveals an interesting subtlety. From

$$[\psi] = \frac{D-1}{2}, \qquad [D_{\mu}] = 1, \qquad [l_{\mu}] = 1, \qquad (3.36)$$

where the square brackets denote the dimension in units of mass i.e. [m] = 1 and ψ an arbitrary fermion field, see e.g. Ref. [38], we obtain that while Eq. (3.35) is

⁹Note that mixing with lower-dimensional operators can arise at NLO-QCD, see e.g. Ref. [173].
dimensionally correct, since both sides have the same mass dimension, the integrals in Eq. (3.35) have now become dimensionful, namely

$$\left[\int \frac{d^4l}{(l^2 - m_q^2 + i\varepsilon)^2}\right] = 0 \quad \rightarrow \quad \left[\int \frac{d^Dl}{(l^2 - m_q^2 + i\varepsilon)^2}\right] = -2\epsilon \,. \tag{3.37}$$

This fictitious dimension is an artefact of the regularisation scheme and does not correspond to any physical parameter. In fact, in this case, the integrals in Eq. (3.35) would be expressed in terms of ill defined logarithms with a dimensionful argument, cf. Eqs. (B.1), (B.23). The origin of this mismatch lies in the different scaling between the Darwin and the four-quark operators with the number of dimensions, see a similar discussion, for the case of $b \to s\gamma$ and $b \to sg$, in Ref. [172], i.e.

$$\left[\vec{\mathcal{O}}_{4q}^{(q)}\right] = 6 - 4\epsilon \quad \neq \quad 6 - 2\epsilon = \left[\mathcal{O}_{\rho_D}\right]. \tag{3.38}$$

In order to keep the integrals in Eq. (3.35) dimensionless, we must introduce a scale factor of $\mu^{2\epsilon}$ in the integration measure. Eq. (3.35) then becomes

$$\langle \mathcal{O}_{4q,j}^{(q)} \rangle_{1-loop} = -\mu^{-2\epsilon} \langle \bar{b}_v \bar{\Gamma}_j \mu^{2\epsilon} \int \frac{d^D l}{(2\pi)^D} \Biggl\{ \frac{[D_\nu, [D^\sigma, D^\tau]]}{(l^2 - m_q^2 + i\varepsilon)^3} l^\nu l^\rho \gamma^\mu \epsilon_{\rho\mu\sigma\tau} + \frac{2}{3} \frac{[D_\rho, [D^\rho, D^\mu]]}{(l^2 - m_q^2 + i\varepsilon)^2} \left(\gamma_\mu - l_\mu \frac{\not{l}}{(l^2 - m_q^2 + i\varepsilon)} \right) - \frac{2}{3} [D_\nu, [D_\rho, D_\mu]] \frac{l^\nu l^\rho \gamma^\mu}{(l^2 - m_q^2 + i\varepsilon)^3} \Biggr\} \Gamma_j b_v \rangle + \dots .$$
(3.39)

Using the results in Eqs. (B.21), (B.24), to evaluate the scalar and tensor integrals in Eq. (3.39), we obtain that

$$\langle \mathcal{O}_{4q,j}^{(q)} \rangle_{1-loop} = -\mu^{-2\epsilon} \langle \bar{b}_v \bar{\Gamma}_j \left\{ \frac{1}{4} \mathcal{I}_0(m_q^2) [D_\nu, [D_\sigma, D_\tau]] \epsilon^{\nu\mu\sigma\tau} + \frac{1}{3} \mathcal{I}_0(m_q^2) [D_\rho, [D^\rho, D^\mu]] \right\} \gamma_\mu \Gamma_j b_v \rangle + \dots, \qquad (3.40)$$

where $\mathcal{I}_0(m_q^2)$ is given in Eq. (B.23). The gamma structure on the r.h.s. of Eq. (3.40) reduces to

$$\gamma_{\nu} (1 - \gamma_5) \gamma_{\mu} \gamma^{\nu} (1 - \gamma_5) = 2 (2 - D) \gamma_{\mu} (1 - \gamma_5), \quad j = 1, \quad (3.41)$$

$$(1 - \gamma_5) \gamma_{\mu} (1 + \gamma_5) = 2 \gamma_{\mu} (1 + \gamma_5), \qquad j = 2, \qquad (3.42)$$

and by substituting the expression in Eq. (B.23), it follows that

$$\langle \mathcal{O}_{4q,j}^{(q)} \rangle_{1-loop} = \frac{a_j}{24\pi^2} \left[\mu^{-2\epsilon} \left(\frac{1}{\epsilon} - \gamma_E + \log(4\pi) \right) + \log\left(\frac{\mu^2}{m_q^2} \right) + b_j \right] \\ \times \left[\frac{3}{4} \langle \bar{b}_v [D_\nu, [D_\sigma, D_\tau]] \gamma^\nu \gamma^\sigma \gamma^\tau (\gamma_5 + (-1)^j) b_v \rangle \right. \\ \left. + i \langle \bar{b}_v [D_\rho, [D^\rho, D^\mu]] \gamma_\mu (1 + (-1)^j \gamma_5) b_v \rangle \right] + \dots, \qquad (3.43)$$

with

$$a_1 = 2, \quad b_1 = -1, \qquad a_2 = -1, \quad b_2 = 0.$$
 (3.44)

Note that the tensor decomposition of three gamma matrices Eq. (7) has been used to rewrite $i \epsilon^{\nu \sigma \tau \mu} \gamma_{\mu}$ in the second line of Eq. (3.43), see also Eq. (2.54), and that terms $\mathcal{O}(\epsilon)$ have been already neglected. We emphasise that the presence of the constant terms b_j in Eq. (3.43) depends on the choice of the four-quark operators basis in Eq. (3.28). This will be discussed further at the end of this chapter.

In order to simplify the structure of Eq. (3.43) and identity the relevant operators, first we take into account that due to parity conservation, matrix elements with an odd number of γ_5 vanish, then we recall that the rescaled heavy quark field $b_v(x)$ satisfies $(iv \cdot D)b_v(x) = \mathcal{O}(1/m_b)$, cf. Eq. (1.149), together with $\psi b_v(x) =$ $b_v(x) + \mathcal{O}(1/m_b)$, cf. Eq. (1.148). It follows that the two operators in the second and third line of Eq. (3.43), respectively give

$$\langle \bar{b}_{v}[D_{\nu}, [D_{\sigma}, D_{\tau}]]\gamma^{\nu}\gamma^{\sigma}\gamma^{\tau}b_{v}\rangle$$

$$= \frac{1}{2} \left(\langle \bar{b}_{v}\psi[D_{\nu}, [D_{\sigma}, D_{\tau}]]\gamma^{\nu}\gamma^{\sigma}\gamma^{\tau}b_{v}\rangle + \langle \bar{b}_{v}[D_{\nu}, [D_{\sigma}, D_{\tau}]]\gamma^{\nu}\gamma^{\sigma}\gamma^{\tau}\psi b_{v}\rangle \right) + \dots$$

$$= \frac{1}{2} \langle \bar{b}_{v}[D_{\nu}, [D_{\sigma}, D_{\tau}]] \left(\psi\gamma^{\nu}\gamma^{\sigma}\gamma^{\tau} + \gamma^{\nu}\psi\gamma^{\sigma}\gamma^{\tau} + 2v^{\tau}\gamma^{\nu}\gamma^{\sigma} - 2v^{\sigma}\gamma^{\nu}\gamma^{\tau}\right) b_{v}\rangle + \dots$$

$$= -i \langle \bar{b}_{v}[D_{\nu}, [D_{\sigma}, D_{\tau}]] \left(v^{\nu}\sigma^{\sigma\tau} + v^{\tau}\sigma^{\nu\sigma} - v^{\sigma}\sigma^{\nu\tau}\right) b_{v}\rangle + \dots$$

$$= \mathcal{O} \left(\frac{1}{m_{b}}\right), \qquad (3.45)$$

where in the last step we have taken into account the result in Eq. (1.67), which applies up to order $\mathcal{O}(1/m_b)$, moreover

$$i\langle \bar{b}_{v}[D_{\rho}, [D^{\rho}, D^{\mu}]]\gamma_{\mu}b_{v}\rangle$$

$$= \frac{i}{2} \Big(\langle \bar{b}_{v} \psi[D_{\rho}, [D^{\rho}, D^{\mu}]]\gamma_{\mu}b_{v}\rangle + \langle \bar{b}_{v}[D_{\rho}, [D^{\rho}, D^{\mu}]]\gamma_{\mu}\psi b_{v}\rangle \Big) + \dots$$

$$= \frac{i}{2} \langle \bar{b}_{v}[D_{\rho}, [D^{\rho}, D^{\mu}]]\{\psi, \gamma_{\mu}\}b_{v}\rangle + \dots$$

$$= \langle \bar{b}_{v}[iD_{\rho}, [iD^{\mu}, iD^{\rho}]]v_{\mu}b_{v}\rangle + \dots$$

$$= 2 \langle \mathcal{O}_{\rho_{D}}\rangle_{tree} + \mathcal{O}\left(\frac{1}{m_{b}}\right), \qquad (3.46)$$

with

$$\langle \mathcal{O}_{\rho_D} \rangle_{tree} = \frac{1}{2} \langle B | \bar{b}_v \left[i D_\rho, \left[i v \cdot D, i D^\rho \right] \right] b_v | B \rangle.$$
(3.47)

Using Eqs. (3.45), (3.46), from Eq. (3.43) we finally obtain

$$\langle \mathcal{O}_{4q,j}^{(q)} \rangle_{1-loop} = \frac{a_j}{12\pi^2} \left[\mu^{-2\epsilon} \left(\frac{1}{\epsilon} - \gamma_E + \log(4\pi) \right) + \log\left(\frac{\mu^2}{m_q^2}\right) + b_j \right] \langle \mathcal{O}_{\rho_D} \rangle_{tree} + \dots,$$
(3.48)

and the ellipsis refer to terms of higher order in α_s and $1/m_b$. We can now consider the one-loop matrix element of the colour octet operators $\mathcal{O}_{4q,j+2}^{(q)}$, j = 1, 2, namely

$$\langle \mathcal{O}_{4q,j+2}^{(q)} \rangle_{1-loop} = \langle i \int d^4 z \operatorname{T} \left\{ \vec{\mathcal{O}}_{4q,j+2}^{(q)}, \sum_{q'} \bar{q}'(z) \mathcal{A}(z) q'(z) \right\} \rangle + \dots$$
$$= \langle \bar{b}_v^l \bar{\Gamma}_j t^a_{lm} \left(\int d^4 z \, i S_0^{(q)}(-z) i \mathcal{A}^b(z) t^b_{mr} \, i S_0^{(q)}(z) \right) \Gamma_j t^a_{rs} b^s_v \rangle + \dots$$
(3.49)

where for clarity the colour indices have been explicitly indicated. Taking into account the relations in Eqs. (9), (11), it is straightforward to simplify the product of three colour matrices on the r.h.s. of Eq. (3.49), i.e.

$$(t^{a} \cdot t^{b} \cdot t^{a})_{ls} = (t^{b} \cdot t^{a} \cdot t^{a})_{ls} + if^{abc}(t^{c} \cdot t^{a})_{ls}$$
$$= C_{F}t^{b}_{ls} - \frac{1}{2}f^{bac}f^{dac}t^{d}_{ls}$$
$$= t^{b}_{ls}\left(C_{F} - \frac{C_{A}}{2}\right) = -\frac{1}{2N_{c}}t^{b}_{ls}.$$
(3.50)

Substituting Eq. (3.50) into Eq. (3.49) and comparing with Eq. (3.29) we can readily obtain that

$$\langle \mathcal{O}_{4q,j+2}^{(q)} \rangle_{1-loop} = -\frac{1}{2N_c} \langle \bar{b}_v \bar{\Gamma}_j \left(\int d^4 z \, i S_0^{(q)}(-z) i \mathcal{A}(z) i S_0^{(q)}(z) \right) \Gamma_j b_v \rangle$$
$$= -\frac{1}{2N_c} \langle \mathcal{O}_{4q,j}^{(q)} \rangle_{1-loop} + \dots, \qquad (3.51)$$

and the ellipsis stand for terms of order $\mathcal{O}(\alpha_s)$ and $\mathcal{O}(1/m_b)$. Eq. (3.51) shows that up to higher order corrections, to compute the mixing between the four-quark and the Darwin operators, it is sufficient to know only the contribution of the colour singlet operators $\langle \mathcal{O}_{4q,j}^{(q)} \rangle_{1-loop}$. The basis in Eq. (3.28) is then redundant and the computation would appear simpler if we would had chosen the equivalent basis 10

$$\vec{\mathcal{O}}_{4q}^{(q)\,\prime} = \left(\mathcal{O}_1^{(q)}, \, \mathcal{O}_2^{(q)}, \, \mathcal{O}_3^{(q)}, \, \mathcal{O}_4^{(q)}\right), \tag{3.52}$$

in terms of the colour-rearranged four-quark operators

$$\mathcal{O}_{3}^{(q)} = \left(\bar{b}_{v}^{l}\gamma_{\nu}(1-\gamma_{5})q^{m}\right)\left(\bar{q}^{m}\gamma^{\nu}(1-\gamma_{5})b_{v}^{l}\right), \qquad (3.53)$$

$$\mathcal{O}_{4}^{(q)} = \left(\bar{b}_{v}^{l}(1-\gamma_{5})q^{m}\right)\left(\bar{q}^{m}(1+\gamma_{5})b_{v}^{l}\right).$$
(3.54)

In fact, from

$$\mathcal{O}_{4q,j+2}^{(q)\prime} = \left(2 \,\mathcal{O}_{4q,j+2}^{(q)} + \frac{1}{N_c} \,\mathcal{O}_{4q,j}^{(q)}\right) \,, \tag{3.55}$$

it immediately follows that

$$\langle \mathcal{O}_{4q,j+2}^{(q)\prime} \rangle_{1-loop} = \mathcal{O}\left(\frac{1}{m_b}\right) + \mathcal{O}(\alpha_s).$$
 (3.56)

However, we will continue the computation using the original basis in Eq. (3.28).

The one-loop matrix elements in Eq. (3.48), and then also in Eq. (3.51), are divergent in the limit $\epsilon \to 0$ and need to be renormalised. For the sake of clarity, let us introduce the compact notation for the dimension-six operators in Eq. (3.27) and their corresponding coefficients i.e.

$$\vec{\mathcal{O}} = \left(\mathcal{O}_{\rho_D}, \sum_q \vec{\mathcal{O}}_{4q}^{(q)}\right), \qquad \vec{c} = \left(c_{\rho_D}, \sum_q \vec{c}_{4q}^{(q)}\right) \tag{3.57}$$

so that the operator renormalisation reads

$$\langle \vec{\mathcal{O}} \rangle^{(0)} = \hat{Z} \langle \vec{\mathcal{O}} \rangle,$$
 (3.58)

where $\langle \vec{O} \rangle^{(0)}$ and $\langle \vec{O} \rangle$ denote respectively the bare and the renormalised matrix elements and the \hat{Z} matrix is constructed, by definition, to absorb the divergences of

 $^{^{10}}$ We stress that the computation in Ref. [137] has been performed in this basis.

 $\langle \vec{\mathcal{O}} \rangle^{(0)}$. From

$$\langle \vec{\mathcal{O}} \rangle^{(0)} = \langle \vec{\mathcal{O}} \rangle_{tree} + \langle \vec{\mathcal{O}} \rangle_{1-loop} + \dots, \qquad (3.59)$$

and Eqs. (3.48), (3.51), we can read the expression of the \hat{Z} matrix and of the renormalised matrix elements $\langle \vec{\mathcal{O}} \rangle$, in the $\overline{\text{MS}}$ scheme [174] and at order α_s^0 , namely

$$\hat{Z} = \begin{pmatrix}
1 & 0 & 0 & 0 & 0 \\
Z_{21} & 1 & 0 & 0 & 0 \\
Z_{31} & 0 & 1 & 0 & 0 \\
Z_{41} & 0 & 0 & 1 & 0 \\
Z_{51} & 0 & 0 & 0 & 1
\end{pmatrix} + \mathcal{O}(\alpha_s),$$
(3.60)

with

$$Z_{21} = \frac{a_1}{12\pi^2} \mu^{-2\epsilon} \left(\frac{1}{\epsilon} - \gamma_E + \log(4\pi) \right) , \qquad (3.61)$$

$$Z_{31} = \frac{a_2}{12\pi^2} \mu^{-2\epsilon} \left(\frac{1}{\epsilon} - \gamma_E + \log(4\pi) \right) , \qquad (3.62)$$

$$Z_{41} = -\frac{1}{2N_c} Z_{21}, \quad Z_{51} = -\frac{1}{2N_c} Z_{31}, \qquad (3.63)$$

and

$$\langle \mathcal{O}_{\rho_D} \rangle(\mu) = \langle \mathcal{O}_{\rho_D} \rangle_{tree} + \mathcal{O}(\alpha_s) ,$$
 (3.64)

$$\langle \mathcal{O}_{4q,j}^{(q)} \rangle (\mu) = \langle \mathcal{O}_{4q,j} \rangle_{tree} + \frac{a_j}{12\pi^2} \left(\log\left(\frac{\mu^2}{m_q^2}\right) + b_j \right) \langle \mathcal{O}_{\rho_D} \rangle_{tree} + \mathcal{O}(\alpha_s) , \qquad (3.65)$$

$$\langle \mathcal{O}_{4q,j+2}^{(q)} \rangle(\mu) = \langle \mathcal{O}_{4q,j+2} \rangle_{tree} - \frac{1}{2N_c} \frac{a_j}{12\pi^2} \left(\log\left(\frac{\mu^2}{m_q^2}\right) + b_j \right) \langle \mathcal{O}_{\rho_D} \rangle_{tree} + \mathcal{O}(\alpha_s) ,$$
(3.66)

where j = 1, 2 and a_j, b_j are given in Eq. (3.44). While Eqs. (3.64)-(3.66) describe how the renormalised matrix elements depend on the renormalisation scale μ , the behaviour under a variation of μ is obtained by requiring that the bare matrix elements $\langle \vec{\mathcal{O}} \rangle^{(0)}$ in Eq. (3.58), are scale independent, i.e.

$$\frac{d}{d\log\mu^2}\langle\vec{\mathcal{O}}\rangle^{(0)} = 0\,, \quad \Rightarrow \quad \frac{d\langle\vec{\mathcal{O}}\rangle}{d\log\mu^2} = -\hat{\gamma}\langle\vec{\mathcal{O}}\rangle\,, \tag{3.67}$$

with the anomalous dimension matrix (ADM) $\hat{\gamma}$ defined as

$$\hat{\gamma} = \hat{Z}^{-1} \frac{d}{d \log \mu^2} \hat{Z}, \qquad (3.68)$$

or explicitly, taking into account Eq. (3.60)

$$\hat{\gamma} = \begin{pmatrix} 0 & 0 & 0 & 0 & 0 \\ \gamma_{21} & 0 & 0 & 0 & 0 \\ \gamma_{31} & 0 & 0 & 0 & 0 \\ \gamma_{41} & 0 & 0 & 0 & 0 \\ \gamma_{51} & 0 & 0 & 0 & 0 \end{pmatrix} + \mathcal{O}(\alpha_s), \qquad (3.69)$$

where

$$\gamma_{21} = -\frac{a_1}{12\pi^2}, \quad \gamma_{31} = -\frac{a_2}{12\pi^2}, \quad \gamma_{41} = -\frac{\gamma_{21}}{2N_c}, \quad \gamma_{51} = -\frac{\gamma_{31}}{2N_c}. \tag{3.70}$$

The RGEs for the renormalised matrix elements in Eq. (3.67) lead to the corresponding ones for the renormalised coefficients \vec{c} , since the product

$$\langle \mathrm{Im}\mathcal{T} \rangle \Big|_{d=6} = \vec{c} \cdot \langle \vec{\mathcal{O}} \rangle,$$
 (3.71)

must be scale independent. We then obtain the following system of equations

$$\begin{cases} \frac{d\langle \vec{\mathcal{O}} \rangle}{d \log \mu^2} &= -\hat{\gamma} \langle \vec{\mathcal{O}} \rangle, \\ \frac{d \vec{c}}{d \log \mu^2} &= \hat{\gamma}^T \vec{c}, \end{cases}$$
(3.72)

which can be easily solved since $\hat{\gamma}$ is constant at this order. Integrating from the matching scale $\mu = m_b$ to $\mu = \mu_0$, respectively yields

$$\langle \mathcal{O}_{\rho_D} \rangle(\mu_0) = \langle \mathcal{O}_{\rho_D} \rangle(m_b) + \mathcal{O}(\alpha_s),$$
 (3.73)

$$\langle \vec{\mathcal{O}}_{4q}^{(q)} \rangle(\mu_0) = \langle \vec{\mathcal{O}}_{4q}^{(q)} \rangle(m_b) - \vec{\gamma} \log\left(\frac{\mu_0^2}{m_b^2}\right) \langle \mathcal{O}_{\rho_D} \rangle(m_b) + \mathcal{O}(\alpha_s) , \qquad (3.74)$$

together with

$$\vec{c}_{4q}^{(q)}(\mu_0) = \vec{c}_{4q}^{(q)}(m_b) + \mathcal{O}(\alpha_s), \qquad (3.75)$$

$$c_{\rho_D}(\mu_0) = c_{\rho_D}(m_b) + \sum_q \vec{\gamma} \cdot \vec{c}_{4q}^{(q)}(m_b) \log\left(\frac{\mu_0^2}{m_b^2}\right) + \mathcal{O}(\alpha_s), \qquad (3.76)$$

and we have defined $\vec{\gamma} = (\gamma_{21}, \gamma_{31}, \gamma_{41}, \gamma_{51})$. Substituting Eqs. (3.73)-(3.76) into Eq. (3.27) and taking into account Eqs. (3.64)-(3.66), we arrive at

$$\left\langle \operatorname{Im} \mathcal{T} \right\rangle \Big|_{d=6} = \left[c_{\rho_D}(m_b) + \sum_{q} \vec{\gamma} \cdot \vec{c}_{4q}^{(q)}(m_b) \log\left(\frac{\mu_0^2}{m_b^2}\right) \right] \left\langle \mathcal{O}_{\rho_D} \right\rangle_{tree} + \sum_{q} \sum_{j=1}^{2} \left(c_{4q,j}^{(q)}(m_b) - \frac{c_{4q,j+2}^{(q)}(m_b)}{2N_c} \right) \frac{a_j}{12\pi^2} \left[\log\left(\frac{m_b^2}{m_q^2}\right) + b_j \right] \left\langle \mathcal{O}_{\rho_D} \right\rangle_{tree} + \sum_{q} \vec{c}_{4q}^{(q)}(m_b) \cdot \left[\left\langle \vec{\mathcal{O}}_{4q}^{(q)} \right\rangle_{tree} - \vec{\gamma} \log\left(\frac{\mu_0^2}{m_b^2}\right) \left\langle \mathcal{O}_{\rho_D} \right\rangle_{tree} \right] + \mathcal{O}(\alpha_s) , \quad (3.77)$$

in which the dependence on the renormalisation scale μ_0 cancels, consistently with a calculation of order α_s^0 . Dropping for simplicity the suffix *tree*, we then obtain

$$\left\langle \operatorname{Im} \mathcal{T} \right\rangle \Big|_{d=6} = c_{\rho_D}(m_b) \left\langle \mathcal{O}_{\rho_D} \right\rangle + \sum_q \vec{c}_{4q}^{(q)}(m_b) \cdot \left\langle \vec{\mathcal{O}}_{4q}^{(q)} \right\rangle + \sum_q \sum_{j=1}^2 \left(c_{4q,j}^{(q)}(m_b) - \frac{c_{4q,j+2}^{(q)}(m_b)}{2N_c} \right) \frac{a_j}{12\pi^2} \left[\log\left(\frac{m_b^2}{m_q^2}\right) + b_j \right] \left\langle \mathcal{O}_{\rho_D} \right\rangle + \mathcal{O}(\alpha_s)$$

$$(3.78)$$

The l.h.s. of Eq. (3.78) has been computed in Section 2.2.1 and in Section 3.1 and schematically reads

$$\left\langle \mathrm{Im}\mathcal{T}\right\rangle\Big|_{d=6} = \mathcal{C}_D \left\langle \mathcal{O}_{\rho_D} \right\rangle + \vec{\mathcal{C}}_{\mathrm{WE}} \cdot \left\langle \vec{\mathcal{O}}_{4q}^{(q_3)} \right\rangle + \vec{\mathcal{C}}_{\mathrm{WA}} \cdot \left\langle \vec{\mathcal{O}}_{4q}^{(q_1)} \right\rangle + \vec{\mathcal{C}}_{\mathrm{PI}} \cdot \left\langle \vec{\mathcal{O}}_{4q}^{(q_2)} \right\rangle, \qquad (3.79)$$

where the coefficients C_D have the divergent behaviour shown in Eq. (3.26). Equating the respective r.h.s. of Eqs. (3.78), (3.79), we can finally read the expression of $\vec{c}_{4q}^{(q)}$ and c_{ρ_D} at the matching scale m_b , namely

$$\vec{c}_{4q}^{(q)}(m_b) = \vec{\mathcal{C}}_{\rm WE} \,\delta^{qq_3} + \vec{\mathcal{C}}_{\rm WA} \,\delta^{qq_1} + \vec{\mathcal{C}}_{\rm PI} \,\delta^{qq_2} \,, \tag{3.80}$$

and

$$c_{\rho_D}(m_b) = \mathcal{C}_D - \sum_q \sum_{j=1}^2 \left(c_{4q,j}^{(q)}(m_b) - \frac{1}{2N_c} c_{4q,j+2}^{(q)}(m_b) \right) \frac{a_j}{12\pi^2} \left[\log\left(\frac{m_b^2}{m_q^2}\right) + b_j \right].$$
(3.81)

It is straightforward to check, by taking into account the results for the coefficients of the four-quark operators computed in Section 2.2.1, together with the expressions of the divergent functions listed in Appendix E, that all the dependence on $\log (m_q^2/m_b^2)$ exactly cancels on the r.h.s. of Eq. (3.81), leaving the final coefficient of the Darwin operator c_{ρ_D} , free of any IR divergences. Notice though that as consequence of the operator mixing, C_D and c_{ρ_D} differ also by a finite contribution due to the presence of the constants b_j . The complete expressions for the coefficients c_{ρ_D} are presented in the next section.

3.3 Analytical expressions for the coefficients of the Darwin operator

The final expression for the contribution of the Darwin operator to the inclusive decay width of a *B* meson, induced by the flavour-changing transition $b \to q_1 \bar{q}_2 q_3$, with $q_1, q_2 = \{u, c\}$ and $q_3 = \{d, s\}$, is presented in the following form ¹¹

$$\Gamma_{\rho_D}^{(q_1\bar{q}_2q_3)}(B) = \Gamma_0 \left(N_c C_1^2 c_{\rho_D,11}^{(q_1\bar{q}_2q_3)} + 2 C_1 C_2 c_{\rho_D,12}^{(q_1\bar{q}_2q_3)} + N_c C_2^2 c_{\rho_D,22}^{(q_1\bar{q}_2q_3)} \right) \frac{\rho_D^3(B)}{m_b^3}, \quad (3.82)$$

¹¹Note that the coefficients originally presented in Ref. [137], still depend on the renormalisation scale μ_0 since the running of the matrix element of the corresponding four-quark operators was not explicitly included.

where the non perturbative parameter $\rho_D(B)$ is defined as in Eq. (1.160), Γ_0 is given in Eq. (3.24) and the coefficients $c_{\rho_D,mn}^{(q_1\bar{q}_2q_3)}$ respectively read

$$c_{\rho_D,11}^{(u\bar{u}d)} = 6, \qquad c_{\rho_D,12}^{(u\bar{u}d)} = -\frac{34}{3}, \qquad c_{\rho_D,22}^{(u\bar{u}d)} = 6,$$
(3.83)

for the $b \rightarrow u\bar{u}d \mod d$,

$$c_{\rho_D,11}^{(u\bar{c}s)} = \frac{2}{3}(1-\rho) \bigg[9 + 11\rho - 12\rho^2 \log(\rho) - 24\left(1-\rho^2\right) \log(1-\rho) - 25\rho^2 + 5\rho^3 \bigg],$$
(3.84)

$$c_{\rho_D,12}^{(u\bar{c}s)} = \frac{2}{3} \bigg[-41 - 12 \left(2 + 5\rho + 2\rho^2 - 2\rho^3 \right) \log(\rho) - 48(1-\rho)^2 (1+\rho) \log(1-\rho) \bigg]$$

$$+26\rho - 18\rho^2 + 38\rho^3 - 5\rho^4 \bigg], \qquad (3.85)$$

$$c_{\rho_D,22}^{(u\bar{c}s)} = \frac{2}{3}(1-\rho) \bigg[9 + 11\rho - 12\rho^2 \log(\rho) - 24\left(1-\rho^2\right) \log(1-\rho) - 25\rho^2 + 5\rho^3 \bigg],$$
(3.86)

in the case of $b \rightarrow u \bar{c} s$ transition,

$$c_{\rho_{D},11}^{(c\bar{u}d)} = \frac{2}{3} \bigg[17 + 12\log(\rho) - 16\rho - 12\rho^{2} + 16\rho^{3} - 5\rho^{4} \bigg], \qquad (3.87)$$

$$c_{\rho_{D},12}^{(c\bar{u}d)} = \frac{2}{3} \bigg[-9 + 12\left(1 - 3\rho^{2} + \rho^{3}\right)\log(\rho) + 24(1 - \rho)^{3}\log(1 - \rho) + 50\rho - 90\rho^{2} + 54\rho^{3} - 5\rho^{4} \bigg], \qquad (3.88)$$

$$c_{\rho_{D},22}^{(c\bar{u}d)} = \frac{2}{3}(1 - \rho) \bigg[9 + 11\rho - 12\rho^{2}\log(\rho) + 24\left(1 - \rho^{2}\right)\log(1 - \rho) - 25\rho^{2} + 5\rho^{3} \bigg], \qquad (3.89)$$

for the $b \rightarrow u\bar{c}s$ mode and finally

$$c_{\rho_D,11}^{(c\bar{c}s)} = \frac{2}{3} \left[\sqrt{1 - 4\rho} \left(17 + 8\rho - 22\rho^2 - 60\rho^3 \right) - 12 \left(1 - \rho - 2\rho^2 + 2\rho^3 + 10\rho^4 \right) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) \right], \quad (3.90)$$

$$c_{\rho_D,12}^{(c\bar{c}s)} = \frac{2}{3} \left[\sqrt{1 - 4\rho} \left(-45 + 46\rho - 106\rho^2 - 60\rho^3 \right) + 12 \left(1 + 4\rho^2 - 16\rho^3 - 10\rho^4 \right) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) \right] + 8 \left[\mathcal{M}_{112}(\rho, \eta) - \sqrt{1 - 4\rho} \log(\eta) \right] \Big|_{\eta \to 0},$$
(3.91)

$$c_{\rho_{D},22}^{(c\bar{c}s)} = \frac{2}{3} \left[\sqrt{1 - 4\rho} \left(-3 + 22\rho - 34\rho^{2} - 60\rho^{3} \right) - 24\rho \left(1 + \rho + 2\rho^{2} + 5\rho^{3} \right) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) \right] + 8 \left[\mathcal{M}_{112}(\rho, \eta) - \sqrt{1 - 4\rho} \log(\eta) \right] \Big|_{\eta \to 0}, \qquad (3.92)$$

in the case of $b \to c\bar{c}s$ decay. In Eqs. (3.84)-(3.92) the dimensionless mass parameter is $\rho = m_c^2/m_b^2$, moreover the master integral \mathcal{M}_{112} in Eqs. (3.91), (3.92) is defined as

$$\mathcal{M}_{112}(\rho,\eta) = -\int_{(\sqrt{\rho}+\sqrt{\eta})^2}^{(1-\sqrt{\rho})^2} dt \, \frac{(t^2 - 2(1+\rho)t + (1-\rho)^2)(t-\eta+\rho)}{t\sqrt{(t^2 - 2(1+\rho)t + (1-\rho)^2)(t^2 - 2t(\eta+\rho) + (\eta-\rho)^2)}},$$
(3.93)

with $\eta = m_q^2/m_b^2$. We emphasise however, that the analytical expression for the limits in Eqs. (3.91), (3.92), has been derived in Ref. [138], namely

$$\begin{split} \left[\mathcal{M}_{112}(\rho,\eta) - \sqrt{1 - 4\rho} \, \log(\eta) \right] \Big|_{\eta \to 0} &= 2(1 - \rho) \log\left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}}\right) \\ &+ \sqrt{1 - 4\rho} \left[1 + 2\log(\rho) - 4\log\left(\sqrt{1 - 4\rho}\right) \right], \end{split}$$
(3.94)



Figure 3.5: Relative effect of the Darwin operator with respect to the dimensionthree term for the $b \to c\bar{u}d$ (top left), $b \to u\bar{u}d$ (top right), $b \to c\bar{c}s$ (bottom left), and $b \to u\bar{c}s$ (bottom right) transitions. For each mode, the green dotted, the solid cyan and the dotted-dashed orange lines correspond respectively to the $Q_1 \otimes Q_1$, $Q_1 \otimes Q_2$ and $Q_2 \otimes Q_2$ contributions. The reference values $m_b = 4.5$ GeV and $\rho_D^3 = 0.2$ GeV³ have been used and the dashed vertical line indicates the approximate value $\rho = 0.05$ in the $\overline{\text{MS}}$ scheme.

in terms of which Eqs. (3.91), (3.92), can be simplified as

$$c_{\rho_D,12}^{(c\bar{c}s)} = \frac{2}{3} \left[\sqrt{1 - 4\rho} \left(-33 + 24 \log(\rho) - 24 \log(1 - 4\rho) + 46\rho - 106\rho^2 - 60\rho^3 \right) + 12 \left(3 - 2\rho + 4\rho^2 - 16\rho^3 - 10\rho^4 \right) \log\left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) \right], \quad (3.95)$$

$$c_{\rho_D,22}^{(c\bar{c}s)} = \frac{2}{3} \left[\sqrt{1 - 4\rho} \left(9 + 24 \log(\rho) - 24 \log(1 - 4\rho) + 22\rho - 34\rho^2 - 60\rho^3 \right) + 24 \left(1 - 2\rho - \rho^2 - 2\rho^3 - 5\rho^4 \right) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) \right].$$
 (3.96)

Finally, we stress that for $m_d = m_s = 0$, the following relations hold, i.e.

$$c_{\rho_D,mn}^{(c\bar{u}d)} = c_{\rho_D,mn}^{(c\bar{u}s)}, \qquad c_{\rho_D,mn}^{(c\bar{c}s)} = c_{\rho_D,mn}^{(c\bar{c}d)}, \qquad c_{\rho_D,mn}^{(u\bar{u}d)} = c_{\rho_D,mn}^{(u\bar{u}s)}, \qquad c_{\rho_D,mn}^{(u\bar{c}s)} = c_{\rho_D,mn}^{(u\bar{c}d)}.$$
(3.97)



Figure 3.6: Relative size of the Darwin term compared to the partonic-level contribution respectively for the $b \rightarrow u\bar{u}d$ (dashed orange), $b \rightarrow c\bar{u}d$ (solid purple), $b \rightarrow u\bar{c}s$ (dotted green) and $b \rightarrow c\bar{c}s$ (dot-dashed cyan) modes. The reference values $m_b = 4.5 \text{ GeV}$ and $\rho_D^3 = 0.2 \text{ GeV}^3$ have been used and the dashed vertical line indicates the approximate value $\rho = 0.05$ in the $\overline{\text{MS}}$ scheme.

The relative effect of the Darwin operator with respect to the corresponding partoniclevel contribution $\mathcal{C}_0^{(q_1\bar{q}_2q_3)}$ is given by

$$\Delta_{\rho_D,mn}^{(q_1\bar{q}_2q_3)} = \frac{c_{\rho_D,mn}^{(q_1\bar{q}_2q_3)}}{\mathcal{C}_0^{(q_1\bar{q}_2q_3)}} \frac{\rho_D^3}{m_b^3}.$$
(3.98)

In Figure 3.5, the dependence of the functions in Eq. (3.98) on the dimensionless mass parameter ρ , is plotted for all the three colour structures and the four modes, using for reference the values $m_b = 4.5$ GeV and $\rho_D^3 = 0.2$ GeV³. Furthermore, Figure 3.6 shows the total relative contribution for each mode, namely

$$\Delta_{\rho_D}^{(q_1\bar{q}_2q_3)} = \frac{3 C_1^2 c_{\rho_D,11}^{(q_1\bar{q}_2q_3)} + 2 C_1 C_2 c_{\rho_D,12}^{(q_1\bar{q}_2q_3)} + 3 C_2^2 c_{\rho_D,22}^{(q_1\bar{q}_2q_3)}}{\left(3 C_1^2 + 2 C_1 C_2 + 3 C_2^2\right) \mathcal{C}_0^{(q_1\bar{q}_2q_3)}} \frac{\rho_D^3}{m_b^3}, \qquad (3.99)$$

indicating that the Darwin operator can lead to sizeable corrections to the $b \rightarrow q_1 \bar{q}_2 q_3$ decay width, of the order of 1 - 7% (for $\rho = 0.05$).

A final comment about the numerical effect of the constant terms b_j in Eq. (3.81) for the coefficients $c_{\rho_D,mm}^{(q_1\bar{q}_2q_3)}$. We have already noticed that their values depend on the choice of the four-quark operators basis. Consider as an example the coefficient $c_{\rho_D,12}^{(c\bar{u}d)}$. In our basis given by Eq. (3.28), $c_{\rho_D,12}^{(c\bar{u}d)} = -29.0$, for the reference value $\rho = 0.05$. Had we chosen the same basis as done in Ref. [138], namely with $O_1^{(u)}$ and $O_2^{(u)}$, replaced by

$$\mathcal{O}_1^{(u)} = (\bar{b}_v^i \Gamma^\sigma \gamma^\mu \Gamma^\rho u^i) (\bar{u}^j \Gamma_\sigma \gamma_\mu \Gamma_\rho b_v^j), \qquad \mathcal{O}_2^{(u)} = (\bar{b}_v^i \Gamma^\sigma \psi \Gamma^\rho u^i) (\bar{u}^j \Gamma_\sigma \psi \Gamma_\rho b_v^j), \quad (3.100)$$

we would have obtained in Eq. (3.44), that $a_1 = 8$, $b_1 = -5/4$ and $a_2 = 2$, $b_2 = -3/2$, leading to $c_{\rho_D,12}^{(c\bar{u}d)} = -24.0$, for the same value of ρ . This shift of ~ 17% must be compensated, up to corrections of higher orders, by the different value of the matrix element of the operators defined in these two bases.

Chapter 4

Phenomenology of Lifetime and Mixing

In this last chapter, we consider two phenomenological applications of the HQE in the charm sector, specifically, the study of the lifetime of charmed mesons and of neutral D-meson mixing. In light of the large amount of current and future charm data collected by LHCb [175], BESIII [176], and Belle-II [177], an improvement of the theoretical understanding of charm physics, see Refs. [34, 178–181] for a comprehensive introduction to the subject, is crucial to fully exploit the significant experimental progress in this field. The recent discovery by the LHCb collaboration [182] of direct CP violation in the charm sector, specifically in the non-leptonic decays $D^0 \to \pi^+ + \pi^-$ and $D^0 \to K^+ + K^-$, provides one example of this necessity, since after its announcement, both SM and BSM interpretations of the measurement have been proposed, see Refs. [183–186] for the former, and Refs. [187, 188], partly based on the calculation of Ref. [189], for the latter ¹. Exclusive non-leptonic decays of charm hadrons and even of b-hadrons are among the most challenging observables in quark flavour physics from a theoretical point of view. In the following, we will instead focus on the study of inclusive quantities like the total decay width, for which the HQE provides a systematic theoretical framework, see Section 1.4. However, due to the size of its mass, the charm quark sits at the boundary between the heavy- and light-quark region, making the applicability of the HQE a priory questionable. This is clearly signalled by the fact that contrary to the *b*-sector, lifetime ratios of charmed hadrons can significantly differ from one, which represents the naive expectation in the heavy quark limit. Specifically, lifetimes of charmed hadrons are experimentally determined very precisely [191] and also inclusive semileptonic branching fractions have been measured [191], with a recent update for the D_s -meson released by the

 $^{^{1}}$ A summary of references investigating a previous claim for evidence of CP violation can be found in Ref. [190].

	D^0	D^+	D_s^+	
$\tau [\mathrm{ps}]$	0.4101(15)	1.040(7)	0.504(4)	
$\Gamma [\mathrm{ps}^{-1}]$	2.44(1)	0.96(1)	1.98(2)	
$ au(D_X)/ au(D^0)$	1	2.54(2)	1.20(1)	
$Br(D_X \to Xe^+\nu_e)[\%]$	6.49(11)	16.07(30)	6.30(16)	
$\frac{\Gamma(D_X \to X e^+ \nu_e)}{\Gamma(D^0 \to X e^+ \nu_e)}$	1	0.977(26)	0.790(26)	

Table 4.1: Status of the experimental determinations of the lifetime and the semileptonic branching fractions of the lightest charmed mesons. All values are taken from the PDG [191], apart from the ones for the semileptonic D_s -meson decays, which were recently measured by the BESIII Collaboration [192].

BESIII Collaboration [192]². A summary of the current experimental status for the lightest *D*-mesons, is shown in Table 4.1. Finally, a long-standing puzzle in charm physics, is the theoretical description of mixing of neutral *D* mesons, see e.g. the excellent reviews [194–196]. Charm-mixing is experimentally well established and the HFLAV [197] average of Refs. [182, 198–231] ³ reads

$$x = \frac{\Delta M_D}{\Gamma_{D^0}} = (0.409^{+0.048}_{-0.049})\%, \quad y = \frac{\Delta\Gamma_D}{2\Gamma_{D^0}} = 0.615^{+0.056}_{-0.055}\%, \quad (4.1)$$

where ΔM_D is the mass difference of the neutral D^0 mesons mass eigenstates and $\Delta \Gamma_D$ the corresponding decay rate difference. However, the theoretical predictions for x and y cover a vast range of values, which spread over several orders of magnitude, see e.g. Refs. [232, 233]. Future measurements will not only increase the precision of x and y, but also provide stronger bounds or even evidence for CP violation in mixing [234]. It is clear that having a reliable range of potential SM predictions is necessary in order to benefit from these experimental improvements.

In Section 4.1 we discuss the study of the total decay width of charmed mesons and obtain theoretical predictions for the lifetimes of the D^0 , D^+ , and D_s^+ mesons and their ratios, as well as for the semileptonic branching fractions $Br(D \to Xe^+\nu_e)$, and their ratios. Furthermore, in Section 4.2 we present a possible solution for the discrepancy between previous HQE determinations of *D*-mixing with data. The content of this chapter closely follows the one of Ref. [140] and Ref. [235].

²New results from Belle II have recently been published [193]: $\tau(D^0) = 410.5 \pm 1.1 \pm 0.8$ fs, $\tau(D^+) = 1030.4 \pm 4.7 \pm 3.1$ fs.

 $^{^{3}\}mathrm{Performed}$ in the case of allowed CP violation.

4.1 Theoretical study of the total decay width of charmed mesons

In the present section, we analyse the structure of the HQE in the charm sector, to try to shed further light into the question, whether the expansion parameters $\alpha_s(m_c)$, and $\Lambda_{\rm QCD}/m_c$, are small enough in order to ensure meaningful theoretical predictions for the observables listed in Table 4.1. The Particle Data Group [191] quotes, respectively for the pole and the $\overline{\rm MS}$ mass of the charm quark, the values

$$m_c^{\text{Pole}} = (1.67 \pm 0.07) \text{ GeV}, \qquad \overline{m}_c(\overline{m}_c) = (1.27 \pm 0.02) \text{ GeV}, \qquad (4.2)$$

while the dependence of α_s on both the charm scale and the loop order, obtained using the RunDec package [236], is shown in Table 4.2. We emphasise that in our numerical analysis we use the five-loop running result.

$\alpha_s(m_c)$	$m_c = 1.67 \mathrm{GeV}$	$m_c = 1.48 \mathrm{GeV}$	$m_c = 1.27 \mathrm{GeV}$
two-loop	0.322	0.346	0.373
five-loop	0.329	0.356	0.387

Table 4.2: Numerical values of the strong coupling α_s evaluated at different scales and loop order, obtained using the RunDec package [236].

The relation between the pole and $\overline{\text{MS}}$ mass schemes, up to third order in the strong coupling, reads [237–239]

$$m_c^{\text{Pole}} = \overline{m}_c(\overline{m}_c) \left[1 + \frac{4}{3} \frac{\alpha_s(\overline{m}_c)}{\pi} + 10.43 \left(\frac{\alpha_s(\overline{m}_c)}{\pi} \right)^2 + 116.5 \left(\frac{\alpha_s(\overline{m}_c)}{\pi} \right)^3 \right]$$
$$= \overline{m}_c(\overline{m}_c) \left[1 + 0.1642 + 0.1582 + 0.2176 \right], \qquad (4.3)$$

where we have used the five-loop value of α_s , at the scale 1.27 GeV. The strong dependence of Γ_3 on the charm pole mass, cf. Eq. (3.24), leads to different results according to how higher orders in Eq. (4.3) are treated. Specifically, by truncating the expansion in Eq. (4.3) at first order in α_s , and using $\overline{m}_c(\overline{m}_c) = 1.27$ GeV, we obtain for the pole mass the value $m_c^{\text{Pole}} = 1.479$ GeV, which, respectively yields

$$(m_c^{\text{Pole}})^5 = \overline{m}_c (\overline{m}_c)^5 [1 + 0.1642]^5 = 2.14 \,\overline{m}_c (\overline{m}_c)^5,$$
 (4.4)

by computing the fifth power of m_c^{Pole} , and

$$\left(m_c^{\text{Pole}}\right)^5 \approx \overline{m}_c (\overline{m}_c)^5 \left[1 + 5 \cdot 0.1642\right] = 1.82 \,\overline{m}_c (\overline{m}_c)^5 \,, \tag{4.5}$$

by further expanding the fifth power up to the first order in α_s . Note that the result in Eq. (4.5) is about 15 % smaller than the one in Eq. (4.4). Conversely, by including also all the higher order terms shown in Eq. (4.3), gives

$$\left(m_c^{\text{Pole}}\right)^5 = \overline{m}_c (\overline{m}_c)^5 \left[1 + 0.1642 + 0.1582 + 0.2176\right]^5 = 8.66 \,\overline{m}_c (\overline{m}_c)^5 \,, \qquad (4.6)$$

which is roughly four times larger than the result in Eq. (4.4). In order to deal with this numerical instability, in the following we investigate different scenarios:

- 1. Use Eq. (4.3) to first order in α_s , since this is the order at which most of the Wilson coefficients are known. In this case we fix $m_c^{\text{Pole}} = 1.48$ GeV and $\alpha_s = 0.356$. A further possibility would be to use as input the pole mass value from the PDG, i.e. $m_c^{\text{Pole}} = 1.67$ GeV. However, in this case, our numerical analysis gives results for the decay rates which are roughly 30% larger than the ones obtained in the 1*S* scheme, discussed below. Since we expect this enhancement to be compensated by missing NNLO corrections to the nonleptonic decay rates, we do not present explicit results for $m_c^{\text{Pole}} = 1.67$ GeV.
- 2. Express m_c^{Pole} in terms of the $\overline{\text{MS}}$ mass [174]

$$m_c^{\text{Pole}} = \overline{m}_c(\overline{m}_c) \left[1 + \frac{4}{3} \frac{\alpha_s(\overline{m}_c)}{\pi} \right], \qquad (4.7)$$

using $\overline{m}_c(\overline{m}_c) = 1.27$ GeV [191], and expand consistently up to order α_s . Because of the dependence on the fifth power of the charm quark mass, in this case, Γ_3 receives large corrections ~ 5 × (4/3)(α_s/π).

3. Express m_c^{Pole} in terms of the kinetic mass [104, 240]. The kinetic scheme has been introduced in order to obtain a short distance definition of the heavy quark mass which allows a faster convergence of the perturbative series and which is still valid at small scales $\mu \sim 1$ GeV. The relation between the kinetic scheme and the $\overline{\text{MS}}$ and pole schemes can be found, up to N³LO corrections, in Ref. [241]. At order α_s we have

$$m_c^{\text{Pole}} = m_c^{\text{Kin}} \left[1 + \frac{4\alpha_s}{3\pi} \left(\frac{4}{3} \frac{\mu^{\text{cut}}}{m_c^{\text{Kin}}} + \frac{1}{2} \left(\frac{\mu^{\text{cut}}}{m_c^{\text{Kin}}} \right)^2 \right) \right], \qquad (4.8)$$

where μ^{cut} is the Wilsonian cutoff separating the perturbative and non perturbative regimes. In our numerical analysis we set $\mu^{\text{cut}} = 0.5$ GeV, which gives, at NLO-QCD and using as an input $\overline{m}_c(3 \text{ GeV})$ [241]

$$m_c^{\rm kin}(0.5\,{\rm GeV}) = 1.363\,{\rm GeV}\,.$$
 (4.9)

4. Express m_c^{Pole} in the 1*S*-mass scheme, defined as [242, 243]

$$m_c^{\text{Pole}} = \frac{m_{J/\psi}}{2} \left(1 + \frac{(\alpha_s C_F)^2}{8} \right) ,$$
 (4.10)

where $C_F = 4/3$, and we use $m_{J/\psi} = 3.0969$ GeV [191], so that $m_{J/\psi}/2 \approx 1.55$ GeV. Note that in Eq. (4.10), the NLO correction actually starts at order α_s^2 , see Ref. [242].

4.1.1 Description of the computation

The non-leptonic decay of a charm quark $c \rightarrow q_1 \bar{q}_2 u$, with $q_1, q_2 = \{u, d, s\}$, is described by the following effective Hamiltonian, cf. Section 1.1, i.e.

$$\mathcal{H}_{\text{eff}}^{\text{NL}} = \frac{G_F}{\sqrt{2}} \left[\sum_{q_{1,2}=d,s} \lambda_{q_1q_2} \left(C_1 \, Q_1^{q_1q_2} + C_2 \, Q_2^{q_1q_2} \right) - \lambda_b \sum_{j=3}^6 C_j Q_j \right] + \text{h.c.}, \quad (4.11)$$

where we have defined the CKM factors respectively as $\lambda_{q_1q_2} = V_{cq_1}^* V_{uq_2}$ and $\lambda_b = V_{cb}^* V_{ub}$, and introduced the following notation for the tree-level $\Delta C = 1$ operators:

$$Q_1^{q_1 q_2} = \left(\bar{q}_1^i \gamma_\rho (1 - \gamma_5) c^i\right) \left(\bar{u}^j \gamma^\rho (1 - \gamma_5) q_2^j\right) , \qquad (4.12)$$

$$Q_2^{q_1q_2} = \left(\bar{q}_1^i \gamma_{\rho} (1-\gamma_5) c^j\right) \left(\bar{u}^j \gamma^{\rho} (1-\gamma_5) q_2^i\right) , \qquad (4.13)$$

while Q_j , with j = 3, ..., 6, refer to the penguin operators, which can only arise in the singly Cabibbo suppressed decays $c \to s\bar{s}u$ and $c \to d\bar{d}u$. In Eq. (4.11), $C_i(\mu_1)$, with i = 1, ..., 6, denote the corresponding Wilson coefficients evaluated at the renormalisation scale $\mu_1 \sim m_c$. A comparison of their values respectively at NLO-(LO-)QCD and for different choices of m_c , is shown in Table 4.3.

μ_1 [GeV]	1	1.27	1.36	1.48	1.55	3
$C_1(\mu_1)$	1.25 (1.34)	1.20 (1.27)	1.19 (1.26)	1.18 (1.24)	1.17 (1.23)	1.10 (1.15)
$C_2(\mu_1)$	-0.48 (-0.62)	$-0.39 \\ (-0.50)$	-0.40 (-0.53)	-0.37 (-0.48)	-0.36 (-0.47)	-0.24 (-0.32)
$C_3(\mu_1)$	$ \begin{array}{c} 0.03 \\ (0.02) \end{array} $	$0.02 \\ (0.01)$	$0.02 \\ (0.01)$	$0.01 \\ (0.01)$	$0.01 \\ (0.01)$	$0.00 \\ (0.00)$
$C_4(\mu_1)$	-0.06 (-0.04)	$-0.05 \ (-0.03)$	-0.04 (-0.03)	-0.04 (-0.02)	-0.04 (-0.02)	-0.01 (-0.01)
$C_5(\mu_1)$	$ \begin{array}{c} 0.01 \\ (0.01) \end{array} $	$0.01 \\ (0.01)$	$0.01 \\ (0.01)$	$0.01 \\ (0.01)$	$0.01 \\ (0.01)$	$0.00 \\ (0.00)$
$C_6(\mu_1)$	-0.08 (-0.05)	-0.05 (-0.03)	-0.05 (-0.03)	-0.04 (-0.03)	-0.04 (-0.02)	-0.01 (-0.01)

Table 4.3: Comparison of the Wilson coefficients at NLO-QCD (LO-QCD), for different values of $\mu_1 = m_c$.

We see that the Wilson coefficients C_j , with $j = 3, \ldots, 6$, are very small and additionally strongly CKM suppressed because of the factor $\lambda_b \ll \lambda_{q_1q_2}$. For these reasons, in the following, the contribution due to the penguin operators in Eq. (4.11) is neglected. However, the most general effective Hamiltonian describing all possible c-quark decays is a sum of non-leptonic, semileptonic as well as radiative contributions, namely

$$\mathcal{H}_{\rm eff} = \mathcal{H}_{\rm eff}^{\rm NL} + \mathcal{H}_{\rm eff}^{\rm SL} + \mathcal{H}_{\rm eff}^{\rm rare} \,, \tag{4.14}$$

here, $\mathcal{H}_{\text{eff}}^{\text{NL}}$ is given in Eq. (4.11),

$$\mathcal{H}_{\text{eff}}^{\text{SL}} = \frac{G_F}{\sqrt{2}} \sum_{q=d,s} \sum_{\ell=e,\mu} V_{cq}^* Q^{q\ell} + \text{h.c.}, \qquad (4.15)$$

where we have introduced the semileptonic operator

$$Q^{q\ell} = (\bar{q}\gamma^{\mu}(1-\gamma_5)c) (\bar{\nu}_{\ell}\gamma_{\mu}(1-\gamma_5)\ell) , \qquad \ell = e, \mu , \qquad (4.16)$$

while $\mathcal{H}_{\text{eff}}^{\text{rare}}$ describes decays like $D \to \pi \ell^+ \ell^-$, whose branching fraction is much smaller than those corresponding to the tree-level transitions. Hence, in the following, we also neglect the presence of rare decays and omit to specify further the form of $\mathcal{H}_{\text{eff}}^{\text{rare}}$. Starting from the expression of the effective Hamiltonian in Eq. (4.14), the total decay width of the heavy charm mesons D^0, D^+, D_s^+ , can be computed according to Eq. (1.144), where now we need to set Q = c. The structure of the HQE is schematically given in Eq. (1.164), and its diagrammatic representation can be visualised as in Figure 1.7. The lowest dimensional contributions, namely those due to two-quark operators up dimension-six, and to four-quark operators up dimensionseven, have been discussed in detail in Chapter 2 and Chapter 3. Despite having considered explicitly the case of the B meson, almost all the expressions obtained, can be used also in the charm sector, taking into account the proper replacements i.e. $b \rightarrow c, c \rightarrow s$, etc.. However, we cannot directly use the coefficients of the Darwin operator listed in Eqs. (3.83)-(3.92), since by setting $\rho = m_s^2/m_c^2$, it is straightforward to see that some of the functions diverge in the limit $m_s \rightarrow 0$. The presence of IR divergences reflects the fact that now there are further contributions due to mixing of four-quark operators with external s-quarks and the Darwin operator, that must be included, whereas for the b-system, the corresponding operators with external *c*-quarks did not. This point will be discussed further later on, and we refer to Ref. [169] for more details.

Following Ref. [140], we try to analyse each of the contributions that enter the HQE of a D meson, in order to identify the presence of possible cancellations which might affect the charm system. We start from the leading order term Γ_3 , cf. Eq. (1.164), which, including also NLO-QCD corrections to the short distance coefficients, can be schematically written as

$$\Gamma_3 = \Gamma_0 c_3 = \Gamma_0 \left[3 C_1^2 \mathcal{C}_{3,11} + 2 C_1 C_2 \mathcal{C}_{3,12} + 3 C_2^2 \mathcal{C}_{3,22} + \mathcal{C}_{3,\text{SL}} \right], \qquad (4.17)$$

where a summation over all modes is implied and we stress that now Γ_0 is defined slightly differently compared to Eq. (3.24), i.e.

$$\Gamma_0 = \frac{G_F^2 m_c^5}{192\pi^3} |V_{cs}|^2 \,. \tag{4.18}$$

At LO-QCD, the three non-leptonic coefficients $C_{3,11}$, $C_{3,12}$, and $C_{3,22}$, for each of the $c \to q_1 \bar{q}_2 u$ modes, reduce to $C_0^{(q_1 \bar{q}_2 u)}$, computed in Chapter 3, cf. Eq. (3.23)⁴. The expressions for the QCD corrections to the non-leptonic coefficients $C_{3,11}$ and $C_{3,22}$, as well as to $C_{3,SL}$, are obtained from Ref. [111]. In the latter, the computation has been performed for three arbitrary massive final states of the decaying quark, hence their results can be easily applied to all *c*-quark decay modes, by taking the

⁴Up to the CKM factor $\lambda_{q_1q_2}^2/|V_{cs}|^2$.

	$\Gamma_3^{\rm LO} \ [{\rm ps}^{-1}]$	$\Gamma_3^{\rm NLO} \ [{\rm ps}^{-1}]$
$m_c^{\text{Pole}} = 1.48 \text{ GeV}$	$1.45^{+0.17}_{-0.14}$	$1.52_{-0.16}^{+0.20}$
$\overline{m}_c(\overline{m}_c) = 1.27 \text{ GeV}$	$0.69^{+0.06}_{-0.09}$	$1.32_{-0.03}^{+0.06}$
$m_c^{\rm kin}(0.5{\rm GeV}) = 1.363{\rm GeV}$	$0.97^{+0.10}_{-0.11}$	$1.47_{-0.30}^{+0.27}$
$m_c^{1S} = 1.548 \text{ GeV}$	$1.80^{+0.24}_{-0.16}$	$2.12_{-0.30}^{+0.51}$

Table 4.4: Numerical values of Γ_3 and LO- and NLO-QCD, using different schemes for the *c*-quark mass. The uncertainties are obtained by varying the renormalisation scale μ_1 between 1 GeV and 3 GeV.

appropriate mass limits. For the coefficient $C_{3,12}$, we use the results of Ref. [114], respectively for the $c \to s\bar{d}u$, $c \to d\bar{s}u$ and $c \to d\bar{d}u$ decay channels, and those of Ref. [118] in the case of two massive final states, e.g. $c \to s\bar{s}u$. To compare the size of c_3 between the *b*- and *c*-system, it is interesting to consider the effect of the NLO corrections in the case of dominant CKM non-leptonic and semileptonic modes ⁵, neglecting for simplicity the mass of the final state particles. The result was determined in 1991 in Ref. [112], and reads ⁶

$$c_{3}^{\text{NLO}} - c_{3}^{\text{LO}} = 8 \frac{\alpha_{s}}{4\pi} \left[\underbrace{\left(\frac{25}{4} - \pi^{2}\right)}_{<0} \underbrace{+ (C_{1}^{2} + C_{2}^{2}) \left(\frac{31}{4} - \pi^{2}\right)}_{<0} \underbrace{-\frac{2}{3} C_{1} C_{2} \left(\frac{7}{4} + \pi^{2}\right)}_{\geqslant 0} \right]_{\geqslant 0}.$$

$$(4.19)$$

The first term on the r.h.s. of Eq. (4.19), corresponds to the semileptonic mode while the remaining two terms to the non-leptonic one. For the *b*-quark decay, the NLO corrections are negative, while for the charm system, the third term can dominate over the second one and lead to a positive correction to c_3 . Moreover, there is a sizeable enhancement of the QCD corrections for non-leptonic *b*-quark decays due to finite charm quark mass effects [113–115, 118], whereas the corresponding increase for charm is much less pronounced since $m_c^2/m_b^2 \approx 0.1 \gg m_s^2/m_c^2 \approx 0.005$. A comparison of Γ_3 , both at LO- and NLO-QCD, for different *c*-quark mass schemes is shown in Table 4.4. The range of values between 1.3 ps^{-1} to 2.7 ps^{-1} for the free charm-quark decay at NLO-QCD, is in good agreement with the experimental determinations in Table 4.1, and we find that the effect of a non-vanishing strange quark mass leads to small corrections (< 5%). Interestingly, the NLO-QCD result is affected by strong cancellations. We in fact observe a suppression of the non-

⁵For example $b \to c\bar{u}d$, $b \to c\ell\nu_{\ell}$ and $c \to s\bar{d}u$, $c \to s\ell\nu_{\ell}$ transitions.

 $^{^6\}mathrm{Note}$ that the factor $|V_{ud}|^2\approx 1$ has been omitted for simplicity.

leptonic contribution because of the opposite sign between the NLO corrections to the $\Delta C = 1$ operators and to their Wilson coefficients. Furthermore, a cancellation is present between the semileptonic and the non-leptonic modes. In the $\overline{\text{MS}}$ scheme an additional NLO contribution arises from the conversion factor of m_c^5 , which is the origin of the large shift between the LO and the NLO value. This is explicitly indicated in the following two equations

(Pole)
$$\Gamma_3 = \Gamma_3^{\text{LO}} \left[1 + \left(\underbrace{\overbrace{1.84}^{\text{NL}} - \underbrace{0.74}_{\text{WC}} - \overbrace{0.67}^{\text{SL}} \right) \frac{\alpha_s}{\pi} + \mathcal{O} \left(\frac{\alpha_s}{\pi} \right)^2 \right], \quad (4.20)$$

$$(\overline{\mathrm{MS}}) \qquad \Gamma_{3} = \Gamma_{3}^{\mathrm{LO}} \left[1 + \left(\underbrace{\underbrace{2.10}_{\mathrm{oper.}} - \underbrace{0.70}_{\mathrm{WC}} - \underbrace{0.71}_{\mathrm{WC}} + \underbrace{6.66}_{\mathrm{6.66}} \right) \frac{\alpha_{s}}{\pi} + \mathcal{O} \left(\frac{\alpha_{s}}{\pi} \right)^{2} \right].$$

$$(4.21)$$

To obtain a first indication of the behaviour of the QCD series for Γ_3 at higher orders, the authors of Ref. [140] have compared the results for the NNLO- [127] and NNNLO-[26] QCD corrections to the semi leptonic *b*-quark decay and the preliminary NNLO-QCD corrections for the non-leptonic *b*-quark decay [129], concluding that higher order corrections seem to be crucial for a reliable determination of Γ_3 ⁷.

At order $1/m_c^2$ in the HQE, Eq. (1.164), we find the contribution of the kinetic and the chromo-magnetic operators, defined in Eqs. (1.154), (1.155), and respectively parametrised by the non perturbative input μ_{π}^2 and μ_G^2 , see Eq. (1.159). At this order and at LO-QCD, we can schematically write

$$\Gamma_5 \frac{\langle \mathcal{O}_5 \rangle}{m_c^2} = \Gamma_0 \left[c_{\mu_\pi} \frac{\mu_\pi^2}{m_c^2} + c_G \frac{\mu_G^2}{m_c^2} \right], \qquad (4.22)$$

where now, compared to the corresponding ones introduced in Eq. (3.23), the short distance coefficients $c_{\mu\pi}$, c_G , contain also the contribution due to the semileptonic modes as well as the dependence on the CKM factor $\lambda_{q_1q_2}/|V_{cs}|^2$, due to the different definition of Γ_0 . Their expressions can be then obtained from those of $\mathcal{C}_{G,mn}^{(q_1\bar{q}_2u)}$ and $\mathcal{C}_0^{(q_1\bar{q}_2u)}$ listed in Appendix D. Note that again a summation over all modes is implied. Specifically, we can decompose c_G as

$$c_G = 3 C_1^2 c_{G,11} + 2 C_1 C_2 c_{G,12} + 3 C_2^2 c_{G,22} + c_{G,SL}, \qquad (4.23)$$

 $^{^{7}}$ Note that the results presented in Ref. [129] are not complete and hence cannot be used for phenomenological applications.



Figure 4.1: Scale dependence of the coefficient of the chromo-magnetic operator c_G .

which leads to the following expression if we neglect the strange and muon masses and consider only the dominant CKM modes, i.e.

$$c_G \approx -|V_{ud}|^2 \left[\frac{9}{2} \left(C_1^2 + C_2^2 \right) + 19 C_1 C_2 \right] - 3.$$
 (4.24)

Because of the large coefficient in front of C_1C_2 and of its negative value, Eq. (4.24) can be affected by cancellations. This can be visualised in Figure 4.1, in which we show the dependence of the function c_G in Eq. (4.23), on the renormalisation scale μ_1 , for both LO- and NLO-QCD, $\Delta C = 1$ Wilson coefficients. Note that the latter case in Figure 4.1, is indicated in quotation marks since it does not represent the complete NLO result, as corrections for non-leptonic modes are still missing and their effect could significantly reduce the strong scale dependence. In particular, from Figure 4.1, we see that a change of sign occurs in the region between 1 and 2 GeV, leading to a large uncertainty due to scale variation. The numerical determinations of μ_{π}^2 and μ_G^2 are presented at the end of this section.

We now turn to analyse the contribution of the Darwin operator, which arises at order $1/m_c^3$ in Eq. (1.164) and which has been discussed in detail in Chapter 3. This can be compactly written as

$$\Gamma_6 \frac{\langle \mathcal{O}_6 \rangle}{m_c^3} = \Gamma_0 \, c_{\rho_D} \frac{\rho_D^3}{m_c^3} \,, \tag{4.25}$$

where again a summation over all modes is implied and the coefficient c_{ρ_D} includes the effect of non-leptonic and semileptonic channels, namely

$$c_{\rho_D} = 3 C_1^2 \mathcal{C}_{\rho_D,11} + 2 C_1 C_2 \mathcal{C}_{\rho_D,12} + 3 C_2^2 \mathcal{C}_{\rho_D,22} + \mathcal{C}_{\rho_D,SL}.$$
(4.26)

As already mentioned, the expressions for the non-leptonic coefficients obtained in Eqs. (3.83)-(3.92), cannot be directly applied to the charm sector, since by naively replacing $m_b \to m_c$ and $m_c \to m_s$, some of the functions would develop infrared divergences in the limit $m_s \rightarrow 0$, whereas in the *b*-system the corresponding coefficients were finite functions of $\rho = m_c^2/m_b^2$. In fact, while we can assume $m_b \sim m_c \gg \Lambda_{\rm QCD}$, and neglect the effect of four-quark operators with external c-quarks in matrix elements between B-meson states 8 , see e.g. Ref. [169], in the charm sector, it is $m_c \gg m_s \sim \Lambda_{\rm QCD}$, and there are further contributions due to the mixing of fourquark operators with external s-quarks which must be additionally included. Specifically, this leads to a modification of the coefficients proportional to C_1^2 and C_1C_2 . Using the same procedure as discussed in Ref. [137], the coefficients of the Darwin operator required for the study of *D*-meson decays have been computed in Ref. [140], and the analytical expressions, including the full s-quark mass dependence, however finite in the limit $m_s \rightarrow 0$, are listed in Appendix F, for all non-leptonic modes. The results for $\mathcal{C}_{\rho_D,SL}$, can then be obtained by setting, $N_c = 1, C_1 = 1, C_2 = 0$ and $m_s \to m_\mu$ in the case of $c \to s\mu^+\nu_\mu$ decay. By neglecting the strange and muon masses and by considering only the dominant CKM modes, we have

$$c_{\rho_D} \approx |V_{ud}|^2 \left(18 C_1^2 - \frac{68}{3} C_1 C_2 + 18 C_2^2 \right) + 12.$$
 (4.27)

It is interesting to note that in this combination all terms have the same sign and no cancellations arise. In Figure 4.2 we show the dependence on the renormalisation scale μ_1 of the function c_{ρ_D} in Eq. (4.26), where the quotation marks in the NLO-QCD result reflect again the fact that only corrections due to the $\Delta C = 1$ Wilson coefficients have been included, since also in this case a complete determination of the NLO corrections is still missing. Estimates for the matrix element of the Darwin operator will be presented at the end of this section.

In order to discuss the contribution of four-quark operators at order $1/m_c^3$ in Eq. (1.164), see Section 2.2, we introduce the following basis, in complete analogy

 $^{^8 \}rm We$ recall that we do not consider the case of the B_c meson.



Figure 4.2: Scale dependence of the coefficient of the Darwin operator c_{ρ_D} .

to Eqs. (2.100)-(2.103)⁹, namely

$$O_1^q = \left(\bar{c}\,\gamma_\mu(1-\gamma_5)q\right)\left(\bar{q}\,\gamma^\mu(1-\gamma_5)c\right),\tag{4.28}$$

$$O_2^q = (\bar{c}(1-\gamma_5)q) (\bar{q}(1+\gamma_5)c), \qquad (4.29)$$

$$O_3^q = \left(\bar{c}\,\gamma_\mu (1-\gamma_5)t^a q\right) \left(\bar{q}\,\gamma^\mu (1-\gamma_5)t^a c\right), \tag{4.30}$$

$$O_4^q = \left(\bar{c}(1-\gamma_5)t^a q\right) \left(\bar{q}(1+\gamma_5)t^a c\right),$$
(4.31)

where a summation over colour indices is implied and we have replaced c_v with c, cf. Eq. (1.145). The parametrisation of the matrix element of the operators in Eqs. (4.28)-(4.31) is given in Appendix G. However, by evaluating them in the framework of the HQET, the dependence on the charm quark mass can be further extracted from the *c*-quark field and meson state cf. Section 2.2.2, and in this case, the corresponding dimension-six operators read

$$\mathcal{O}_{1}^{q} = (\bar{h}_{v} \gamma_{\mu} (1 - \gamma_{5}) q) \left(\bar{q} \gamma^{\mu} (1 - \gamma_{5}) h_{v} \right), \qquad (4.32)$$

$$\mathcal{O}_{2}^{q} = (\bar{h}_{v}(1-\gamma_{5})q) \left(\bar{q}(1+\gamma_{5})h_{v}\right), \qquad (4.33)$$

$$\mathcal{O}_{3}^{q} = (\bar{h}_{v} \gamma_{\mu} (1 - \gamma_{5}) t^{a} q) \left(\bar{q} \gamma^{\mu} (1 - \gamma_{5}) t^{a} h_{v} \right), \qquad (4.34)$$

$$\mathcal{O}_{4}^{q} = (\bar{h}_{v}(1-\gamma_{5})t^{a}q)\left(\bar{q}(1+\gamma_{5})t^{a}h_{v}\right), \qquad (4.35)$$

⁹Note however that now, we do not use the tilde to denote the colour-octet operators.

where h_v denotes the effective heavy quark field, see Section 1.2. The parametrisation in HQET of the matrix element of the operators in Eqs. (4.32)-(4.35), see also Appendix G, can be written as

$$\langle D_q | \mathcal{O}_i^q | D_q \rangle = F^2(m_c) m_{D_q} \left(\tilde{B}_i^q + \tilde{\delta}_i^{qq} \right) = f_{D_q}^2 m_{D_q}^2 \left(1 + \frac{4}{3} \frac{\alpha_s(m_c)}{\pi} \right) \left(\tilde{B}_i^q + \tilde{\delta}_i^{qq} \right) ,$$

$$(4.36)$$

$$\langle D_q | \mathcal{O}_i^{q'} | D_q \rangle = F^2(m_c) \, m_{D_q} \tilde{\delta}_i^{qq'} = f_{D_q}^2 m_{D_q}^2 \left(1 + \frac{4}{3} \frac{\alpha_s(m_c)}{\pi} \right) \tilde{\delta}_i^{qq'}, \quad q \neq q', \quad (4.37)$$

where q, q' = u, d, s, \tilde{B}_i^q are the Bag parameters computed in HQET, while $F(\mu)$ and f_{D_q} correspond respectively to HQET and QCD decay constants, defined, by ¹⁰

$$\langle 0|\bar{q}\gamma^{\mu}\gamma_5 c|D_q(v)\rangle_{\rm QCD} = if_{D_q} p^{\mu}, \qquad (4.38)$$

with $p^{\mu} = m_D v^{\mu}$, and

$$\langle 0|\bar{q}\gamma^{\mu}\gamma_5 h_v|D_q(v)\rangle_{\rm HQET} = i F(\mu) \sqrt{m_{D_q}} v^{\mu}.$$
(4.39)

The relation between f_D and $F(\mu)$ up to QCD and power corrections, can be found e.g. in Refs. [244, 245]. At the scale $\mu = m_c$, it reads

$$f_D = \frac{F(m_c)}{\sqrt{m_D}} \left(1 - \frac{2}{3} \frac{\alpha_s(m_c)}{\pi} + \frac{G_1(m_c)}{m_c} + 6 \frac{G_2(m_c)}{m_c} - \frac{1}{2} \frac{\bar{\Lambda}}{m_c} \right), \tag{4.40}$$

where $\bar{\Lambda} = m_D - m_c$, and the parameters G_1 and G_2 characterise matrix elements of non-local operators. Note that in Eqs. (4.36), (4.37), by expressing the HQET decay constant in terms of the one defined in QCD, we have included only corrections due to α_s , which become part of the NLO-QCD contribution at dimension-six. In fact, as we will discuss, the power corrections can be absorbed in the contribution of some of the dimension-seven operators appearing in HQET.

In vacuum insertion approximation (VIA), the Bag parameters of the coloursinglet operators are equal to one, i.e. $\tilde{B}_{1,2}^q = 1$, and those of the colour-octet

 $^{^{10}{\}rm The}$ subscript 'QCD' or 'HQET' on the states is usually omitted, however for clarity it is specified in the definition of the decay constant.



Figure 4.3: Diagram describing the eye-contractions.

operators vanish, i.e. $\tilde{B}_{3,4}^q = 0$. Note that we assume isospin symmetry, so that $\tilde{B}_i^u = \tilde{B}_i^d$. The parameters $\tilde{\delta}_i^{qq}$, $\tilde{\delta}_i^{qq'}$, in (4.37), describe subleading effects in the non perturbative matrix elements, compared to the dominant \tilde{B}_i , and correspond to the so called eye-contractions, shown in Figure 4.3. In VIA, the contribution of all eye-contractions vanish i.e. $\tilde{\delta}_i^{qq} = \tilde{\delta}_i^{qq'} = 0$. However, beyond vacuum insertion approximation, the matrix element of four-quark operators with external q' quarks, differ from zero even when the spectator quark q in the D_q meson does not coincide with q', as it is indicated in Eq. (4.37) and in Figure 4.3. Again, due to isospin symmetry, we assume $\tilde{\delta}_i^{uq'} = \tilde{\delta}_i^{dq'}$ and $\tilde{\delta}_i^{qu} = \tilde{\delta}_i^{qd}$. The Bag parameters \tilde{B}_i and $\tilde{\delta}_i^{qq'}$ have been computed using HQET sum rules, specifically, the formed were obtained for the $D^{+,0}$ mesons in Ref. [246], while corrections due to the strange quark mass as well as the contribution of the eye-contractions, have been determined for the first time in Ref. [247]. The numerical values of the HQET Bag parameters are listed in Table 4.10.

By considering only the dominant CKM modes and by neglecting the effect of the eye-contractions, at LO-QCD and at dimension-six, the contribution of four-quark operators to the decay rate of the D^0, D^+ and D_s^+ mesons, cf. Eq. (1.164), reads

$$16\pi^{2}\tilde{\Gamma}_{6}^{D^{0}}\frac{\langle\tilde{O}_{6}\rangle^{D^{0}}}{m_{c}^{3}} = \Gamma_{0}|V_{ud}^{*}|^{2} 16\pi^{2}\frac{M_{D^{0}}f_{D^{0}}^{2}}{m_{c}^{3}}(1-x_{s})^{2} \left\{ C_{WE}^{S} \left[(\tilde{B}_{2}^{u}-\tilde{B}_{1}^{u}) \right] + x_{s} \left(2\tilde{B}_{2}^{u}-\frac{\tilde{B}_{1}^{u}}{2} \right) + C_{WE}^{O} \left[(\tilde{B}_{4}^{u}-\tilde{B}_{3}^{u}) + x_{s} \left(2\tilde{B}_{4}^{u}-\frac{\tilde{B}_{3}^{u}}{2} \right) \right] \right\},$$

$$(4.41)$$

which corresponds to the WE topology,

$$16\pi^{2}\tilde{\Gamma}_{6}^{D^{+}}\frac{\langle\tilde{O}_{6}\rangle^{D^{+}}}{m_{c}^{3}} = \Gamma_{0}|V_{ud}^{*}|^{2}16\pi^{2}\frac{M_{D^{+}}f_{D^{+}}^{2}}{m_{c}^{3}}(1-x_{s})^{2}\left\{C_{\mathrm{PI}}^{S}\tilde{B}_{1}^{d}+C_{\mathrm{PI}}^{O}\tilde{B}_{3}^{d}\right\}, \quad (4.42)$$

describing the PI contribution, and

$$16\pi^{2}\tilde{\Gamma}_{6}^{D_{s}^{+}}\frac{\langle\tilde{O}_{6}\rangle^{D_{s}^{+}}}{m_{c}^{3}} = \Gamma_{0}|V_{ud}^{*}|^{2}16\pi^{2}\frac{M_{D_{s}^{+}}f_{D_{s}^{+}}^{2}}{m_{c}^{3}}\left\{\left(C_{\mathrm{WA}}^{S}+\frac{2}{|V_{ud}^{*}|^{2}}\right)\left(\tilde{B}_{2}^{s}-\tilde{B}_{1}^{s}\right)\right.\right.$$
$$\left.+C_{\mathrm{WA}}^{O}\left(\tilde{B}_{4}^{s}-\tilde{B}_{3}^{s}\right)\right\},\tag{4.43}$$

due to WA. Here $x_s = m_s^2/m_c^2$, and we have introduced the following notation for the combinations of Wilson coefficients, cf. Eqs. (2.81), (2.89), and (2.95), namely

$$C_{\rm WE}^S = \frac{1}{3}C_1^2 + 2C_1C_2 + 3C_2^2, \qquad C_{\rm WE}^O = 2C_1^2, \qquad (4.44)$$

$$C_{\rm PI}^S = C_1^2 + 6 C_1 C_2 + C_2^2, \qquad C_{\rm PI}^O = 6 \left(C_1^2 + C_2^2 \right), \qquad (4.45)$$

$$C_{\rm WA}^S = 3 C_1^2 + 2 C_1 C_2 + \frac{1}{3} C_2^2, \qquad C_{\rm WA}^O = 2 C_2^2, \qquad (4.46)$$

where the superscript "S" and "O" refers to coefficient in front of the colour-singlet and of colour-octet Bag parameters, respectively. Note that in Eq. (4.43), the contribution due to the muon mass in the semileptonic decay $c \rightarrow s\mu^+\nu_{\mu}$, has been neglected. The expressions in Eqs. (4.41)-(4.43) lead to some interesting numerical effects. First, in the charm system, one expects that the contribution due to the spectator quark is of similar size compared to the leading term Γ_3 in the HQE, unless some additional cancellations are present. Using the pole mass $m_c^{\text{Pole}} = 1.48 \text{ GeV}$ and Lattice QCD values for the decay constants [248], roughly yields

$$16\pi^2 \frac{M_{D^0} f_{D^0}^2}{m_c^3} = 4.1 \approx \mathcal{O}(c_3), \qquad 16\pi^2 \frac{M_{D_s^+} f_{D_s^+}^2}{m_c^3} = 6.0 \approx \mathcal{O}(c_3). \tag{4.47}$$

This result has led the authors of Ref. [249] to propose a different way to rearrange the HQE series in the charm sector. However, to investigate further the size of four-quark contributions at dimension-six, we consider the combinations of Wilson coefficients that appear in Eqs. (4.41) - (4.43). A comparison of these coefficients at LO- and NLO-QCD, for different values of the renormalisation scale μ_1 is shown in Table 4.5. The first observation is that C_{WE}^S is strongly suppressed. Moreover, in

$\mu_1 \; [\text{GeV}]$	1	1.206	1.27	1.48	1.67	3
$C_{\rm WE}^{\rm S}({\rm LO})$	0.09	0.04	0.03	0.01	0.00	0.01
$C_{\rm WE}^{\rm S}({\rm NLO})$	-0.03	-0.03	-0.03	-0.02	-0.02	0.04
$C_{\rm WE}^{\rm O}({\rm LO})$	3.57	3.30	3.24	3.08	2.98	2.63
$C_{\rm WE}^{\rm O}({\rm NLO})$	3.11	2.93	2.89	2.77	2.70	2.44
$C_{\rm PI}^{\rm S}({\rm LO})$	-2.80	-2.25	-2.12	-1.79	-1.57	-0.79
$C_{\rm PI}^{\rm S}({\rm NLO})$	-1.74	-1.36	-1.28	-1.04	-0.88	-0.27
$C_{\rm PI}^{\rm O}({\rm LO})$	13.0	11.7	11.4	10.6	10.1	8.50
$C_{\rm PI}^{\rm O}({\rm NLO})$	10.6	9.73	9.55	9.05	8.72	7.60
$C_{\rm WA}^{\rm S}({\rm LO})$	3.82	3.65	3.61	3.51	3.45	3.24
$C_{\mathrm{WA}}^{\mathrm{S}}(\mathrm{NLO})$	3.57	3.45	3.42	3.35	3.31	3.16
$C_{\rm WA}^{\rm O}({\rm LO})$	0.77	0.59	0.55	0.46	0.40	0.21
$C_{\rm WA}^{\rm O}(\rm NLO)$	0.41	0.32	0.30	0.24	0.21	0.10

Table 4.5: Comparison of the combinations $C_{\text{WE,PI,WA}}^{S,O}$, respectively at LO- and NLO-QCD, for different values of the renormalisation scale μ_1 .

Eq. (4.44) the Bag parameters of the colour-singlet operators exactly cancel in VIA. On the other side, the combination of Wilson coefficients in front of the colour-octet operators is not suppressed for weak exchange, indicating that both colour structures might be equally important in this case. For Pauli interference, the combinations of Wilson coefficients multiplying the colour-singlet operators are significantly enhanced compared to those in WE, the same holds for the colour-octet operators. Note that $C_{\rm PI}^O$ and $C_{\rm PI}^S$ get large modifications, and even a change of sign, compared to the case $C_1 = 1$ and $C_2 = 0$ revealing the importance of gluon radiative corrections. Moreover $C_{\rm PI}^{O}$ is enhanced compared to $C_{\rm PI}^{S}$, again indicating that both colour structures might be equally important for PI. In the case of weak annihilation, C_{WA}^S is large. On the other hand, the Bag parameters of the colour-singlet operators exactly cancel in VIA. The above arguments show that by neglecting the effect of the colour-octet operators in VIA, one might be led to misleading conclusions, and therefore an accurate determination of the deviation of the Bag parameters from their VIA values, using non-perturbative methods like HQET sum rules or lattice simulations, is necessary. Finally, by including all CKM modes as well as NLO-QCD corrections, the contribution of four-quark operators to the total decay width at order $1/m_c^3$, schematically reads

scheme	D^0	D^+	D_s^+					
VIA								
Pole	$\underbrace{-0.06}_{\text{NLO}} = \underbrace{0.00}_{\text{LO}} \underbrace{-0.06}_{\text{\Delta NLO}}$	$\underbrace{-11.3}_{\text{NLO}} = \underbrace{-7.18}_{\text{LO}} \underbrace{-4.13}_{\text{\DeltaNLO}}$	$\underbrace{-0.85}_{\text{NLO}} = \underbrace{-0.51}_{\text{LO}} \underbrace{-0.34}_{\text{\DeltaNLO}}$					
$\overline{\mathrm{MS}}$	$\underbrace{-0.09}_{\text{NLO}} = \underbrace{0.00}_{\text{LO}} \underbrace{-0.09}_{\text{\Delta NLO}}$	$\underbrace{-22.9}_{\text{NLO}} = \underbrace{-11.3}_{\text{LO}} \underbrace{-11.5}_{\text{\DeltaNLO}}$	$\underbrace{-1.66}_{\text{NLO}} = \underbrace{-0.77}_{\text{LO}} \underbrace{-0.89}_{\text{\DeltaNLO}}$					
Kinetic	$\underbrace{-0.08}_{\text{NLO}} = \underbrace{0.00}_{\text{LO}} \underbrace{-0.08}_{\Delta\text{NLO}}$	$\underbrace{-16.3}_{\text{NLO}} = \underbrace{-9.18}_{\text{LO}} \underbrace{-7.14}_{\text{\Delta NLO}}$	$\underbrace{-1.21}_{\text{NLO}} = \underbrace{-0.64}_{\text{LO}} \underbrace{-0.57}_{\Delta\text{NLO}}$					
1S	$\underbrace{-0.05}_{\text{NLO}} = \underbrace{0.00}_{\text{LO}} \underbrace{-0.05}_{\text{\Delta NLO}}$	$\underbrace{-10.1}_{\text{NLO}} = \underbrace{-6.27}_{\text{LO}} \underbrace{-3.82}_{\text{\DeltaNLO}}$	$\underbrace{-0.76}_{\text{NLO}} = \underbrace{-0.45}_{\text{LO}} \underbrace{-0.31}_{\text{\DeltaNLO}}$					
	HQET SR							
Pole	$\underbrace{0.06}_{\text{NLO}} = \underbrace{0.10}_{\text{LO}} \underbrace{-0.04}_{\Delta\text{NLO}}$	$\underbrace{-12.3}_{\text{NLO}} = \underbrace{-7.97}_{\text{LO}} \underbrace{-4.37}_{\Delta\text{NLO}}$	$\underbrace{-0.93}_{\text{NLO}} = \underbrace{-0.69}_{\text{LO}} \underbrace{-0.24}_{\Delta\text{NLO}}$					
MS	$\underbrace{0.23}_{\text{NLO}} = \underbrace{0.17}_{\text{LO}} \underbrace{+0.06}_{\Delta\text{NLO}}$	$\underbrace{-24.9}_{\text{NLO}} = \underbrace{-12.6}_{\text{LO}} \underbrace{-12.3}_{\Delta\text{NLO}}$	$\underbrace{-1.78}_{\text{NLO}} = \underbrace{-1.06}_{\text{LO}} \underbrace{-0.72}_{\Delta\text{NLO}}$					
Kinetic	$\underbrace{0.12}_{\text{NLO}} = \underbrace{0.13}_{\text{LO}} \underbrace{-0.01}_{\text{\Delta NLO}}$	$\underbrace{-17.8}_{\text{NLO}} = \underbrace{-10.2}_{\text{LO}} \underbrace{-7.61}_{\text{\Delta NLO}}$	$\underbrace{-1.31}_{\text{NLO}} = \underbrace{-0.87}_{\text{LO}} \underbrace{-0.43}_{\text{\DeltaNLO}}$					
1S	$\underbrace{0.06}_{\text{NLO}} = \underbrace{0.09}_{\text{LO}} \underbrace{-0.03}_{\Delta\text{NLO}}$	$\underbrace{-11.0}_{\text{NLO}} = \underbrace{-6.97}_{\text{LO}} \underbrace{-4.05}_{\text{\Delta NLO}}$	$\underbrace{-0.83}_{\text{NLO}} = \underbrace{-0.60}_{\text{LO}} \underbrace{-0.23}_{\text{\DeltaNLO}}$					

Table 4.6	: Dimensio	on-six co	ntributions	to the	D-mesons	decay	width	norma	lised
by Γ_0 and	l split up i	nto LO-Q	QCD and N	ILO-QC	D correctio	ons wit	hin difl	ferent	mass
schemes a	nd using b	oth VIA	and HQE	Γ SR val	lues for the	Bag p	aramet	cers.	

$$\tilde{\Gamma}_{6}^{D_{q}} \frac{\langle \tilde{\mathcal{O}}_{6} \rangle^{D_{q}}}{m_{c}^{3}} = \frac{\Gamma_{0}}{|V_{cs}|^{2}} \sum_{q_{1},q_{2}=d,s} |\lambda_{q_{1}q_{2}}|^{2} \sum_{i=1}^{4} \left[A_{i,q_{1}q_{2}}^{\text{WE}} \frac{\langle D_{q} | \mathcal{O}_{i}^{u} | D_{q} \rangle}{m_{c}^{3}} + A_{i,q_{1}q_{2}}^{\text{PI}} \frac{\langle D_{q} | \mathcal{O}_{i}^{q_{2}} | D_{q} \rangle}{m_{c}^{3}} \right] \\ + A_{i,q_{1}q_{2}}^{\text{WA}} \frac{\langle D_{q} | \mathcal{O}_{i}^{q_{1}} | D_{q} \rangle}{m_{c}^{3}} \right] + \frac{\Gamma_{0}}{|V_{cs}|^{2}} \sum_{q_{1}=d,s} |V_{cq_{1}}|^{2} \sum_{\ell=e,\mu} \sum_{i=1}^{4} \left[A_{i,q_{1}\ell}^{\text{WA}} \frac{\langle D_{q} | \mathcal{O}_{i}^{q_{1}} | D_{q} \rangle}{m_{c}^{3}} \right],$$

$$(4.48)$$

where the matrix elements of the four-quark operators are given in Eqs. (4.36), (4.37), and the short-distance coefficients for the WE, PI and WA topologies are denoted by $A_{i,q_1q_2}^{\text{WE}}$, $A_{i,q_1q_2}^{\text{PI}}$ and $A_{i,q_1q_2}^{\text{WA}}$, $A_{i,q_1\ell}^{\text{WA}}$, respectively. Their expressions at LO-QCD have been derived in Section 2.2.1, while NLO corrections to $A_{i,q_1q_2}^{\text{WE}}$ and $A_{i,q_1q_2}^{\text{PI}}$ have been computed in Ref. [142]. The corresponding results for $A_{i,q_1q_2}^{\text{WA}}$ can be obtained by using Eq. (2.93), since the Fierz symmetry is respected also at one-loop.

For the semileptonic modes, the coefficients $A_{i,q_1\ell}^{\text{WA}}$ have been determined in Ref. [143]. Note that in our analysis, we treat the contribution of the $\tilde{\delta}_i^{qq'}$ parameters effectively as a NLO effect, therefore their coefficients are included only at LO-QCD. Finally, in Table 4.6, we compare the size of the LO- and NLO-QCD corrections in Eq. (4.48), normalised by Γ_0 , both in VIA and using HQET SR results for the Bag parameters. The NLO-QCD corrections turn out to have an essential numerical effect for the contribution of four-quark at order $1/m_c^3$. In particular, in the case of the D^0 and D_s^+ mesons, they lift the helicity suppression present in the weak exchange and weak annihilation topologies at LO-QCD and in VIA. Note that for the D_s^+ meson, in addition to the CKM dominant WA contribution, there is a correction due to the CKM suppressed, but nevertheless large PI topology. In the case of the D^+ meson the NLO corrections to Pauli interference are very large, 50% - 100% depending on the mass scheme. Already in the B system they were found to be of the order of 30% for the ratio $\tau(B^+)/\tau(B_d)$, in the pole scheme, see e.g. Ref. [141]. We conclude that, neglecting these contributions in the study of charm lifetime, as it has been previously done in Ref. [250], is clearly not justified and the determination of higher order corrections would be highly desirable.

We now consider the contribution of four-quark operators at order $1/m_c^4$ to the HQE in Eq. (1.164), which have been discussed in detail in Section 2.2.2. By expanding $p^{\mu} = p_c^{\mu} \pm p_q^{\mu}$, only in the small momentum of the light spectator quark $p_q \sim \Lambda_{QCD}$, leads to the following basis for the dimension-seven operators ¹¹

$$P_1^q = m_q \left(\bar{c}(1 - \gamma_5)q \right) \left(\bar{q}(1 - \gamma_5)c \right), \qquad (4.49)$$

$$P_2^q = \frac{1}{m_c} (\bar{c} D_\nu \gamma_\mu (1 - \gamma_5) D^\nu q) (\bar{q} \gamma^\mu (1 - \gamma_5) c) , \qquad (4.50)$$

$$P_3^q = \frac{1}{m_c} (\bar{c} D_\nu (1 - \gamma_5) D^\nu q) (\bar{q} (1 + \gamma_5) c) , \qquad (4.51)$$

together with the corresponding \tilde{P}_1^q , \tilde{P}_2^q , \tilde{P}_3^q , containing the generators t^a . Due to the presence in Eqs. (4.50), (4.51), of a covariant derivative acting on the charm quark field, which scales as m_c at this order, there is no immediate power counting for these operators, contrary to those defined in HQET, cf. Eqs. (4.53), (4.54). Moreover, note that this basis differs from the one used in Ref. [144] for the computation of the dimension-seven and dimension-eight contributions. By evaluating the

¹¹Note that the basis used e.g. in Ref. [143], is redundant, since it contains also the additional operator denoted by P_2^q , related to P_1^q by hermitean conjugation, namely $P_2^q = m_q (\bar{c}(1+\gamma_5)q)(\bar{q}(1+\gamma_5)c) = (P_1^q)^{\dagger}$. Leading to the same matrix element, we do not include this operator in our basis.

matrix elements of the dimension-seven four-quark operators using the framework of the HQET, we have to further expand the charm quark momentum, according to $p_c^{\mu} = m_c v^{\mu} + k^{\mu}$, see Section 1.2, as well as to include $1/m_c$ corrections to the effective heavy quark field and to the HQET Lagrangian, cf. Section 2.2.2. In this case, we obtain the following basis, in complete analogy to Eqs. (2.113)-(2.121), namely

$$\mathcal{P}_1^q = m_q \left(\bar{h}_v (1 - \gamma_5) q \right) \left(\bar{q} (1 - \gamma_5) h_v \right), \qquad (4.52)$$

$$\mathcal{P}_{2}^{q} = (\bar{h}_{v}\gamma_{\mu}(1-\gamma_{5})(iv\cdot D)q)(\bar{q}\gamma^{\mu}(1-\gamma_{5})h_{v}), \qquad (4.53)$$

$$\mathcal{P}_{3}^{q} = (\bar{h}_{v}(1-\gamma_{5})(iv \cdot D)q)(\bar{q}(1+\gamma_{5})h_{v}), \qquad (4.54)$$

due to the contribution of the light spectator quark momentum,

$$\mathcal{R}_{1}^{q} = (\bar{h}_{v}\gamma_{\mu}(1-\gamma_{5})q)(\bar{q}\gamma^{\mu}(1-\gamma_{5})(iD)h_{v}), \qquad (4.55)$$

$$\mathcal{R}_{2}^{q} = (\bar{h}_{v}(1-\gamma_{5})q)(\bar{q}(1+\gamma_{5})(i\not\!\!D)h_{v}), \qquad (4.56)$$

due to $1/m_c$ corrections to the effective heavy quark field h_v , and

$$\mathcal{M}_{1}^{q} = i \int d^{4}y \,\mathrm{T}\left[\mathcal{O}_{1}^{q}(0), (\bar{h}_{v}(iD)^{2}h_{v})(y)\right], \qquad (4.57)$$

$$\mathcal{M}_{2}^{q} = i \int d^{4}y \operatorname{T} \left[\mathcal{O}_{1}^{q}(0), \frac{1}{2} g_{s} \left(\bar{h}_{v} \sigma_{\alpha\beta} G^{\alpha\beta} h_{v} \right) (y) \right], \qquad (4.58)$$

$$\mathcal{M}_{3}^{q} = i \int d^{4}y \,\mathrm{T}\left[\mathcal{O}_{2}^{q}(0), (\bar{h}_{v}(iD)^{2}h_{v})(y)\right], \qquad (4.59)$$

$$\mathcal{M}_{4}^{q} = i \int d^{4}y \,\mathrm{T}\left[\mathcal{O}_{2}^{q}(0), \frac{1}{2}g_{s}\left(\bar{h}_{v}\sigma_{\alpha\beta}G^{\alpha\beta}h_{v}\right)(y)\right],\tag{4.60}$$

due to $1/m_c$ corrections to the HQET Lagrangian, which we have explicitly indicated, cf. Eq. (1.69), (1.70). Moreover, the set of operators in Eqs. (4.52)-(4.60), are supplemented by the corresponding colour octet ones. To parametrise the matrix element of the dimension-seven operators in HQET, we use VIA and account for deviations from it by including the corresponding Bag parameters, as it is explicitly shown in Appendix G. However, since for these matrix elements, does not exist a non perturbative evaluation available yet, in our analysis we have to rely only on VIA. It follows that, at LO-QCD the matrix element of the dimension-seven operators listed above, can be expressed in terms of the HQET non perturbative parameters $F(\mu)$, $G_1(\mu)$, $G_2(\mu)$, and $\bar{\Lambda}$, so far determined only with large uncertainties. For this reason, we prefer to use as an input the QCD decay constant f_D , which is computed very precisely using Lattice QCD [248]. In doing so, we obtain that in VIA and at the matching scale $\mu = m_c$, the contribution of the local operators $\mathcal{R}_{1,2}^q$, as well as that of the non-local \mathcal{M}_1^q , \mathcal{M}_2^q , \mathcal{M}_3^q and \mathcal{M}_4^q , can be entirely absorbed in the QCD decay constant f_D , cf. Eq. (4.40), more precisely, in the QCD matrix element of the dimension-six operators in Eqs. (4.28), (4.31), which are proportional to f_D , and the only remaining $1/m_c$ contribution is due to the operators $\mathcal{P}_{1,2,3}^q$, analogously to the QCD case ¹². To make this point more clear, we consider as an example the contribution due to Pauli interference at LO-QCD and up to order $1/m_c^4$, in the case of $c \to s \bar{d}u$ transition, which constitutes the dominant correction to $\Gamma(D^+)$, namely

$$\operatorname{Im} \mathcal{T}^{\mathrm{PI}} = \Gamma_0 |V_{ud}^*|^2 \frac{32\pi^2}{m_c^3} (1 - x_s)^2 \left[C_{\mathrm{PI}}^S \left(\mathcal{O}_1^d + \frac{\mathcal{R}_1^d}{m_c} + \frac{\mathcal{M}_1^d}{m_c} + \frac{\mathcal{M}_2^d}{m_c} + 2\frac{1 + x_s}{1 - x_s} \frac{\mathcal{P}_3^q}{m_c} \right) + (\operatorname{singlet} \to \operatorname{octet}) \right].$$

$$(4.61)$$

By evaluating the matrix element of $\text{Im}\mathcal{T}^{\text{PI}}$ in VIA, the contribution due to the colour-octet operators vanishes. Moreover, using the parametrisation for the matrix elements of the four-quark operators given in Eq. (4.36) and in Appendix G, we obtain in VIA and setting $\mu = m_c$, that

$$\langle \mathcal{O}_1^d + \frac{\mathcal{R}_1^d}{m_c} + \frac{\mathcal{M}_1^d}{m_c} + \frac{\mathcal{M}_2^d}{m_c} \rangle_{\text{HQET}} = F^2(m_c) \, m_{D^+} \left[1 - \frac{\bar{\Lambda}}{m_c} + \frac{2 \, G_1(m_c)}{m_c} + \frac{12 \, G_2(m_c)}{m_c} \right]$$
$$= f_D^2 \, m_{D^+}^2 = \langle O_1^d \rangle_{\text{QCD}},$$
(4.62)

where in the second line we have used the conversion between the QCD and HQET decay constants given in Eq. (4.40). From Eq. (4.62) we see that the contribution of the local operators \mathcal{R}_1^q and non-local operators \mathcal{M}_1^q and \mathcal{M}_2^q , is entirely absorbed by using the QCD decay constant. Note that, by neglecting the effect due to the strange quark mass and using VIA we reproduce the result in Eq. (19) of Ref. [249].

 $^{^{12}\}text{In}$ the matrix element of $\tilde{P}^q_{1,2,3}$ one can replace the HQET decay constant with the QCD one, up to higher order corrections.

The same argument applies also to the remaining topologies i.e. WE and WA. However, it is worth remarking that in VIA and neglecting the strange quark mass, the contribution of WE and WA exactly vanishes at LO-QCD, due to the helicity suppression. This suppression is lifted once the s-quark mass or perturbative gluon corrections are included, and in this case it becomes again manifest that the contribution of \mathcal{R}_i^q , and \mathcal{M}_i^q in HQET, can be completely absorbed in f_D by evaluating the matrix elements in VIA.¹³. We stress that in our numerical analysis, we employ this argument also when using the results of the Bag parameters determined from HQET SR, by neglecting the small deviation from their corresponding VIA values. Note that a detailed analysis of the dimension-seven contributions within the HQET has been performed in Ref. [245] for the case of $B - \overline{B}$ -mixing. Specifically, it was found that in VIA, subleading power corrections due to non-local operators can be entirely absorbed in the definition of the QCD decay constant, and that the residual $1/m_b$ corrections, due to the running of the local dimension-seven operators from the scale m_b to $\mu \sim 1$ GeV, is numerically small (~ 5% for Ref. [245]).¹⁴ Finally, by summing over all the CKM modes, at LO-QCD, the dimension-seven contribution can be presented as

$$16\pi^{2}\tilde{\Gamma}_{7}^{D_{q}}\frac{\langle\tilde{\mathcal{O}}_{7}\rangle^{D_{q}}}{m_{c}^{4}} = \frac{\Gamma_{0}}{|V_{cs}|^{2}}\sum_{q_{1},q_{2}=d,s} \left\{ |\lambda_{q_{1}q_{2}}|^{2}\sum_{i=1}^{3} \left[G_{i,q_{1}q_{2}}^{WE}\frac{\langle D_{q}|\mathcal{P}_{i}^{u}|D_{q}\rangle}{m_{c}^{4}} + G_{i,q_{1}q_{2}}^{PI}\frac{\langle D_{q}|\mathcal{P}_{i}^{q_{2}}|D_{q}\rangle}{m_{c}^{4}} \right] + G_{i,q_{1}q_{2}}^{WA}\frac{\langle D_{q}|\mathcal{P}_{i}^{q_{1}}|D_{q}\rangle}{m_{c}^{4}} + (\text{colour-octet part}) \right\},$$

$$(4.63)$$

and we confirm the results for the short-distance coefficients $G_{i,q_1q_2}^{\text{WE}}$, $G_{i,q_1q_2}^{\text{PI}}$ and $G_{i,q_1q_2}^{\text{WA}}$, $G_{i,q_1\ell}^{\text{WA}}$ presented in Ref. [143]. Note that, due to the current accuracy of the analysis, at dimension-seven we include only the contribution of the valence-quark therefore e.g. $\langle D^0 | \mathcal{P}_i^s | D^0 \rangle = 0$.

Having presented each of the contributions that enter the HQE of *D*-mesons, we now turn to discuss the numerical evaluation of the corresponding matrix elements. For most of the non perturbative parameters in the charm-sector there is no determination available yet, contrary to the *b*-system, where e.g. the value of μ_G^2 ,

¹³Note, that for the operator O_2^q the contribution of \mathcal{R}_2^q is absorbed by the combination $(m_D f_D/m_c)^2 \approx (1 + 2 \bar{\Lambda}/m_c) f_D^2$.

¹⁴By neglecting the effect of running down to a lower scale, from Ref. [245] one can see that in VIA the QCD decay constant entirely absorbs all the $1/m_b$ contributions.

 μ_{π}^2 , and ρ_D^3 , has been extracted performing fits to experimental data for B^0 and B^+ semileptonic decays [251]. Specifically, for μ_G^2 they find [251]

$$\mu_G^2(B) = (0.332 \pm 0.062) \,\text{GeV}^2 \,. \tag{4.64}$$

By using the heavy quark symmetry, we could expect the corresponding parameter in the *D* system to have a similar size. The value of μ_G^2 can also be obtained taking into account the spectroscopy relation [252]

$$\mu_G^2(D_{(s)}) = \frac{3}{2} m_c \left(M_{D_{(s)}^*} - M_{D_{(s)}} \right) , \qquad (4.65)$$

which holds up to power corrections. Using the value for the meson masses given in the PDG [191] and setting $m_c = 1.27 \,\text{GeV}$, we obtain the following estimates

$$\mu_G^2(D) = (0.268 \pm 0.107) \,\text{GeV}^2, \qquad \mu_G^2(D_s) = (0.274 \pm 0.110) \,\text{GeV}^2, \qquad (4.66)$$

where we have conservatively added an uncertainty of 40% due to unknown power corrections of order $1/m_c$. The values in Eq. (4.66) are roughly 19% smaller than those obtained from experimental fits for semileptonic *B*-meson decays, see Eq. (4.64). Moreover, Eq. (4.65) leads to a tiny amount of $SU(3)_f$ -symmetry breaking of $\approx 2\%$, which might, however, be enhanced by the neglected power corrections. In the literature instead of Eq. (4.65) it is often adopted the relation [52, 70]

$$\mu_G^2(D_{(s)}) = \frac{3}{4} \left(M_{D_{(s)}^*}^2 - M_{D_{(s)}}^2 \right) , \qquad (4.67)$$

which coincides with Eq. (4.65) up to corrections of order $1/m_c$. Numerically we find that Eq. (4.67) yields

$$\mu_G^2(D) = 0.41 \,\text{GeV}^2, \qquad \mu_G^2(D_s^+) = 0.44 \,\text{GeV}^2,$$
(4.68)

which are roughly 23% higher than the values in Eq. (4.64). In our numerical analysis we take the average of the two determinations in Eq. (4.66) and Eq. (4.68). This gives

$$\mu_G^2(D) = (0.34 \pm 0.10) \,\text{GeV}^2, \qquad \mu_G^2(D_s^+) = (0.36 \pm 0.10) \,\text{GeV}^2, \qquad (4.69)$$

which agrees well with the one in Eq. (4.64). From Eq. (4.22), we expect correc-
Source	LQCD [253]	$\begin{array}{c} \text{LQCD} \\ [254] \end{array}$	Exp. fit [251]	QCD SR $[255]$	QCD SR $[256]$
$\mu_{\pi}^2 [\text{GeV}^2]$	0.05(22)	0.314(15)	0.465(68)	0.10(5)	0.6(1)

Table 4.7: Different determinations of $\mu_{\pi}^2(B)$ available in the literature.

tions to the total decay rate due to the chromo-magnetic operator, $c_G \mu_G^2/(c_3 m_c^2)$ ranging between -6% and +8% with respect to the leading free-quark decay contribution. A large part of the sizeable uncertainty derives from the cancellations in the coefficient c_G , as shown in Figure 4.1, which could be reduced with a complete determination of the NLO-QCD corrections. For semileptonic rates the contribution of the chromomagnetic operator can be even of the order of 20%. An experimental determination of $\mu_G^2(D)$ from inclusive semileptonic *D*-meson decays could further reduce the uncertainties and could in particular give some insight into the numerical size of $SU(3)_f$ breaking.

For the matrix element of the kinetic operator, there are several predictions of μ_{π}^2 , available in the literature for the *B*-meson, which cover a large range of values, see Table 4.7. Assuming heavy quark symmetry we can again use the determination in Ref. [251]

$$\mu_{\pi}^{2}(B) = (0.465 \pm 0.068) \,\mathrm{GeV}^{2} \,, \tag{4.70}$$

to obtain the following estimate in the case of D-meson

$$\mu_{\pi}^2(D) = (0.465 \pm 0.198) \,\text{GeV}^2,\tag{4.71}$$

where we have added a conservative uncertainty of 40% in order to account for the breaking of the heavy quark symmetry. This value fulfils the theoretical bound $\mu_{\pi}^2 \ge \mu_G^2$, see e.g. the review [257]. We then expect, from Eq. (4.22), corrections due to the kinetic operator of the order of -10%. The $SU(3)_f$ breaking effects for the matrix element of the kinetic operator have been estimated in Refs. [143,258], i.e.

$$\mu_{\pi}^2(D_s^+) - \mu_{\pi}^2(D^0) \approx 0.09 \,\mathrm{GeV}^2\,,$$
(4.72)

leading to the following estimate in the case of the D_s meson

$$\mu_{\pi}^2(D_s^+) = (0.555 \pm 0.232) \,\mathrm{GeV}^2. \tag{4.73}$$

Again a more precise experimental determination of μ_{π}^2 from fits to semileptonic

 D^+ , D^0 and D_s^+ meson decays, as it has been done for the B^+ and B^0 decays, would be very desirable.

For the matrix element of the Darwin operator, we can again assume the validity of the heavy quark symmetry and use the corresponding value obtained from fits of the semileptonic B decays [251], namely

$$\rho_D^3(B) = (0.170 \pm 0.038) \,\text{GeV}^3,$$
(4.74)

which, by adding quadratically an uncertainty of 40% to account for the breaking of the heavy-quark symmetry, leads to a first estimate of

$$\rho_D^3(D)^{\rm I} = (0.17 \pm 0.07) \,{\rm GeV}^3 \,.$$
(4.75)

Alternatively the Darwin parameter can be related to the matrix elements of the dimension-six four-quark operators through the equation of motion for the gluon field. At leading order in $1/m_Q$, we obtain

$$\rho_D^3(H_Q) = \frac{g_s^2}{18} f_{H_Q}^2 m_{H_Q} \left[2 \,\tilde{B}_2^q - \tilde{B}_1^q + \frac{3}{4} \tilde{B}_3^q - \frac{3}{2} \tilde{B}_4^q + \sum_{q'=u,d,s} \left(2 \tilde{\delta}_2^{q'q} - \tilde{\delta}_1^{q'q} + \frac{3}{4} \tilde{\delta}_3^{q'q} - \frac{3}{2} \tilde{\delta}_4^{q'q} \right) \right],$$
(4.76)

where H_Q is a heavy hadron with the mass m_{H_Q} and the decay constant f_{H_Q} , q = u, d, s, is the light valence quark in H_Q , and the Bag parameters have bee introduced in Eqs. (4.36). The strong coupling $g_s^2 = 4\pi\alpha_s$, should be evaluated at a non perturbative scale and e.g. Ref. [259] suggests to set $\alpha_s = 1$. From the input listed

	$\mu = 1$.5 GeV	$\mu = 1$	$1.0 \mathrm{GeV}$	$\alpha_s = 1$		
$ ho_D^3 [{ m GeV^3}]$	VIA	HQET	VIA	HQET	VIA	HQET	
B^+, B_d	0.048	0.047	0.066	0.064	0.133	0.129	
B_s	0.072	0.070	0.098	0.095	0.199	0.193	
D^+, D^0	0.021	0.020	0.027	0.026	0.059	0.056	
D_s^+	0.030	0.029	0.040	0.038	0.086	0.082	

Table 4.8: Values of $\rho_D^3(H)$ for *B*- and *D*-mesons in VIA and using HQET SR for the Bag parameters for three different choices of α_s in Eq. (4.76).

in Table 4.9 and the expression in Eq. (4.76), we can estimate the size of ρ_D^3 for both the *B*- and *D*-mesons and using the VIA as well as the HQET SR values for the Bag parameters. The results are summarised in Table 4.8 for the three different choices, namely $\alpha_s(\mu = 1.5 \text{ GeV})$, $\alpha_s(\mu = 1 \text{ GeV})$ and $\alpha_s = 1$. By setting $\alpha_s = 1$ in Eq. (4.76), we obtain values for $\rho_D^3(B)$ that are close to the one in Eq. (4.74), indicating $1/m_b$ -corrections in Eq. (4.76) of the order of +30%. Moreover, the difference between using VIA and HQET sum rule is small. We emphasise that because of the sizeable $SU(3)_F$ breaking in the decay constants, Eq. (4.76) leads also to a sizeable $SU(3)_F$ breaking for the non-perturbative parameters $\rho_D^3(D)$, $\rho_D^3(D_s^+)$. By setting $\alpha_s = 1$ and using HQET SR results for the Bag parameters we arrive at the second estimate, cf. last column in Table 4.8

$$\rho_D^3(D)^{II} = (0.056 \pm 0.022) \,\text{GeV}^3 \,, \quad \rho_D^3(D_s^+)^{II} = (0.82 \pm 0.033) \,\text{GeV}^3 \,, \qquad (4.77)$$

where we have again added 40% uncertainty. Finally, another possibility to extract $\rho_D^3(D)$ is to substitute in Eq. (4.76) the values of the Bag parameters in VIA, which gives

$$\rho_D^3(H_Q) \approx \frac{g_s^2}{18} f_{H_Q}^2 m_{H_Q}.$$
(4.78)

Assuming the strong coupling to have a similar size for both the B- and D-meson matrix elements, from Eq. (4.78) we obtain

$$\rho_D^3(D) \approx \frac{f_D^2 m_D}{f_B^2 m_B} \rho_D^3(B), \quad \rho_D^3(D_s) \approx \frac{f_{D_s}^2 m_{D_s}}{f_B^2 m_B} \rho_D^3(B).$$
(4.79)

Using the most precise determination of the decay constants from Lattice QCD [248], and of the meson masses from PDG [191] and taking into account the value of $\rho_D^3(B)$ in Eq. (4.74), leads to the third estimate

$$\rho_D^3(D)^{III} = (0.075 \pm 0.034) \,\text{GeV}^3, \quad \rho_D^3(D_s)^{III} = (0.110 \pm 0.050) \,\text{GeV}^3, \quad (4.80)$$

where we again assign in addition a conservative 40% uncertainty due to missing power corrections. These values are consistent with the numbers shown in Table 4.8 for $\alpha_s = 1$. Contrary to the case of the dimension-five matrix elements, in Eq. (4.80) we observe a large $SU(3)_f$ symmetry breaking of $\approx 46\%$, and similarly of $\approx 49\%$ for the $B_{(s)}$ -mesons, as already stated above, mostly stemming from the ratios f_{B_s}/f_{B_d} and $f_{D_s^+}/f_{D^0}$. In our numerical analysis we use the values shown in Eq. (4.80), which lie between the estimates obtained in Eq. (4.75) and Eq. (4.77). Again, a more precise experimental determination of ρ_D^3 from fits to semileptonic D^+ , D^0 and D_s^+ meson decays, as it has been done for the B^+ and B^0 decays , would be very desirable and could have a significant effect on the phenomenology of inclusive charm decays.

Finally, the dimension-six Bag parameters of the D^+ and D^0 mesons have been determined using HQET sum rules in Ref. [246]. Corrections due to the inclusion of the strange quark mass, needed in the case of the D_s^+ meson, as well as the effect of the eye-contractions, have been computed for the first time in Ref. [247], again using HQET sum rules. The results, collected in Table 4.10, show only a small deviation from the corresponding VIA values. For the dimension-seven Bag parameters we use only VIA. In HQET the matrix elements of dimension-seven operators depend also on the parameters $\bar{\Lambda}_{(s)} = m_{D_{(s)}} - m_c$, for which we use the following range of values [247]

$$\bar{\Lambda} = (0.5 \pm 0.1) \,\text{GeV}, \qquad \bar{\Lambda}_s = (0.6 \pm 0.1) \,\text{GeV}.$$
 (4.81)

4.1.2 Numerical results

In this section, using all the ingredients described above, we present the theoretical predictions for the total and semileptonic decay rates of the D^0 , D^+ and D_s^+ mesons, and for their ratios. All the input included in our numerical analysis are collected in Table 4.9. For each observable we investigate several quark mass schemes, using as default the kinetic and the 1S scheme, and compare the corresponding results with both VIA and HQET SR values for the dimension-six Bag parameters. The uncertainties quoted are obtained by varying all the input parameters within their intervals. For the renormalisation scales, we fix the central values to $\mu_1 = \mu_0 = 1.5 \,\text{GeV}^{15}$, and vary both of them independently between 1 and 3 GeV. Moreover, we add an estimated uncertainty due to missing higher power and QCD corrections.

We start by considering the total decay rates, which are expected to have sizeable uncertainties due to the dependence of the free quark decay on the fifth power of the charm quark mass and due to large perturbative and power corrections. A

¹⁵The renormalisation scale μ_0 enters in the NLO-QCD corrections to the dimension-six coefficients as well as in the running of the Bag parameters.

Parameter	Value	Source	
$\alpha_s(M_Z)$	0.1179 ± 0.0010	PDG [191]	
$ V_{us} $	$0.224834^{+0.000252}_{-0.000059}$		
$ert V_{ub} ert / ert V_{cb} ert$	$0.088496\substack{+0.001885\\-0.002244}$		
$ V_{cb} $	$0.04162^{+0.00026}_{-0.00080}$		
δ	$\left(65.80^{+0.94}_{-1.29} ight)^\circ$		
$\overline{m}_c(\overline{m}_c)$	$(1.27\pm0.02){\rm GeV}$	PDG [191]	
$m_c^{\rm kin}(0.5{ m GeV})$	$1.306{ m GeV}$	[241]	
$m_{J/\psi}$	$3.0969~{\rm GeV}$	PDG [191]	
m_s	(93^{+11}_{-5}) MeV	PDG [191]	
M_{D^0}	$1.86493{ m GeV}$		
M_{D^+}	$1.86965{ m GeV}$	PDG [191]	
$M_{D_s^+}$	$1.96834{\rm GeV}$		
f_D	$(0.2120 \pm 0.0007) \text{ GeV}$	Lattice OCD [248]	
f_{D_s}	$(0.2499 \pm 0.0005) \text{ GeV}$		
$\mu_{\pi}^2(D)$	$(0.465 \pm 0.198) \text{ GeV}^2$	Exp. fit $[251]$ and HQ symmetry	
$\mu_{\pi}^2(D_s)$	$(0.555 \pm 0.232) \text{ GeV}^2$	$SU(3)_f$ -breaking [258] and HQ symmetry	
$\mu_G^2(D)$	$(0.339 \pm 0.098) \ { m GeV^2}$	Spectroscopy relations [70, 252]	
$\mu_G^2(D_s)$	$(0.357 \pm 0.104) \text{ GeV}^2$		
$\rho_D^3(D)$	$(0.075 \pm 0.034) \text{ GeV}^3$	Exp. fit $[251]$ and E \cap M relations	
$ ho_D^3(D_s)$	$(0.110 \pm 0.050) \ { m GeV^3}$	[201] and $[201]$ and $[201]$ and $[201]$	

Table 4.9: Numerical input used in the numerical analysis.

	HQET		$ ilde{B}_1$		\tilde{B}_2		$ ilde{B}_3$		$ ilde{B}_4$	
	$D^{+,0}$	1	$.0000^{+0.0020}_{-0.0006}$		$1.0000\substack{+0.0007\\-0.0000}$	_	$-0.0161^{+0.0115}_{-0.0206}$		$-0.0007^{+0.0104}_{-0.0170}$	
	D_s^+	1	$.0000\substack{+0.0014\\-0.0003}$		$1.0000\substack{+0.0007\\-0.0000}$	_	$-0.0094^{+0.0103}_{-0.0171}$		$-0.0001^{+0.0104}_{-0.0169}$	
	HQET		$ ilde{\delta}_1$		$ ilde{\delta}_2$		$ ilde{\delta}_3$		$ ilde{\delta}_4$	
$\langle $	$D_q \tilde{O}^q D_q$	\rangle	$0.0026^{+0.000}_{-0.000}$	49	$-0.0018^{+0.000}_{-0.000}$	5 2	$-0.0004^{+0.000}_{-0.000}$)1)1	$0.0003^{+0.000}_{-0.000}$)0)1
$\langle $	$D_s \tilde{O}^q D_s$	\rangle	$0.0025^{+0.000}_{-0.000}$	48	$-0.0018^{+0.000}_{-0.000}$	15 12	$-0.0004^{+0.000}_{-0.000}$)1)1	$0.0003\substack{+0.000\\-0.000}$)0)1
$\langle $	$D_q \tilde{O}^s D_q]$	\rangle	$0.0017\substack{+0.000\\-0.000}$	5 9	$-0.0012^{+0.000}_{-0.000}$	15 13	$-0.0003\substack{+0.000\\-0.000}$)1)1	$0.0002\substack{+0.000\\-0.000}$)1)1
\langle	$D_s \tilde{O}^s D_s $	\rangle	$0.0023^{+0.000}_{-0.000}$	5	$-0.0017^{+0.000}_{-0.000}$	15 12	$-0.0004^{+0.000}_{-0.000}$)2)1	$0.0003\substack{+0.000\\-0.000}$)0)1

Table 4.10: Numerical values of the HQET Bag parameters [246, 247], at the renormalisation scale $\mu_0 = 1.5 \text{ GeV}$.

comparison of the central values for the HQE prediction of the decay widths in several mass schemes is shown in the three first rows of Table 4.11, using VIA for the Bag parameters and of Table 4.12 using the HQET sum rules results. In Table 4.13 we present the complete theoretical prediction including the corresponding uncertainties, using the 1*S* and kinetic scheme for the quark masses and the HQET SR values for the dimension-six Bag parameters, the same results can be visualised also in Figure 4.4. In each table, the corresponding experimental determinations are listed in the last column. For the D_s^+ meson there is an additional subtlety due to the large branching fraction of the leptonic decay $D_s^+ \rightarrow \tau^+ \nu_{\tau}$, which however is not included in the HQE, since the tau lepton is more massive than the charm quark. Using the experimental value of the leptonic branching ratio [191] (online update), we obtain

$$Br(D_s^+ \to \tau^+ \nu_\tau) = (5.48 \pm 0.23)\%, \qquad (4.82)$$

accordingly, we define the reduced decay rate $\overline{\Gamma}(D_s^+)$, as

$$\bar{\Gamma}(D_s^+) \equiv \Gamma(D_s^+) - \Gamma(D_s^+ \to \tau^+ \nu_\tau) = (1.88 \pm 0.02) \,\mathrm{ps}^{-1} \,, \tag{4.83}$$

which leads to the reduced lifetime ratio

$$\frac{\bar{\tau}(D_s^+)}{\tau(D^0)} = 1.30 \pm 0.01$$
. (4.84)

VIA					
Observable	Pole	$\overline{\mathrm{MS}}$	Kinetic	1S	Exp. value
$\Gamma(D^0)[\mathrm{ps}^{-1}]$	1.68	1.47	1.56	2.31	2.44
$\Gamma(D^+)[\mathrm{ps}^{-1}]$	0.19	-0.03	0.09	0.56	0.96
$\bar{\Gamma}(D_s^+)[\mathrm{ps}^{-1}]$	1.72	1.48	1.58	2.34	1.88
$\tau(D^+)/\tau(D^0)$	2.55	2.56	2.53	2.82	2.54
$\bar{\tau}(D_s^+)/\tau(D^0)$	0.99	1.00	0.99	0.99	1.30
$B^{D^0}_{sl}[\%]$	5.31	6.46	6.03	8.48	6.49
$B^{D^+}_{sl}[\%]$	13.5	16.4	15.3	21.5	16.07
$B^{D^+_s}_{sl}[\%]$	6.88	8.24	7.74	10.8	6.30
$\Gamma^{D^+}_{sl}/\Gamma^{D^0}_{sl}$	1.000	1.000	1.000	1.000	0.985
$\Gamma^{D_s^+}_{sl}/\Gamma^{D^0}_{sl}$	1.04	1.04	1.04	1.05	0.790

Table 4.11: Central values of the charm observables in different quark mass schemes using VIA for the matrix elements of the four-quark operators compared to the corresponding experimental values (last column).

The first and main result we can derive from Table 4.13 and from Figure 4.4, is that the HQE can reproduce the experimental values of $\Gamma(D^0)$, $\Gamma(D^+)$, and $\Gamma(D_s^+)$, within very large uncertainties. Moreover, we find that in the 1S scheme we obtain larger values for the decay rates, while the kinetic and the $\overline{\rm MS}$ scheme typically result in smaller values, close to the pole scheme. Within the uncertainties the predictions in the different mass schemes are compatible with each other, however, to a large extent. Given the current precision then, to consider only one quark mass scheme might lead to considerably underestimate the uncertainties. Due to the fact that the values of the HQET Bag parameters [246, 247] are close to the corresponding ones in VIA, the predictions shown in Table 4.11 and in Table 4.12 do not differ much. A peculiar role is played by the D^+ meson, for which we obtain huge theoretical uncertainties because of the large negative value of the Pauli interference contribution at dimension-six. This term actually dominates the total decay rate. Furthermore, the large negative value is enhanced by the NLO-QCD corrections, but partly compensated by the dimension-seven contribution. In this respect, having an independent determination of the HQET sum rule results, e.g. with a lattice QCD computation, as well as higher order QCD corrections to dimension-six and seven

HQET SR					
Observable	Pole	$\overline{\mathrm{MS}}$	Kinetic	1S	Exp. value
$\Gamma(D^0)[\mathrm{ps}^{-1}]$	1.71	1.50	1.59	2.34	2.44
$\Gamma(D^+)[\mathrm{ps}^{-1}]$	-0.05	-0.25	-0.14	0.29	0.96
$\bar{\Gamma}(D_s^+)[\mathrm{ps}^{-1}]$	1.70	1.46	1.56	2.32	1.88
$\tau(D^+)/\tau(D^0)$	2.83	2.83	2.80	3.14	2.54
$\bar{\tau}(D_s^+)/\tau(D^0)$	1.01	1.02	1.01	1.01	1.30
$B^{D^0}_{sl}[\%]$	5.18	6.37	5.93	8.34	6.49
$B^{D^+}_{sl}[\%]$	13.2	16.2	15.1	21.2	16.07
$B_{sl}^{D_s^+}[\%]$	6.79	8.19	7.67	10.7	6.30
$\Gamma^{D^+}_{sl}/\Gamma^{D^0}_{sl}$	1.002	1.001	1.001	1.002	0.985
$\Gamma^{D_s^+}_{sl}/\Gamma^{D^0}_{sl}$	1.05	1.05	1.05	1.05	0.790

Table 4.12: Central values of the charm observables in different quark mass schemes using HQET sum rule results [246, 247] for the matrix elements of the four-quark operators compared to the corresponding experimental values (last column).

might could significantly bring more insights. For the determination of the lifetime ratios, in order to eliminate the contribution of the free-quark decay, we use

$$\frac{\tau(D_{(s)}^+)}{\tau(D^0)} = 1 + \left(\Gamma^{\text{HQE}}(D^0) - \Gamma^{\text{HQE}}(D_{(s)}^+)\right)\tau^{\exp}(D_{(s)}^+).$$
(4.85)

In Eq. (4.85), Γ_3 cancels exactly and Γ_5 and Γ_6 cancel up to isospin or $SU(3)_f$ breaking corrections in the corresponding non-perturbative matrix elements. The lifetime ratios should then be dominated by the contribution of four-quark operators. The results for the HQE prediction of the lifetime ratios, in several mass schemes, are shown in the fourth and fifth rows of Table 4.11, Table 4.12, Table 4.13 as well as in Figure 4.4. We observe that the large lifetime ratio $\tau(D^+)/\tau(D^0)$ is well reproduced in all schemes considered, while in the case of $\tau(D_s^+)/\tau(D^0)$ the HQE result lies closer to one, compared to the experimental value. In the latter case, the theoretical prediction is dominated by $SU(3)_f$ -symmetry breaking effects in the non-perturbative parameters μ_{π}^2 , μ_G^2 and ρ_D^3 , which are only very roughly known. Future and more precise determinations of their values can considerably improve our

Observable	Kinetic scheme	1S-scheme	Exp. value
$\Gamma(D^0)[\mathrm{ps}^{-1}]$	$1.590 \pm 0.242^{+0.451}_{-0.365}{}^{+0.002}_{-0.002}$	$2.348 \pm 0.247^{+0.651}_{-0.489}{}^{+0.002}_{-0.001}$	2.44 ± 0.01
$\Gamma(D^+)[\mathrm{ps}^{-1}]$	$-0.138 \pm 0.572^{+0.581}_{-0.273}{}^{+0.252}_{-0.273}{}^{+0.252}_{-0.102}$	$0.293 \pm 0.664^{+0.939}_{-0.453}{}^{+0.360}_{-0.161}$	0.96 ± 0.01
$\bar{\Gamma}(D_s^+)[\mathrm{ps}^{-1}]$	$1.572 \pm 0.309^{+0.508}_{-0.399}{}^{+0.018}_{-0.004}$	$2.330 \pm 0.349^{+0.734}_{-0.540}{}^{+0.027}_{-0.009}$	1.88 ± 0.02
$\boxed{\tau(D^+)/\tau(D^0)}$	$2.798 \pm 0.606^{+0.001}_{-0.135}{}^{+0.109}_{-0.263}$	$3.137 \pm 0.691^{+0.023}_{-0.299}{}^{+0.169}_{-0.376}$	2.54 ± 0.02
$\bar{\tau}(D_s^+)/\tau(D^0)$	$1.010 \pm 0.105^{+0.018}_{-0.030}{}^{+0.003}_{-0.010}$	$1.010 \pm 0.118^{+0.027}_{-0.044}{}^{+0.006}_{-0.015}$	1.30 ± 0.01
$B_{sl}^{D^0}[\%]$	$5.94 \pm 1.15^{+0.33}_{-0.28}$	$8.36 \pm 1.31^{+0.23}_{-0.08}$	6.49 ± 0.11
$B_{sl}^{D^+}[\%]$	$15.1 \pm 2.91^{+0.83}_{-0.72}$	$21.2 \pm 3.32^{+0.58}_{-0.19}$	16.07 ± 0.30
$B_{sl}^{D_s^+}[\%]$	$7.73 \pm 1.80^{+0.45}_{-0.40}$	$10.76 \pm 2.18^{+0.33}_{-0.04}$	6.30 ± 0.16
$\Gamma^{D^+}_{sl}/\Gamma^{D^0}_{sl}$	$1.002 \pm 0.002 \pm 0.001$	$1.002 \pm 0.003 \pm 0.001$	0.985 ± 0.028
$\Gamma^{D^+_s}_{sl}/\Gamma^{D^0}_{sl}$	$1.053 \pm 0.130^{+0.006}_{-0.007}$	$1.060 \pm 0.164^{+0.006}_{-0.007}$	0.790 ± 0.026

Table 4.13: HQE predictions for all the ten observables in the kinetic (second column) and in the 1*S*-schemes (third column), using HQET SR results for the Bag parameters. The first uncertainty is parametric one, second and third uncertainties are due to μ_1 - and μ_0 -scales variation, respectively. The results are compared with the corresponding experimental measurements (fourth column).

conclusion for these lifetime ratios. In the case of the inclusive semileptonic decays, we introduce the shorthand notations $\Gamma_{sl}^D \equiv \Gamma(D \to Xe^+\nu_e)$ and $B_{sl}^D \equiv \text{Br}(D \to Xe^+\nu_e)$. The theoretical predictions are then obtained as

$$B_{sl}^{D,\text{HQE}} = \Gamma_{sl}^{D,\text{HQE}} \cdot \tau(D)^{\text{Exp.}} .$$
(4.86)

The HQE results in several mass schemes are shown in the sixth, seventh and eighth row of Table 4.11, Table 4.12 and Table 4.13, as well as in Figure 4.4. In the kinetic scheme all HQE predictions for the semi-leptonic branching fractions cover the experimental values, while the results in the 1*S* scheme tend to be too large. It is interesting to note, that by adding NNLO-QCD corrections to the semileptonic decays, the difference between the two quark mass schemes is considerably reduced, see Figure 4.5. Using the experimental values respectively for the D^0 lifetime and the semileptonic branching fraction, we determine the semileptonic ratios in the following way



Figure 4.4: A comparison of the HQE prediction for the charm observables in the kinetic scheme (magenta) and in the 1S scheme (blue), with the corresponding experimental data (green). Note that all the quantities are normalised to the corresponding experimental central values.

$$\frac{\Gamma_{sl}^{D^+}}{\Gamma_{sl}^{D^0}} = 1 + \left[\Gamma_{sl}^{D^+} - \Gamma_{sl}^{D^0}\right]^{\text{HQE}} \left[\frac{\tau(D^0)}{B_{sl}^{D^0}}\right]^{\text{exp}}, \qquad (4.87)$$

$$\frac{\Gamma_{sl}^{D_s^+}}{\Gamma_{sl}^{D^0}} = 1 + \left[\Gamma_{sl}^{D_s^+} - \Gamma_{sl}^{D^0}\right]^{\text{HQE}} \left[\frac{\tau(D^0)}{B_{sl}^{D^0}}\right]^{\text{exp}} .$$
(4.88)

The HQE results for these ratios are shown in the ninth and tenth row of Table 4.11, Table 4.12 and Table 4.13 and in Figure 4.4. In agreement with the experimental data, the HQE leads to values for $\Gamma_{sl}^{D^+}/\Gamma_{sl}^{D^0}$ very close to one. Also for $\Gamma_{sl}^{D^+}/\Gamma_{sl}^{D^0}$ the corresponding theoretical prediction is close to one, however, the corresponding experimental number is as low as 0.79, confirming the necessity of having better control over the $SU(3)_f$ -symmetry breaking effects in the non-perturbative parameters μ_G^2 , μ_{π}^2 and ρ_D^3 for the D mesons.

4.2 Charm mixing

In Section 4.1 we have shown that the HQE is able to reproduce, within large theoretical uncertainties, the experimental pattern for the lifetime of charmed mesons.



Figure 4.5: A comparison of the HQE prediction for the charm observables in the kinetic scheme (magenta) and in the 1S scheme (blue) with the corresponding experimental data (green). All the quantities are normalised to the corresponding experimental central values. In comparison to Fig. 4.4 now the NNLO corrections to the semileptonic branching fractions are included, taken from the talk of Matteo Fael at the CHARM-2020 conference.

However, a naive application of the HQE yields results for the decay rate difference of neutral D mesons that are four orders of magnitude smaller than the experimental ones. It is well known that this huge suppression results from severe Glashow-Iliopoulos-Maiani (GIM) cancellations [261]. Following Ref. [235], we discuss a possible explanation for the large discrepancy between the theoretical prediction for D-mixing and experimental data. We stress though, that we do not present a detailed derivation of the fundamentals of the theory of mixing, for which, instead, we refer to the comprehensive reviews [194–196].

4.2.1 GIM cancellations in *D*-mixing

Because of the weak interaction, neutral mesons, here we consider the case of the D^0 meson, can mix with their corresponding antiparticles through the box diagrams shown in Figure 4.6. The process is described by a 2 × 2 Hamiltonian matrix with non vanishing off-diagonal entries M_{12} and Γ_{12} . By diagonalising the mixing matrix of the D^0 and the \bar{D}^0 mesons, we can obtain the two eigenstates with definite mass and decay width. The corresponding observables ΔM_D and $\Delta \Gamma_D$, denoting respectively the mass and decay width difference between the two eigenstates, are



Figure 4.6: Box diagrams contributing to mixing of neutral *D*-mesons.

functions of Γ_{12} and M_{12} . Moreover we define

$$x = \frac{\Delta M_D}{\Gamma_{D^0}}, \qquad y = \frac{\Delta \Gamma_D}{2\Gamma_{D^0}}, \qquad (4.89)$$

and

$$x_{12} = \frac{2|M_{12}|}{\Gamma_{D^0}}, \qquad y_{12} = \frac{2|\Gamma_{12}|}{\Gamma_{D^0}}, \qquad \phi_{12} = \arg\left(\frac{M_{12}}{\Gamma_{12}}\right).$$
(4.90)

In the following we only discuss the computation of Γ_{12} and, by taking into account the bound $\Delta\Gamma_D \leq 2 |\Gamma_{12}|$, see e.g. Refs. [195, 262], we derive the theoretical prediction for $\Delta\Gamma_D$. In fact, we do not consider the calculation of M_{12} , and hence we can only determine one contribution to the mixing phase ϕ_{12} in Eq. (4.90). Γ_{12} corresponds to the absorptive part of the mixing amplitude of the $D^0 - \bar{D}^0$ system, and it is then obtained by computing the imaginary part of the matrix element of the effective Hamiltonian describing the *c*-quark decay, between the D^0 and \bar{D}^0 states. Using the formalism described in Section 1.4, Γ_{12} can be expanded in inverse powers of the heavy *c*-quark mass, leading to

$$\Gamma_{12} = \left[\Gamma_6^{(0)} + \frac{\alpha_s}{4\pi}\Gamma_6^{(1)} + \ldots\right] \frac{\langle Q_6 \rangle}{m_c^3} + \ldots , \qquad (4.91)$$

where the ellipsis stand for terms of higher order and we have explicitly shown the perturbative expansion of the short distance coefficient Γ_6 , cf. Eq. (1.165). Eq. (1.164), is diagrammatically represented in Figure 4.7. The product of $\Delta C = 1$ operators in the effective Hamiltonian, i.e. in the "full" theory, cf. Eq. (4.11), is now matched into a series of local $\Delta C = 2$ operators Q_d of increasing dimension $d \ge 6$, with the short distance coefficients denoted by Γ_d . The expressions for $\Gamma_6^{(0)}$ can be easily derived in complete analogy to what it has been done in Section 2.2, while those for $\Gamma_6^{(1)}$ can be obtained from the corresponding ones for *B*-mixing determined in Refs. [263–268] by replacing $m_b \to m_c$, $m_c \to m_s$, etc. Furthermore, the matrix elements of the dimension-six operators have been computed e.g. in Refs. [246, 269]. The experimental value of the decay rate difference reads ¹⁶

$$\Delta \Gamma_D^{\text{Exp}} = 2y/\tau(D^0) = (0.032 \pm 0.003) \,\text{ps}^{-1}, \tag{4.92}$$

which, at one standard deviation, leads to the following bound

$$\Delta \Gamma_D^{\text{Exp}} \ge 0.028 \,\text{ps}^{-1} \,. \tag{4.93}$$

In order to compare the theoretical predictions with the experimental determinations, we investigate the quantities

$$\alpha = -\arg(\Gamma_{12}), \qquad \Omega = \frac{2 |\Gamma_{12}|^{\text{SM}}}{0.028 \, \text{ps}^{-1}},$$
(4.94)

where α contributes to CP violation in mixing and values of Ω smaller than one signal a discrepancy between the theoretical and experimental description of D-mixing, within the one sigma range. A naive application of the HQE leads to $\Omega = 3.4 \cdot 10^{-5}$ at LO-QCD and to $6.2 \cdot 10^{-5}$ at NLO-QCD, showing that the theoretical prediction for the decay rate difference is more than four orders of magnitude smaller than the corresponding experimental number. Moreover, the phase α is very large, i.e. $\alpha = 93^{\circ}$ at LO-QCD and $\alpha = 99^{\circ}$ at NLO-QCD. By default in our numerical analysis we have used PDG [270] values for the quark, in the $\overline{\text{MS}}$ scheme, and meson masses, as well as for the strong coupling, CKM input from Ref. [260], the results of Ref. [246] for the non-perturbative matrix elements and Ref. [248] for the D^0 decay constant. In order to analyse the peculiarities of D-mixing, we decompose Γ_{12} according to the flavour of the internal quark pair, cf. Figure 4.7. We denote the corresponding three contributions by Γ_{12}^{ss} , Γ_{12}^{dd} , and Γ_{12}^{sd} , i.e.

$$\Gamma_{12} = -\left(\lambda_s^2 \Gamma_{12}^{ss} + 2 \lambda_s \lambda_d \Gamma_{12}^{sd} + \lambda_d^2 \Gamma_{12}^{dd}\right) = -\lambda_s^2 \left(\Gamma_{12}^{ss} - 2\Gamma_{12}^{sd} + \Gamma_{12}^{dd}\right) + 2\lambda_s \lambda_b \left(\Gamma_{12}^{sd} - \Gamma_{12}^{dd}\right) - \lambda_b^2 \Gamma_{12}^{dd},$$
(4.95)

where the CKM factors are defined as $\lambda_q = V_{cq}V_{uq}^*$, and we have used the unitarity relation $\lambda_d + \lambda_s + \lambda_b = 0$, to eliminate λ_d in the second line of Eq. (4.95). Taking

¹⁶Note that for consistency we present the numbers used in the analysis of Ref. [235] based on the previous determination $y = 0.68^{+0.06}_{-0.07}$ %, however, using the new value quoted in Eq. (4.1), would not lead to any significant difference, and we would have instead $\Delta\Gamma_D^{\text{Exp}} \ge 0.027 \text{ ps}^{-1}$.



Figure 4.7: Diagrams describing mixing of neutral D mesons via intermediate $s\bar{s}$, $s\bar{d}$, $d\bar{s}$, and $d\bar{d}$, states in the "full" theory at LO-QCD (left) and NLO-QCD (center) and at NLO-QCD in the HQE (right). The crossed circles denote the insertion of $\Delta C = 1$ operators of the effective Hamiltonian describing the charm-quark decay, while the full dot indicates the insertion of $\Delta C = 2$ operators in the HQE.

into account the numerical value of the CKM elements, we see that Eq. (4.95) shows the presence of a very pronounced hierarchy, namely

$$-\lambda_s^2 = -4.791 \cdot 10^{-2} + 3.094 \cdot 10^{-6}i, \qquad (4.96)$$

$$+2\lambda_s\lambda_b = +2.751 \cdot 10^{-5} + 6.121 \cdot 10^{-5}i, \qquad (4.97)$$

$$-\lambda_b^2 = +1.560 \cdot 10^{-8} - 1.757 \cdot 10^{-8}i.$$
(4.98)

The CKM factor in the first term of Eq. (4.95) has considerably the largest real part, whereas the second term has the largest imaginary part and it should then be important for the determination of the potential size of CP violation in *D*-mixing. The relative size between the imaginary and real part, is much larger in λ_b than in λ_s and we therefore suggest to include all terms in Eq. (4.95). Moreover, extreme GIM cancellations [261] affect the short distance coefficients of the CKM elements in Eq. (4.95). By expanding in the small mass parameter $z = m_s^2/m_c^2$, we obtain at LO- (top line) and at NLO-QCD (lower line), respectively

$$\Gamma_{12}^{ss} = \begin{cases} 1.62 - 2.34 \, z - 5.07 \, z^2 + \dots, \\ 1.42 - 4.30 \, z - 12.45 \, z^2 + \dots, \end{cases}$$
(4.99)

$$\Gamma_{12}^{sd} - \Gamma_{12}^{dd} = \begin{cases} -1.17 \, z - 2.53 \, z^2 + \dots, \\ -2.15 \, z - 6.26 \, z^2 + \dots, \end{cases}$$
(4.100)

$$\Gamma_{12}^{ss} - 2\Gamma_{12}^{sd} + \Gamma_{12}^{dd} = \begin{cases} -13.38 \, z^3 + \dots, \\ 0.07 \, z^2 - 29.72 \, z^3 + \dots \end{cases}$$
(4.101)

Note that in the NLO result in Eq. (4.101), the GIM suppression is lowered by one power of z, as it has been observed before [271, 272]. We conclude that the peculiarity of Eq. (4.95) lies in the fact that the CKM dominant factor λ_s^2 multiplies the extremely GIM suppressed term given in Eq. (4.101), the CKM suppressed factor $\lambda_s \lambda_b$ multiplies the GIM suppressed term given in Eq. (4.100), while the very CKM suppressed factor λ_b^2 multiplies Γ_{12}^{dd} , obtained taking the limit $z \to 0$ in Eq. (4.99), in which no GIM suppression is present. Therefore, the three contributions in Eq. (4.95) have actually a similar size, in fact

$$\Gamma_{12} = \left(2.08 \cdot 10^{-7} - 1.34 \cdot 10^{-11}i\right) \text{(1st term)} - \left(3.74 \cdot 10^{-7} + 8.31 \cdot 10^{-7}i\right) \text{(2nd term)} + \left(2.22 \cdot 10^{-8} - 2.5 \cdot 10^{-8}i\right) \text{(3rd term)}.$$

$$(4.102)$$

Because of Eq. (4.97), it also follows that a sizeable contribution to the mixing phase can only arise if the slightly GIM suppressed term in Eq. (4.100) is enhanced. In order to explain the mismatch between the HQE prediction and experimental determination, in the literature different solutions have been proposed.

- i) At higher order in the HQE, the GIM suppression could be less pronounced [273-275]. First estimates of the dimension-nine contribution to *D*-mixing, performed in Ref. [276], show indeed such an enhancement, but not on a scale sufficient to reproduce the experimental result. For a final statement about this possibility, the complete determination of the dimension-nine and twelve contributions, would be necessary.
- ii) The discrepancy is a signal of the violation of quark hadron duality. However, while it was originally suggested that deviations of quark hadron duality should be as large as 10^5 , because of $\Omega \approx 10^{-5}$, this seems unlikely given the many successful tests of the HQE. In fact in Ref. [262] it was shown that violations as small as 20 per cent could be sufficient to explain the experimental data.
- iii) The HQE is not applicable and we have to consider different methods, like to sum over all the exclusive decays channels contributing to the decay rate difference, see e.g. Refs. [277–279].



Figure 4.8: Comparison of μ_1 -dependence of $|\Gamma_{12}|$ at LO-QCD (dotted blue) and NLO-QCD (solid pink).

4.2.2 Alternative scale setting

 Γ_{12} depends on the two scales μ_1 and μ_2 . The former denotes the renormalisation scale of the $\Delta C = 1$ operators, and it is explicitly present in the expressions of the corresponding Wilson coefficients in the effective Hamiltonian, the latter is the renormalisation scale of the $\Delta C = 2$ operators that arise in the HQE and it appears also in their short distance coefficients. Up to higher order terms, the dependence on μ_1 and μ_2 must cancel between the local matrix elements and the corresponding short distance functions. Without discussing the dependence on μ_2 , for which the cancellation is very effective, in the following we consider only that on μ_1 . In the B_s system the cancellation is numerically only weakly realised when moving from LO- to NLO-QCD, see Refs. [280, 281], indicating the importance of higher order corrections. First steps in this direction show indeed large NNLO-QCD effects [280, 281]. In the D system a reduction of the μ_1 -dependence, when moving from LO- to NLO-QCD, is present in the individual contributions $\Gamma_{12}^{ss,sd,dd}$, but not in Γ_{12} , see Figure 4.8, which seems to be again consequence of the severe GIM cancellations. By explicitly showing the scale dependence in Γ_{12} , we can write

$$\Gamma_{12} = \sum_{q_1q_2 = ss, sd, dd} \Gamma_6^{q_1q_2}(\mu_1^{q_1q_2}, \mu_2^{q_1q_2}) \langle Q_6 \rangle (\mu_2^{q_1q_2}) \frac{1}{m_c^3} + \dots$$
(4.103)

In general different internal quark pairs contribute to different decay channels of the $D^0(\bar{D}^0)$ meson e.g. $s\bar{s}$ to a K^+K^- final state and $s\bar{d}$ to a π^+K^- final state. For each of these observables the choice of the renormalisation scales is a priori arbitrary, nevertheless typically one fixes $\mu_x^{ss} = \mu_x^{sd} = \mu_x^{dd} = \mu$, which is then chosen to be equal to the mass of the decaying heavy quark, i.e. $\mu = m_Q$ for Q quark decays, in order to minimise the effect of the logarithmic terms $\alpha_s(\mu) \log(\mu^2/m_Q^2)$. Uncertainties due to

unknown higher order corrections are estimated varying μ between $m_Q/2$ and $2 m_Q$ and in the case of the charm quark we fix the lower bound to 1 GeV to still ensure reliable perturbative results.

We consider two alternative possibilities to treat the renormalisation scale $\mu_1^{q_1q_2}$, which both allow to reduce the discrepancy between the theoretical and the experimental determination of *D*-mixing, while leaving the other HQE predictions unchanged, namely

- A) We fix the central value of the three scales μ_1^{ss} , μ_1^{sd} , and μ_1^{dd} , to m_c , but we vary them independently between 1 GeV and $2m_c$.
- B) We choose different central values for the three scales μ_1^{ss} , μ_1^{sd} , and μ_1^{dd} , according to the size of the available phase space. In particular we evaluate Γ_6^{ss} at the scale $\mu_1^{ss} = \mu 2\epsilon$, Γ_6^{sd} at the scale $\mu_1^{sd} = \mu \epsilon$, and Γ_6^{dd} at the scale $\mu_1^{dd} = \mu$, where ϵ is an unknown parameter, related to the kinematics of the decays.

If ϵ is not too large, both methods yield results for the individual Γ_6^{ss} , Γ_6^{sd} and Γ_6^{dd} , which lie within the usually quoted theory uncertainties obtained following the prescription stated above, but they affect in a sizeable way the severe GIM cancellations in Eqs. (4.100) and (4.101). The first method gives a considerably enhanced range of values for Ω in Eq. (4.93), i.e.

$$\Omega \in \left[4.6 \cdot 10^{-5}, 1.3\right],\tag{4.104}$$

which nicely covers also the experimental determination of the decay rate difference. Scanning independently over μ_1^{ss} , μ_1^{sd} , and μ_1^{dd} , in 11 equidistant steps we find that out of the 1331 points only 14 give a value of $\Omega < 0.001$, while 984 give a value of $\Omega > 0.1$. The large discrepancy between the theoretical and the experimental determinations, seems then to be an artefact of fixing the scales μ_1^{ss} , μ_1^{sd} , and μ_1^{dd} , to be the same. The range of values shown in Eq. (4.104) does not change significantly if we use the pole scheme for the quark masses, lattice results instead of the HQET sum rule results, or a different $\Delta C = 2$ operator basis. In all these cases we can obtain $\Omega \ge 1$. For α in Eq. (4.94), we observe that the results lie in the range $[-\pi, \pi]$. A closer look however, shows that for $\Omega > 0.5$ only values of $\alpha < 0.1^{\circ}$ are allowed, and conversely large values of α correspond to results for Ω inconsistent with the experimental data.

The second method for the scale setting requires the introduction of a mass scale ϵ . A possible estimate for the size of this parameter could be the strange quark mass $\epsilon = m_s \approx 0.1$ GeV or the phase space difference of the corresponding exclusive



Figure 4.9: Comparison of the ϵ dependence of Ω at LO-QCD (blue) and NLO-QCD (pink) for different values of μ : the dashed line corresponds to $\mu = m_c$ while the two solid lines to $\mu = 1$ GeV and $\mu = 2m_c$.

decay channels, specifically, by comparing the energy release of $D^0 \to K^+ K^-$, $M_{D^0} - 2M_{K^+} = 0.88$ GeV, with that of $D^0 \to \pi^+\pi^-$, $M_{D^0} - 2M_{\pi^+} = 1.59$ GeV we might expect that $\epsilon \approx 0.35$ GeV. In Figure 4.9 it is shown how the HQE prediction of Ω would be affected in this scenario. Also in this case, it appears possible that the theoretical prediction could reach the experimental value for $\epsilon \approx 0.2$ GeV.

Finally, we have to consider how other HQE predictions would be affected by choosing a different scale setting procedure. In the case of observables in which GIM-like cancellations are not present, like the lifetime, both in the charm and bottom system, and the decay rate difference $\Delta\Gamma_s$, there is no significant change, but only a shift within the usually quoted theory uncertainties. However, the semileptonic CP asymmetries are governed by the weakly GIM suppressed contribution in B_s -mixing. The SM predictions read

$$\operatorname{Re}\left(\frac{\Gamma_{12}^{q}}{M_{12}^{q}}\right)^{\mathrm{SM}} = -\frac{\Delta\Gamma_{q}}{\Delta M_{q}} = \begin{cases} -(49.9 \pm 6.7) \cdot 10^{-4} & q = s \\ -(49.7 \pm 6.8) \cdot 10^{-4} & q = d \end{cases},$$
$$\operatorname{Im}\left(\frac{\Gamma_{12}^{q}}{M_{12}^{q}}\right)^{\mathrm{SM}} = a_{sl}^{q} = \begin{cases} (+2.2 \pm 0.2) \cdot 10^{-5} & q = s \\ (-5.0 \pm 0.4) \cdot 10^{-4} & q = d \end{cases},$$
(4.105)

while in the scenario B we obtain

$\epsilon ({\rm GeV})$	Γ^s_{12}/M^s_{12}	Γ^d_{12}/M^d_{12}
0.	-0.00499 + 0.000022i	-0.00497 - 0.00050i
0.2.	-0.00494 + 0.000023i	-0.00492 - 0.00053i
0.5.	-0.00484 + 0.000026i	-0.00482 - 0.00059i
1.0	-0.00447 + 0.000037i	-0.00448 - 0.00084i
1.5.	-0.00287 + 0.000091i	-0.00309 - 0.0021i

We see that for values of $\epsilon \leq 1$ GeV, the predictions for the real part lie within the usually quoted theory uncertainties, indicated in bold type. However, they would be increased by almost 100% in correspondence of larger values of ϵ .

Our conclusion is that, by modifying the usually adopted scale setting, the theoretical uncertainty of y, within the HQE, becomes larger than previously thought and it can cover the experimental value. However, this does not represent a complete solution and more precise estimates of higher power corrections to the HQE, as well as full NNLO-QCD corrections to the leading dimension-six term, could bring further insights. The alternative scale setting procedure shows that a small contribution to CP violation in mixing stemming from the decay rate can be up to one per mille within in the SM, which agrees with estimates made in Refs. [282, 283]. For a prediction of CP violation in mixing, the contribution coming from M_{12} needs to be determined in addition. This might be done in future via the help of dispersion relations, see e.g. Refs. [278, 283, 284]. We would like to note that our suggested procedure still respects the GIM mechanism, because for vanishing internal strange quark mass, also the parameter ϵ is zero. Finally this alternative scale setting does not affect quantities like $\tau(D^+)/\tau(D^0)$, b-hadron lifetimes and $\Delta\Gamma_s$ outside the range of their quoted theoretical errors, but it affects the semileptonic CP asymmetries and we get enhanced SM ranges

$$a_{sl}^d \in [-9.2; -4.6] \cdot 10^{-4}, \qquad a_{sl}^s \in [2.0; 4.0] \cdot 10^{-5}.$$
 (4.106)

Note that in Ref. [14], the CP violating effects responsible for creating the baryon asymmetry stem actually from a_{sl}^d and a_{sl}^s .

Conclusion

As discussed at the beginning this work, indirect BSM searches with quark flavour observables, represent a promising route to improve the current understanding of the fundamental laws of physics. However, this strategy strongly relies on the ability to systematically increase both the experimental and the theoretical precision. In the present thesis we have analysed the theoretical status for the study of the inclusive decay widths of heavy hadrons, which define observables of primary phenomenological importance in heavy flavour physics. In particular, we have discussed the computation of higher power corrections to the HQE, and tested its applicability in the charm sector, for the case of inclusive quantities like lifetimes, semileptonic branching fractions and mixing observables. Specifically, the first part of this thesis has been dedicated to presenting the main ingredients required for the computation. We have started by introducing the weak effective Hamiltonian and the heavy quark effective theory, which constitute the two effective theories that allow to disentangle a multi-scale problem like the weak decay of heavy hadrons, by progressively integrating out the heavier degrees of freedom, respectively, the W-boson and the massive component of the heavy quark field. We stress that our exposure, far from being exhaustive, has only covered the aspects relevant for the subsequent discussions and has mostly followed the excellent reviews available in the literature. We have then performed a detailed derivation of the expansion of the quark propagator in the external gluon field using the Fock-Schwinger gauge, which provides a gauge covariant parametrisation of the soft interaction with the non perturbative QCD field, for a quark propagating with large momentum inside the hadronic state. The corresponding expressions, up to terms proportional to one covariant derivative of the gluon field strength tensor, have been computed both in momentum and in coordinate space. Finally, we have presented a pedagogical introduction to the heavy quark expansion, the theoretical framework in which all of the remaining computations and discussions are embedded. In the second part of this work, we have shown the explicit calculation of the lowest dimensional contributions to the HQE of a Bmeson, namely those of two-quark operators up to order $1/m_b^2$ and of four-quark operators up to order $1/m_b^4$. For the former, the computation has been performed

using the coordinate representation of the quark propagator and considering the single decay mode $b \rightarrow c\bar{u}d$. Moreover, we have reproduced the results given in the literature and in some cases derived more general expressions. Also in this case we have tried to provide a very comprehensive and detailed presentation. The most important results, from a technical point of view, have been discussed in the third part of this work. Here, we have outlined the detailed computation of the contribution of two-quark operators up to order $1/m_b^3$, for arbitrary non-leptonic decays of the heavy b-quark and using the momentum representation of the quark propagator in the external gluon field. Particular emphasis has been put in describing the mixing between four-quark operators and the Darwin operator at dimensionsix, which ensures the cancellations of the infrared divergences otherwise present in the coefficients of the Darwin operator, due to the emission of a soft gluon from a light quark propagator. The contribution of the Darwin operator for non-leptonic bquark decays, has been only recently determined and found to be sizeable, hence its effect, previously neglected, might have important consequences for b-physics phenomenology. In the last part of this work, we have considered two phenomenological applications of the HQE in the charm sector, specifically the study of the inclusive decay width of D-mesons and of the Glashow-Iliopoulos-Maiani (GIM) cancellations in *D*-meson mixing. Due to the value of its mass, the charm quark sits at the boundary between the heavy- and light-quark region, and the applicability of the HQE in the charm sector is a priori questionable. In fact, both the perturbative and the power corrections might not describe a well converging series. For this reason, the charm system can be considered an important testing ground for the theoretical framework here discussed. We have then performed a comprehensive study of the structure of the HQE for the inclusive decay width of charmed mesons, including for the first time the contribution of the Darwin operator and in addition clarified some inconsistencies related to the contribution of dimension-seven four-quark operators. Our predictions appear to be consistent with the corresponding experimental pattern, albeit with large theoretical uncertainties. Given the current poor knowledge of many of the non perturbative input in the charm sector, as well as the absence of determinations of higher order perturbative and power corrections, we conclude that our numerical analysis does not show signals for a breakdown of the HQE in the charm system. Finally, we have also attempted to clarify the long standing puzzle due to the big discrepancy between the experimental determination for neutral Dmeson mixing and the corresponding HQE prediction, which might naively point at a complete failure of the HQE in the charm sector. In this respect, we have proposed a novel procedure to treat the renormalisation scale for observables affected by GIM cancellations. Our results show that the experimental value can be accommodated within the HQE, albeit again with very large theoretical uncertainties.

In conclusion, we have improved the current theoretical status of the HQE, by computing the contribution of the Darwin operator to non-leptonic decays of heavy quarks and by testing its applicability in the charm system. The same framework can in future be applied to the B system, to improve the theoretical prediction for lifetimes and mixing observables. A precise determination of e.g. $\tau(B_s)/\tau(B_d)$, could in fact increase the bounds on the size of potential new physics contributions in the decay $b \rightarrow s\tau\tau$, which are predicted by some of the current BSM models that explain the B-anomalies, or could further constrain the baryogengesis model discussed in Ref. [14].

Acknowledgements

First and outmost I thank Prof. Dr. Alexander Lenz for his precious guidance and the inestimable support provided during the last three years. I also thank him for being an inexhaustible source of energy and motivation and I consider myself very lucky to have had the possibility to learn from him.

A special thanks goes to my collaborator and dear friend Dr. Aleksey Rusov, he has deeply contributed to my technical growth and I am grateful for the chance had to work together. Moreover, I sincerely thank him for the patience and the constant support offered in the last few months while writing this thesis.

I warmly thank Prof. Dr. Michael Spannowsky and Prof. Dr. Frank Krauss for giving me the opportunity to start my PhD program at the IPPP in Durham and Dr. Daniel Maitre for the supervision during the first year there.

A big thanks to everyone met in the past four years, first in Durham and now in Siegen, for each of them has contributed in some way to this journey.

Arrived at this point, I feel obliged to also thank all the teachers and professors I have had the fortune to find on my lifelong learning path, many of them have certainly inspired me and motivated me to continue studying.

Last but not least, I thank my family, my friends, and Peter, for understanding the many times I have been absent and for taking care of me despite the distance.

Appendix A

Expansion of operators containing the heavy quark momentum

Here we discuss a general procedure to generate operators of higher dimension, bilinear in the heavy quark field, starting from expressions containing lower dimensional operators with coefficients proportional to the heavy quark momentum. For definiteness we consider the case of the *b*-quark in order to make the connection to Chapter 2 and Chapter 3 easier. We start by recalling that inside a heavy hadron, the momentum of the heavy quark can be conveniently parametrised as $p_b^{\mu} = m_b v^{\mu} + k^{\mu}$. Correspondingly the rescaled heavy quark field $b_v(x)$ is defined by

$$b(x) = e^{-im_b v \cdot x} b_v(x), \qquad (A.1)$$

where the phase factor removes the large 'kinetic' part of the heavy quark momentum so that a derivative acting on $b_v(x)$ returns only the residual component k. Consider now the following expression $p_b^{\mu} \bar{b}b$, with the b-fields evaluated at the origin x = 0. By taking into account Eq. (A.1), we can write, see Refs. [130, 131]

$$p_b^{\mu}\bar{b}b = \lim_{x \to 0} \bar{b}(x)i\partial^{\mu}b(x) = \lim_{x \to 0} \bar{b}_v(x)(m_bv^{\mu} + i\partial^{\mu})b_v(x), \qquad (A.2)$$

here the partial derivative is acting on the right. Using that in the FS gauge, see Section 1.3, the gauge field satisfies the useful property $A^{\mu}(0) = 0$, cf. Eq. (1.87), it follows that on the r.h.s. of Eq. (A.2), the partial derivative acting on $b_v(x)$ can be replaced by the corresponding covariant derivative in the limit $x \to 0$, namely

$$p_b^{\mu}\bar{b}b = \lim_{x \to 0} \bar{b}_v(x) \big(m_b v^{\mu} + i\partial^{\mu} + A^{\mu}(x) \big) b_v(x) = \bar{b}_v(m_b v^{\mu} + iD^{\mu}) b_v \,, \tag{A.3}$$

which generates the dimension-four operator $\bar{b}_v i D^\mu b_v$. If the original expression contains more than one power of the heavy quark momentum, to keep track of the order of the covariant derivatives we can symmetrise the action of the partial derivatives, e.g. the case of $p_b^\mu p_b^\nu \bar{b}b$ gives

$$p_{b}^{\mu}p_{b}^{\nu}\bar{b}b = \frac{1}{2}\lim_{x\to 0}\bar{b}(x)\left(i\partial^{\mu}i\partial^{\nu} + i\partial^{\nu}i\partial^{\mu}\right)b(x)$$
$$= \frac{1}{2}\lim_{x\to 0}\bar{b}_{v}(x)\left((m_{b}v^{\mu} + i\partial^{\mu})(m_{b}v^{\nu} + i\partial^{\nu}) + (m_{b}v^{\nu} + i\partial^{\nu})(m_{b}v^{\mu} + i\partial^{\mu})\right)b_{v}(x).$$
(A.4)

Here again we would like to replace the action of the partial derivative with that of the covariant derivative. Note that in this case, apart from terms containing the gauge field on the most left, which vanish when taking the limit $x \to 0$, we introduce also terms with derivatives of the gauge field evaluated at the origin, that are in general non zero. From Eq. (1.87), it follows that $\partial^{\mu}A^{\nu}(0) = (1/2)G^{\mu\nu}(0)$, however, due to the antisymmetry of the gluon field strength tensor, these contributions cancel in the symmetric combination in Eq. (A.4), namely

$$\frac{1}{2} \bar{b}_{v} \left(iD^{\mu}iD^{\nu} + iD^{\nu}iD^{\mu} \right) b_{v}$$

$$= \frac{1}{2} \lim_{x \to 0} \bar{b}_{v}(x) \left(\left(i\partial^{\mu}i\partial^{\nu} + A^{\mu}(x)i\partial^{\nu} + i\partial^{\mu}A^{\nu}(x) + A^{\nu}(x)i\partial^{\mu} + A^{\mu}(x)A^{\nu}(x) \right) \right) + \left(i\partial^{\nu}i\partial^{\mu} + A^{\nu}(x)i\partial^{\mu} + i\partial^{\nu}A^{\mu}(x) + A^{\mu}(x)i\partial^{\nu} + A^{\nu}(x)A^{\mu}(x) \right) \right) b_{v}(x)$$

$$= \frac{1}{2} \bar{b}_{v} \left(i\partial^{\mu}i\partial^{\nu} + i\partial^{\nu}i\partial^{\mu} + \frac{i}{2}G^{\mu\nu}(0) + \frac{i}{2}G^{\nu\mu}(0) \right) b_{v}$$

$$= \frac{1}{2} \bar{b}_{v} \left(i\partial^{\mu}i\partial^{\nu} + i\partial^{\nu}i\partial^{\mu} + i\partial^{\nu}i\partial^{\mu} \right) b_{v}.$$
(A.5)

We then obtain that

$$p_b^{\mu} p_b^{\nu} \bar{b}b = \frac{1}{2} b_v \Big\{ (m_b v^{\mu} + iD^{\mu}), (m_b v^{\nu} + iD^{\nu}) \Big\} b_v \,, \tag{A.6}$$

where the curly brackets denote the anticommutator. Notice that we generate operators of dimension-four and dimension-five with respectively one and two covariant derivatives. In the case of three powers of the heavy quark momentum, $p_b^{\mu} p_b^{\nu} p_b^{\rho} \bar{b} b$, in rewriting this expression in terms of partial derivatives we consider all the permutations in the three Lorentz indices $\mu\nu\rho$. The replacement $i\partial^{\mu} \rightarrow iD^{\mu}$ again follows from the symmetric combination of antisymmetric tensors, since Eq. (1.87) now gives $\partial^{\mu}\partial^{\nu}A^{\rho}(0) = (1/2)\partial^{\mu}G^{\nu\rho}(0) + (1/3)D^{\mu}G^{\nu\rho}(0)$. For *n*-powers of the heavy quark momentum, we use the general expression

$$p_b^{\mu_1} \dots p_b^{\mu_n} \bar{b}b = \frac{1}{n!} \sum_{\sigma \in S_n} \bar{b}_v \left(m_b v + iD \right)^{\sigma(\mu_1)} \dots \left(m_b v + iD \right)^{\sigma(\mu_n)} b_v , \qquad (A.7)$$

where S_n is the group of all permutations of n elements. Furthermore consider the following function

$$f(r) = a + b(r)r + c(r)r^{2} + d(r)r^{3} + e(r)r^{4}, \qquad (A.8)$$

with the dependence on the heavy quark momentum contained in the argument $r = m^2/p_b^2$. In this case, we can use that $p_b^2 = m_b^2 (1 + X)$, where

$$X = 2 \frac{v \cdot k}{m_b} + \frac{k^2}{m_b^2} \ll 1 , \qquad (A.9)$$

and expand Eq. (A.8) in series, i.e.

$$f(r) = \sum_{n=0}^{\infty} g_n(\rho) (-X)^n, \qquad (A.10)$$

here $g_0(\rho) = f(\rho)$ and the dimensionless mass parameter $\rho = m^2/m_b^2$. The series in Eq. (A.10) can be truncated at a certain value of n, leading to corrections up to order $1/m_b^{2n}$. Expressing X in Eq. (A.10) back in terms of the four-momentum, we then obtain

$$f(r) = \sum_{n=0}^{\infty} g_n(\rho) \left(1 - \frac{p_b^2}{m_b^2}\right)^n,$$

from which it follows that $f(r)\bar{b}b$ can be expanded according to Eq. (A.7). Finally, for the computation of the coefficient of the Darwin operator discussed in Chapter 3,

we need to expand also the structure $p_b^{\rho} \bar{b} G^{\mu\nu} b$. In this case we write

$$p_{b}^{\rho}\bar{b}G^{\mu\nu}b = \frac{1}{2}\lim_{x\to 0} \left(\bar{b}(x)(-i\vec{\partial}^{\rho})G^{\mu\nu}(x)b(x) + \bar{b}(x)G^{\mu\nu}(x)(i\vec{\partial}^{\rho})b(x)\right)$$

$$= \frac{1}{2}\lim_{x\to 0} \left(\bar{b}_{v}(x)(m_{b}v^{\rho} - i\vec{\partial}^{\rho})G^{\mu\nu}(x)b_{v}(x) + \bar{b}_{v}(x)G^{\mu\nu}(x)(m_{b}v^{\rho} + i\vec{\partial}^{\rho})b_{v}(x)\right)$$

$$= \frac{1}{2}\lim_{x\to 0} \left(\bar{b}_{v}(x)(m_{b}v^{\rho} + i\vec{\partial}^{\rho})G^{\mu\nu}(x)b_{v}(x) + \bar{b}_{v}(x)G^{\mu\nu}(x)(m_{b}v^{\rho} + i\vec{\partial}^{\rho})b_{v}(x)\right)$$

$$= \frac{1}{2}\left(\bar{b}_{v}(m_{b}v^{\rho} + iD^{\rho})G^{\mu\nu}b_{v} + \bar{b}_{v}G^{\mu\nu}(m_{b}v^{\rho} + iD^{\rho})b_{v}\right), \qquad (A.11)$$

where in the third step we have applied the chain rule for the derivative operator acting on the left and dropped the corresponding term with a total derivative since it does not contribute in forward matrix elements with zero momentum transfer.

In general, by expressing powers of the heavy quark momentum in terms of operators acting on the heavy quark field, we generate a set of higher dimensional operators with multiple covariant derivatives. Specifically, in order to compute the contributions discussed in Chapter 2 and Chapter 3, it is sufficient to consider the expansions up to three covariant derivatives. The operators obtained in this way, must be then evaluated between external B meson states for the calculation of the total decay width. This can be easily done by taking into account the results presented in Ref. [105], in which the complete parametrisation of these matrix elements, up to order $1/m_b^4$ was derived.

Appendix B

Results for one-loop integrals in dimensional regularisation

Here we list some useful results for the computation of one-loop integrals in dimensional regularisation [47–50] with $D = 4 - 2\epsilon$. All the expressions presented can be found in standard QFT textbooks like Refs. [38,81,285]. Note that in the following we use the notation

$$\mu^{2\varepsilon} \int \frac{d^D l}{(2\pi)^D} \equiv \int_l \,. \tag{B.1}$$

The scalar tadpole and bubble integrals are respectively given by

$$\int_{l} \frac{1}{(l^2 - m^2 + i\varepsilon)} = \mathcal{A}_0(m^2), \qquad (B.2)$$

and

$$\int_{l} \frac{1}{(l^2 - m_1^2 + i\varepsilon)((l - p)^2 - m_2^2 + i\varepsilon)} = \mathcal{B}_0(p^2, m_1^2, m_2^2), \qquad (B.3)$$

with

$$\mathcal{A}_0(m^2) = \frac{i m^2}{16\pi^2} \left(\frac{1}{\epsilon} - \gamma_E + \log(4\pi) + 1 + \log\left(\frac{m^2}{\mu^2}\right) \right) + \mathcal{O}(\epsilon) , \qquad (B.4)$$

$$\mathcal{B}_0(p^2, m_1^2, m_2^2) = \frac{i}{16\pi^2} \left(\frac{1}{\epsilon} - \gamma_E + \log(4\pi) + f(p^2, m_1^2, m_2^2) \right) + \mathcal{O}(\epsilon) , \qquad (B.5)$$

where γ_E is the Euler-Mascheroni constant and

$$f(p^2, m_1^2, m_2^2) = -\int_0^1 dx \, \log\left(\frac{(1-x)m_1^2 + xm_2^2 - x(1-x)p^2 - i\varepsilon}{\mu^2}\right), \qquad (B.6)$$

Note that as $\varepsilon \to 0$, the function $f(p^2, m_1^2, m_2^2)$ is real and analytic in $s \in \mathbb{C}$, with $\operatorname{Re}(s) = p^2$, only for $p^2 < (m_1 + m_2)^2 \lor 0 \leqslant x \leqslant 1$, while it exhibits a branch point when the argument of the logarithm in the integrand on the r.h.s of Eq. (B.6) vanishes. This leads to a cut in the complex *s*-plane, in correspondence of the points $p^2 > (m_1 + m_2)^2$ in which the argument of logarithm becomes negative. By introducing ε , the computation of the integral, performed above and below the branch cut, gives see e.g. the lecture notes [286, 287]

Disc
$$f(p^2, m_1^2, m_2^2) = 2i\pi \frac{\sqrt{\lambda(p^2, m_1^2, m_2^2)}}{p^2} \theta \left(p^2 - \left(m_1 + m_2\right)^2\right)$$
, (B.7)

and $\theta(x)$ is the Heaviside function. From Eqs. (B.4)-(B.7) it follows that

$$\operatorname{Im} i\mathcal{A}_0(m^2) = 0, \qquad (B.8)$$

while

$$\operatorname{Im} i\mathcal{B}_0(p^2, m_1^2, m_2^2) = -\frac{1}{16\pi^2} \operatorname{Im} f(p^2, m_1^2, m_2^2) = -\frac{1}{16\pi} \frac{\sqrt{\lambda(p^2, m_1^2, m_2^2)}}{p^2}, \quad (B.9)$$

for $p^2 \ge (m_1 + m_2)^2$. The corresponding one- and two-point one-loop tensor integrals of rank r can be computed, using the Passarino-Veltman reduction algorithm [155], as a linear combination of tensors of the same rank, built from the metric $g^{\mu\nu}$ and the external momentum p^{μ} , with coefficients proportional to the scalar integrals in Eqs. (B.2), (B.3). Let us consider explicitly the cases r = 1, 2. Rank-1 integrals can only depend on the four-vector p^{μ} , namely

$$\int_{l} \frac{l^{\rho}}{(l^2 - m_1^2 + i\varepsilon)((l-p)^2 - m_2^2 + i\varepsilon)} = B_{11} p^{\rho}, \qquad (B.10)$$

where the coefficient B_{11} is obtained contracting both sides of Eq. (B.10) with p_{ρ} i.e.

$$B_{11} = \frac{1}{p^2} \int_l \frac{l \cdot p}{(l^2 - m_1^2 + i\varepsilon)((l - p)^2 - m_2^2 + i\varepsilon)}.$$
 (B.11)

By substituting the identity

$$l \cdot p = \frac{1}{2} \left(l^2 + p^2 - (l - p)^2 + m_1^2 - m_1^2 + m_2^2 - m_2^2 \right),$$
(B.12)

into Eq. (B.11), evidently yields

$$B_{11} = \frac{1}{2p^2} \int_l \frac{(l^2 - m_1^2) - ((l - p)^2 - m_2^2) + (p^2 + m_1^2 - m_2^2)}{(l^2 - m_1^2 + i\varepsilon)((l - p)^2 - m_2^2 + i\varepsilon)}$$
$$= \frac{1}{2p^2} \left(\mathcal{A}_0(m_2^2) - \mathcal{A}_0(m_1^2) + \left(p^2 + m_1^2 - m_2^2\right) \mathcal{B}_0(p^2, m_1^2, m_2^2) \right), \qquad (B.13)$$

where we have taken into account Eqs. (B.2), (B.3). In the case of rank-2 integrals, we can build two independent tensors, $g^{\rho\sigma}$ and $p^{\rho}p^{\sigma}$, namely

$$\int_{l} \frac{l^{\rho} l^{\sigma}}{(l^2 - m_1^2 + i\varepsilon)((l-p)^2 - m_2^2 + i\varepsilon)} = B_{21} p^{\rho} p^{\sigma} + B_{22} p^2 g^{\rho\sigma}, \qquad (B.14)$$

here, the coefficients B_{21} , B_{22} , are the solutions of the system of two equations obtained contracting both sides of Eq. (B.14) respectively with p_{ρ} and $g_{\rho\sigma}$, i.e.

$$\int_{l} \frac{l^2}{(l^2 - m_1^2 + i\varepsilon)((l-p)^2 - m_2^2 + i\varepsilon)} = B_{21} p^2 + p^2 B_{22} D, \qquad (B.15)$$

$$\int_{l} \frac{(l \cdot p) p^{\sigma}}{(l^2 - m_1^2 + i\varepsilon)((l - p)^2 - m_2^2 + i\varepsilon)} = B_{21} p^2 p^{\sigma} + p^2 B_{22} p^{\sigma}.$$
 (B.16)

Adding and subtracting m_1^2 in the numerator of Eq. (B.15) and substituting Eq. (B.12) into Eq. (B.16), we obtain ¹

$$\left(B_{21} + D B_{22}\right)p^2 = \mathcal{A}_0(m_2^2) + m_1^2 \mathcal{B}_0(p^2, m_1^2, m_2^2), \qquad (B.17)$$

$$\left(B_{21} + B_{22}\right)p^2 = \frac{1}{2}\left(\mathcal{A}_0(m_2^2) + \left(p^2 + m_1^2 - m_2^2\right)B_{11}\right),\tag{B.18}$$

solved by

¹Note that integrals of odd functions of l^{μ} vanish due to parity.

$$B_{21} = \frac{1}{D-1} \frac{1}{2p^2} \left(\mathcal{A}_0(m_2^2)(D-2) - 2m_1^2 \mathcal{B}_0(p^2, m_1^2, m_2^2) + \frac{D(m_1^2 - m_2^2 + p^2) \left(\mathcal{A}_0(m_2^2) - \mathcal{A}_0(m_1^2) + (p^2 + m_1^2 - m_2^2) \mathcal{B}_0(p^2, m_1^2, m_2^2) \right)}{2p^2} \right),$$
(B.19)

and

$$B_{22} = \frac{1}{D-1} \frac{1}{2p^2} \left(\mathcal{A}_0(m_2^2) + 2m_1^2 \mathcal{B}_0(p^2, m_1^2, m_2^2) - \frac{(m_1^2 - m_2^2 + p^2) \left(\mathcal{A}_0(m_2^2) - \mathcal{A}_0(m_1^2) + (p^2 + m_1^2 - m_2^2) \mathcal{B}_0(p^2, m_1^2, m_2^2) \right)}{2p^2} \right),$$
(B.20)

where Eq. (B.13) has been used. The expressions in Eq. (B.2)-(B.20) are needed for the computation of the four-quark operators contribution discussed in Chapter 2.

Finally we derive some useful results used in the calculation of the one-loop diagram in Figure 3.4, describing the mixing of four-quark operators with two-quark operators at dimension-six, discussed in Chapter 3. Consider the following scalar integrals

$$\int_{l} \frac{1}{(l^2 - m^2 + i\varepsilon)^2} = \mathcal{I}_0(m^2), \qquad (B.21)$$

and

$$\int_{l} \frac{l^2}{(l^2 - m^2 + i\varepsilon)^3} = \frac{D}{4} \mathcal{I}_0(m^2), \qquad (B.22)$$

with

$$\mathcal{I}_0(m^2) = \frac{i}{16\pi^2} \left(\frac{1}{\epsilon} - \gamma_E + \log(4\pi) + \log\left(\frac{\mu^2}{m^2}\right) \right) + \mathcal{O}(\epsilon) \,. \tag{B.23}$$

Rank-2 integrals, since now there is no external momentum they can depend on, can be only parametrised by the metric tensor i.e.

$$\int_{l} \frac{l^{\mu} l^{\nu}}{(l^2 - m^2 + i\varepsilon)^3} = g(m^2) g^{\mu\nu}, \qquad (B.24)$$

where the coefficient $g(m^2)$ is obtained by contracting both sides of Eq. (B.24) with $g_{\mu\nu}$. This gives

$$\int_{l} \frac{l^2}{(l^2 - m^2 + i\varepsilon)^3} = D g(m^2), \qquad (B.25)$$

and from Eq. (B.22) it then follows that

$$g(m^2) = \frac{1}{4} \mathcal{I}_0(m^2).$$
 (B.26)
Appendix C

Decomposition of tensor integrals

In the following section we describe how to reduce rank-r integrals to a linear combination of tensors of rank-r with coefficients given by scalar integrals. We consider explicitly integrals of the sunset type encountered in Chapter 3 although the same procedure can be generalised to any tensor integral without loss of generality. We work again in dimensional regularisation with $D = 4 - 2\varepsilon$ and to simplify the notation we define

$$\int_{l_1 l_2} \equiv \int_{l_1} \int_{l_2} \frac{1}{(l_1^2 - m_1^2 + i\varepsilon) (l_2^2 - m_2^2 + i\varepsilon)((l_1 + l_2 - p)^2 - m_3^2 + i\varepsilon)}, \quad (C.1)$$

with \int_l defined as in Eq. (B.1). Rank-1 integrals can only be parametrised in terms of the external momentum p^{μ} , hence

$$\int_{l_1 l_2} l_j^{\mu} = a_j(p^2, m_i^2) \, p^{\mu} \,, \tag{C.2}$$

where j = 1, 2 and the coefficient $a_j(p^2, m_i^2)$ is obtained by contracting both sides of Eq. (C.2) with p_{μ} i.e.

$$a_j(p^2, m_i^2) = \frac{1}{p^2} \int_{l_1 l_2} (l_j \cdot p) \,.$$
 (C.3)

In the case of rank-2 integrals, we can build two independent tensors, namely $p^2 g^{\mu\nu}$ and $p^{\mu}p^{\nu}$, hence

$$\int_{l_1 l_2} l_j^{\mu} l_k^{\nu} = b_{jk}(p^2, m_i^2) \, p^2 g^{\mu\nu} + c_{jk}(p^2, m_i^2) \, p^{\mu} p^{\nu} \,, \tag{C.4}$$

where j, k = 1, 2 and the coefficients $b_{jk}(p^2, m_i^2), c_{jk}(p^2, m_i^2)$ are the solutions of the

system of two equations obtained by contracting both sides of Eq. (C.4) respectively with $g_{\mu\nu}$ and $p_{\mu}p_{\nu}$ i.e.

$$b_{jk}(p^2, m_i^2) = \frac{1}{D-1} \frac{1}{p^4} \int_{l_1 l_2} \left((l_j \cdot l_k) p^2 - (l_j \cdot p) (l_k \cdot p) \right),$$
(C.5)

$$c_{jk}(p^2, m_i^2) = \frac{1}{D-1} \frac{1}{p^4} \int_{l_1 l_2} \left(D \left(l_j \cdot p \right) (l_k \cdot p) - (l_j \cdot l_k) p^2 \right).$$
(C.6)

In the case of rank-3 integrals, there are four independent tensors $p^2 g^{\{\mu\nu} p^{\rho\}}$ with the curly brackets denoting all possible permutation of the Lorentz indices, hence

$$\int_{l_1 l_2} l_j^{\mu} l_k^{\nu} l_m^{\rho} = d_{1,jkm}(p^2, m_i^2) p^2 g^{\mu\nu} p^{\rho} + d_{2,jkm}(p^2, m_i^2) p^2 g^{\rho\mu} p^{\nu} + d_{3,jkm}(p^2, m_i^2) p^2 g^{\nu\rho} p^{\mu} + e_{jkm}(p^2, m_i^2) p^{\mu} p^{\nu} p^{\rho} ,$$
(C.7)

where j, k, m = 1, 2 and the coefficients $d_{r,jkm}(p^2, m_i^2)$, $e_{jkm}(p^2, m_i^2)$ are the solutions of the system of four equations obtained by contracting both sides of Eq. (C.7) respectively with $g_{\mu\nu}p_{\rho}$, $g_{\rho\mu}p_{\nu}$, $g_{\nu\rho}p_{\mu}$ and $p_{\mu}p_{\nu}p_{\rho}$. Defining for simplicity

$$d_{jkm}(p^2, m_i^2) = \frac{1}{D-1} \frac{1}{p^6} \int_{l_1 l_2} \left((l_j \cdot l_k)(l_m \cdot p)p^2 - (l_j \cdot p)(l_k \cdot p)(l_m \cdot p) \right), \quad (C.8)$$

leads to

$$d_{1,jkm}(p^2, m_i^2) = d_{jkm}(p^2, m_i^2), \qquad (C.9)$$

$$d_{2,jkm}(p^2, m_i^2) = d_{kmj}(p^2, m_i^2), \qquad (C.10)$$

$$d_{3,jkm}(p^2, m_i^2) = d_{mjk}(p^2, m_i^2), \qquad (C.11)$$

while

$$e_{jkm}(p^2, m_i^2) = \frac{1}{D-1} \frac{1}{p^6} \int_{l_1 l_2} \left((D+2) (l_j \cdot p) (l_k \cdot p) (l_m \cdot p) - (l_j \cdot l_k) (l_m \cdot p) p^2 - (l_k \cdot l_m) (l_j \cdot p) p^2 - (l_m \cdot l_j) (l_k \cdot p) p^2 \right).$$
(C.12)

Finally in the case of rank-4 integrals there are 10 independent tensor structures we can build, namely $p^4 g^{\{\mu\nu}g^{\rho\sigma\}}$, $p^2 g^{\{\mu\nu}p^{\rho}p^{\sigma\}}$, and $p^{\mu}p^{\nu}p^{\rho}p^{\sigma}$, hence

$$\int_{l_{1}l_{2}} l_{j}^{\mu} l_{k}^{\nu} l_{m}^{\rho} l_{n}^{\sigma} = f_{1,jkmn}(p^{2}, m_{i}) p^{4} g^{\mu\nu} g^{\rho\sigma} + f_{2,jkmn}(p^{2}, m_{i}) p^{4} g^{\mu\rho} g^{\nu\sigma}
+ f_{3,jkmn}(p^{2}, m_{i}) p^{4} g^{\mu\sigma} g^{\nu\rho} + g_{1,jkmn}(p^{2}, m_{i}) p^{2} g^{\mu\nu} p^{\rho} p^{\sigma}
+ g_{2,jkmn}(p^{2}, m_{i}) p^{2} g^{\mu\rho} p^{\nu} p^{\sigma} + g_{3,jkmn}(p^{2}, m_{i}) p^{2} g^{\mu\sigma} p^{\nu} p^{\rho}
+ g_{4,jkmn}(p^{2}, m_{i}) p^{2} g^{\nu\rho} p^{\mu} p^{\sigma} + g_{5,jkmn}(p^{2}, m_{i}) p^{2} g^{\nu\sigma} p^{\mu} p^{\rho}
+ g_{6,jkmn}(p^{2}, m_{i}) p^{2} g^{\rho\sigma} p^{\mu} p^{\nu} + h_{jkmn}(p^{2}, m_{i}) p^{\mu} p^{\nu} p^{\rho} p^{\sigma},$$
(C.13)

where the coefficients $f_{r,jmk}(p^2, m_i)$, $g_{s,jmk}(p^2, m_i)$ and $h_{jmk}(p^2, m_i)$ are the solutions of the system of 10 equations obtained by contracting both sides of Eq. (C.13) with each of the tensor structures. Defining

$$f_{jkmn}(p^{2}, m_{i}) = \frac{1}{(D^{2} - 1)(D - 2)} \frac{1}{p^{8}} \int_{l_{1}l_{2}} \left[D\left((l_{j} \cdot p)(l_{k} \cdot p)(l_{m} \cdot p)(l_{n} \cdot p) + (l_{j} \cdot l_{k})\left((l_{m} \cdot l_{n}) - (l_{m} \cdot p)(l_{n} \cdot p) \right) - (l_{j} \cdot p)(l_{k} \cdot p)(l_{m} \cdot l_{n}) \right) \right. \\ \left. + (l_{j} \cdot l_{m})(l_{k} \cdot p)(l_{n} \cdot p) + (l_{j} \cdot l_{n})(l_{k} \cdot p)(l_{m} \cdot p) + (l_{j} \cdot p)(l_{m} \cdot p)(l_{m} \cdot p)(l_{k} \cdot l_{n}) + (l_{j} \cdot p)(l_{m} \cdot p)(l_{k} \cdot l_{n}) + (l_{j} \cdot l_{n})(l_{k} \cdot l_{n}) + (l_{j} \cdot l_{n})(l_{k} \cdot l_{m}) + 2(l_{j} \cdot p)(l_{k} \cdot p)(l_{m} \cdot p)(l_{n} \cdot p) \right],$$

$$\left. \left. \left. \left((l_{j} \cdot l_{m})(l_{k} \cdot l_{n}) + (l_{j} \cdot l_{n})(l_{k} \cdot l_{m}) + 2(l_{j} \cdot p)(l_{k} \cdot p)(l_{m} \cdot p)(l_{n} \cdot p) \right) \right] \right],$$

$$\left. \left((C.14) \right) \right\}$$

and

$$g_{jkmn}(p^{2}, m_{i}) = -\frac{1}{(D^{2} - 1)(D - 2)} \frac{1}{p^{8}} \int_{l_{1}l_{2}} \left[-D^{2}(l_{j} \cdot l_{k})(l_{m} \cdot p)(l_{n} \cdot p) + D\left(D(l_{j} \cdot p)(l_{k} \cdot p)(l_{m} \cdot p)(l_{n} \cdot p) + (l_{j} \cdot l_{k})(l_{m} \cdot l_{n}) - (l_{j} \cdot p)(l_{k} \cdot p)(l_{m} \cdot l_{n}) \right) \right. \\ \left. + 2(l_{j} \cdot l_{k})(l_{m} \cdot p)(l_{n} \cdot p) + (l_{j} \cdot l_{m})(l_{k} \cdot p)(l_{n} \cdot p) + (l_{j} \cdot l_{n})(l_{k} \cdot p)(l_{m} \cdot p) \right. \\ \left. + (l_{j} \cdot p)(l_{n} \cdot p)(l_{k} \cdot l_{m}) + (l_{j} \cdot p)(l_{m} \cdot p)(l_{k} \cdot l_{n}) \right. \\ \left. - \left((l_{j} \cdot l_{m})(l_{k} \cdot l_{n}) + (l_{j} \cdot l_{n})(l_{k} \cdot l_{m}) + 4(l_{j} \cdot p)(l_{m} \cdot p)(l_{m} \cdot p)(l_{n} \cdot p) \right) \right],$$

$$\left. (C.15) \right]$$

it follows that

$$f_{1,jkmn}(p^2, m_i) = f_{jkmn}(p^2, m_i),$$
 (C.16)

$$f_{2,jkmn}(p^2, m_i) = f_{jmkn}(p^2, m_i),$$
 (C.17)

$$f_{3,jkmn}(p^2, m_i) = f_{jnkm}(p^2, m_i),$$
 (C.18)

$$g_{1,jkmn}(p^2, m_i) = g_{jkmn}(p^2, m_i),$$
 (C.19)

$$g_{2,jkmn}(p^2, m_i) = g_{jmkn}(p^2, m_i),$$
 (C.20)

$$g_{3,jkmn}(p^2, m_i) = g_{jnkm}(p^2, m_i),$$
 (C.21)

$$g_{4,jkmn}(p^2, m_i) = g_{kmjn}(p^2, m_i),$$
 (C.22)

$$g_{5,jkmn}(p^2, m_i) = g_{knjm}(p^2, m_i),$$
 (C.23)

$$g_{6,jkmn}(p^2, m_i) = g_{mnjk}(p^2, m_i),$$
 (C.24)

while

$$h_{jkmn}(p^{2}, m_{i}) = -\frac{1}{D^{2} - 1} \frac{1}{p^{8}} \int_{l_{1}l_{2}} \left[-(D + 4)(D + 2)(l_{j} \cdot p)(l_{k} \cdot p)(l_{m} \cdot p)(l_{n} \cdot p) + (D + 2)\left((l_{j} \cdot l_{k})(l_{m} \cdot p)(l_{n} \cdot p) + (l_{j} \cdot l_{m})(l_{k} \cdot p)(l_{n} \cdot p) + (l_{j} \cdot l_{n})(l_{k} \cdot p)(l_{n} \cdot p) + (l_{j} \cdot l_{n})(l_{j} \cdot p)(l_{n} \cdot p) + (l_{k} \cdot l_{m})(l_{j} \cdot p)(l_{n} \cdot p) + (l_{k} \cdot l_{n})(l_{j} \cdot p)(l_{k} \cdot p)\right) - \left((l_{j} \cdot l_{k})(l_{m} \cdot l_{n}) + (l_{j} \cdot l_{m})(l_{k} \cdot l_{n}) + (l_{j} \cdot l_{n})(l_{k} \cdot l_{m})\right)\right].$$
(C.25)

Appendix D

Expressions for the coefficients $C_0^{(q_1\bar{q}_2q_3)}$ and $C_{G,mn}^{(q_1\bar{q}_2q_3)}$

Here we present the analytic expressions for the coefficients $C_0^{(q_1\bar{q}_2q_3)}$ and $C_{G,mn}^{(q_1\bar{q}_2q_3)}$ with mn = 11, 12, 22, introduced in Eq. (3.23). They read respectively

$$\mathcal{C}_{0}^{(u\bar{u}d)} = 1, \qquad \mathcal{C}_{G,11}^{(u\bar{u}d)} = -\frac{3}{2} = \mathcal{C}_{G,22}^{(u\bar{u}d)}, \qquad \mathcal{C}_{G,12}^{(u\bar{u}d)} = -\frac{19}{2}, \qquad (D.1)$$

$$\mathcal{C}_0^{(u\bar{c}s)} = 1 - 8\rho - 12\rho^2 \log(\rho) + 8\rho^3 - \rho^4, \qquad (D.2)$$

$$\mathcal{C}_{G,11}^{(u\bar{c}s)} = -\frac{1}{2} \left(3 - 8\rho + 12\rho^2 \log(\rho) + 24\rho^2 - 24\rho^3 + 5\rho^4 \right) = \mathcal{C}_{G,22}^{(u\bar{c}s)}, \tag{D.3}$$

$$\mathcal{C}_{G,12}^{(u\bar{c}s)} = -\frac{1}{2} \left(19 + 16\rho + 12\rho(\rho+4) \log(\rho) - 24\rho^2 - 16\rho^3 + 5\rho^4 \right) , \qquad (D.4)$$

$$\mathcal{C}_0^{(c\bar{u}d)} = 1 - 8\rho - 12\rho^2 \log(\rho) + 8\rho^3 - \rho^4, \qquad (D.5)$$

$$\mathcal{C}_{G,11}^{(c\bar{u}d)} = -\frac{1}{2} \left(3 - 8\rho + 12\rho^2 \log(\rho) + 24\rho^2 - 24\rho^3 + 5\rho^4 \right) = \mathcal{C}_{G,22}^{(c\bar{u}d)}, \quad (D.6)$$

$$\mathcal{C}_{G,12}^{(c\bar{u}d)} = -\frac{1}{2} \left(19 - 56\rho + 12\rho^2 \log(\rho) + 72\rho^2 - 40\rho^3 + 5\rho^4 \right) , \tag{D.7}$$

$$\mathcal{C}_{0}^{(c\bar{c}s)} = \sqrt{1 - 4\rho} \left(1 - 14\rho - 2\rho^{2} - 12\rho^{3} \right) + 24\rho^{2} (1 - \rho^{2}) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right),$$
(D.8)

$$\mathcal{C}_{G,11}^{(c\bar{c}s)} = -\frac{1}{2} \left[\sqrt{1 - 4\rho} \left(3 - 10\rho + 10\rho^2 + 60\rho^3 \right) - 24\rho^2 (1 - 5\rho^2) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) \right] = \mathcal{C}_{G,22}^{(c\bar{c}s)}, \quad (D.9)$$

$$\mathcal{C}_{G,12}^{(c\bar{c}s)} = -\frac{1}{2} \left[\sqrt{1 - 4\rho} \left(19 - 2\rho + 58\rho^2 + 60\rho^3 \right) - 24\rho \left(2 + \rho - 4\rho^2 - 5\rho^3 \right) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) \right], \quad (D.10)$$

Appendix E

Expressions for the divergent functions $\mathcal{D}_{mn}^{(q_1\bar{q}_2q_3)}$

Here we list the expressions for the divergent functions $\mathcal{D}_{nm}^{(q_1\bar{q}_2q_3)}$ given in Eq. (3.26).

$$\mathcal{D}_{11}^{(u\bar{u}d)} = 8 \log\left(\frac{m_u^2}{m_b^2}\right) \,,\tag{E.1}$$

$$\mathcal{D}_{12}^{(u\bar{u}d)} = 8 \left[\log \left(\frac{m_d^2}{m_b^2} \right) - \log \left(\frac{m_u^2}{m_b^2} \right) \right], \tag{E.2}$$

$$\mathcal{D}_{22}^{(u\bar{u}d)} = 8 \log\left(\frac{m_d^2}{m_b^2}\right) \,, \tag{E.3}$$

$$\mathcal{D}_{11}^{(u\bar{c}s)} = 8 \left(1 - \rho\right)^2 \left(1 + \rho\right) \log\left(\frac{m_u^2}{m_b^2}\right) \,, \tag{E.4}$$

$$\mathcal{D}_{12}^{(u\bar{c}s)} = 8\left(1-\rho\right)^2 \left(1+\rho\right) \left[\log\left(\frac{m_u^2}{m_b^2}\right) + \log\left(\frac{m_s^2}{m_b^2}\right)\right],$$
(E.5)

$$\mathcal{D}_{22}^{(u\bar{c}s)} = 8 \left(1-\rho\right)^2 \left(1+\rho\right) \log\left(\frac{m_s^2}{m_b^2}\right) \,, \tag{E.6}$$

$$\mathcal{D}_{12}^{(c\bar{u}d)} = -16\left(1-\rho\right)^2 \log\left(\frac{m_u^2}{m_b^2}\right) + 8\left(1-\rho\right)^2 (1+\rho) \log\left(\frac{m_d^2}{m_b^2}\right), \quad (E.7)$$

$$\mathcal{D}_{22}^{(c\bar{u}d)} = 8\left(1-\rho\right)^2 (1+\rho) \log\left(\frac{m_d^2}{m_b^2}\right) \,, \tag{E.8}$$

$$\mathcal{D}_{12}^{(c\bar{c}s)} = 8\sqrt{1-4\rho} \log\left(\frac{m_s^2}{m_b^2}\right) \,,\tag{E.9}$$

$$\mathcal{D}_{22}^{(c\bar{c}s)} = 8\sqrt{1-4\rho} \log\left(\frac{m_s^2}{m_b^2}\right) \,. \tag{E.10}$$

Appendix F

Coefficients of the Darwin operator for the charm system

The coefficients $C_{\rho_D,mn}^{(q_1q_2)}$ of the Darwin operator corresponding to the $c \to q_1 \bar{q}_2 u$ decya, needed for the analysis of *D*-meson decays, including the full *s*-quark mass dependence with $\rho = m_s^2/m_c^2$, are given by the following expressions

$$\mathcal{C}_{\rho_D,11}^{(d\bar{d})} = 6, \qquad \mathcal{C}_{\rho_D,12}^{(d\bar{d})} = -\frac{34}{3}, \qquad \mathcal{C}_{\rho_D,22}^{(d\bar{d})} = 6,$$
(F.1)

$$\mathcal{C}_{\rho_D,11}^{(d\bar{s})} = \frac{2}{3}(1-\rho) \bigg[9 + 11\rho - 12\rho^2 \log(\rho) - 24\left(1-\rho^2\right) \log(1-\rho) - 25\rho^2 + 5\rho^3 \bigg],$$
(F.2)

$$\mathcal{C}_{\rho_D,12}^{(d\bar{s})} = -\frac{2}{3} \bigg[17 + 12\rho \left(5 + 2\rho - 2\rho^2 \right) \log(\rho) + 48(1-\rho)(1-\rho^2) \log(1-\rho) - 26\rho + 18\rho^2 - 38\rho^3 + 5\rho^4 \bigg],$$
(F.3)

$$\mathcal{C}_{\rho_D,22}^{(d\bar{s})} = \frac{2}{3}(1-\rho) \bigg[9 + 11\rho - 12\rho^2 \log(\rho) - 24\left(1-\rho^2\right) \log(1-\rho) - 25\rho^2 + 5\rho^3 \bigg],$$
(F.4)

$$\mathcal{C}_{\rho_D,11}^{(s\bar{d})} = \frac{2}{3} \left[9 - 16\rho - 12\rho^2 + 16\rho^3 - 5\rho^4 \right],\tag{F.5}$$

$$\mathcal{C}_{\rho_D,12}^{(s\bar{d})} = -\frac{2}{3} \bigg[17 + 12\,\rho^2 \,(3-\rho)\log(\rho) - 24(1-\rho)^3\log(1-\rho) - 50\rho + 90\rho^2 - 54\rho^3 + 5\rho^4 \bigg], \quad (F.6)$$

$$\mathcal{C}_{\rho_D,22}^{(s\bar{d})} = \frac{2}{3} (1-\rho) \bigg[9 + 11\rho - 12\rho^2 \log(\rho) - 24 \left(1-\rho^2\right) \log(1-\rho) - 25\rho^2 + 5\rho^3 \bigg],$$
(F.7)

$$\mathcal{C}_{\rho_D,11}^{(s\bar{s})} = \frac{2}{3} \Biggl[\sqrt{1 - 4\rho} \left(17 + 8\rho - 22\rho^2 - 60\rho^3 \right) - 4 \left(2 - 3\rho + \rho^3 \right) + - 12 \left(1 - \rho - 2\rho^2 + 2\rho^3 + 10\rho^4 \right) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) - 12 \left(1 - \rho \right) (1 - \rho^2) \log(\rho) \Biggr],$$
(F.8)

$$\mathcal{C}_{\rho_D,12}^{(s\bar{s})} = \frac{2}{3} \Biggl[\sqrt{1 - 4\rho} \left(-33 + 24 \log(\rho) - 24 \log(1 - 4\rho) + 46\rho - 106\rho^2 - 60\rho^3 \right) + 12 \left(3 - 2\rho + 4\rho^2 - 16\rho^3 - 10\rho^4 \right) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) + 4 \left(1 - \rho \right)^2 \left(4 + 3(1 - \rho) \log(\rho) - \rho \right) \Biggr],$$
(F.9)

$$\mathcal{C}_{\rho_D,22}^{(s\bar{s})} = \frac{2}{3} \left[\sqrt{1 - 4\rho} \left(9 + 24 \log(\rho) - 24 \log(1 - 4\rho) + 22\rho - 34\rho^2 - 60\rho^3 \right) + 24 \left(1 - 2\rho - \rho^2 - 2\rho^3 - 5\rho^4 \right) \log \left(\frac{1 + \sqrt{1 - 4\rho}}{1 - \sqrt{1 - 4\rho}} \right) \right].$$
 (F.10)

Appendix G

Parametrisation of matrix elements of fourquark operators

The matrix elements of the dimension-six operators in QCD are parametrised as

$$\left\langle D_q | O_i^q | D_q \right\rangle = A_i f_{D_q}^2 m_{D_q}^2 B_i^q \,, \tag{G.1}$$

$$\langle D_q | O_i^{q'} | D_q \rangle = A_i f_{D_q}^2 m_{D_q}^2 \delta_i^{qq'}, \qquad q \neq q' \tag{G.2}$$

where

$$A_1^q = A_3^q = 1, \qquad A_2^q = A_4^q = \frac{m_D^2}{(m_c + m_q)^2}.$$
 (G.3)

In VIA the Bag parameters reduce to $B_1^q = B_2^q = 1$, $B_3^q = 0$, $B_4^q = 0$ and $\delta_i^{qq'} = 0$. The matrix elements of the dimension-seven four-quark operators in Eqs. (4.52) - (4.60) in HQET are parametrised in the following way

$$\langle D_q | \mathcal{P}_1^q | D_q \rangle = -m_q F^2(\mu_0) m_D \, \tilde{B}_{P,1}^q \,,$$
 (G.4)

$$\langle D_q | \mathcal{P}_2^q | D_q \rangle = -F^2(\mu_0) \, m_D \, \bar{\Lambda} \, \tilde{B}_{P,2}^q \,, \tag{G.5}$$

$$\left\langle D_q | \mathcal{P}_3^q | D_q \right\rangle = -F^2(\mu_0) \, m_D \, \bar{\Lambda} \, \tilde{B}_{P,3}^q \,, \tag{G.6}$$

$$\langle D_q | \mathcal{R}_1^q | D_q \rangle = -F^2(\mu_0) \, m_D \left(\bar{\Lambda} - m_q\right) \tilde{B}_{R,1}^q \,, \tag{G.7}$$

$$\left\langle D_q | \mathcal{R}_2^q | D_q \right\rangle = F^2(\mu_0) \, m_D \left(\bar{\Lambda} - m_q \right) \tilde{B}_{R,1}^q \,, \tag{G.8}$$

with $\bar{\Lambda} = m_D - m_c$, and

$$\langle D_q | \mathcal{M}_1^q | D_q \rangle = 2 F^2(\mu_0) m_D G_1(\mu_0) \tilde{B}_{M,1}^q,$$
 (G.9)

$$\langle D_q | \mathcal{M}_2^q | D_q \rangle = 12 F^2(\mu_0) m_D G_2(\mu_0) \tilde{B}_{M,2}^q,$$
 (G.10)

$$\langle D_q | \mathcal{M}_3^q | D_q \rangle = 2 F^2(\mu_0) m_D G_1(\mu_0) \tilde{B}_{M,3}^q ,$$
 (G.11)

$$\langle D_q | \mathcal{M}_4^q | D_q \rangle = 12 F^2(\mu_0) m_D G_2(\mu_0) \tilde{B}_{M,4}^q,$$
 (G.12)

and similar expressions for the colour-octet operators. Again, in VIA, the dimensionseven Bag parameters are $\tilde{B}_{P,i}^q = 1$, $\tilde{B}_{R,i}^q = 1$, and $\tilde{L}_{M,i}^q = 1$, while the corresponding colour-octet Bag parameters vanish. The expressions in Eqs. (G.4) - (G.8) have been obtained using the general parametrisation in HQET of the matrix element of heavy-quark currents between a heavy pseudo-scalar meson and the vacuum, see e.g. Ref. [52], namely

$$\langle 0|\bar{q}\,\Gamma\,h_v|M(v)\rangle = \frac{i}{2}F(\mu)\,\mathrm{Tr}[\Gamma\mathcal{M}(v)]\,,$$
 (G.13)

$$\langle 0|\bar{q}\,\Gamma\,iD_{\alpha}h_{v}|M(v)\rangle = -\frac{i}{6}(\bar{\Lambda}-m_{q})F(\mu)\,\mathrm{Tr}[(v_{\alpha}+\gamma_{\alpha})\Gamma\mathcal{M}(v)]\,,\qquad(\mathrm{G.14})$$

$$\langle 0|\bar{q}(-i\overleftarrow{D}_{\alpha})\Gamma h_{v}|M(v)\rangle = -\frac{i}{6}F(\mu)\operatorname{Tr}[((4\bar{\Lambda}-m_{q})v_{\alpha}+(\bar{\Lambda}-m_{q})\gamma_{\alpha})\Gamma\mathcal{M}(v)],$$
(G.15)

while for the non-local operators

$$\langle 0|i \int d^4 y \operatorname{T}\left[(\bar{q} \,\Gamma \,h_v)(0), (\bar{h}_v(iD)^2 h_v)(y)\right] |\mathcal{M}(v)\rangle = F(\mu) \,G_1(\mu) \operatorname{Tr}[\Gamma \mathcal{M}(v)],$$
(G.16)

$$\langle 0|i \int d^4 y \,\mathrm{T}\left[(\bar{q}\,\Gamma\,h_v)(0), \frac{1}{2}g_s\left(\bar{h}_v\sigma_{\alpha\beta}G^{\alpha\beta}h_v\right)(y) \right] |\mathcal{M}(v)\rangle = 6\,F(\mu)\,G_2(\mu)\,\mathrm{Tr}[\Gamma\mathcal{M}(v)],\tag{G.17}$$

where Γ is a generic Dirac structure, and

$$\mathcal{M}(v) = -\sqrt{m_D} \,\frac{(1+\psi)}{2} \gamma_5 \,. \tag{G.18}$$

Bibliography

- S. Weinberg, "A Model of Leptons," *Phys. Rev. Lett.*, vol. 19, pp. 1264–1266, 1967.
- [2] S. L. Glashow, "Partial Symmetries of Weak Interactions," Nucl. Phys., vol. 22, pp. 579–588, 1961.
- [3] A. Salam, "Weak and Electromagnetic Interactions," Conf. Proc. C, vol. 680519, pp. 367–377, 1968.
- [4] P. Langacker, The standard model and beyond. 2010.
- [5] P. W. Higgs, "Broken Symmetries and the Masses of Gauge Bosons," Phys. Rev. Lett., vol. 13, pp. 508–509, 1964.
- [6] F. Englert and R. Brout, "Broken Symmetry and the Mass of Gauge Vector Mesons," *Phys. Rev. Lett.*, vol. 13, pp. 321–323, 1964.
- [7] G. S. Guralnik, C. R. Hagen, and T. W. B. Kibble, "Global Conservation Laws and Massless Particles," *Phys. Rev. Lett.*, vol. 13, pp. 585–587, 1964.
- [8] G. Aad *et al.*, "Observation of a new particle in the search for the Standard Model Higgs boson with the ATLAS detector at the LHC," *Phys. Lett. B*, vol. 716, pp. 1–29, 2012, 1207.7214.
- [9] S. Chatrchyan *et al.*, "Observation of a New Boson at a Mass of 125 GeV with the CMS Experiment at the LHC," *Phys. Lett. B*, vol. 716, pp. 30–61, 2012, 1207.7235.
- [10] A. D. Sakharov, "Violation of CP Invariance, C asymmetry, and baryon asymmetry of the universe," *Pisma Zh. Eksp. Teor. Fiz.*, vol. 5, pp. 32–35, 1967.
- [11] N. Cabibbo, "Unitary Symmetry and Leptonic Decays," *Phys. Rev. Lett.*, vol. 10, pp. 531–533, 1963. [,648(1963)].
- [12] M. Kobayashi and T. Maskawa, "CP Violation in the Renormalizable Theory of Weak Interaction," *Prog. Theor. Phys.*, vol. 49, pp. 652–657, 1973.

- [13] M. B. Gavela, P. Hernandez, J. Orloff, O. Pene, and C. Quimbay, "Standard model CP violation and baryon asymmetry. Part 2: Finite temperature," *Nucl. Phys. B*, vol. 430, pp. 382–426, 1994, hep-ph/9406289.
- [14] G. Alonso-Alvarez, G. Elor, and M. Escudero, "Collider signals of baryogenesis and dark matter from B mesons: A roadmap to discovery," *Phys. Rev. D*, vol. 104, no. 3, p. 035028, 2021, 2101.02706.
- [15] F. R. Klinkhamer and N. S. Manton, "A Saddle Point Solution in the Weinberg-Salam Theory," Phys. Rev. D, vol. 30, p. 2212, 1984.
- [16] K. Kajantie, M. Laine, K. Rummukainen, and M. E. Shaposhnikov, "Is there a hot electroweak phase transition at m_H larger or equal to m_W?," Phys. Rev. Lett., vol. 77, pp. 2887–2890, 1996, hep-ph/9605288.
- [17] M. L. Piscopo, M. Spannowsky, and P. Waite, "Solving differential equations with neural networks: Applications to the calculation of cosmological phase transitions," *Phys. Rev. D*, vol. 100, no. 1, p. 016002, 2019, 1902.05563.
- [18] A. A. Petrov and A. E. Blechman, *Effective Field Theories*. WSP, 2016.
- [19] G. C. Branco, P. M. Ferreira, L. Lavoura, M. N. Rebelo, M. Sher, and J. P. Silva, "Theory and phenomenology of two-Higgs-doublet models," *Phys. Rept.*, vol. 516, pp. 1–102, 2012, 1106.0034.
- [20] O. Atkinson, M. Black, A. Lenz, A. Rusov, and J. Wynne, "Cornering the Two Higgs Doublet Model Type II," 7 2021, 2107.05650.
- [21] A. Dainese, M. Mangano, A. B. Meyer, A. Nisati, G. Salam, and M. A. Vesterinen, eds., *Report on the Physics at the HL-LHC, and Perspectives for the HE-LHC*, vol. 7/2019 of CERN Yellow Reports: Monographs. Geneva, Switzerland: CERN, 2019.
- [22] Y. S. Amhis *et al.*, "Averages of b-hadron, c-hadron, and τ-lepton properties as of 2018," *Eur. Phys. J. C*, vol. 81, no. 3, p. 226, 2021, 1909.12524.
- [23] A. Khodjamirian, Hadron Form Factors: From Basic Phenomenology to QCD Sum Rules. March 2020.
- [24] A. Buras, *Gauge Theory of Weak Decays*. Cambridge University Press, 6 2020.
- [25] A. V. Manohar and M. B. Wise, *Heavy quark physics*, vol. 10. 2000.
- [26] M. Fael, K. Schönwald, and M. Steinhauser, "Third order corrections to the semi-leptonic $b \rightarrow c$ and the muon decays," 11 2020, 2011.13654.

- [27] A. Lenz, "Lifetimes and heavy quark expansion," Int. J. Mod. Phys. A, vol. 30, no. 10, p. 1543005, 2015, 1405.3601.
- [28] T. Mannel, S. Turczyk, and N. Uraltsev, "Higher Order Power Corrections in Inclusive B Decays," *JHEP*, vol. 11, p. 109, 2010, 1009.4622.
- [29] M. A. Shifman, A. I. Vainshtein, and V. I. Zakharov, "QCD and Resonance Physics: Applications," *Nucl. Phys. B*, vol. 147, pp. 448–518, 1979.
- [30] M. A. Shifman, A. I. Vainshtein, and V. I. Zakharov, "QCD and Resonance Physics. Theoretical Foundations," *Nucl. Phys. B*, vol. 147, pp. 385–447, 1979.
- [31] I. I. Balitsky, V. M. Braun, and A. V. Kolesnichenko, "Radiative Decay $\Sigma^+ \rightarrow p\gamma$ in Quantum Chromodynamics," *Nucl. Phys. B*, vol. 312, pp. 509–550, 1989.
- [32] K. G. Wilson, "Confinement of Quarks," Phys. Rev. D, vol. 10, pp. 2445–2459, 1974.
- [33] I. I. Bigi and A. I. Sanda, CP violation, vol. 9. Cambridge University Press, 9 2009.
- [34] A. Lenz and G. Wilkinson, "Mixing and CP violation in the charm system," 11 2020, 2011.04443.
- [35] M. Artuso, G. Borissov, and A. Lenz, "CP violation in the B⁰_s system," *Rev. Mod. Phys.*, vol. 88, no. 4, p. 045002, 2016, 1511.09466. [Addendum: Rev.Mod.Phys. 91, 049901 (2019)].
- [36] J. Albrecht, D. van Dyk, and C. Langenbruch, "Flavour anomalies in heavy quark decays," Prog. Part. Nucl. Phys., vol. 120, p. 103885, 2021, 2107.04822.
- [37] M. Algueró, B. Capdevila, S. Descotes-Genon, J. Matias, and M. Novoa-Brunet, "b → sℓℓ global fits after Moriond 2021 results," in 55th Rencontres de Moriond on QCD and High Energy Interactions, 4 2021, 2104.08921.
- [38] C. Itzykson and J. B. Zuber, *Quantum Field Theory*. International Series In Pure and Applied Physics, New York: McGraw-Hill, 1980.
- [39] R. K. Ellis, W. J. Stirling, and B. R. Webber, QCD and Collider Physics. Cambridge Monographs on Particle Physics, Nuclear Physics and Cosmology, Cambridge University Press, 1996.
- [40] M. A. Shifman, "Wilson Loop in Vacuum Fields," Nucl. Phys. B, vol. 173, pp. 13–31, 1980.

- [41] A. J. Buras, "Weak Hamiltonian, CP violation and rare decays," in Les Houches Summer School in Theoretical Physics, Session 68: Probing the Standard Model of Particle Interactions, pp. 281–539, 6 1998, hep-ph/9806471.
- [42] G. Buchalla, A. J. Buras, and M. E. Lautenbacher, "Weak decays beyond leading logarithms," *Rev. Mod. Phys.*, vol. 68, pp. 1125–1144, 1996, hepph/9512380.
- [43] G. Buchalla, "Heavy quark theory," in 55th Scottish Universities Summer School in Physics: Heavy Flavor Physics (SUSSP 2001), 2 2002, hepph/0202092.
- [44] A. Grozin, "Effective weak Lagrangians in the Standard Model and B decays," in Helmholtz International Summer School on Physics of Heavy Quarks and Hadrons, pp. 78–98, 2014, 1311.0550.
- [45] K. G. Wilson, "Nonlagrangian models of current algebra," Phys. Rev., vol. 179, pp. 1499–1512, 1969.
- [46] K. G. Wilson and J. B. Kogut, "The Renormalization group and the epsilon expansion," *Phys. Rept.*, vol. 12, pp. 75–199, 1974.
- [47] G. 't Hooft and M. J. G. Veltman, "Regularization and Renormalization of Gauge Fields," Nucl. Phys. B, vol. 44, pp. 189–213, 1972.
- [48] C. G. Bollini and J. J. Giambiagi, "Dimensional Renormalization: The Number of Dimensions as a Regularizing Parameter," *Nuovo Cim. B*, vol. 12, pp. 20–26, 1972.
- [49] G. M. Cicuta and E. Montaldi, "Analytic renormalization via continuous space dimension," *Lett. Nuovo Cim.*, vol. 4, pp. 329–332, 1972.
- [50] J. F. Ashmore, "A Method of Gauge Invariant Regularization," Lett. Nuovo Cim., vol. 4, pp. 289–290, 1972.
- [51] G. 't Hooft, "Dimensional regularization and the renormalization group," Nucl. Phys. B, vol. 61, pp. 455–468, 1973.
- [52] M. Neubert, "Heavy quark symmetry," Phys. Rept., vol. 245, pp. 259–396, 1994, hep-ph/9306320.
- [53] E. Eichten and F. L. Feinberg, "Spin Dependent Forces in Heavy Quark Systems," *Phys. Rev. Lett.*, vol. 43, p. 1205, 1979.

- [54] E. V. Shuryak, "Hadrons Containing a Heavy Quark and QCD Sum Rules," Nucl. Phys. B, vol. 198, pp. 83–101, 1982.
- [55] M. B. Voloshin and M. A. Shifman, "On the annihilation constants of mesons consisting of a heavy and a light quark, and $B^0 \leftrightarrow \overline{B}^{-0}$ oscillations," Sov. J. Nucl. Phys., vol. 45, p. 292, 1987.
- [56] M. A. Shifman and M. B. Voloshin, "On Production of d and D* Mesons in B Meson Decays," Sov. J. Nucl. Phys., vol. 47, p. 511, 1988.
- [57] H. David Politzer and M. B. Wise, "Effective field theory approach to processes involving both light and heavy fields," *Physics Letters B*, vol. 208, no. 3, pp. 504–507, 1988.
- [58] E. Eichten and B. R. Hill, "An Effective Field Theory for the Calculation of Matrix Elements Involving Heavy Quarks," *Phys. Lett. B*, vol. 234, pp. 511– 516, 1990.
- [59] N. Isgur and M. B. Wise, "Weak decays of heavy mesons in the static quark approximation," *Physics Letters B*, vol. 232, no. 1, pp. 113–117, 1989.
- [60] H. Georgi, "An effective field theory for heavy quarks at low energies," *Physics Letters B*, vol. 240, no. 3, pp. 447–450, 1990.
- [61] B. Grinstein, "The Static Quark Effective Theory," Nucl. Phys. B, vol. 339, pp. 253–268, 1990.
- [62] A. F. Falk, H. Georgi, B. Grinstein, and M. B. Wise, "Heavy Meson Formfactors From QCD," Nucl. Phys. B, vol. 343, pp. 1–13, 1990.
- [63] M. E. Luke, "Effects of subleading operators in the heavy quark effective theory," *Phys. Lett. B*, vol. 252, pp. 447–455, 1990.
- [64] T. Mannel, "Heavy quark effective field theory," Rept. Prog. Phys., vol. 60, pp. 1113–1172, 1997.
- [65] A. V. Manohar and M. B. Wise, *Heavy Quark Physics*. Cambridge Monographs on Particle Physics, Nuclear Physics and Cosmology, Cambridge University Press, 2000.
- [66] T. Mannel, "Recent progress in the theory of heavy flavor decays," J. Phys., vol. 21, pp. 1007–1042, 1995.

- [67] I. Bigi, M. Shifman, and N. Uraltsev, "Aspects of heavy-quark theory," Annual Review of Nuclear and Particle Science, vol. 47, no. 1, pp. 591–661, 1997, https://doi.org/10.1146/annurev.nucl.47.1.591.
- [68] T. Mannel, W. Roberts, and Z. Ryzak, "A Derivation of the heavy quark effective Lagrangian from QCD," Nucl. Phys. B, vol. 368, pp. 204–217, 1992.
- [69] H. Georgi, B. Grinstein, and M. B. Wise, " Λ_b semileptonic decay form-factors for m_c does not equal infinity," *Phys. Lett. B*, vol. 252, pp. 456–460, 1990.
- [70] A. F. Falk and M. Neubert, "Second order power corrections in the heavy quark effective theory. 1. Formalism and meson form-factors," *Phys. Rev. D*, vol. 47, pp. 2965–2981, 1993, hep-ph/9209268.
- [71] M. A. Shifman, "Lectures on heavy quarks in quantum chromodynamics," in Theoretical Advanced Study Institute in Elementary Particle Physics (TASI 95): QCD and Beyond, 10 1995, hep-ph/9510377.
- [72] J. S. Schwinger, "On gauge invariance and vacuum polarization," *Phys. Rev.*, vol. 82, pp. 664–679, 1951.
- [73] V. A. Novikov, M. A. Shifman, A. I. Vainshtein, and V. I. Zakharov, "Operator expansion in quantum chromodynamics beyond perturbation theory," *Nucl. Phys. B*, vol. 174, pp. 378–396, 1980.
- [74] V. Fock, "Proper time in classical and quantum mechanics," Phys. Z. Sowjetunion, vol. 12, pp. 404–425, 1937.
- [75] M. S. Dubovikov and A. V. Smilga, "Analytical Properties of the Quark Polarization Operator in an External Selfdual Field," *Nucl. Phys. B*, vol. 185, pp. 109–132, 1981.
- [76] E. V. Shuryak and A. I. Vainshtein, "Theory of Power Corrections to Deep Inelastic Scattering in Quantum Chromodynamics. 1. Q² Effects," Nucl. Phys. B, vol. 199, pp. 451–481, 1982.
- [77] A. V. Smilga, "Calculation of the degree corrections in fixed point gauge. (in Russian)," Sov. J. Nucl. Phys., vol. 35, pp. 271–277, 1982.
- [78] V. A. Novikov, M. A. Shifman, A. I. Vainshtein, and V. I. Zakharov, "Calculations in External Fields in Quantum Chromodynamics. Technical Review," *Fortsch. Phys.*, vol. 32, p. 585, 1984.

- [79] V. I. Zakharov, "Gluon condensate and beyond," Int. J. Mod. Phys. A, vol. 14, pp. 4865–4880, 1999, hep-ph/9906264.
- [80] The Fock-Schwinger Gauge. Theses, University of Tasmania, 1991.
- [81] M. E. Peskin and D. V. Schroeder, An Introduction to quantum field theory. Reading, USA: Addison-Wesley, 1995.
- [82] R. Soldati, "Introduction to Quantum Field Theory. A Basic Course for Primary Education," on line notes for the I Semester Course, http://www.robertosoldati.com.
- [83] I. S. Gradshteyn and I. M. Ryzhik, Table of integrals, series, and products. Elsevier/Academic Press, Amsterdam, seventh ed., 2007.
- [84] M. Abramowitz and I. A. Stegun, Handbook of Mathematical Functions with Formulas, Graphs, and Mathematical Tables. New York: Dover, ninth ed., 1964.
- [85] V. M. Belyaev and B. Y. Blok, "Charmed baryons in quantum chromodynamics," Z. Phys. C, vol. 30, p. 151, 1986.
- [86] B. Blok, R. D. Dikeman, and M. A. Shifman, "Calculation of 1/m³_c terms in the total semileptonic width of *D* mesons," *Phys. Rev. D*, vol. 51, pp. 6167–6176, 1995, hep-ph/9410293.
- [87] M. A. Shifman and M. B. Voloshin, "Preasymptotic Effects in Inclusive Weak Decays of Charmed Particles," Sov. J. Nucl. Phys., vol. 41, p. 120, 1985.
- [88] M. A. Shifman, "Theory of weak inclusive decays and lifetimes of heavy hadrons," in 27th International Conference on High-energy Physics, 7 1994, hep-ph/9409359.
- [89] R. Aaij et al., "Measurement of the Ω_c^0 baryon lifetime," Phys. Rev. Lett., vol. 121, no. 9, p. 092003, 2018, 1807.02024.
- [90] R. Aaij *et al.*, "Precision measurement of the Λ_c^+ , Ξ_c^+ and Ξ_c^0 baryon lifetimes," *Phys. Rev. D*, vol. 100, no. 3, p. 032001, 2019, 1906.08350.
- [91] N. N. Nikolaev, "Gauge model for weak interaction and supercharged hadrons," *Pisma Zh. Eksp. Teor. Fiz.*, vol. 18, pp. 447–451, 1973.
- [92] M. K. Gaillard, B. W. Lee, and J. L. Rosner, "Search for Charm," *Rev. Mod. Phys.*, vol. 47, pp. 277–310, 1975.

- [93] R. L. Kingsley, S. B. Treiman, F. Wilczek, and A. Zee, "Weak Decays of Charmed Hadrons," *Phys. Rev. D*, vol. 11, p. 1919, 1975.
- [94] J. R. Ellis, M. K. Gaillard, and D. V. Nanopoulos, "On the Weak Decays of High Mass Hadrons," *Nucl. Phys. B*, vol. 100, p. 313, 1975. [Erratum: Nucl.Phys.B 104, 547 (1976)].
- [95] G. Altarelli, N. Cabibbo, and L. Maiani, "Possibility that Charmed Vector Mesons are Lighter than Charmed Pseudoscalars," *Phys. Rev. Lett.*, vol. 35, pp. 635–638, 1975.
- [96] V. A. Novikov, M. A. Shifman, A. I. Vainshtein, and V. I. Zakharov, "Wilson's Operator Expansion: Can It Fail?," *Nucl. Phys. B*, vol. 249, pp. 445–471, 1985.
- [97] J. Chay, H. Georgi, and B. Grinstein, "Lepton energy distributions in heavy meson decays from QCD," *Phys. Lett. B*, vol. 247, pp. 399–405, 1990.
- [98] I. I. Y. Bigi, N. G. Uraltsev, and A. I. Vainshtein, "Nonperturbative corrections to inclusive beauty and charm decays: QCD versus phenomenological models," *Phys. Lett. B*, vol. 293, pp. 430–436, 1992, hep-ph/9207214. [Erratum: Phys.Lett.B 297, 477–477 (1992)].
- [99] I. I. Y. Bigi, M. A. Shifman, N. G. Uraltsev, and A. I. Vainshtein, "QCD predictions for lepton spectra in inclusive heavy flavor decays," *Phys. Rev. Lett.*, vol. 71, pp. 496–499, 1993, hep-ph/9304225.
- [100] B. Blok, L. Koyrakh, M. A. Shifman, and A. I. Vainshtein, "Differential distributions in semileptonic decays of the heavy flavors in QCD," *Phys. Rev.* D, vol. 49, p. 3356, 1994, hep-ph/9307247. [Erratum: Phys.Rev.D 50, 3572 (1994)].
- [101] T. Mannel, "Operator product expansion for inclusive semileptonic decays in heavy quark effective field theory," *Nucl. Phys. B*, vol. 413, pp. 396–412, 1994, hep-ph/9308262.
- [102] M. A. Shifman, "Theory of preasymptotic effects in weak inclusive decays," in Workshop on Continuous Advances in QCD, 2 1994, hep-ph/9405246.
- [103] I. I. Y. Bigi, B. Blok, M. A. Shifman, N. G. Uraltsev, and A. I. Vainshtein, "A QCD 'manifesto' on inclusive decays of beauty and charm," in 7th Meeting of the APS Division of Particles Fields, 11 1992, hep-ph/9212227.

- [104] I. I. Y. Bigi, M. A. Shifman, N. G. Uraltsev, and A. I. Vainshtein, "Sum rules for heavy flavor transitions in the SV limit," *Phys. Rev. D*, vol. 52, pp. 196– 235, 1995, hep-ph/9405410.
- [105] B. M. Dassinger, T. Mannel, and S. Turczyk, "Inclusive semi-leptonic B decays to order $1/m_{h}^{4}$," *JHEP*, vol. 03, p. 087, 2007, hep-ph/0611168.
- [106] A. Alberti, P. Gambino, K. J. Healey, and S. Nandi, "Precision determination of the cabibbo-kobayashi-maskawa elementvcb," *Physical Review Letters*, vol. 114, Feb 2015.
- [107] V. A. Khoze and M. A. Shifman, "Heavy quarks," Sov. Phys. Usp., vol. 26, p. 387, 1983.
- [108] M. A. Shifman and M. B. Voloshin, "Hierarchy of Lifetimes of Charmed and Beautiful Hadrons," Sov. Phys. JETP, vol. 64, p. 698, 1986. [Zh. Eksp. Teor. Fiz.91,1180(1986)].
- [109] M. Neubert and C. T. Sachrajda, "Spectator effects in inclusive decays of beauty hadrons," Nucl. Phys., vol. B483, pp. 339–370, 1997, hep-ph/9603202.
- [110] N. G. Uraltsev, "On the problem of boosting nonleptonic b baryon decays," *Phys. Lett. B*, vol. 376, pp. 303–308, 1996, hep-ph/9602324.
- [111] Q. Ho-kim and X.-Y. Pham, "Exact One Gluon Corrections for Inclusive Weak Processes," Annals Phys., vol. 155, p. 202, 1984.
- [112] G. Altarelli and S. Petrarca, "Inclusive beauty decays and the spectator model," *Phys. Lett.*, vol. B261, pp. 303–310, 1991.
- [113] M. B. Voloshin, "QCD radiative enhancement of the decay $b \rightarrow c\bar{c}s$," Phys. Rev., vol. D51, pp. 3948–3951, 1995, hep-ph/9409391.
- [114] E. Bagan, P. Ball, V. M. Braun, and P. Gosdzinsky, "Charm quark mass dependence of QCD corrections to nonleptonic inclusive B decays," *Nucl. Phys.*, vol. B432, pp. 3–38, 1994, hep-ph/9408306.
- [115] E. Bagan, P. Ball, B. Fiol, and P. Gosdzinsky, "Next-to-leading order radiative corrections to the decay $b \rightarrow c\bar{c}s$," *Phys. Lett.*, vol. B351, pp. 546–554, 1995, hep-ph/9502338.
- [116] A. Lenz, U. Nierste, and G. Ostermaier, "Penguin diagrams, charmless B decays and the missing charm puzzle," *Phys. Rev.*, vol. D56, pp. 7228–7239, 1997, hep-ph/9706501.

- [117] A. Lenz, U. Nierste, and G. Ostermaier, "Determination of the CKM angle gamma and |V_{ub}/V_{cb}| from inclusive direct CP asymmetries and branching ratios in charmless B decays," *Phys. Rev.*, vol. D59, p. 034008, 1999, hepph/9802202.
- [118] F. Krinner, A. Lenz, and T. Rauh, "The inclusive decay $b \rightarrow c\bar{c}s$ revisited," Nucl. Phys., vol. B876, pp. 31–54, 2013, 1305.5390.
- [119] A. Czarnecki and K. Melnikov, "Two loop QCD corrections to semileptonic b decays at maximal recoil," *Phys. Rev. Lett.*, vol. 78, pp. 3630–3633, 1997, hep-ph/9703291.
- [120] A. Czarnecki and K. Melnikov, "Two loop QCD corrections to semileptonic b decays at an intermediate recoil," *Phys. Rev.*, vol. D59, p. 014036, 1999, hep-ph/9804215.
- [121] T. van Ritbergen, "The Second order QCD contribution to the semileptonic $b \rightarrow u$ decay rate," *Phys. Lett.*, vol. B454, pp. 353–358, 1999, hep-ph/9903226.
- [122] K. Melnikov, " $\mathcal{O}(\alpha_s^2)$ corrections to semileptonic decay $b \to c l \bar{\nu}_l$," Phys. Lett., vol. B666, pp. 336–339, 2008, 0803.0951.
- [123] A. Pak and A. Czarnecki, "Heavy-to-heavy quark decays at NNLO," Phys. Rev., vol. D78, p. 114015, 2008, 0808.3509.
- [124] A. Pak and A. Czarnecki, "Mass effects in muon and semileptonic $b \rightarrow c$ decays," *Phys. Rev. Lett.*, vol. 100, p. 241807, 2008, 0803.0960.
- [125] M. Dowling, A. Pak, and A. Czarnecki, "Semi-Leptonic b-decay at Intermediate Recoil," *Phys. Rev.*, vol. D78, p. 074029, 2008, 0809.0491.
- [126] R. Bonciani and A. Ferroglia, "Two-Loop QCD Corrections to the Heavy-to-Light Quark Decay," JHEP, vol. 11, p. 065, 2008, 0809.4687.
- [127] S. Biswas and K. Melnikov, "Second order QCD corrections to inclusive semileptonic $b \to X(c) l \bar{\nu}_l$ decays with massless and massive lepton," *JHEP*, vol. 02, p. 089, 2010, 0911.4142.
- [128] M. Brucherseifer, F. Caola, and K. Melnikov, "On the $O(\alpha_s^2)$ corrections to $b \rightarrow X_u e \bar{\nu}$ inclusive decays," *Phys. Lett.*, vol. B721, pp. 107–110, 2013, 1302.0444.
- [129] A. Czarnecki, M. Slusarczyk, and F. V. Tkachov, "Enhancement of the hadronic b quark decays," *Phys. Rev. Lett.*, vol. 96, p. 171803, 2006, hepph/0511004.

- [130] B. Blok and M. A. Shifman, "The Rule of discarding $1/N_c$ in inclusive weak decays. 1.," *Nucl. Phys.*, vol. B399, pp. 441–458, 1993, hep-ph/9207236.
- [131] B. Blok and M. A. Shifman, "The Rule of discarding 1/N_c in inclusive weak decays. 2.," Nucl. Phys., vol. B399, pp. 459–476, 1993, hep-ph/9209289.
- [132] A. Alberti, P. Gambino, and S. Nandi, "Perturbative corrections to power suppressed effects in semileptonic B decays," *JHEP*, vol. 01, p. 147, 2014, 1311.7381.
- [133] T. Mannel, A. A. Pivovarov, and D. Rosenthal, "Inclusive semileptonic B decays from QCD with NLO accuracy for power suppressed terms," *Phys. Lett.*, vol. B741, pp. 290–294, 2015, 1405.5072.
- [134] T. Mannel, A. A. Pivovarov, and D. Rosenthal, "Inclusive weak decays of heavy hadrons with power suppressed terms at NLO," *Phys. Rev.*, vol. D92, no. 5, p. 054025, 2015, 1506.08167.
- [135] M. Gremm and A. Kapustin, "Order $1/m_b^3$ corrections to $B \to X(c)$ lepton anti-neutrino decay and their implication for the measurement of $\bar{\Lambda}$ and λ_1 ," *Phys. Rev.*, vol. D55, pp. 6924–6932, 1997, hep-ph/9603448.
- [136] T. Mannel and A. A. Pivovarov, "QCD corrections to inclusive heavy hadron weak decays at $\Lambda_{\rm QCD}^3/m_Q^3$," *Phys. Rev.*, vol. D100, no. 9, p. 093001, 2019, 1907.09187.
- [137] A. Lenz, M. L. Piscopo, and A. V. Rusov, "Contribution of the Darwin operator to non-leptonic decays of heavy quarks," *JHEP*, vol. 12, p. 199, 2020, 2004.09527.
- [138] T. Mannel, D. Moreno, and A. Pivovarov, "Heavy quark expansion for heavy hadron lifetimes: completing the $1/m_b^3$ corrections," *JHEP*, vol. 08, p. 089, 2020, 2004.09485.
- [139] D. Moreno, "Completing $1/m_b^3$ corrections to non-leptonic bottom-to-up-quark decays," *JHEP*, vol. 01, p. 051, 2021, 2009.08756.
- [140] D. King, A. Lenz, M. L. Piscopo, T. Rauh, A. V. Rusov, and C. Vlahos, "Revisiting Inclusive Decay Widths of Charmed Mesons," 9 2021, 2109.13219.
- [141] M. Beneke, G. Buchalla, C. Greub, A. Lenz, and U. Nierste, "The $B^+ B_d^0$ Lifetime Difference Beyond Leading Logarithms," *Nucl. Phys.*, vol. B639, pp. 389–407, 2002, hep-ph/0202106.

- [142] E. Franco, V. Lubicz, F. Mescia, and C. Tarantino, "Lifetime ratios of beauty hadrons at the next-to-leading order in QCD," *Nucl. Phys.*, vol. B633, pp. 212– 236, 2002, hep-ph/0203089.
- [143] A. Lenz and T. Rauh, "D-meson lifetimes within the heavy quark expansion," *Phys. Rev.*, vol. D88, p. 034004, 2013, 1305.3588.
- [144] F. Gabbiani, A. I. Onishchenko, and A. A. Petrov, "Spectator effects and lifetimes of heavy hadrons," *Phys. Rev.*, vol. D70, p. 094031, 2004, hepph/0407004.
- [145] B. Blok, M. A. Shifman, and D.-X. Zhang, "An Illustrative example of how quark hadron duality might work," *Phys. Rev. D*, vol. 57, pp. 2691–2700, 1998, hep-ph/9709333. [Erratum: Phys.Rev.D 59, 019901 (1999)].
- [146] B. Chibisov, R. D. Dikeman, M. A. Shifman, and N. Uraltsev, "Operator product expansion, heavy quarks, QCD duality and its violations," Int. J. Mod. Phys. A, vol. 12, pp. 2075–2133, 1997, hep-ph/9605465.
- [147] M. A. Shifman, "Quark hadron duality," in 8th International Symposium on Heavy Flavor Physics, vol. 3, (Singapore), pp. 1447–1494, World Scientific, 7 2000, hep-ph/0009131.
- [148] M. Beneke and G. Buchalla, "The B_c Meson Lifetime," Phys. Rev. D, vol. 53, pp. 4991–5000, 1996, hep-ph/9601249.
- [149] J. Aebischer and B. Grinstein, "Standard Model prediction of the B_c lifetime," 5 2021, 2105.02988.
- [150] J. Aebischer and B. Grinstein, "A novel determination of the B_c lifetime," 8 2021, 2108.10285.
- [151] G. C. Wick, "The Evaluation of the Collision Matrix," Phys. Rev., vol. 80, pp. 268–272, 1950.
- [152] N. N. Bogolyubov and D. V. Shirkov, Introduction to the theory of quantized fields, vol. 3. 1959.
- [153] H. Georgi, "Heavy quark effective field theory," in Theoretical Advanced Study Institute in Elementary Particle Physics (TASI 91): Perspectives in the Standard Model, 8 1991.
- [154] W. R. Inc., "Mathematica, Version 12.3.1." Champaign, IL, 2021.

- [155] G. Passarino and M. J. G. Veltman, "One Loop Corrections for e⁺e⁻ Annihilation Into mu⁺mu⁻ in the Weinberg Model," Nucl. Phys. B, vol. 160, pp. 151–207, 1979.
- [156] R. Mertig, M. Bohm, and A. Denner, "FEYN CALC: Computer algebraic calculation of Feynman amplitudes," *Comput. Phys. Commun.*, vol. 64, pp. 345– 359, 1991.
- [157] R. N. Lee, "Presenting LiteRed: a tool for the Loop InTEgrals REDuction," 2012, 1212.2685.
- [158] R. N. Lee, "LiteRed 1.4: a powerful tool for reduction of multiloop integrals," J. Phys. Conf. Ser., vol. 523, p. 012059, 2014, 1310.1145.
- [159] O. V. Tarasov, "Generalized recurrence relations for two loop propagator integrals with arbitrary masses," *Nucl. Phys. B*, vol. 502, pp. 455–482, 1997, hep-ph/9703319.
- [160] M. Caffo, H. Czyz, S. Laporta, and E. Remiddi, "The Master differential equations for the two loop sunrise selfmass amplitudes," *Nuovo Cim. A*, vol. 111, pp. 365–389, 1998, hep-th/9805118.
- [161] E. Remiddi and L. Tancredi, "Differential equations and dispersion relations for Feynman amplitudes. The two-loop massive sunrise and the kite integral," *Nucl. Phys.*, vol. B907, pp. 400–444, 2016, 1602.01481.
- [162] A. A. Pivovarov, N. N. Tavkhelidze, and V. F. Tokarev, "The Schwinger model and sum rules," *Phys. Lett.*, vol. 132B, pp. 402–406, 1983.
- [163] S. Groote and A. A. Pivovarov, "Threshold expansion of Feynman diagrams within a configuration space technique," *Nucl. Phys.*, vol. B580, pp. 459–484, 2000, hep-ph/0003115.
- [164] J. Broedel, C. Duhr, F. Dulat, and L. Tancredi, "Elliptic polylogarithms and iterated integrals on elliptic curves. Part I: general formalism," *JHEP*, vol. 05, p. 093, 2018, 1712.07089.
- [165] J. Broedel, C. Duhr, F. Dulat, and L. Tancredi, "Elliptic polylogarithms and iterated integrals on elliptic curves II: an application to the sunrise integral," *Phys. Rev.*, vol. D97, no. 11, p. 116009, 2018, 1712.07095.
- [166] T. Mannel and K. K. Vos, "Reparametrization Invariance and Partial Re-Summations of the Heavy Quark Expansion," *JHEP*, vol. 06, p. 115, 2018, 1802.09409.

- [167] P. Gambino, G. Ossola, and N. Uraltsev, "Hadronic mass and q² moments of charmless semileptonic *B* decay distributions," *JHEP*, vol. 09, p. 010, 2005, hep-ph/0505091.
- [168] I. I. Bigi, N. Uraltsev, and R. Zwicky, "On the nonperturbative charm effects in inclusive $B \rightarrow X_c \ell \nu$ decays," *Eur. Phys. J. C*, vol. 50, pp. 539–556, 2007, hep-ph/0511158.
- [169] C. Breidenbach, T. Feldmann, T. Mannel, and S. Turczyk, "On the Role of 'Intrinsic Charm' in Semi-Leptonic B-Meson Decays," *Phys. Rev. D*, vol. 78, p. 014022, 2008, 0805.0971.
- [170] R. Grigjanis, P. J. O'Donnell, M. Sutherland, and H. Navelet, "QCD radiative corrections to $B \rightarrow X_s e^+ e^-$ processes," *Phys. Lett. B*, vol. 223, pp. 239–244, 1989.
- [171] M. Misiak, "QCD corrected effective Hamiltonian for the $b \rightarrow s\gamma$ decay," Phys. Lett. B, vol. 269, pp. 161–168, 1991.
- [172] M. Ciuchini, E. Franco, L. Reina, and L. Silvestrini, "Leading order QCD corrections to b → sγ and b → sg decays in three regularization schemes," Nucl. Phys. B, vol. 421, pp. 41–64, 1994, hep-ph/9311357.
- [173] C. W. Bauer, A. F. Falk, and M. E. Luke, "Resumming phase space logarithms in inclusive semileptonic *B* decays," *Phys. Rev. D*, vol. 54, pp. 2097–2107, 1996, hep-ph/9604290.
- [174] W. A. Bardeen, A. J. Buras, D. W. Duke, and T. Muta, "Deep Inelastic Scattering Beyond the Leading Order in Asymptotically Free Gauge Theories," *Phys. Rev. D*, vol. 18, p. 3998, 1978.
- [175] R. Aaij et al., "Physics case for an LHCb Upgrade II Opportunities in flavour physics, and beyond, in the HL-LHC era," 2018, 1808.08865.
- [176] M. Ablikim *et al.*, "Future Physics Programme of BESIII," *Chin. Phys.*, vol. C44, no. 040001, 2020, 1912.05983.
- [177] W. Altmannshofer *et al.*, "The Belle II Physics Book," *PTEP*, vol. 2019, no. 12, p. 123C01, 2019, 1808.10567. [Erratum: PTEP2020,no.2,029201(2020)].
- [178] G. Bellini, I. I. Y. Bigi, and P. J. Dornan, "Lifetimes of charm and beauty hadrons," *Phys. Rept.*, vol. 289, pp. 1–155, 1997.

- [179] S. Bianco, F. L. Fabbri, D. Benson, and I. Bigi, "A Cicerone for the physics of charm," *Riv. Nuovo Cim.*, vol. 26, no. 7-8, pp. 1–200, 2003, hep-ex/0309021.
- [180] M. Artuso, B. Meadows, and A. A. Petrov, "Charm Meson Decays," Ann. Rev. Nucl. Part. Sci., vol. 58, pp. 249–291, 2008, 0802.2934.
- [181] M. Gersabeck, "Brief Review of Charm Physics," Mod. Phys. Lett. A, vol. 27, p. 1230026, 2012, 1207.2195.
- [182] R. Aaij et al., "Observation of CP Violation in Charm Decays," Phys. Rev. Lett., vol. 122, no. 21, p. 211803, 2019, 1903.08726.
- [183] H.-N. Li, C.-D. Lu, and F.-S. Yu, "Implications on the first observation of charm CPV at LHCb," 3 2019, 1903.10638.
- [184] Y. Grossman and S. Schacht, "The emergence of the $\Delta U = 0$ rule in charm physics," *JHEP*, vol. 07, p. 020, 2019, 1903.10952.
- [185] H.-Y. Cheng and C.-W. Chiang, "Revisiting CP violation in $D \rightarrow PP$ and VP decays," *Phys. Rev. D*, vol. 100, no. 9, p. 093002, 2019, 1909.03063.
- [186] A. Soni, "Charm CP: ΔA_{CP} and Radiative decays," in 37th International Symposium on Lattice Field Theory, 1 2020, 2001.10014.
- [187] M. Chala, A. Lenz, A. V. Rusov, and J. Scholtz, " ΔA_{CP} within the Standard Model and beyond," *JHEP*, vol. 07, p. 161, 2019, 1903.10490.
- [188] A. Dery and Y. Nir, "Implications of the LHCb discovery of CP violation in charm decays," JHEP, vol. 12, p. 104, 2019, 1909.11242.
- [189] A. Khodjamirian and A. A. Petrov, "Direct CP asymmetry in $D \to \pi^- \pi^+$ and $D \to K^- K^+$ in QCD-based approach," *Phys. Lett. B*, vol. 774, pp. 235–242, 2017, 1706.07780.
- [190] A. Lenz, "What did we learn in theory from the ΔA_{CP}-saga?," in Proceedings, 6th International Workshop on Charm Physics (Charm 2013): Manchester, UK, August 31-September 4, 2013, 11 2013, 1311.6447.
- [191] P. Zyla et al., "Review of Particle Physics," PTEP, vol. 2020, no. 8, p. 083C01, 2020.
- [192] M. Ablikim *et al.*, "Measurement of the absolute branching fraction of inclusive semielectronic D_s^+ decays," 4 2021, 2104.07311.

- [193] F. Abudinén *et al.*, "Precise measurement of the D^0 and D^+ lifetimes at Belle II," 8 2021, 2108.03216.
- [194] B physics at the Tevatron: Run II and beyond, 12 2001, hep-ph/0201071.
- [195] U. Nierste, "Three Lectures on Meson Mixing and CKM phenomenology," in Heavy quark physics. Proceedings, Helmholtz International School, HQP08, Dubna, Russia, August 11-21, 2008, pp. 1–38, 2009, 0904.1869.
- [196] L. Silvestrini, "Effective Theories for Quark Flavour Physics," 5 2019, 1905.00798.
- [197] Y. S. Amhis *et al.*, "Averages of *b*-hadron, *c*-hadron, and τ -lepton properties as of 2018," 2019, 1909.12524.
- [198] E. Aitala *et al.*, "Search for D0 anti-D0 mixing in semileptonic decay modes," *Phys. Rev. Lett.*, vol. 77, pp. 2384–2387, 1996, hep-ex/9606016.
- [199] C. Cawlfield *et al.*, "Limits on neutral D mixing in semileptonic decays," *Phys. Rev. D*, vol. 71, p. 077101, 2005, hep-ex/0502012.
- [200] B. Aubert *et al.*, "Search for $D^0 \overline{D}^0$ mixing using doubly flavor tagged semileptonic decay modes," *Phys. Rev. D*, vol. 76, p. 014018, 2007, 0705.0704.
- [201] U. Bitenc *et al.*, "Improved search for $D^0 \overline{D}^0$ mixing using semileptonic decays at Belle," *Phys. Rev. D*, vol. 77, p. 112003, 2008, 0802.2952.
- [202] E. Aitala *et al.*, "A Search for D⁰ D
 ⁰ mixing and doubly Cabibbo suppressed decays of the D⁰ in hadronic final states," *Phys. Rev. D*, vol. 57, pp. 13–27, 1998, hep-ex/9608018.
- [203] R. Godang *et al.*, "Search for $D^0 \overline{D}^0$ mixing," *Phys. Rev. Lett.*, vol. 84, pp. 5038–5042, 2000, hep-ex/0001060.
- [204] J. Link *et al.*, "Measurement of the doubly Cabibbo suppressed decay $D^0 \rightarrow K^+\pi^-$ and a search for charm mixing," *Phys. Lett. B*, vol. 618, pp. 23–33, 2005, hep-ex/0412034.
- [205] L. Zhang *et al.*, "Improved constraints on $D^0 \overline{D}^0$ mixing in $D^0 \to K^+\pi^-$ decays at BELLE," *Phys. Rev. Lett.*, vol. 96, p. 151801, 2006, hep-ex/0601029.
- [206] B. Aubert *et al.*, "Evidence for D⁰ D
 ⁰ Mixing," *Phys. Rev. Lett.*, vol. 98, p. 211802, 2007, hep-ex/0703020.

- [207] T. A. Aaltonen *et al.*, "Observation of $D^0 \overline{D}^0$ Mixing Using the CDF II Detector," *Phys. Rev. Lett.*, vol. 111, no. 23, p. 231802, 2013, 1309.4078.
- [208] B. Ko *et al.*, "Observation of D⁰ D
 ⁰ Mixing in e⁺e⁻ Collisions," *Phys. Rev. Lett.*, vol. 112, no. 11, p. 111801, 2014, 1401.3402. [Addendum: Phys.Rev.Lett. 112, 139903 (2014)].
- [209] R. Aaij *et al.*, "Updated determination of $D^0-\overline{D}^0$ mixing and CP violation parameters with $D^0 \rightarrow K^+\pi^-$ decays," *Phys. Rev. D*, vol. 97, no. 3, p. 031101, 2018, 1712.03220.
- [210] B. Aubert *et al.*, "Measurement of $D^0 \overline{D}^0$ mixing from a time-dependent amplitude analysis of $D^0 \rightarrow K^+\pi^-\pi^0$ decays," *Phys. Rev. Lett.*, vol. 103, p. 211801, 2009, 0807.4544.
- [211] R. Aaij *et al.*, "First observation of $D^0 \overline{D}^0$ oscillations in $D^0 \to K^+ \pi^- \pi^+ \pi^-$ decays and measurement of the associated coherence parameters," *Phys. Rev. Lett.*, vol. 116, no. 24, p. 241801, 2016, 1602.07224.
- [212] E. Aitala *et al.*, "Measurements of lifetimes and a limit on the lifetime difference in the neutral D meson system," *Phys. Rev. Lett.*, vol. 83, pp. 32–36, 1999, hep-ex/9903012.
- [213] J. Link et al., "A Measurement of lifetime differences in the neutral D meson system," Phys. Lett. B, vol. 485, pp. 62–70, 2000, hep-ex/0004034.
- [214] S. Csorna *et al.*, "Lifetime differences, direct CP violation and partial widths in D⁰ meson decays to K⁺K⁻ and pi⁺pi⁻," Phys. Rev. D, vol. 65, p. 092001, 2002, hep-ex/0111024.
- [215] J. Lees *et al.*, "Measurement of $D^0 \overline{D}^0$ Mixing and CP Violation in Two-Body D^0 Decays," *Phys. Rev. D*, vol. 87, no. 1, p. 012004, 2013, 1209.3896.
- [216] T. A. Aaltonen *et al.*, "Measurement of indirect CP-violating asymmetries in $D^0 \rightarrow K^+K^-$ and $D^0 \rightarrow \pi^+\pi^-$ decays at CDF," *Phys. Rev. D*, vol. 90, no. 11, p. 111103, 2014, 1410.5435.
- [217] M. Ablikim *et al.*, "Measurement of y_{CP} in $D^0 \overline{D}^0$ oscillation using quantum correlations in $e^+e^- \rightarrow D^0\overline{D}^0$ at $\sqrt{s} = 3.773 \text{ GeV}$," *Phys. Lett. B*, vol. 744, pp. 339–346, 2015, 1501.01378.
- [218] R. Aaij *et al.*, "Measurement of indirect *CP* asymmetries in $D^0 \rightarrow K^-K^+$ and $D^0 \rightarrow \pi^-\pi^+$ decays using semileptonic *B* decays," *JHEP*, vol. 04, p. 043, 2015, 1501.06777.

- [219] M. Staric *et al.*, "Measurement of $D^0 \overline{D^0}$ mixing and search for CP violation in $D^0 \rightarrow K^+ K^-, \pi^+ \pi^-$ decays with the full Belle data set," *Phys. Lett. B*, vol. 753, pp. 412–418, 2016, 1509.08266.
- [220] R. Aaij *et al.*, "Measurement of the *CP* violation parameter A_{Γ} in $D^0 \rightarrow K^+K^-$ and $D^0 \rightarrow \pi^+\pi^-$ decays," *Phys. Rev. Lett.*, vol. 118, no. 26, p. 261803, 2017, 1702.06490.
- [221] R. Aaij *et al.*, "Measurement of the Charm-Mixing Parameter y_{CP} ," *Phys. Rev. Lett.*, vol. 122, no. 1, p. 011802, 2019, 1810.06874.
- [222] B. Aubert *et al.*, "Search for CP violation in the decays $D^0 \rightarrow K^-K^+$ and $D^0 \rightarrow \pi^-\pi^+$," *Phys. Rev. Lett.*, vol. 100, p. 061803, 2008, 0709.2715.
- [223] A. Di Canto, "CP Violation in charm decays at CDF," Nuovo Cim. C, vol. 036, no. 01, pp. 26–28, 2013, 1208.2517.
- [224] P. del Amo Sanchez *et al.*, "Measurement of $D^0 \overline{D}^0$ mixing parameters using $D^0 \rightarrow K_S^0 \pi^+ \pi^-$ and $D^0 \rightarrow K_S^0 K^+ K^-$ decays," *Phys. Rev. Lett.*, vol. 105, p. 081803, 2010, 1004.5053.
- [225] T. Peng *et al.*, "Measurement of $D^0 \overline{D}^0$ mixing and search for indirect CP violation using $D^0 \rightarrow K_S^0 \pi^+ \pi^-$ decays," *Phys. Rev. D*, vol. 89, no. 9, p. 091103, 2014, 1404.2412.
- [226] R. Aaij *et al.*, "Model-independent measurement of mixing parameters in D⁰ $\rightarrow K_S^0 \pi^+\pi^-$ decays," *JHEP*, vol. 04, p. 033, 2016, 1510.01664.
- [227] R. Aaij et al., "Measurement of the mass difference between neutral charmmeson eigenstates," Phys. Rev. Lett., vol. 122, no. 23, p. 231802, 2019, 1903.03074.
- [228] A. Zupanc *et al.*, "Measurement of y_{CP} in D^0 meson decays to the $K_S^0 K^+ K^-$ final state," *Phys. Rev. D*, vol. 80, p. 052006, 2009, 0905.4185.
- [229] J. Lees *et al.*, "Measurement of the neutral D meson mixing parameters in a time-dependent amplitude analysis of the $D^0 \rightarrow \pi^+\pi^-\pi^0$ decay," *Phys. Rev.* D, vol. 93, no. 11, p. 112014, 2016, 1604.00857.
- [230] D. Asner *et al.*, "Updated Measurement of the Strong Phase in $D^0 \to K^+\pi^-$ Decay Using Quantum Correlations in $e^+e^- \to D^0\bar{D}^0$ at CLEO," *Phys. Rev.* D, vol. 86, p. 112001, 2012, 1210.0939.

- [231] R. Aaij *et al.*, "Observation of the mass difference between neutral charmmeson eigenstates," 6 2021, 2106.03744.
- [232] H. N. Nelson, "Compilation of D⁰ → D
 ⁰ mixing predictions," in Lepton and photon interactions at high energies. Proceedings, 19th International Symposium, LP'99, Stanford, USA, August 9-14, 1999, 8 1999, hep-ex/9908021.
- [233] A. A. Petrov, "Charm physics: Theoretical review," *eConf*, vol. C030603,
 p. MEC05, 2003, hep-ph/0311371.
- [234] A. Cerri et al., Report from Working Group 4: Opportunities in Flavour Physics at the HL-LHC and HE-LHC, vol. 7, pp. 867–1158. 12 2019, 1812.07638.
- [235] A. Lenz, M. L. Piscopo, and C. Vlahos, "Renormalization scale setting for D-meson mixing," Phys. Rev. D, vol. 102, no. 9, p. 093002, 2020, 2007.03022.
- [236] F. Herren and M. Steinhauser, "Version 3 of RunDec and CRunDec," Comput. Phys. Commun., vol. 224, pp. 333–345, 2018, 1703.03751.
- [237] K. Chetyrkin and M. Steinhauser, "The Relation between the MS-bar and the on-shell quark mass at order α³_s," Nucl. Phys. B, vol. 573, pp. 617–651, 2000, hep-ph/9911434.
- [238] K. Chetyrkin and M. Steinhauser, "Short distance mass of a heavy quark at order α_s^3 ," *Phys. Rev. Lett.*, vol. 83, pp. 4001–4004, 1999, hep-ph/9907509.
- [239] K. Melnikov and T. v. Ritbergen, "The Three loop relation between the MSbar and the pole quark masses," *Phys. Lett. B*, vol. 482, pp. 99–108, 2000, hep-ph/9912391.
- [240] I. I. Y. Bigi, M. A. Shifman, N. Uraltsev, and A. I. Vainshtein, "High power n of m_b in beauty widths and $n = 5 \rightarrow \infty$ limit," *Phys. Rev. D*, vol. 56, pp. 4017–4030, 1997, hep-ph/9704245.
- [241] M. Fael, K. Schönwald, and M. Steinhauser, "Relation between the MS and the kinetic mass of heavy quarks," *Phys. Rev. D*, vol. 103, no. 1, p. 014005, 2021, 2011.11655.
- [242] A. H. Hoang, Z. Ligeti, and A. V. Manohar, "B decay and the Upsilon mass," *Phys. Rev. Lett.*, vol. 82, pp. 277–280, 1999, hep-ph/9809423.
- [243] A. H. Hoang, Z. Ligeti, and A. V. Manohar, "B decays in the upsilon expansion," *Phys. Rev. D*, vol. 59, p. 074017, 1999, hep-ph/9811239.

- [244] M. Neubert, "Symmetry breaking corrections to meson decay constants in the heavy quark effective theory," *Phys. Rev. D*, vol. 46, pp. 1076–1087, 1992.
- [245] W. Kilian and T. Mannel, "QCD corrected $1/m_b$ contributions to $B \bar{B}$ mixing," *Phys. Lett. B*, vol. 301, pp. 382–392, 1993, hep-ph/9211333.
- [246] M. Kirk, A. Lenz, and T. Rauh, "Dimension-six matrix elements for meson mixing and lifetimes from sum rules," *JHEP*, vol. 12, p. 068, 2017, 1711.02100.
- [247] D. King, A. Lenz, and T. Rauh, "to appear,"
- [248] S. Aoki et al., "FLAG Review 2019," Eur. Phys. J. C, vol. 80, no. 2, p. 113, 2020, 1902.08191.
- [249] T. Mannel, D. Moreno, and A. A. Pivovarov, "The Heavy Quark Expansion for the Charm Quark," 3 2021, 2103.02058.
- [250] H.-Y. Cheng, "Phenomenological Study of Heavy Hadron Lifetimes," JHEP, vol. 11, p. 014, 2018, 1807.00916.
- [251] A. Alberti, P. Gambino, K. J. Healey, and S. Nandi, "Precision Determination of the Cabibbo-Kobayashi-Maskawa Element V_{cb}," *Phys. Rev. Lett.*, vol. 114, no. 6, p. 061802, 2015, 1411.6560.
- [252] N. Uraltsev, "On the chromomagnetic expectation value μ_G^2 and higher power corrections in heavy flavor mesons," *Phys. Lett. B*, vol. 545, pp. 337–344, 2002, hep-ph/0111166.
- [253] A. Bazavov et al., "Up-, down-, strange-, charm-, and bottom-quark masses from four-flavor lattice QCD," Phys. Rev. D, vol. 98, no. 5, p. 054517, 2018, 1802.04248.
- [254] P. Gambino, A. Melis, and S. Simula, "Extraction of heavy-quark-expansion parameters from unquenched lattice data on pseudoscalar and vector heavylight meson masses," *Phys. Rev.*, vol. D96, no. 1, p. 014511, 2017, 1704.06105.
- [255] M. Neubert, "QCD sum rule calculation of the kinetic energy and chromo interaction of heavy quarks inside mesons," *Phys. Lett.*, vol. B389, pp. 727– 736, 1996, hep-ph/9608211.
- [256] P. Ball and V. M. Braun, "Next-to-leading order corrections to meson masses in the heavy quark effective theory," *Phys. Rev.*, vol. D49, pp. 2472–2489, 1994, hep-ph/9307291.

- [257] I. I. Y. Bigi, M. A. Shifman, and N. Uraltsev, "Aspects of heavy quark theory," Ann. Rev. Nucl. Part. Sci., vol. 47, pp. 591–661, 1997, hep-ph/9703290.
- [258] I. I. Bigi, T. Mannel, and N. Uraltsev, "Semileptonic width ratios among beauty hadrons," *JHEP*, vol. 09, p. 012, 2011, 1105.4574.
- [259] I. I. Y. Bigi, M. A. Shifman, N. G. Uraltsev, and A. I. Vainshtein, "On the motion of heavy quarks inside hadrons: Universal distributions and inclusive decays," *Int. J. Mod. Phys. A*, vol. 9, pp. 2467–2504, 1994, hep-ph/9312359.
- [260] J. Charles, A. Hocker, H. Lacker, S. Laplace, F. Le Diberder, J. Malcles, J. Ocariz, M. Pivk, and L. Roos, "CP violation and the CKM matrix: Assessing the impact of the asymmetric *B* factories," *Eur. Phys. J. C*, vol. 41, no. 1, pp. 1–131, 2005, hep-ph/0406184.
- [261] S. L. Glashow, J. Iliopoulos, and L. Maiani, "Weak Interactions with Lepton-Hadron Symmetry," Phys. Rev., vol. D2, pp. 1285–1292, 1970.
- [262] T. Jubb, M. Kirk, A. Lenz, and G. Tetlalmatzi-Xolocotzi, "On the ultimate precision of meson mixing observables," *Nucl. Phys. B*, vol. 915, pp. 431–453, 2017, 1603.07770.
- [263] M. Beneke, G. Buchalla, and I. Dunietz, "Width Difference in the $B_s \bar{B}_s$ System," *Phys. Rev. D*, vol. 54, pp. 4419–4431, 1996, hep-ph/9605259. [Erratum: Phys.Rev.D 83, 119902 (2011)].
- [264] M. Beneke, G. Buchalla, C. Greub, A. Lenz, and U. Nierste, "Next-to-leading order QCD corrections to the lifetime difference of B_s mesons," *Phys. Lett. B*, vol. 459, pp. 631–640, 1999, hep-ph/9808385.
- [265] A. Dighe, T. Hurth, C. Kim, and T. Yoshikawa, "Measurement of the lifetime difference of B_d mesons: Possible and worthwhile?," Nucl. Phys. B, vol. 624, pp. 377–404, 2002, hep-ph/0109088.
- [266] M. Beneke, G. Buchalla, A. Lenz, and U. Nierste, "CP asymmetry in flavor specific B decays beyond leading logarithms," *Phys. Lett. B*, vol. 576, pp. 173– 183, 2003, hep-ph/0307344.
- [267] M. Ciuchini, E. Franco, V. Lubicz, F. Mescia, and C. Tarantino, "Lifetime differences and CP violation parameters of neutral B mesons at the next-toleading order in QCD," *JHEP*, vol. 08, p. 031, 2003, hep-ph/0308029.
- [268] A. Lenz and U. Nierste, "Theoretical update of $B_s B_s$ mixing," *JHEP*, vol. 06, p. 072, 2007, hep-ph/0612167.

- [269] A. Bazavov *et al.*, "Short-distance matrix elements for D^0 -meson mixing for $N_f = 2 + 1$ lattice QCD," *Phys. Rev.*, vol. D97, no. 3, p. 034513, 2018, 1706.04622.
- [270] M. Tanabashi *et al.*, "Review of Particle Physics," *Phys. Rev. D*, vol. 98, no. 3, p. 030001, 2018.
- [271] E. Golowich and A. A. Petrov, "Short distance analysis of D^0 D^0 mixing," *Phys. Lett. B*, vol. 625, pp. 53–62, 2005, hep-ph/0506185.
- [272] M. Bobrowski, A. Lenz, J. Riedl, and J. Rohrwild, "How Large Can the SM Contribution to CP Violation in $D^0 \overline{D}^0$ Mixing Be?," *JHEP*, vol. 03, p. 009, 2010, 1002.4794.
- [273] H. Georgi, "D anti-D mixing in heavy quark effective field theory," Phys. Lett. B, vol. 297, pp. 353–357, 1992, hep-ph/9209291.
- [274] T. Ohl, G. Ricciardi, and E. H. Simmons, "D D̄ mixing in heavy quark effective field theory: The Sequel," Nucl. Phys. B, vol. 403, pp. 605–632, 1993, hep-ph/9301212.
- [275] I. I. Bigi and N. G. Uraltsev, " $D^0 \overline{D}^0$ oscillations as a probe of quark hadron duality," Nucl. Phys. B, vol. 592, pp. 92–106, 2001, hep-ph/0005089.
- [276] M. Bobrowski, A. Lenz, and T. Rauh, "Short distance D D mixing," in Proceedings, 5th International Workshop on Charm Physics (Charm 2012): Honolulu, Hawaii, USA, May 14-17, 2012, 8 2012, 1208.6438.
- [277] A. F. Falk, Y. Grossman, Z. Ligeti, and A. A. Petrov, "SU(3) breaking and D0
 anti-D0 mixing," *Phys. Rev. D*, vol. 65, p. 054034, 2002, hep-ph/0110317.
- [278] H.-Y. Cheng and C.-W. Chiang, "Long-Distance Contributions to $D^0 \overline{D}^0$ Mixing Parameters," *Phys. Rev. D*, vol. 81, p. 114020, 2010, 1005.1106.
- [279] H.-Y. Jiang, F.-S. Yu, Q. Qin, H.-n. Li, and C.-D. Lu, "D⁰-D⁰ mixing parameter y in the factorization-assisted topological-amplitude approach," Chin. Phys. C, vol. 42, no. 6, p. 063101, 2018, 1705.07335.
- [280] H. Asatrian, A. Hovhannisyan, U. Nierste, and A. Yeghiazaryan, "Towards next-to-next-to-leading-log accuracy for the width difference in the $B_s \bar{B}_s$ system: fermionic contributions to order $(m_c/m_b)^0$ and $(m_c/m_b)^1$," *JHEP*, vol. 10, p. 191, 2017, 1709.02160.
- [281] H. M. Asatrian, H. H. Asatryan, A. Hovhannisyan, U. Nierste, S. Tumasyan, and A. Yeghiazaryan, "Penguin contribution to width difference and CP asymmetry in B_q - \bar{B}_q mixing at order $\alpha_s^2 N_f$," 6 2020, 2006.13227.
- [282] A. L. Kagan and L. Silvestrini, "Dispersive and Absorptive CP Violation in $D^0 \overline{D^0}$ Mixing," 1 2020, 2001.07207.
- [283] H.-N. Li, H. Umeeda, F. Xu, and F.-S. Yu, "D meson mixing as an inverse problem," 1 2020, 2001.04079.
- [284] A. F. Falk, Y. Grossman, Z. Ligeti, Y. Nir, and A. A. Petrov, "The D⁰ D
 ⁰ mass difference from a dispersion relation," *Phys. Rev. D*, vol. 69, p. 114021, 2004, hep-ph/0402204.
- [285] M. D. Schwartz, Quantum Field Theory and the Standard Model. Cambridge University Press, 3 2014.
- [286] B. A. Kniehl, "Dispersion relations in loop calculations," Acta Phys. Polon. B, vol. 27, pp. 3631–3644, 1996, hep-ph/9607255.
- [287] R. Soldati, "Intermediate Quantum Field Theory. A Next-to-Basic Course for Primary Education," on line notes for the II Semester Course, http://www.robertosoldati.com.